Instabilities, Chaos and Turbulence

An Introduction to Nonlinear Dynamics and Complex Systems

Imperial College Pres

Instabilities, Chaos and Turbulence

An Introduction to Nonlinear Dynamics and Complex Systems

This page intentionally left blank

Instabilities, Chaos and Turbulence

An Introduction to Nonlinear Dynamics and Complex Systems



Paul Manneville Ecole Polytechnique, France

Imperial College Press

Published by

Imperial College Press 57 Shelton Street Covent Garden London WC2H 9HE

Distributed by

World Scientific Publishing Co. Pte. Ltd.
5 Toh Tuck Link, Singapore 596224
USA office: 27 Warren Street, Suite 401–402, Hackensack, NJ 07601
UK office: 57 Shelton Street, Covent Garden, London WC2H 9HE

British Library Cataloguing-in-Publication Data A catalogue record for this book is available from the British Library.

INSTABILITIES, CHAOS AND TURBULENCE An Introduction to Nonlinear Dynamics and Complex Systems

Copyright © 2004 by Imperial College Press

All rights reserved. This book, or parts thereof, may not be reproduced in any form or by any means, electronic or mechanical, including photocopying, recording or any information storage and retrieval system now known or to be invented, without written permission from the Publisher.

For photocopying of material in this volume, please pay a copying fee through the Copyright Clearance Center, Inc., 222 Rosewood Drive, Danvers, MA 01923, USA. In this case permission to photocopy is not required from the publisher.

ISBN 1-86094-483-3 ISBN 1-86094-491-4 (pbk) To Jacqueline,

Jean-Baptiste, Sébastien, Alexis, and Claire.

This page intentionally left blank

Preface

During the last quarter of the Twentieth Century, the study of nonlinear and complex systems has experienced unprecedented development. In the following pages, we present a first approach to standard results and recent advances in this field, applied to phenomena at our scale in our surrounding world. The pretext chosen is mainly instabilities in out-of-equilibrium systems, and more specifically the transition to turbulence in flowing fluids. It should however be clear that the methods used are of fully general use.

The book is based on lecture notes for a short optional course given to second-year students in an engineering school, École Nationale Supérieure de Techniques Avancées, in Paris. This school is devoted to the training of high level engineers in fields including applied mathematics, mechanics and hydrodynamics, electronics,..., oceanography, and management. At the time of the course, students have not yet chosen their specialty, so the course has to be sufficiently general and without too specific requisites. Accordingly, the book should be of interest to nearly any science-oriented undergraduate student and, potentially, to everybody wanting to learn about recent advances in the field of applied nonlinear dynamics. Technicalities are not completely avoided but they are explained as simply as possible using heuristic arguments and specific worked examples, while openings on different topics can be gained by solving exercises at the end of each chapter using the same methods as those explained in the text.

At first, the problem of *chaos* that one has to face very early in this field may seem abstract and difficult. Even if the treatment of examples is not complete, the reader should get a concrete and operational mastery of concepts and techniques to be used from them. As far as the difficulty is concerned, our aim is to transmit the knowledge rather informally and without full mathematical rigor. With respect to mathematics and physics, only basic understanding is required, at the level of what is currently known after one or two years of undergraduate training. In mathematics, this does not go further than elementary algebraic calculus, basic notions of linear algebra and ordinary differential calculus. As far as physics is concerned, it should suffice to follow one's intuition and to admit a few fundamental equations without discussion. So, adaptation of the approach to any other field of interest should thus be envisaged without excessive anxiety.

A first brief chapter situates the context of the study, that of evolutionary problems involving a specific independent variable called *time*, distinguishing *discrete* systems governed by finite sets of ordinary differential equations and *continuous* media described by partial differential equations. It serves as an introduction to the rest of the book, explaining in particular that continuous media driven out of equilibrium may experience instabilities inducing structures that further break down, leading to turbulence.

The second chapter is devoted to a preliminary study of dynamical systems with a small number of degrees of freedom. The archetype of such systems is the oscillator which serves to introduce the first manifestations of nonlinear effects, *e.g.* the occurrence of self-oscillations or the relation between amplitude and frequency.

The way to complex behavior is then apparently left at a too early stage, before the occurrence of chaos. In Chapter 3 we indeed turn to a specific but particularly simple and intuitive physical problem, the stability of a fluid layer heated from below and entering a convection regime. The first part of the chapter is devoted to the analysis of the instability mechanism and an approximate determination of the threshold. In the second part, a description is given of the "death" of the so-formed *dissipative structure* (pattern of convection rolls) and of the steps toward turbulence.

After this detour, we come back to the mathematics of the transition within the dynamical systems framework. A preliminary step of the reduction of the dynamics has to be performed, resting on the distinction between *driving* and *enslaved* modes, and on elimination of the latter modes. The emergence of complexity is then analyzed as a result of the increase of the number of driving modes. This is done in two steps. The first one constitutes the last part of Chapter 4 where we introduce *scenarios* of transition to *temporal chaos* and present some of the tools used to identify it and measure its amount from an empirical point of view. However, as already observed when studying convection, *confinement* effects play an important role, and all that precedes relates to the case when they are strong enough to freeze the spatial structure of the modes. Otherwise, scenarios relevant to extended systems, the second step, involve large scale modulations and *spatio-temporal chaos*, both introduced in Chapter 5.

In common language, the third element of the title, *turbulence*, refers to the irregular, highly fluctuating, behavior of most of the flows surrounding us (the opposite situation of so-called laminar flows is rather exceptional). This problem is tackled in two steps. Instability and transition of *open flows* is examined in Chapter 6. By contrast with systems considered in Chapter 3, where the fluid remained confined to an enclosure, it now circulates from upstream to downstream, the consequences of which will be sketchily discussed. In Chapter 7, we consider *developed turbulence*, again along two paths. First we analyze the different scales, from the largest where energy is injected eddies by instability mechanisms to the smallest where it is consumed *via* viscous dissipation. In a second instance, we turn to the statistical problem of predicting the average properties of a given turbulence flow, of utmost practical interest for an engineer.

Chapter 8 recapitulates the results and opens the perspective toward a complex dynamical system of contemporary interest, the climate of the Earth, and the problem of understanding/modeling its past and present trends.

A first appendix is devoted to a summary of linear algebra results that are useful throughout the book. As far as the understanding of nonlinear phenomena is concerned, the recourse to computers has been of considerable help at different levels. This is the reason why a second appendix, introducing hands-on computer sessions, is devoted to elementary methods that can be developed, with sufficient common sense and no superfluous specialization, to extract useful information from numerical simulations of simple, even simplistic, but well-designed generic models of nonlinear dynamics and pattern formation. The course is completed by laboratory sessions on topics, the theory of which is considered in some exercises.

Palaiseau, June 2004.

Notations.

Here are some indications about the conventions that, with as few exceptions as possible, we will be using throughout. First, upper-case bold letters will denote points in the spaces that will be considered, e.g. $\mathbf{X} \in \mathbb{X}$, \mathbb{X} will most often be a real vector space \mathbb{R}^d . Components will be $X_1, X_2, ..., X_d$, collectively denoted $\{X_i\}$. We will also possibly use the bold and normal upper-case Greek letters in the same context.

For operators working in these spaces, we will use the bold 'cal' TEX font, e.g. \mathcal{F} , and for their components the normal 'cal' font with indices, e.g. \mathcal{F}_i , hence $\mathbf{Y} = \mathcal{F}(\mathbf{X})$ and $Y_j = \mathcal{F}_j(\{X_i\}), i, j = 1, \ldots, d$.

In the case of a linear operator, e.g. \mathcal{L} , we will rather write $\mathbf{Y} = \mathcal{L}\mathbf{X}$. In general, we will not distinguish the operator from the matrix that represents it in a given basis, \mathcal{L} will denote, with little ambiguity, either the operator or the matrix with elements $l_{jj'}$. The above equation would then read in developed form: $Y_j = \sum_i l_{ji} X_i$. Some elements of linear algebra are recalled in Appendix A.

With respect to differentiation, for the ordinary derivative with respect to some variable U we will use d(...)/dU. With respect to time t, we usually prefer dots surmounting the variable, one for each differentiation order: $dX/dt \equiv \dot{X}$ and $d^2X/dt^2 \equiv \ddot{X}$. A short-hand notation for the partial derivative with respect to variable X, $\partial(...)/\partial X$, will be $\partial_X(...)$.

Points in physical space will usually be noted **x** with coordinates x, y, z, corresponding unit vectors $\hat{\mathbf{x}}, \hat{\mathbf{y}}, \hat{\mathbf{z}}$, and partial derivatives $\partial_x, \partial_y, \partial_z$. In the same way, a velocity vector will be noted **V** or **v** with components V_x, V_y, V_z or v_x, v_y, v_z . On some occasions we will rather take x_j, v_j , and $\partial_j \boxminus \partial_{x_j}$, j = 1, 2, 3 allowing us to use the Einstein convention of implicit sum over repeated indices.

The complex conjugate of a complex number $Z = Z_r + iZ_i$, $\mathcal{R}e(Z) = Z_r$, $\mathcal{I}m(Z) = Z_i$, will be noted as Z^* .

For quantities that play the role of control parameters, we will *not* follow the convention used in fluid mechanics to take two letters, *e.g.* 'Re' for the Reynolds number of a given flow, 'Ra' for the Rayleigh number in convection, etc., but a simple ordinary letter, upper or lower case, most usually R or r, to stress the fact that control parameters are ordinary variables (ambiguities will be raised on a case-by-case basis).

Numerical illustrations have most often been obtained using MATLAB. We will occasionally use some of its conventions.

Contents

Pro	eface		vii
1.	Inti	roduction and Overview	1
	1.1	Dynamical Systems as a Context	2
	1.2	Continuous Media as a Subject	5
	1.3	From Simple to Complex	9
		1.3.1 Thermal convection: the instability mechanism	10
		1.3.2 Nonlinear convection and dynamical systems	12
		1.3.3 Stability and instability of open flows	15
		1.3.4 Beyond the transition: fully developed turbulence .	17
	1.4	Conclusion	20
	1.5	Exercises	20
2.	First Steps in Nonlinear Dynamics		
	2.1	From Oscillators to Dynamical Systems	25
		2.1.1 First definitions	25
		2.1.2 Formalism of analytical mechanics	32
		2.1.3 Gradient systems	33
	2.2	Stability and Linear Dynamics	35
		2.2.1 Formulation of the linear stability problem	36
		2.2.2 Two-dimensional linear systems	37
		2.2.3 Stability of a time-independent regime	41
	2.3	Two-dimensional Nonlinear Systems	43
		2.3.1 Two examples of oscillators	44
		2.3.2 Amplitude and phase of nonlinear oscillators	50
	2.4	What Next?	60
	2.5	Exercises	61

3.	Life	and D	eath of Dissipative Structures	71
	3.1	Emer	gence of Dissipative Structures	71
		3.1.1	Qualitative analysis of the instability mechanism	71
		3.1.2	Simplified model	73
		3.1.3	Normal mode analysis, general perspective	75
		3.1.4	Back to the model \ldots	78
		3.1.5	Vicinity of the threshold: linear stage	82
		3.1.6	Classification of unstable modes	84
	3.2	Disint	tegration of Dissipative Structures	86
		3.2.1	Simplified model of nonlinear convection	86
		3.2.2	$Transition \ to \ turbulence \ of \ convection \ cells . \ . \ .$	90
		3.2.3	Transition toward chaos in confined systems \ldots .	94
		3.2.4	Dynamics of "textures" in extended systems	97
		3.2.5	$Turbulent \ convection \ . \ . \ . \ . \ . \ . \ . \ . \ . \ $	101
	3.3	Exerc	ises	103
4.	Nor	linear	Dynamics: from Simple to Complex	115
	4.1	Redu	ction of the Number of Degrees of Freedom	116
		4.1.1	Role of aspect ratios	116
		4.1.2	Low dimensional effective dynamics	118
		4.1.3	Center manifolds and normal forms	121
	4.2 Transition to Chaos			125
		4.2.1	First steps: time-independent and periodic regimes .	126
		4.2.2	Quasi-periodicity and resonances	130
		4.2.3	Quasi-periodicity and lockings	136
	4.3	Chara	acterization of Chaotic Regimes	147
		4.3.1	Instability of trajectories and Lyapunov exponents .	148
		4.3.2	Fractal aspects	150
	4.4	Empi	rical Approach of Chaotic Systems	154
		4.4.1	Standard analysis by means of Fourier transform	156
		4.4.2	Reconstruction by the method of delays	157
		4.4.3	Sampling frequency and embedding dimension	160
		4.4.4	Application	165
	4.5	Concl	usion	168
	4.6	Exerc	ises	168

Con	te	n	ts
-----	----	---	----

5.	Nonlinear Dynamics of Patterns		
	5.1	Quasi-one-dimensional Cellular Structures	182
		5.1.1 Steady states	182
		5.1.2 Amplitude equation	185
	5.2	Dissipative Crystals	187
	5.3	Short Term Selection of Patterns	190
	5.4	Modulations and Envelope Equations	191
		5.4.1 Quasi-one-dimensional cellular patterns	191
		5.4.2 2D modulations of quasi-1D cellular patterns	193
		5.4.3 Quasi-two-dimensional cellular patterns	195
		5.4.4 Oscillatory patterns and dissipative waves	197
		5.4.5 Universal long-wavelength instabilities	199
	5.5	What Lies Beyond?	204
	5.6	Exercises	205
6.	Open Flows: Instability and Transition		
	6.1	Base Flow Profiles	213
		6.1.1 Strictly one-dimensional flows	213
		6.1.2 More general velocity profiles	215
		6.1.3 Extension to arbitrary profiles	218
	6.2	Linear Stability	219
		6.2.1 General framework	219
		6.2.2 Inviscid flows	223
		6.2.3 Viscous flows	231
		6.2.4 Instability and downstream transport	234
	6.3	Transition to Turbulence	241
		6.3.1 Nonlinear development of instabilities	241
		6.3.2 Inviscidly unstable flows	244
		6.3.3 Inviscidly stable flows	249
		6.3.4 Turbulent spots and intermittency	253
	6.4	Exercises	265
7.	Developed Turbulence		
	7.1	Scales in Developed Turbulence	272
		7.1.1 Production scale	272
		7.1.2 Inertial scales and Kolmogorov spectrum	274
		7.1.3 Dissipation scales	276
		7.1.4 Remarks	278
		7.1.4 Remarks	278

	7.2	Mean Flow and Fluctuations	279
		7.2.1 Statistical approach	279
		7.2.2 Reynolds averaged Navier-Stokes (RANS) equations	280
		7.2.3 Energy exchanges in a turbulent flow	283
	7.3	Mean Flow and Effective Diffusion	284
		7.3.1 Mixing length and eddy viscosity	284
		7.3.2 Application to the determination of the mean flow .	286
	7.4	Beyond the Elementary Approach	290
		7.4.1 Turbulence modeling	292
		7.4.2 Large eddy simulations	294
	7.5	Exercises	299
8.	Sun	nmary and Perspectives	303
	81	Dynamics Stability and Chaos	304
	82	Continuous Media Instabilities and Turbulence	308
	0.2 Q 2	Approach to a Complex System: the Earth's Climate	312
	8.0	Exercise: Ice ages as catastrophes	328
	0.1		020
Apj	pend	ix A Linear Algebra	331
	A.1	Vector Spaces, Bases, and Linear Operators	331
	A.2	Structure of a Linear Operator	334
	A.3	Metric Properties of Linear Operators	343
	A.4	Exercises	348
Apj	pend	ix B Numerical Approach	351
	B.1	Treatment of the Time Dependence	352
	B.2	Treatment of Space Dependence in PDEs	358
		B.2.1 Finite difference methods	358
		B.2.2 Spectral methods	363
	B.3	Exercises	369
	B.4	Case studies	371
		B.4.1 ODEs 1: Forced pendulum	371
		B.4.2 ODEs 2: Lorenz model	373
		B.4.3 ODEs 3: Rössler and Chua models	376
		B.4.4 PDEs 1: SH model, finite differences	378
		B.4.5 PDEs 2: SH model, pseudo-spectral method	380
Rih	linar	anhu	383
Ind	or or	×L1a3	387
1164	ີພ		001

Chapter 1

Introduction and Overview

In a linear world, the effects being always proportional to their causes, everything would be "simple" since the tools that would allow us to represent it as a *superposition* of elementary states are rather well mastered. Unfortunately the world is "complex." As a matter of fact, we often observe effects which saturate in spite of an increase of their causes, or which go in different and somehow unexpected ways. All this is ascribed to nonlinearities. Of course, if we succeed in determining the state of a nonlinear system, we immediately try to go back to a problem that we know how to handle by linearizing the dynamics around it and treating small departures from it by a perturbative approach, and next to reproduce this scheme as far as possible in order to reach other fully nonlinear states, eventually "far from" the initial one.

Most recently, our science of intrinsically nonlinear phenomena has made much progress (even though it rests in part on the use of linear tools) and investing this knowledge further in applications is of primary interest. The aim of these lecture notes is therefore to introduce the reader to this breakthrough by taking the problem of macroscopic instabilities as a pretext. In thermodynamic systems close to equilibrium the response to excitations is essentially linear, *i.e.* proportional to the (sufficiently small) amplitude of the applied stress with a proportionality factor called a *susceptibility*. As a consequence of linearity, the regime that develops is *unique*. When the applied stress increases, the system is driven farther from equilibrium and nonlinearities can no longer be neglected. This opens the possibility of *bifurcations* towards different regimes that can coexist and compete. As a result of such "catastrophes," losing reference to the initial state the dynamics becomes increasingly complicated, typically from "laminar" to "turbulent flow."

1.1 Dynamical Systems as a Context

In order to study the causes of the complexity induced by nonlinearities, we will concentrate our attention on problems defined in terms of the evolution of a set of *state variables* functions of a single independent variable called *time* (the general concept of dynamical system). Mechanical systems in the ordinary sense obviously belong to this class. Their archetype is the oscillator, and in its simplest linear expression, the *harmonic oscillator*¹ (Figure 1.1, left)

$$m\ddot{X} = F = -kX = 0, \qquad (1.1)$$

where X measures the departure from the equilibrium length of the spring, m is the mass attached to its end, and F is the restoring force, here taken proportional to X via the stiffness coefficient k. As a result the oscillation period is independent of the amplitude: $T = T_0 = 2\pi/\omega_0$, with $\omega_0^2 = k/m$.

At this stage, nonlinearities can introduce themselves in two ways. First it is easy to imagine that the stiffness coefficient may not be a constant but rather a function of X itself. Assuming $k = k_0(1 + \alpha X^2)$ leads to what is known as the *Duffing oscillator*. Another possibility comes from the existence of mechanical constraints. The ideal rigid *pendulum*, a mass m at the end of a weightless rod of length l and revolving around a horizontal axis in the gravity field, is a good example (Figure 1.1, right). Parameterizing its position by the angle θ it forms with the vertical axis, one obtains:

$$J\ddot{\theta} + mgl\sin\theta = 0\,,\tag{1.2}$$

where $J = ml^2$ is the moment of inertia and $-mgl\sin\theta$ the torque exerted by its weight. The nonlinearity built in the sine function is a consequence of the topological constraint keeping the mass rigidly at a constant distance from the rotation axis and plays a role in case of large deviations. As an outcome, several equilibrium positions exist. The 'down' position is a stable equilibrium point, with small amplitude oscillations around it governed by the same equation as the harmonic oscillator. The 'up' position is unstable and the pendulum departs from it upon perturbations of any amplitude, even infinitesimal. Furthermore low energy states oscillate around the 'down' position with a period depending on the energy, while at larger energies, the system rotates always in the same direction, accelerating and slowing down periodically as it passes through the 'down' and 'up'

 $^{^{1}}$ As indicated at the end of the preface, when the independent variable is time, differentiation is denoted by dots on top of dependent variables.



Fig. 1.1 Left: Linear spring. Right: Rigid pendulum.

positions, respectively. The behavior of nonlinear oscillators will be further considered in Chapter 2.

Such mechanical examples can serve us to introduce a geometrical description of the dynamics in a space called the *phase space*, here with coordinates the positions X or θ and the momenta $m\dot{X}$ or $J\dot{\theta}$, and to account for the dynamics in terms of trajectories in that space. In practice, any system involving a finite set of dependent variables serving to characterize its state, its *state variables*, governed by differential equations, belongs to the class we consider here. Examples are intensities and potential differences in an electrical circuit (Chapter 2, §2.3.1.2), concentrations of reactants in chemistry (Exercise 1.5.1), population densities in ecology (Exercise 1.5.2),... In analytical mechanics, a pair 'position + conjugate momentum' is called a *degree of freedom*. In a more general context, what is called a degree of freedom is simply a state variable.

In general, it is advantageous to write down the dynamical equations as a system of first order differential equations

$$X_j = \mathcal{F}_j (\{X_i(t); i = 1, ..., d\}, t), \qquad j = 1, ..., d,$$

where the integer d is, in this context, called the *dimension* of the system.² Using the notations X and \mathcal{F} for the sets $\{X_j, j = 1...d\}$, and $\{\mathcal{F}_j, j = 1...d\}$, we have thus:

$$\dot{\mathbf{X}} = \mathcal{F}(\mathbf{X}, t) \,. \tag{1.3}$$

²Accordingly, the dimension of the phase space is twice the number of degrees of freedom in analytical mechanics whereas otherwise it is just the number of state variables. In the following we will try to avoid ambiguities arising from this terminology. See later, Chapter 2, §2.1.2.

For a system such as (1.3) time is a *continuous* variable. By contrast, a *discrete-time* system is defined as an iteration that we can write in the form

$$X_{j,k+1} = \Phi_j (\{X_{i,k}; i = 1, \dots, d\}, k), \qquad j = 1, \dots, d.$$

or, formally:

$$\mathbf{X}_{k+1} = \mathbf{\Phi}\left(\mathbf{X}_k\right). \tag{1.4}$$

For such a system, time is the discrete index k that serves to monitor the evolution. While appearing as a topic in itself in mathematics, the study of discrete-time dynamical systems turns out to be essential in physics and engineering owing to their occurrence as a result of the stroboscopic analysis of periodically forced continuous-time dynamical systems, or of a Poincaré section in self-oscillating systems. They can also introduce themselves as the outcome of modeling effort of specific phenomena, *e.g.* seasonal counts in population dynamics. All this will be considered in detail in Chapter 4.

Systems considered here all carry out the intuitive concept of determinism which imply the prediction of a state at time t or k beyond some time t_0 or k_0 at which some initial condition is specified. This *initial value problem* presents itself formally or explicitly as a computational problem, directly if it is defined as an iteration (1.4) or indirectly through some numerical approximation for a continuous-time system (1.3), *e.g.* Euler extrapolation,

$$\mathbf{X}(t_{k+1}) = \mathbf{X}(t_k) + \Delta t \, \mathcal{F}\left(\mathbf{X}(t_k), t_k\right), \qquad (1.5)$$

since analytical integration is rarely possible. See Appendix B, B.1, for an introduction.

Resuming the geometrical perspective introduced above, we are now interested in the properties of *trajectories* followed by the system in its phase space. This study rapidly points to the key role played by the concept of *stability* taken in the broad sense of *resistance to perturbations*. In practice, this vague definition refers to two different viewpoints:

The first one, rather quantitative, applies to specific trajectories: the solutions found, which depend on initial conditions, have to withstand small perturbations imposed either at the start or during the subsequent evolution, owing to unavoidable disturbances, either intrinsic (thermodynamic fluctuations) or extrinsic (noise). From this first viewpoint, only *stable* solutions (that resist) are physically observable. Further, a sufficiently large set of experimental conditions should make them attainable. The second viewpoint is mode qualitative. It refers to the very definition of the system itself: when compared to the real world, any abstract implementation is intrinsically blurred with numerous approximations and its control parameters are not determined with infinite precision. Accordingly, in order to be of help, the analytical model must be robust, *i.e.* its predictions must not be too sensitive to these different sources of inaccuracy. This property indeed fails at bifurcation points where the system experiences qualitative changes of behavior. At such points one says that it is structurally unstable: the nature of the state depends sensitively on the perturbation.

It can be easily understood that these two facets of the term 'stability' are equally important for the applications. They will be at the heart of the most abstract part of the course, first in Chapter 2 and next in Chapter 4 where we will give a more precise meaning to the word 'prediction' when the considered system evolves chaotically, that is to say in a way *unpredictable* in the long term in spite of short term *determinism*, due to an instability of trajectories that is the essence of *chaos*.

1.2 Continuous Media as a Subject

Dynamical systems considered up to now were endowed with a supposedly small number of dependent variables, thus living in spaces of low enough dimensions. Once the microscopic structure of matter has been recognized, in principle one should turn to the study of systems made of a large number of components at the molecular scale, each with its own degrees of freedom in the mechanical sense. However a refined description of the microscopic configurations is generally useless (due to chaos at this scale) and our ignorance of dynamical details can be circumvented by adopting a statistical point of view from which the less probable (specific initial condition) evolves into the most probable (equilibrium compatible with conservation laws), founding the thermodynamic approach. In this framework, microscopic variables are replaced by average quantities such as energy, entropy, temperature or pressure.

In practice, global thermodynamic equilibrium is not of much interest since it accounts for a world eaten by the worm of the equiprobability of microscopic states, and in some sense completely dead. On the contrary, an inhomogeneous world traversed by various fluxes making it alive (which by the way allows us to study it!) is much richer and more interesting. We are indeed confronted to a large class of transport processes in media that are out of equilibrium on a macroscopic scale and there is a wide range of time and space scales for which the concept of *continuous medium* is appropriate, that is to say the description of a system according to which its state variables are functions of the position in physical (three-dimensional) space, $\mathbf{X}(\mathbf{x}, t)$, governed by partial differential equations:

$$\partial_t \mathbf{X} = \mathcal{G}(\mathbf{X}, \boldsymbol{\nabla} \mathbf{X}, t) \,. \tag{1.6}$$

From a mathematical perspective, such systems are infinite-dimensional since we need to specify the value of these variables at every point in space. On the other hand the validity of the description relies on the definition of a mesoscopic scale inbetween the microscopic (molecular) level and the (macroscopic) size of the system taken as a whole. The concept of *material point* and the assumption of *local equilibrium* make sense precisely on this intermediate scale, sufficiently small to be considered as infinitesimal but large enough to contain as many molecules as necessary for the laws of thermodynamic to be relevant.

Equations such as (1.6) are in general derived from the macroscopic balance of extensive thermodynamic variables, say Z, that, in differential form, read

$$\partial_t \rho_z + \boldsymbol{\nabla} \cdot \mathbf{J}_z = \Sigma_z \,, \tag{1.7}$$

where ρ_z is the density of Z, \mathbf{J}_z its flux, and Σ_z a source term that cancels when Z is a conserved quantity (momentum, energy). The simplest example is the Fourier heat equation governing thermal diffusion:

$$C\partial_t T = \chi \nabla^2 T \,, \tag{1.8}$$

governing the temperature field in a solid submitted to moderate gradients. It derives from the Fourier law

$$\mathbf{J}_{Q} = -\chi \boldsymbol{\nabla} T \,. \tag{1.9}$$

that links the heat flux to the temperature gradient through the thermal conductivity χ . Quantity C in (1.8) is the specific heat defined through the thermodynamic relation $\delta U = C\delta T$. Inserting this into the balance equation for the internal energy $\partial_t \rho_U + \nabla \cdot \mathbf{J}_Q = 0$, where ρ_U is the corresponding density obtained as the limit of $\delta U/\delta V$ as the volume element $\delta V \to 0$, immediately yields (1.8).

The quantity $\kappa = \chi/C$ is the thermal diffusivity. Dimensionally, this coefficient is homogeneous to $[L]^2/[T]$, like the diffusivity D appearing in the Fick law governing molecular diffusion. In this context, an important quantity is the time scale τ for diffusive relaxation over a typical space scale ℓ , directly obtained from the dimensional relation above as $\tau = \ell^2/\kappa$. The relaxation of T is examined in the most elementary case in Exercise 1.5.3 which further illustrates a rare situation where one can find an analytic transform allowing the exact linearization of a parent nonlinear problem.

Equation (1.8) is indeed linear provided that the specific heat C and the thermal conductivity χ are constant, which is reasonable not to far from equilibrium. The Fourier law (1.9) is the prototype of a phenomenological relation linking a flux (the effect) to a gradient (the cause) at a linear level. In practice, the validity range of such linear law is generally rather wide, because constraints that we are able to apply to continuous media are usually weak when compared to molecular interactions. As counter-examples one could cite electronic systems containing active elements with threshold effects (diodes), or chemically reacting media, and a fortiori biological systems. However, in macroscopic media strong nonlinearities can in fact arise from global considerations. This is particularly the case in fluid systems where macroscopic flow deeply affects the transport properties, rendering a locally linear medium effectively nonlinear.

So, in hydrodynamics the concept of the *material point* transforms itself into that of *fluid particle*, the position of which becomes a function of time:

$$\mathbf{M}(t) = (x_{\mathbf{M}}(t), y_{\mathbf{M}}(t), z_{\mathbf{M}}(t)) \ .$$

It is most often useful to pass from this Lagrangian description, to an Eulerian approach in terms of velocity field:³

$$\mathbf{v}\equiv (v_x,v_y,v_z)\equiv \left(rac{\mathrm{d}}{\mathrm{d}t}x_{\mathbf{M}},rac{\mathrm{d}}{\mathrm{d}t}y_{\mathbf{M}},rac{\mathrm{d}}{\mathrm{d}t}z_{\mathbf{M}}
ight)$$
 .

The two descriptions are linked by the definition of the material derivative measuring the evolution of any physical quantity Z attached to the fluid particle as it is followed during its motion: $Z = Z(x_M, y_M, z_M, t)$. Mathematically, material differentiation is thus a total differentiation with respect to time, $\frac{d}{dt}Z$, that is:

$$\frac{\mathrm{d}}{\mathrm{d}t}Z = \partial_t Z + \partial_x Z \,\frac{\mathrm{d}}{\mathrm{d}t} x_{\mathsf{M}} + \partial_y Z \,\frac{\mathrm{d}}{\mathrm{d}t} y_{\mathsf{M}} + \partial_z Z \,\frac{\mathrm{d}}{\mathrm{d}t} z_{\mathsf{M}} = \partial_t Z + \mathbf{v} \cdot \boldsymbol{\nabla} Z \,. \tag{1.10}$$

³Here we no longer denote time differentiation by dots but come back with explicit derivatives in order to avoid ambiguities.

This relation allows a simple expression of the balance equation in differential form provided that the flux of Z is properly split into an irreversible diffusive part and a reversible part linked to the macroscopic motion:

$$\mathbf{J}_{z} = \mathbf{J}_{z,\text{diff}} + \rho_{z} \mathbf{v}.$$

When studying the mechanism of natural convection in fluids originally at rest we will demonstrate the role of a global nonlinearity played by the advection term in spite of its apparent linearity in ρ_z , but before considering this example, let us recapitulate the equations governing the motion of a simple Newtonian fluid, *i.e.* a single component fluid with a constant viscosity. The first one is the continuity equation accounting for the conservation of matter. In full generality it reads:

$$\partial_t \rho + \boldsymbol{\nabla} \cdot \mathbf{J}_{\rho} = 0$$

(no source term). Noticing that diffusion of matter within itself does not make sense, we get $\mathbf{J}_{\rho} = \rho \mathbf{v}$ which leads to

$$\partial_t \rho + \mathbf{v} \cdot \nabla \rho + \rho \nabla \cdot \mathbf{v} \equiv \frac{\mathrm{d}}{\mathrm{d}t} \rho + \rho \nabla \cdot \mathbf{v} = 0.$$
 (1.11)

In the following we will consider *incompressible flows* characterized by $\frac{d}{dt}\rho = 0$, so that the continuity equation more simply reads

$$\boldsymbol{\nabla} \cdot \mathbf{v} = 0. \tag{1.12}$$

On the other hand, for an incompressible fluid the compression viscosity drops out and there only remains a shear viscosity which, for an isotropic Newtonian fluid, relates the viscous stress tensor to the rate of strain in a linear way:

$$\sigma_{ij} = \mu(\partial_i v_j + \partial_j v_i),$$

where μ is the dynamical viscosity. Once inserted into the momentum conservation equation, this relation leads to the Navier-Stokes equation

$$\rho(\partial_t + \mathbf{v} \cdot \nabla)\mathbf{v} = -\nabla p + \mu \nabla^2 \mathbf{v}. \qquad (1.13)$$

The kinematic viscosity $\nu = \mu/\rho$ is the transport coefficient accounting for the diffusive relaxation of velocity gradients (Stokes law). At the inviscid limit ($\nu \rightarrow 0$) the Navier-Stokes equation is called the *Euler* equation. The opposite limit where the viscous dissipation dominates is called the *Stokes approximation*. Finally if the heating generated by viscous friction is negligible, energy conservation yields the heat equation (1.8) where $\partial_t T$ is just replaced by $\frac{d}{dt}T \equiv (\partial_t + \mathbf{v} \cdot \nabla)T$, hence:

$$\partial_t T + \mathbf{v} \cdot \boldsymbol{\nabla} T = \kappa \boldsymbol{\nabla}^2 T \,. \tag{1.14}$$

1.3 From Simple to Complex

In a narrow vicinity of thermodynamic equilibrium, the response of a continuous system is simply proportional to the strength of applied constraints and displays the same symmetries, stationary or periodic in time, uniform or periodic in space, for example. However, far from equilibrium, when nonlinearities can no longer be neglected, the system can bifurcate towards solutions that breaks some of these symmetries.

The nature of the regime that develops depends on the value of control parameters measuring the relative intensity of the different contributions to the dynamics. In fluid mechanics, it will most often be the *Reynolds number* defined as $R = U\ell/\nu$ where U and ℓ represent a typical velocity and a length scale both characteristic of the flow under consideration while ν , the kinematic viscosity introduced above, is a fluid's property. In line with what has been said about the thermal diffusion time, it is enlightening to analyze the Reynolds number as the ratio of the viscous relaxation time over the length ℓ , $\tau_{\rm v} = \ell^2/\nu$, to the advection time required to carry velocity fluctuations over the same distance, $\tau_{\rm a} = \ell/U$. When the Reynolds number $R = \tau_{\rm v}/\tau_{\rm a}$ is small, viscosity has time to wipe out inhomogeneities, whereas when it is large, thermodynamic dissipation is too slow and the mechanical contribution of advection dominates.

The study of the stability of a given regime called the *base state* rests on the definition of perturbations, *i.e.* departures from this state, and their subsequent evolution. Figure 1.2 illustrate the general situation. Below a value of the control parameter R denoted R_g , 'g' for global, the base state is unconditionally stable: whatever the shape and the amplitude of the perturbation, it decays and the system returns to its base state (sufficient condition of stability). Above a second value R_c , 'c' for critical, the system is sensitive to at least one unavoidable perturbation and unconditionally departs from the base state. The determination of R_c appeals to the socalled *linear stability theory* dealing with the evolution of fluctuations that are *mathematically* infinitesimal. It is indeed sufficient that there exist such a perturbation that is amplified for the base state to be unstable. It may happen that R_g coincide with R_c , in which case $R \geq R_c$ is a necessary and



Fig. 1.2 Stability of the base state as a function of some control parameter R, e.g. the Reynolds number.

sufficient condition of instability but in general the interval $[R_g, R_c]$ has a finite width, which defines a range of *conditional stability*: the stability of the base state depends on the shape and amplitude of the (finite amplitude) perturbations to which it is submitted. Most of the time this fundamentally nonlinear problem remains unsolved.

In some favorable cases, one can succeed in computing the state that sets in above the threshold of the *primary* instability by a perturbation method. This so-obtained state is then promoted as a new base state, the stability of which is of interest. This state can in turn become unstable with respect to a *secondary* instability that complicates the dynamics, and so on up to a chaotic state which, for fluid systems, is usually called *turbulent*. In this perspective, turbulence is considered as resulting from various modes at the end of a *cascading process*. Instead of staying at this formal viewpoint let us rather examine a particularly simple and intuitive concrete mechanism responsible for the instability of a liquid layer heated from below and initially at rest.

1.3.1 Thermal convection: the instability mechanism

Let us consider a layer of fluid at rest presenting a density stratification due to heating from below (Figure 1.3). A temperature difference $\Delta T = T_{\rm b} - T_{\rm t} > 0$ is thus applied between the bottom 'b' and the top 't' of the layer. Since, with few exceptions, the density decreases with increasing temperature and this stratification with heavier fluid on top of lighter fluid is potentially unstable in a vertical gravity field (like the 'up' position of the pendulum).

When ΔT is small, the fluid stays at rest since the gravitational potential energy that would be gained by moving the heavier fluid to the bottom is not sufficient to counterbalance the energy loss by dissipation in that motion. More precisely, as long as the fluid is at rest, the heat transfer is entirely carried out by *conduction*. The temperature profile in the base



Fig. 1.3 Left: Instability mechanism of thermal convection. Right: Convection cells with wavelength $\lambda_c = 2\pi/k_c$.

state $T_0(z)$ is obtained from the Fourier equation (1.14) that simply reduces itself to $d^2T_0/dz^2 = 0$ since $\mathbf{v} \equiv 0$, hence $T_0(z) = T_b - \beta z$, where $\beta = \Delta T/h$ is the applied temperature gradient, h being the height of the layer.

Let us suppose a fluctuation localized inside a tiny droplet marked with a slightly higher temperature (perturbation $\theta > 0$). Since this bubble of hotter fluid is surrounded with colder denser fluid, it experiences an upward differential buoyancy force which makes it moving up, encountering ever colder fluid, which reinforce the motion: this is the destabilizing part of the mechanism. However, two stabilizing processes tend to oppose this process. First, the so-induced velocity tends naturally to decay owing to viscous friction. Second, thermal diffusion aims at ironing out the horizontal temperature gradient accompanying fluctuation θ . The fluid layer stays at rest as long as dissipation dominates but convection develops when the destabilization is sufficient, *i.e.* when ΔT is larger than some critical value ΔT_c called the *instability threshold*.

On the other hand, the stabilizing dissipative processes are diffusive in essence, so that their efficiency depends on the horizontal space dependence of the velocity and temperature modulations. Damping is fast on small scales and slow on large scales. It thus depends on the length-scale of the perturbation and it turns out that the onset of the instability corresponds to some optimum scale. For dimensional reasons this scale is of the order of the height h, as implied by the sketch in the left part of Figure 1.3. Accordingly the bifurcated convection state often sets in as a system of periodic rolls, with spatial period $\lambda_c \approx 2h$, as illustrated in its right part. The quantity $k_c = 2\pi/\lambda_c$ which appears when representing the plan-form in Fourier series is called the *critical wavevector*. The pattern formed has been called a *dissipative structure* by Prigogine [Glansdorff and Prigogine (1971)].

1.3.2 Nonlinear convection and dynamical systems

Here we account for the dynamics of convection close to the threshold by heuristic arguments that could be entirely supported by a rather complex detailed theoretical approach yielding the concrete value of the coefficients that we will introduce phenomenologically.

The space-time coherence introduced in the system by the instability mechanism allows us to describe the evolution of the convecting layer using a simple variable A(t) measuring the amplitude of convection. We thus assume that, as far as their horizontal dependence is concerned, the velocity and temperature fields can be taken in the form

$$(v_z, \theta) \propto A(t) \sin(k_c x),$$
 (1.15)

where A plays the role of an *effective degree of freedom* and k_c is the critical wavevector introduced earlier. We are now interested in the phenomenological derivation of an evolution equation for this amplitude.

At the infinitesimal stage, the perturbation is governed by a differential equation that should read

$$\frac{\mathrm{d}}{\mathrm{d}t}A = \sigma A\,,\tag{1.16}$$

where coefficient σ accounts for the growth rate, negative below the threshold (damping) and positive beyond (amplification). Let us define a reduced control parameter $r = (\Delta T - \Delta T_c)/\Delta T_c$ and assume that the behavior of σ as a function of r is not singular, so that its expression can be restricted to the first term of its Taylor expansion in powers of r:

$$\sigma = r/\tau_0 \,, \tag{1.17}$$

where τ_0 , homogeneous to a time, characterizes of the natural evolution of relevant fluctuations.

Equation (1.16) where σ is given by (1.17) accounts for the evolution of coupled velocity-temperature fluctuations as long as the amplitude A is sufficiently small. This does not raise difficulties when r < 0 since A decays but when r > 0, in the unstable domain, it is exponentially amplified and does not stay small for a long time. In order to model the nonlinear effects, we will assume that (1.16) remains valid but that σ has to be corrected and becomes a function of A itself. Noticing that the change ' $A \mapsto -A$ ' corresponds to a change of the spinning direction of the rolls (or to a translation of the structure by $\lambda_c/2$) and that the physics of the problem should not be sensitive to such a change, we are led to assume that the effective growth rate σ_{eff} is an even function of A. At lowest order of an expansion in powers of A, it can be taken as:

$$\sigma_{\rm eff} = (r - gA^2)/\tau_0 \,, \tag{1.18}$$

where g is called the *Landau constant*. In the simple case considered here, g is a positive quantity, *i.e.* $g = 1/\bar{A}^2$, \bar{A} measuring the typical amplitude at which nonlinearities become effective. The experiments show (and the theory demonstrates) that the convection mechanism is indeed selflimiting, so that the effective growth rate decreases when the amplitude of convection increases. The nonlinear model accounting for convection in the neighborhood of the threshold then reads:

$$\tau_0 \frac{\mathrm{d}}{\mathrm{d}t} A = \mathcal{F}_r(A) = rA - gA^3 \,. \tag{1.19}$$

Its study is elementary and will be resumed later in a more general context. Here it is our first example of *effective dynamical system*. It governs a simple scalar variable A representing the macroscopic evolution of our system. Upon rescaling t by τ and A by \overline{A} , equation (1.19) can be recast in a universal form:

$$\frac{\mathrm{d}}{\mathrm{d}t}A = rA - A^3 \,. \tag{1.20}$$

The evolution of A from an initial value $A_0 = 10^{-3}$ for r = 0.1 is illustrated in Figure 1.4. As long as $rA \gg A^3$ the growth remains exponential, which appears graphically linear on the lin-log plot in the left part of the figure. When A increases, the slope of the curve σ_{eff} decreases. Ultimately, A saturates to a value $A_* = \sqrt{r}$. Taking a negative initial value would have led to $-A_*$, owing to the symmetry of the equation. The graph giving the nontrivial state as a function of the control parameter, depicted in the right part of Figure 1.4, is called a *bifurcation diagram*.

In the terminology of dynamical systems, variable A is the degree of freedom.⁴ Accordingly, the real axis is here the phase space, while $\mathcal{F}_r(A)$ is the vector field defined on this space that governs the dynamics. Stationary solutions A_* are called fixed points, a stable fixed point being a simple example of attractor. The transition 'conduction \rightarrow convection' is continuous with a stable nontrivial A_* tending to zero as the threshold is approached from above. One then speaks of a supercritical bifurcation. Owing to the shape of the graph in Figure 1.4 (right), it is further called a fork bifurcation. Equation (1.20) is thus the normal form for a supercritical fork bifurcation.

⁴Remember that we have left the field of analytical mechanics, see Note 2, p. 3.



Fig. 1.4 Left: Evolution of the amplitude A as a function of time for model (1.20) with r = 0.1 (lin-log plot). Right: Bifurcation diagram giving the amplitude A of the steady state as a function of r. Conventionally, the unstable state A = 0 for r > 0 is indicated by dashes while a continuous thick line is used for the stable states, A = 0 for r < 0 and $A = A_* = \pm \sqrt{r}$ for r > 0.

In the opposite case, *i.e.* g < 0 in (1.19), the nontrivial solution appears below the threshold and is unstable, the bifurcation is then *subcritical*, and equation (1.20) has to be further completed to yield a meaningful model. This abstract approach will be resumed in Chapter 2 and 4.

Convection rolls that develop beyond the threshold of the first instability can themselves become unstable with respect to other mechanisms. One thus has to study the flow that results from the superposition of the saturated *primary* mode to the initial base state, using a similar but technically much more complicated approach. *Secondary* modes can then be detected yielding, upon saturation, new base states ready for subsequent destabilization as the control parameter is further increased.

The convecting layer is a continuous medium described by field variables. The reduction of the dynamics to a single scalar variable A, or a small set of similar variables when other instabilities have taken place, calls for justifications that will be examined in Chapter 4 as a precondition to the use of the concept of *chaos* for interpreting the increasing complexity of the dynamics all along the cascade of instabilities leading to turbulence. This approach will be all the more relevant when there are few convection cells, well confined by lateral walls at distances of the order of the height h itself. Its limitations will also have to be studied when *confinement effects* by far-apart lateral boundaries are too weak (Chapter 5).

1.3.3 Stability and instability of open flows

Natural convection has been the subject of many academic studies since it offers an ideal testing ground for ideas developed in nonlinear dynamics. However, systems of interest in engineering studies often display a supplementary feature: they involve *open flows* characterized by the existence of a mean current from upstream to downstream. Whereas the linear stability theory of open flows has a long history, dating back to the end of the Nineteenth Century, difficulties linked to this specific feature have delayed the nonlinear approach. Rather than a thorough account of results obtained in this field, Chapter 6 should thus be considered as a preliminary presentation aiming at better situating the problem in a nonlinear perspective and making its study easier.

Purely kinetic effects play an important role in open flows and can by themselves already be at the origin of instabilities. Accordingly, a first distinction can be made between flows that are *mechanically unstable* and those that are *mechanically stable*. The former display an inflection point in the base velocity profile which makes them unstable in the absence of viscous dissipation according to *Rayleigh's criterion*. Their prototype is the *mixing layer* illustrated in Figure 1.5 (left). By contrast, mechanically stable flows may become unstable only due to subtle feedbacks involving a transfer of momentum of viscous origin. A good example is given by the Blasius *boundary layer* flow depicted in Figure 1.5 (right). This classification will affect the whole process of transition to turbulence in an important way (Chapter 6, §6.2).

A second distinction has to be made from the very existence of a global flow of matter entering the region of interest and leaving it at the outlet,



Fig. 1.5 Typical open flows. Left: Mixing layer created at the merging of two fluid veins with different velocities and maintained separated upstream by a splitting plate. Right: Boundary layer developing along a plate at some distance of its leading edge.



Fig. 1.6 Kármán vortex street emitted by a cylinder (courtesy P. Le Gal, IRPHE, Marseilles).

which makes implicit reference to a frame rigidly attached to the laboratory (obstacle, rigid wall). An important difference indeed appears between instabilities said to be *convective*, those that develop along the flow but are evacuated downstream and do not succeed in going upstream, and instabilities called *absolute* that are sufficiently intense to develop in spite of the global downstream transport. In the first case, the systems behaves as a *noise amplifier* and the result downstream essentially depends on the level of background fluctuations (residual turbulence). In the second case, one has rather to deal with a genuine *self-sustained oscillator* analogous to those studied in Chapter 2.

However, the situation is even more complicated by the fact that the intensity of instability mechanisms may vary in space. This is due to the fact that the velocity profile usually evolves downstream as a consequence of viscous dissipation that tends to smooth out the velocity gradients present at the entrance of the flow. Depending on the *local* intensity of the mechanism (linked to the magnitude of the shear) one can observe a change of the character of the mode usually from absolute to convective, hence the possibility of bifurcated states localized in a given region of space and usually called *global modes*. The wake of a blunt body inserted in an otherwise uniform stream is such an example (Figure 1.6). A recirculation takes place close to the obstacle, rendering the overall downstream transport locally sufficiently weak that regularly shed vortices can develop and stay attached to it (Kármán vortex street).

The understanding of this combined problem 'convective/absolute + local/global + linear/nonlinear' has made great progress recently but it requires sophisticated mathematical tools, especially in complex analysis, that would bring us far beyond the limited purpose of the present notes. The problem will just be evoked at a physically intuitive level in Chapter 6.

In the above presentation the instabilities were tacitly assumed to be linear and supercritical. It may however happen that the bifurcation is subcritical and that several different flow regimes coexist in some range of control parameter, the bifurcated state saturating only "far from" the base state as for the plane channel flow. The base state may also happen to remain stable against infinitesimal fluctuations so that the nontrivial (turbulent) regime cannot be reached by perturbation. Examples are the pipe Poiseuille flow and the plane Couette flow. Though these flows are known to be linearly stable for all Reynolds numbers, they are expected to become turbulent as the shear rate increases beyond all limit. The transition indeed happens and take a rather explosive turn: turbulent bursts developing intermittently from localized finite amplitude perturbations show up during the process, coexist with still laminar flow and then merge to fill the systems with fully developed turbulence. Figure. 1.7 displays such a turbulent spot in plane Couette flow. It shows that the pocket of turbulence is not structureless but, on the contrary, present a streamwise streaky pattern. This type of structure turns out to be omnipresent in turbulent wall flows. Their production and sustainment mechanisms are still the subject of current research (Chapter 6, $\S6.3.4$).



Fig. 1.7 Mature turbulent spot induced by a localized perturbation of finite amplitude in plane Couette flow (courtesy S. Bottin and O. Dauchot, GIT, CE Saclay).

1.3.4 Beyond the transition: fully developed turbulence

Problems considered up to now all relate to the steps of the cascade leading from a simple and regular base state to a complex and irregular flow still partly ordered but chaotic, that may not yet be called turbulent. It seems indeed important to understand the steps of the laminar-turbulent transition just sketched in a deterministic framework. The underlying aim is of course to control it as best as we can, delay it if it is harmful, or advance it if we need better mixing. However it turns out that beyond the transition we have to change our mind and take a fully statistical viewpoint. As a matter of fact, when we pull the considered system (flow) ever farther from equilibrium, more and more degrees of freedom become excited and it no longer makes sense to focus on their individual dynamics, like when passing from few-body systems well described by analytical mechanics to gases for which the thermodynamic approach is more appropriate.

This remark could lead to understand developed turbulence as a new macroscopic state of matter at a scale where the fluid would have transport properties very different from those of ordinary fluids in their laminar state where molecular chaos still control diffusion. Things are unfortunately less simple. The usual distinction between microscopic and macroscopic scales works well for thermodynamics because there is a wide gap between them, the precise reason for which the concept of material point (or fluid particle) makes sense. By contrast, in turbulence, relevant scales belong to a continuous range from the size of the flow domain to small scales. While the smallest eddies seem to evolve randomly like molecules in a gas, they are still coupled and, in fact, driven by the scales above them so that no decoupling is truly legitimate.

In Chapter 7, we will approach the theory of turbulence in a very preliminary way only, without trying to compete with numerous excellent books dealing with it, from conceptual problems to applications. Our limited aim will be to illustrate the idea of an *inertial cascade* (cf. Fig. 1.8) transferring energy from large scales where instability mechanisms generate the eddies, down to the smallest scales where viscosity successfully irons out the fluctuations. It is not difficult to catch an idea about this transfer by the advection term $\mathbf{v} \cdot \nabla \mathbf{v}$ of the Navier–Stokes equations. Considering an eddy motion locally described by a simple trigonometric line $v_x \sim \sin(kx)$, we get $\sin(kx)\partial_x \sin(kx) = k \sin(kx) \cos(kx) = \frac{1}{2}k \sin(2kx)$ that presents



Fig. 1.8 Illustration of the Kolmogorov cascade from large scales to small scales where fluctuating eddies are blurred by viscous dissipation.

itself as a source term in the evolution equation for a mode with a spatial scale half of the initial one. The result of this transfer is a repartition of energy according to the celebrated $k^{-5/3}$ Kolmogorov law that will be derived heuristically.

In a second part, we will turn to the concrete problem of predicting the lowest order statistics of specific turbulent flows, obtaining the so-called Reynolds averaged equations governing the mean flow, which immediately opens the problem of the closure of the statistical description. As a matter of fact, the averaging of primitive equations introduces new higher order statistical quantities called the *Reynolds stresses* that remains to be evaluated in one way or another. We will then exploit the already mentioned analogy between the kinetic theory of gases and the statistical theory of turbulence to introduce a disputable but heuristically valuable approach to turbulent flows resting on the concept of eddy diffusivity patterned on that of molecular diffusivity (see Figure 1.9). This will serve us to make a first evaluation of the average properties of a turbulent flow, taking the turbulent boundary layer as an example, and obtaining the classical Kármán logarithmic law. The main shortcoming of this approach derives from the absence of scale decoupling mentioned above but we will not do much more than mentioning it, suggesting further that progress can be obtained in concrete situations by numerical simulations, and observing that the latter always imply the crucial step of sub-grid-scale modeling, *i.e.* the modeling of smallest scales that cannot be explicitly accounted for in the numerics.



Fig. 1.9 Visualization of a turbulent flow downstream a grid. The fluid passing through a given mesh is seeded with a tracer that demonstrates turbulent diffusion in the flow. After a color picture by J.L. Balint, M. Ayrault and J.P. Schon (École Centrale de Lyon), courtesy M. Lesieur, in "La turbulence développée," La Recherche, n° 139 (1982), with permission.

1.4 Conclusion

Numerous books develop the topics to be touched upon. A partial list is given in the bibliography. At the risk of repetition, let us emphasize that we have tried to present a first approach to current problems, using a wide range of techniques, linear, nonlinear, deterministic, statistical, each with its own qualities and limitations, in order to lucidly face complicated situations encountered in our familiar environment, industrial or natural. Accordingly, in the concluding chapter we will evoke the problem of the Earth's climate as a concentration of the kind of topics examined at one or another moment, with special reference to the predictability problem and modeling issues involved in this nonlinear dynamical system with heterogenous space-time scales. Appendix A recalls some elementary and not so elementary results of linear algebra, while Appendix B works in roughly the same direction by giving rudiments of numerical simulation techniques expected to be of help in the understanding of complex processes (provided that we are able to build simplified models with valuable metaphoric value).

1.5 Exercises

1.5.1 Chemical reactions

Chemical kinetics is a field that has contributed much to the development of applied nonlinear dynamics in out-of-equilibrium systems. Here we consider a simplistic kinetic model introduced by Prigogine and Lefever in 1968, called the Bruxellator and accounting for a hypothetical reaction between two components A and B with two end products C and D, four steps and two intermediate compounds X and Y. Concentrations A and B of reactants A and B are the control parameters whereas those X and Y of the intermediate species X and Y are the variables. The global reaction:

 $A + B \rightarrow C + D$ can be decomposed into successive steps:

Assuming that the kinetic constants of each reaction is equal to 1, write down the equations for each step and show that the reaction is governed by

$$\dot{X} = A - (B+1)X + X^2Y, \qquad \dot{Y} = BX - X^2Y$$

This system will be further studied in Exercise 4.6.7.

Elements of chemical kinetics. Consider:

$$\sum n_i A_i \quad
ightarrow \sum n'_i A_i$$

an elementary step between reactants A_i with concentration A_i [if specie A_i is absent from the left (right) hand side then $n_i = 0$ $(n'_i = 0)$]. The reaction rate corresponding to this step measures the number of reacting collisions per unit time. A collision is the meeting at a single physical point of all the components on the l.h.s. producing all those of the r.h.s. in the proportions given by the stoichiometric coefficients. The probability for reactant A_i to be there is proportional to the concentration and, if one neglects the correlations between species, one expects the probability of the collision as a product of individual probabilities and thus a reaction rate $k \prod_i A_i^{n_i}$, involving only participants on the l.h.s. and where k is the corresponding rate constant. When such a reaction takes place, the number of molecules of component A_i varies by $n'_i - n_i$, so that one gets:

$$\dot{A}_i = (n_i'-n_i)k\prod_i A_i^{n_i} \ .$$

The total variation of the concentration of a given specie A_i for a compound reaction is the sum of the variations at each step.

An oscillatory reaction, discovered by Belousov in the fifties and further studied by Zhabotinsky in the sixties (BZ reaction) has been the subject of intense laboratory work. It corresponds to the oxydization of a organic reducer (malonic acid) by BrO_3^- ions with a redox couple (e.g. Ce^{3+}/Ce^{4+}) as a catalyst. It involves about 15 chemical species coupled by an equivalent number of intermediate reaction steps. Simplified models of this reaction have been devised with only 5 steps and 3 intermediate variables (Field et Noyes, 1974).

1.5.2 Prey-predator systems

Prey and predators interact in much the same way as the molecules considered in the previous exercise. The dynamics of the population X of some specie X is usually described in terms of an effective growth rate:

$$\dot{X} = \alpha_{\text{eff}} X \tag{1.21}$$

where the expression of α_{eff} accounts for the balance between birth and death, which depends on control parameters and the population itself.
1) Assume that the effective growth rate is simply proportional to the amount of available food and interpret the different terms appearing in $\alpha_{\text{eff}} = \alpha - \beta X$, where α and β are two positive constants. The corresponding evolution equation is usually called the (continuous-time) logistic equation. What is the meaning of quantity $X_* = \alpha/\beta$. Derive the solution X(t) of this 'limited growth' model starting with initial condition $X = X_0 > 0$ at t = 0 by explicit time integration for $X_0 < X_*$ and $X_0 > X_*$. Extract the long-term behavior $(t \to \infty)$. Observe also that the evolution of X here is the same as that of A^2 where A is governed by (1.19) for r > 0, apart from scale changes on X and t to be determined [see Figure 1.4 (left) that would correspond to the case $X_0 < X_*$].

2) In the model considered above, X is the population of a predator X and the available food is determined by the population Y of its prey Y, the food is thus not indefinitely available and the term $-\beta X$ is no longer necessary. A simple assumption is that the growth of the population is associated with a food consumption that is proportional to the presence of prey (the equivalent of a 'reactive collision' in chemistry) and that its decay is due to some natural death rate. Argue in favor of expression

$$lpha_{
m eff} = lpha' Y - lpha_0$$

for α_{eff} in (1.21). What should be the signs of the constants α' and α_0 . 3) In order to close the system we need to write down an equation for the effective growth rate of Y. By arguments similar to previous ones, justify the expression

$$\dot{Y} = \gamma Y - \delta X Y$$
.

This coupled system will be studied in Exercise 2.5.12.

Try to generalize the approach to describe more complicated prey-predator models, called Lotka–Volterra systems, by changing the assumptions, *e.g.* that the growth of the prey is also limited by competition for food.

1.5.3 Diffusion equation

We are interested in the decay of a solution to the diffusion equation [e.g. the heat equation (1.8)] in space dimension one:

$$\partial_t \vartheta = \kappa \partial_{xx} \vartheta, \tag{1.22}$$

starting from some initial condition $\vartheta_0(x)$.

1) Determine by substitution the relation between the growth rate s and the wavevector k of a periodic fluctuation taken in the form $\vartheta_k(x,t) = \overline{\vartheta} \exp(st) \exp(ikx)$ and derive its evolution as time goes on.

2) Adding boundary conditions $\vartheta(0,t) = \vartheta(\ell,t) = 0$, solve the evolution problem formally for a solution starting with an initial condition that can be expanded as a sine series: $\vartheta(x,0) = \sum_{n=1}^{\infty} \vartheta_n \sin(\pi n x/\ell)$. Extract its asymptotic solution in the limit $t \to \infty$.

3) Going back to the infinite medium, consider functions ϑ that are taken in the form

$$\vartheta(x,t) = \exp(\alpha w(x,t)). \tag{1.23}$$

Derive the partial differential equation governing w(x,t), and next that governing $v = \partial_x w$. Find the value of α bringing this equation in the form

$$\partial_t v + v \partial_x v = \kappa \partial_{xx} v \tag{1.24}$$

called the Burgers equation.⁵ The inverse of function change (1.23), called the Hopf–Cole transformation, offers an exact analytical way to linearize (1.24) and to put it back in the form (1.22).

⁵J.M. Burgers, *The nonlinear diffusion equation* (Reidel, 1974), see U. Frisch & J. Bec, "Burgulence," in [Lesieur *et al.* (2001)] for a recent review.

This page intentionally left blank

Chapter 2

First Steps in Nonlinear Dynamics

In this chapter we consider the dynamics of systems with a very small number of variables, as a prerequisite to the study of chaos, postponed to Chapter 4 after an examination of convection in Chapter 3, which will give us a physical motivation.

2.1 From Oscillators to Dynamical Systems

2.1.1 First definitions

Newtonian mechanics is the archetype of deterministic dynamical theories. Governed by an equation of the form

$$m\ddot{X} = F, \qquad (2.1)$$

where X represents the position of a particle with mass m submitted to a force F, it accounts for processes that are invariants under a change of the arrow of time.

A traditional example of linear system is the harmonic oscillator (Fig. 1.1, left, p. 3) that describes the motion of a mass attached to an ideal spring with a restoring force proportional to the extension F = -kX (Hooke's law). This elastic force, of internal origin, derives from a potential:

$$F=-rac{\partial \mathcal{V}}{\partial X} \quad ext{with} \quad \mathcal{V}=rac{1}{2}kX^2$$

depicted in Figure 2.1 (left). Here we have

$$m\ddot{X} + kX = 0. \tag{2.2}$$



Fig. 2.1 Left: Potential for the harmonic oscillator. Right: Vector field and trajectories in the phase plane, the plane of variables $X_1 \equiv X$ and $X_2 \equiv P$.

The evolution is uniquely determined when initial conditions are specified:

$$X = X^{(0)}$$
 and $X = V^{(0)}$ at $t = 0$.

(In the following, we will systematically use the superscript (0) to denote initial conditions.) The solution then reads:

$$X(t) = X^{(0)} \cos(\omega_0 t) + (V^{(0)}/\omega_0) \sin(\omega_0 t) \quad \text{with} \quad \omega_0^2 = k/m \,.$$

Right now, it turns out advantageous to substitute a geometrical description (Fig. 2.1, right) to this analytical description. So, let us consider *trajectories* in the *phase space* where coordinates are the *position* X and the *momentum*

$$P = mV = m\dot{X}. \tag{2.3}$$

Equation (2.2) now reads

$$\dot{P} = -kX. \tag{2.4}$$

One can trace back the origin of this representation from the need to turn a high order differential equation into a first order differential system. On general grounds, this is done by introducing the successive derivatives as intermediate variables. For example, equation:

$$\mathrm{d}^{n}X/\mathrm{d}t^{n} = \mathcal{F}\left(\mathrm{d}^{n-1}X/\mathrm{d}t^{n-1}X,\ldots,\ddot{X},\dot{X},X;t\right)$$
(2.5)

is reduced to a first order system by setting $X_1 \equiv X, X_2 \equiv \dot{X}, \ldots, X_n \equiv d^{n-1}X/dt^{n-1}$, which yields:

$$\dot{X}_1 = X_2, \quad \dot{X}_2 = X_3, \dots, \quad \dot{X}_n = \mathcal{F}(X_n, \dots, X_3, X_2, X_1; t).$$

This operation is preliminary to any numerical implementation in view of simulation (Appendix B).

There is a slight disadvantage to start from mechanics¹ for an introduction of basic concepts of the theory of dynamical systems since Newton's equations endow the *phase space* with a specific structure and give it an even number of dimensions, d = 2n. As seen above, this space is indeed constructed as the product of the configuration space (variables X_i , i = 1, ..., n and of the momentum space (variables $P_i, i = 1, ..., n$), each pair of conjugate variables (X_i, P_i) forming a degree of freedom. In the general case when the state of a system is specified using a supposedly sufficient number of variables, no longer grouped by pairs, we will call any of these state variable a 'degree of freedom'. Accordingly, the 'phase space' will then be the space where these variables "live" and the dimension of this space will just be the number of these variables. In agreement with the intuitive concept of determinism, the dimension d is also exactly the number of conditions necessary to specify any evolution uniquely. A way to avoid ambiguities would be to keep the terms 'phase space' and 'degrees of freedom' to mechanics in a strict sense and to use state variables and state space in all other cases, which we will not do since the risk of confusion is limited.

Anyway, let us consider the general case of a phase space X with dimension d spanned by d variables $\{X_1, \ldots, X_d\} \equiv X$. The evolution of the so-defined variables is governed by a system that symbolically reads

$$\dot{\mathbf{X}} = \mathcal{F}(\mathbf{X}; t) \,, \tag{2.6}$$

where \mathcal{F} is a set of *d* functions representing the components of a *vector* field defined on X that specifies the "velocity" of the point representing the system in its phase space.

When the properties of \mathcal{F} guarantee the existence and uniqueness of the solution to the initial value problem, in practice when the vector field \mathcal{F} is \mathcal{C}^1 (differentiable with continuous first derivative), one says that it defines a *flow* on phase space.

When t is explicitly absent from the definition of \mathcal{F} , the system is said to be *autonomous*, otherwise it is *forced*. In practice, among forced systems, only periodically forced systems will be of interest to us, *i.e.* systems such that $\mathcal{F}(\mathbf{X}; t + T) \equiv \mathcal{F}(\mathbf{X}; t)$ for some minimal time interval T called the *period*. Within this class one often distinguishes *parametric forcing* for

¹See later, §2.1.2, for a brief presentation of the formalism of analytical mechanics.

which the expression of \mathcal{F} changes in time, *e.g.* the parametric linear oscillator (Mathieu equation):

$$\ddot{X} + (1 + a\sin(\omega t))X = 0, \qquad T = 2\pi/\omega,$$

from *external forcing* where an otherwise autonomous system is submitted to a periodic force independent of its state, *i.e.*

$$\ddot{X}+X=f\sin(\omega t)$$
 .

Noisy systems can be understood as particular forced systems with a *random* forcing. In case of additive noise, this defines the so-called *Langevin* equation

$$\dot{\mathbf{X}} = \mathcal{F}(\mathbf{X}) + \boldsymbol{\Xi}(t) \,,$$

where $\Xi(t)$ is a random vector function. Quite different tools of statistical essence are then required which will not be introduced here since we want to stick to the deterministic point of view.

During its evolution, the system follows a phase space trajectory starting at $\mathbf{X}^{(0)}$ when $t = t^{(0)}$ and obtained by integration of (2.6):

$$\mathbf{X}(t) = \mathbf{X}^{(0)} + \int_{t^{(0)}}^{t} \mathcal{F}(\mathbf{X}(t'); t') \,\mathrm{d}t' \,. \tag{2.7}$$

The orbit is the set of points in X visited by the system in the course of a given trajectory. The description of a system's dynamics in terms of sets of orbits, called its *phase portrait*, as a function of its *control parameters*, is the field of *qualitative dynamics*.

Relation (2.7) allows us to define a map of X onto itself. Upon specifying an integration time τ , we get a *time*- τ map

$$\mathbf{X}(t+ au)\equiv \mathbf{\Phi}_{ au}ig(\mathbf{X}(t)ig)=\mathbf{X}(t)+\int_{t}^{t+ au}\mathrm{d}t'\, \mathcal{F}ig(\mathbf{X}(t');t'ig),$$

and, starting with an initial condition $\mathbf{X}^{(0)}$, we obtain a discrete sampling of the trajectory, $\mathbf{X}_0 \equiv \mathbf{X}^{(0)}$, $\mathbf{X}_1 = \mathbf{\Phi}_{\tau}(\mathbf{X}_0)$, $\mathbf{X}_2 = \mathbf{\Phi}_{\tau}(\mathbf{X}_1)$,..., $\mathbf{X}_{k+1} = \mathbf{\Phi}_{\tau}(\mathbf{X}_k)$...

The time- τ map is really interesting only when τ corresponds to some characteristics of the system. The most important case corresponds to periodic forcing with period $T = \tau$, in which case Φ_T performs a *stroboscopic analysis* of the dynamics, *i.e.* takes pictures of the system at the period of a strobe signal in phase with the forcing. Later we will see another related way to arrive at such *discrete-time* systems as already mentioned in

Chapter 1, p. 4. The concepts of trajectories and phase portraits transpose immediately to systems written as first-order iterations $\mathbf{X}_{k+1} = \mathbf{\Phi}(\mathbf{X}_k; k)$.

From a practical viewpoint the determination of the trajectory issued from some initial condition $\mathbf{X}^{(0)}$ by numerical integration can be viewed as resulting from the iteration of a map $\boldsymbol{\Phi}_{\tau}$ integrating the field $\boldsymbol{\mathcal{F}}$ over a time interval $\tau = \Delta t$ chosen from accuracy considerations specific to the numerical scheme (cf. Appendix B).

For the moment, let us consider autonomous systems and especially the simple case of the harmonic oscillator. Its phase space is the plane (X, P) isomorphic to \mathbb{R}^2 . The right part of Figure 2.1 displays the corresponding vector field.² In this representation, trajectories follow elliptic orbits.

On general grounds it turns out useful to scale the variables as much as possible in order to cast the system into its most universal form, hiding its specificities inside the details of the variable change. Here such a transformation simply yields

$$\dot{X}_1 = X_2, \qquad \dot{X}_2 = -X_1.$$
 (2.8)

Reversibility is one of the fundamental characteristics of ideal mechanical systems (*i.e.* without friction). The change ' $t \mapsto -t$ ' indeed leaves the Newton equations invariant. This property is associated to energy conservation. Defining $E = \frac{1}{2}X_1^2 + \frac{1}{2}X_2^2$ and computing dE/dt using (2.8) one can check that E is a constant of motion. Things are different for a damped system that *dissipates* its energy. At a linear stage, the introduction of a *viscous friction* proportional to the rate of change of the variable leads to

$$\ddot{X} + 2\eta \dot{X} + X = 0, \qquad (2.9)$$

where $\eta > 0$ measures the strength of the damping. Here the change $t \mapsto t/\omega_0$ has been performed in order to normalize the period of the ideal oscillator to 2π .

Trajectories are obtained in parametric form as

$$X(t) = \bar{X} \exp(-\eta t) \cos(\omega t - \varphi),$$

where \bar{X} and φ can be computed from the initial conditions $X^{(0)}$ and $\dot{X}^{(0)}$ at t = 0 by identification, and where $\omega = (1 - \eta^2)^{1/2}$ corresponds to the angular frequency of a damped oscillation only when the friction is

²Here it has been obtained using the MATLAB macro quiver, explicitly:

X=[-0.8:0.1:0.8]; U=ones(1,size(X,2)); P=X'; FX=P*U; FP=-U'*X; quiver(X,P,FX,FP).



Fig. 2.2 Vector field of the non-ideal linear oscillator in its phase plane (X_1, X_2) for $\eta = 0.5$. Left: Damped oscillator $(\eta > 0)$. Right: Driven oscillator $(\eta < 0)$.

sufficiently weak, *i.e.* when $\eta < 1$ (the over-damped case $\eta > 1$ will be considered as an exercise). In the phase plane, equation (2.9) reads

$$\dot{X}_1 = X_2, \qquad \dot{X}_2 = -2\eta X_2 - X_1, \qquad (2.10)$$

and the orbits now take the aspect of spirals converging towards the origin, see Figure 2.2 (left) that displays the corresponding vector field. The instantaneous dissipation rate of the total energy, still defined as $E = \frac{1}{2}(X_1^2 + X_2^2)$, is now given by $dE/dt = -2\eta X_2^2 < 0$.

For an excited system that would receive energy from the exterior world by viscous *driving*, thus governed by (2.9) but with $\eta < 0$, the integration of the vector field would give diverging spiral trajectories easily imagined from the vector field depicted in Figure 2.2 (right).

In fact, when defining a state physically, one never considers a mathematical point in phase space but rather a small, physically infinitesimal domain around this point. This remark suggests we focus on the future of sets of systems that at a given time belong to volume elements in phase space (surface elements in the specific two-dimensional case of interest here). For a general dynamical system (2.6) defined on a *d*-dimensional phase space X, one then shows that the local evolution rate of the volume Δ of an infinitesimal domain around some point **X** is given by the divergence of the vector field \mathcal{F} computed at this point (cf. Exercise 2.5.1):

$$\frac{1}{\Delta} \frac{\mathrm{d}\Delta}{\mathrm{d}t} = \operatorname{div} \mathcal{F} = \sum_{j=1}^{d} \frac{\partial \mathcal{F}_j}{\partial X_j}.$$
(2.11)

For an ideal oscillator (2.8), one immediately obtains that the divergence is



Fig. 2.3 Evolution of volumes in phase space. Left: Conservative system. Right: Dissipative system.

zero: the surface of a small surface element is thus conserved (Fig. 2.3, left). This property can be generalized to all frictionless mechanical systems, that are called *conservative* for that reason. The same calculation for the damped oscillator leads to div $\mathcal{F} = -2\eta < 0$, thus to an indefinite erosion of areas (Fig. 2.3, right). This reduction is characteristic of *dissipative* systems whose permanent regimes, asymptotic states at the limit $t \to \infty$ after the damping out of transients, are described by *attractors*.

As suggested above, the phase space volumes give a measure of the number of accessible states. Asymptotically the damped oscillator always ends at its rest position $X_1 = X_2 \equiv 0$, a single state whatever the initial energy. The attractor is here a *fixed point* in phase space, 'fixed point' because the orbit of the trajectory starting there is just reduced to that point. By anticipation, we can say that it is a *stable* fixed point since trajectories starting in its vicinity converge to it as t tends to infinity. By contrast, the fixed point at the origin of the phase space associated to an driven oscillator in not an attractor but a *repellor*, it is unstable and trajectories move apart from it.

General systems are not conservative. They can be fueled in energy in some regions of their phase space while dissipating it in other regions. This feature, that cannot be achieved in the framework of linear systems with constant coefficients (sign of div \mathcal{F} fixed once for all), will turn out to be essential to the existence of self-sustained oscillations and more complicated time behavior.

2.1.2 Formalism of analytical mechanics

In order to ease the solution to some exercises, we give here an introduction to the analytical formalism of classical mechanics that allows one to pass from the Newton equations (second order in time) to the Hamilton equations (first order in time) giving to the intermediate variables so-introduced their status of conjugate momenta to the generalized coordinates.

For a system of n_N (subscript 'N' for Newton) material points with masses m_i and positions $\mathbf{X}_i \equiv (x_i, y_i, z_i)$ submitted to forces \mathbf{f}_i deriving from a potential $\mathcal{V}({\mathbf{X}_i}; t)$, the Newton equations read:

$$m_i \ddot{\mathbf{X}}_i = \mathbf{f}_i = -\frac{\partial \mathcal{V}}{\partial \mathbf{X}_i}, \quad (i = 1, \dots, n_N).$$
 (2.12)

The kinetic energy is defined by:

$$\mathcal{T} = rac{1}{2}\sum_{i=1}^{n_{\mathrm{N}}}m_i \left(\dot{\mathbf{X}}_i
ight)^2,$$

and one easily checks that the total energy

$$E = T + V$$

is conserved. But this so-called Newtonian formulation makes the treatment of systems with constraints somehow awkward. The Lagrangian formalism answers this problem by introducing $n_{\rm L}$ ('L' for Lagrange) generalized coordinates.³ Let

$$\mathbf{q} \equiv \{q_j; j = 1, \dots, n_{\rm L}\}, \quad q_j = q_j(\{\mathbf{X}_i; i = 1, \dots, n_{\rm N}\}; t), \qquad (2.13)$$

be the change of variables from the X_i s to the q_j s expressing the constraints. The Lagrangian is then defined by

$$\mathcal{L} = \mathcal{L}(\mathbf{q}, \dot{\mathbf{q}}) = \mathcal{T} - \mathcal{V}\,,$$

so that (2.12) can be rewritten as a set of Lagrange equations:

$$rac{\mathrm{d}}{\mathrm{d}t}rac{\partial\mathcal{L}}{\partial(\dot{q}_j)}-rac{\partial\mathcal{L}}{\partial q_j}=0\,,\qquad j=1,\ldots,n_\mathrm{L}\,.$$

 $^{{}^{3}}n_{\rm L} < n_{\rm N}$ and sometimes $\ll n_{\rm N}$, think of a solid body with many $(n_{\rm N})$ rigidly linked particles and characterized just by the position of its center of mass (three coordinates) and its orientation (three Euler angles).

At this stage, the system is still governed by second order differential equations. One then defines the momenta $\mathbf{p} \equiv \{p_j\}$ conjugated to the coordinates $\mathbf{q} \equiv \{q_j\}$ by:

$$p_j \equiv \frac{\partial \mathcal{L}}{\partial(\dot{q}_j)},$$
 (2.15)

and the Hamiltonian by:

$$\mathcal{H}(\mathbf{q},\mathbf{p}) = \sum_{j} p_{j} \dot{q}_{j} - \mathcal{L}. \qquad (2.16)$$

The dynamical equations (2.14) then turn into the Hamilton equations:

$$\dot{q}_j = \frac{\partial \mathcal{H}}{\partial p_j}, \qquad \dot{p}_j = -\frac{\partial \mathcal{H}}{\partial q_j}, \qquad j = 1, \dots, n_{\rm L}.$$
 (2.17)

The phase space X is the Cartesian product of the configuration space with coordinates q_j and the momentum space with coordinates p_j , thus with dimension $d = 2n_{\rm L}$. The vector field governing the dynamics then reads

$$\mathcal{F}_{q_j} \equiv \frac{\partial \mathcal{H}}{\partial p_j} \quad \text{and} \quad \mathcal{F}_{p_j} \equiv -\frac{\partial \mathcal{H}}{\partial q_j}, \quad (2.18)$$

expressions that provide it with a so-called 'symplectic' structure insuring the conservation of phase space volumes automatically (div $\mathcal{F} \equiv 0$, the Liouville theorem).

2.1.3 Gradient systems

Above, we spoke of 'forces deriving from a potential' in a strictly mechanical framework. Unfortunately, there is a risk of confusion when using the word 'derive' and 'potential' without care, because in some branches of dynamical systems theory, they have a somewhat different meaning that now warrants specification.

To do so, let us first consider an autonomous systems with a single real variable X:

$$\dot{X} = \mathcal{F}(X) \,. \tag{2.19}$$

Defining \mathcal{G} from \mathcal{F} by

$$\mathcal{G}(X) = -\int \mathcal{F}(X) \,\mathrm{d}X\,,$$

we can rewrite (2.19) as:

$$\dot{X} = -\mathrm{d}\mathcal{G}/\mathrm{d}X\,,\qquad(2.20)$$

where \mathcal{G} presents itself as a 'potential' from which the vector field \mathcal{F} can be 'derived'. In higher dimensions, the natural extension of (2.20) for a function of several variables $\mathcal{G}(X_1, \ldots, X_d)$ through

$$\dot{X}_j = \mathcal{F}_j(X_1, \dots, X_d) = -\partial \mathcal{G}/\partial X_j, \quad j = 1, \dots, d,$$
 (2.21)

defines a large class of systems called *gradient flows*. It is also said that such systems are 'relaxational'. The origin of the latter term is to be found in the remark that (2.20) immediately leads to \mathcal{G} being a monotonicly decreasing function of time:

$$\dot{\mathcal{G}} = \left(\mathrm{d}\mathcal{G}/\mathrm{d}X \right) \dot{X} = -\left(\mathrm{d}\mathcal{G}/\mathrm{d}X \right)^2 \le 0 \,. \tag{2.22}$$

Accordingly, the system evolves from almost all initial conditions so as to 'relax' toward one of the local minima of \mathcal{G} where its stops asymptotically. In the *d*-dimensional case, the vector field \mathcal{F} is seen from (2.21) to be everywhere perpendicular to the level curves of \mathcal{G} so that:

$$\dot{\mathcal{G}} = \sum_{j} \left(\partial \mathcal{G} / \partial X_{j} \right) \dot{X}_{j} = -\sum_{j} \left(\partial \mathcal{G} / \partial X_{j} \right)^{2} \leq 0,$$
 (2.23)

which again expresses the 'relaxation' toward one of its local minima.

Figure 2.4 displays the level lines of a two dimensional potential⁴ $\mathcal{G} = -aX_1 - \frac{1}{2}(bX_1^2 + cX_2^2) + \frac{1}{4}(X_1^2 + X_2^2)^2$, with a = 3/2, b = 13/4, c = 5/4. The absolute minimum $\mathcal{G} = -5.5$ is at $\mathbf{M}_1 = (2,0)$. There is a relative minimum $\mathcal{G} \simeq -0.14$ at $\mathbf{M}_2 = (-1.5,0)$, a relative maximum $\mathcal{G} \simeq 0.36$ at $\mathbf{M}_3 = (0.5,0)$, while points \mathbf{S}_1 and \mathbf{S}_2 at $(-3/4, \pm 0.83)$, later called *saddle points*, belong to the level line $\mathcal{G} = 0.171875...$

In mathematics, gradient flows appear in the theory of elementary *catastrophes* (see, *e.g.* [Poston and Stewart (1978)]). In thermodynamics, they offer a good framework for the Landau theory of *phase transitions*, see later the remark on p. 127 and also [Stanley (1988)].

Now, if $\dot{\mathbf{X}} = \mathcal{F}(\mathbf{X})$ derives from a potential \mathcal{G} in the sense of (2.21), the components of \mathcal{F} fulfill the relations:

$$\partial \mathcal{F}_j / \partial X_{j'} = \partial \mathcal{F}_{j'} / \partial X_j, \qquad \forall j, j',$$
 (2.24)

⁴For more detail, see P. M. & L. Tuckerman, "Phenomenological modeling of the first bifurcations of the spherical Couette flow," J. Physique **48** (1987) 1461–1469.



Fig. 2.4 For a gradient flow, the vector field \mathcal{F} is everywhere perpendicular to the level curves of the potential \mathcal{G} from which it derives.

as results from the Schwartz identity:

$$\partial^2 \mathcal{G} / \partial X_j \partial X_{j'} = \partial^2 \mathcal{G} / \partial X_{j'} \partial X_j.$$

Conditions (2.24) are consequently necessary to the existence of a potential. In the general case, the components of the vector field \mathcal{F} do not fulfill such conditions and one can expect an evolution that is richer than a "simple" relaxation towards a local minimum of some putative \mathcal{G} . This is of course the case of ideal mechanical systems for which the total energy is a constant of the motion so that the vector field given by the Hamilton equations is everywhere parallel to the surfaces of constant total energy in phase space (Figure 2.1, right). For profound reasons to be discussed later, we will have to wait for the study of three-dimensional systems before any dynamics more complicated than a relaxation towards fixed points or periodic oscillations can be observed.

2.2 Stability and Linear Dynamics

The first piece of information of interest about the regime attained by a given system under specific conditions relates to its stability, *i.e.* the way it reaches the state and responds to perturbations. In practice one can only

exceptionally determine the response to perturbations of arbitrary amplitude. Except for Lyapunov functions generalizing the potentials introduced in Section 2.1.3 (see also Exercise 2.5.2), we have no tools to attack the problem in full generality and we must restrict ourselves to a study of the evolution of infinitesimal perturbations, for which all the resources of *linear* analysis are available.

2.2.1 Formulation of the linear stability problem

In what follows, we are mainly interested in specific time-independent states of autonomous systems:

$$\dot{\mathbf{X}} = \boldsymbol{\mathcal{F}}(\mathbf{X}) \,. \tag{2.25}$$

These states are thus solutions to:

$$\mathcal{F}(\mathbf{X}_{\mathbf{f}}) = 0 \tag{2.26}$$

and are represented by fixed points in phase space, hence the subscript 'f' serving to denote them. Equation (2.26) is the formal writing of a system of d nonlinear equations with d unknowns that has a discrete and finite set of solutions in general.

In out-of-equilibrium macroscopic systems, especially continuous media, it is natural to first consider the solution that belongs to the *thermodynamic branch*, defined as the branch of solutions that can be followed from thermodynamic equilibrium by continuity, but what will be said also holds for any other time-independent solution, even if it is more difficult to obtain.

Let X_f be the fixed point of interest. Inserting $X = X_f + X'$ in (2.25), one expands that equation in powers of X'. Noting that order 0 is identically fulfilled by the fixed point condition and keeping only first order terms, one gets:

$$\frac{\mathrm{d}}{\mathrm{d}t}\mathbf{X}' = \mathcal{L}\mathbf{X}'\,.\tag{2.27}$$

The (Jacobian) operator \mathcal{L} resulting from this linearization is represented by a matrix with elements

$$l_{ij} = \partial \mathcal{F}_i / \partial X_j \big|_{\mathbf{X}_f},$$

all partial derivatives being evaluated at the fixed point \mathbf{X}_{f} as indicated by the notation. In this context, the linear stability problem amounts to an integration of equation (2.27) and thus to the evaluation of the action of operator $\exp(t\mathcal{L})$ on some initial condition $\mathbf{X}'^{(0)}$. Infinitesimal perturbations are said to live in the *tangent space* at the fixed point. Their evolution is governed by the *tangent dynamics* (2.27).

A simple way to justify the introduction of the exponential of operator \mathcal{L} consists in considering the scalar equation $\dot{X} = aX$ with $X(0) = X^{(0)}$, in writing recursively

$$\begin{aligned} X(t) &= X^{(0)} + \int_0^t a X(t') dt' \\ &= X^{(0)} + \int_0^t \left[X^{(0)} + \int_0^{t'} a X(t'') dt'' \right] dt' = \dots, \end{aligned}$$

and in integrating explicitly all what can be integrated. This yields:

$$X(t) = X^{(0)} + atX^{(0)} + \frac{1}{2}a^2t^2X^{(0)} + \dots + \frac{1}{n!}a^nt^nX^{(0)} + \dots = \exp(at)X^{(0)}.$$

The extension of this approach to solving (2.27) involves what is precisely defined as the exponential $\exp(t\mathcal{L})$, *i.e.* the limit of a power series. In general, the solution of (2.27) rests on turning \mathcal{L} to its diagonal form, or more precisely to its Jordan normal form. See Appendix A, §A.2, for a reminder.

2.2.2 Two-dimensional linear systems

We first turn to the case of two variables since it contains the essentials of nontrivial aspects of the problem and allows us to introduce the core of the terminology. The extension to dimension d will be alluded to in the next subsection. Thus consider the linear dynamical system:

$$\dot{X}_1 = a_{11}X_1 + a_{12}X_2, \qquad (2.28)$$

$$\dot{X}_2 = a_{21}X_1 + a_{22}X_2, \qquad (2.29)$$

with initial condition, $X_j = X_j^{(0)}$, j = 1, 2, at t = 0 as a generalization of the harmonic oscillator studied previously. Since this system can be studied for itself and not as a tangent problem, we have dropped the primes indicating that its variables originally measured the departures from the fixed point of some primitive nonlinear system and forgotten that the coefficients derive from the evaluation of some Jacobian operator \mathcal{L} by changing the notations to anonymous *as.* Solutions to (2.28, 2.29) are searched in the form

$$X_i = \bar{X}_i \exp(st).$$

By mere substitution, this leads to an *eigenvalue problem*:

$$sar{X}_1 = a_{11}ar{X}_1 + a_{12}ar{X}_2 , \ sar{X}_2 = a_{21}ar{X}_1 + a_{22}ar{X}_2 ,$$

that has nontrivial solutions $\bar{\mathbf{X}} \neq 0$ only if the the growth rate s satisfies the compatibility condition:

$$s^{2} - (a_{11} + a_{22}) s + a_{11}a_{22} - a_{12}a_{21} = 0, \qquad (2.30)$$

also called the *characteristic equation*. In full generality, this quadratic equation has two roots, distinct or not, real or complex conjugate (see Appendix A, Exercise A.4, p. 348).

Eigen-solutions corresponding to eigenvalues either positive or complex with positive real part depart exponentially fast from the origin. In the negative case they converge to it.

The general solution is a superposition of eigen-solutions. Hence when both roots are negative or complex with negative real parts, the origin is said to be *linearly stable* whereas it is sufficient that one of the roots be real and positive or complex with positive real part to render the origin instable, which immediately yields a universal classification.

2.2.2.1 Two real distinct roots

Without changing notations, in the eigen-basis we have:

$$\dot{X}_j = s_j X_j \qquad \Rightarrow \quad X_j(t) = X_j^{(0)} \exp(s_j t)\,, \qquad j=1,2\,,$$

and we can distinguish two sub-cases:

i) The two roots have the same sign $(s_1s_2 > 0)$: the origin is called a node, stable when s_1 and s_2 are negative, unstable in the opposite case. Orbits around the origin have a parabolic shape obtained by elimination of t between the different components of the solution, here between $X_1(t) = X_1^{(0)} \exp(s_1 t)$ and $X_2(t) = X_2^{(0)} \exp(s_2 t)$, which yields $X_2/X_2^{(0)} = (X_1/X_1^{(0)})^{s_2/s_1}$ with a positive exponent s_2/s_1 . They open in the direction of X_1 or X_2 according to the relative magnitude of s_1 and s_2 , see Figure 2.5(a).

ii) The two roots have opposite signs $(s_1s_2 < 0)$: the origin is a *saddle*, stable along the direction of the negative root and unstable along the other one and thus always unstable. Orbits have a hyperbolic shape as depicted in Figure 2.5(b).



Fig. 2.5 Real roots s_1 and s_2 . (a) Stable Node, $s_2 < s_1 < 0$. (b) Saddle, $s_2 < 0 < s_1$.

2.2.2.2 A pair of complex conjugate roots

Let us write the roots as $s = \sigma \pm i\omega$. In such a case there are no real eigenvectors but by a linear variable change one can cast the system into the form

$$\dot{X}_1 = \sigma X_1 + \omega X_2 \,, \tag{2.31}$$

$$\dot{X}_2 = -\omega X_1 + \sigma X_2 \,, \tag{2.32}$$

which, by integration, leads to

$$X_1(t) = \exp(\sigma t) \left[X_1^{(0)} \cos(\omega t) + X_2^{(0)} \sin(\omega t) \right], \qquad (2.33)$$

$$X_2(t) = \exp(\sigma t) \left[-X_1^{(0)} \sin(\omega t) + X_2^{(0)} \cos(\omega t) \right].$$
 (2.34)

Trajectories thus spiral around the origin that is called a *spiral point*, or a *focus*, stable or unstable according to the sign of the real part σ , see Figure 2.6(a).

In the marginal case $\sigma = 0$, the fixed point is called a *center* or an *elliptic* point, Figure 2.6(b). This is of course the case of the harmonic oscillator considered previously. One immediately observes that this property is not robust. The introduction of a damping or a driving, as weak as they could be, converts the center into a focus, stable or unstable [cf. (2.9): eigenvalues $s = -\eta \pm i(1-\eta^2)^{1/2}$, pure imaginary only for $\eta = 0$]. In such a point of the space of control parameters, the system is said to be structurally unstable.



Fig. 2.6 Complex roots $s = \sigma \pm i\omega$. (a) Focus. (b) Center ($\sigma = 0$).

2.2.2.3 Double roots

The last case is when the characteristic equation has (real) double roots, which happens when $(a_{11} - a_{22})^2 + 4a_{12}a_{21} = 0$. Usually, the system has only one eigen-direction so that is cannot be cast into diagonal form by a linear variable change but only into what is called its *Jordan normal form*, here:

$$\dot{X}_1 = sX_1 + X_2 \,, \tag{2.35}$$

$$\dot{X}_2 = sX_2.$$
 (2.36)

Integrating (2.36) one gets $X_2(t) = X_2^{(0)} \exp(st)$, which is further inserted in (2.35) that now reads:

$$\dot{X}_1 - sX_1 = X_2^{(0)} \exp(st),$$
 (2.37)

which point out the resonant character of the right hand side since it evolves at the same rate as that defined by the left hand side. Equation (2.37) is easily integrated by the Lagrange method of variation of the constant: solving the homogeneous problem, one gets $X_1 = \tilde{X} \exp(st)$, where \tilde{X} is an integration constant. Assuming that this "constant" is now a function of time, $\tilde{X}(t)$ and introducing this expression into (2.37) one gets

$$\frac{\mathrm{d}}{\mathrm{d}t}\tilde{X}=X_2^{(0)},$$

which leads to

$$\tilde{X} = X_2^{(0)} t + X_1^{(0)},$$



Fig. 2.7 Double real root. (a) Improper node. (b) Star (diagonalizable case).

and thus to

$$X_1(t) = \left(X_1^{(0)} + X_2^{(0)}t\right) \exp(st)$$
(2.38)

where the second, sub-dominant, term inside the parentheses is called a $secular \ term.^5$

This fixed point is called an *improper node*. The corresponding phase portrait is displayed in Figure 2.7(a) which shows that it is indeed intermediate between a node and a focus.

When the eigenvalue is double but the system still diagonalizable, which occurs when the operator \mathcal{L} is symmetric, all directions in the plane (X_1, X_2) are eigen-directions, which is another limit case of a node called a *star*, see Figure 2.7(b).

2.2.3 Stability of a time-independent regime

Let us come back to a *d*-dimensional system linearized around one of its fixed points X_f . At the linear stage, the general solution of the perturbation problem can be expressed as a *superposition* of solutions corresponding to each eigenvalue s_j , with dominant exponential behavior $\exp(s_j t)$. These eigenvalues can be ordered by decreasing value of their real part, and while in the short term one may observe a complicated evolution of generic perturbations made of arbitrary superpositions of eigenmodes, due to linear

 $^{^{5}}$ The word originates from the Nineteenth Century studies of the period of planets by perturbation methods, when corrections to the relative position of the planets were found to show up on time scales of the order of hundreds of years.



Fig. 2.8 Stability of the fixed point \mathbf{X}_{f} of a hypothetical 5-dimensional system $\dot{\mathbf{X}} = \mathcal{F}(\mathbf{X})$: The operator \mathcal{L} obtained by linearization of \mathcal{F} around \mathbf{X}_{f} has 5 real or complex eigenvalues and \mathbf{X}_{f} can be linearly stable (a), unstable against a stationary mode (b), unstable against an oscillatory mode (c). Instability against several modes is also possible but one is usually interested in the first instability of a state, *i.e.* the transition from (a) to (b) or (c), which can be achieved by varying just one control parameter in a sufficiently limited range.

interferences between them (cf. A.3, p. 343), in the long term only the contribution corresponding to the eigenvalue with largest real part survives.

The considered base state associated to the fixed point is thus linearly stable if all the eigenvalues have negative real parts, Figure 2.8(a), and linearly unstable if at least one of the eigenvalues has positive real part, Figure 2.8(b,c). When the unstable root is real, the corresponding unstable mode is said to be *stationary*, Figure 2.8(b), and when it is complex, one speaks of *oscillatory* mode, Figure 2.8(c).

Here we have made use of the concept of asymptotic stability, the amplitude of the perturbation tending to zero as t increases. When the eigenvalue is zero or has zero real part, the corresponding mode is *marginal* or *neutral*. Linear theory then does not allow us to draw any conclusion about the stability of the base state and nonlinearities have to be taken into account. Methods with a more global flavor have to be used, in the spirit of the energy method introduced in Exercise 2.5.2. In the long time limit, perturbations evolve more slowly than exponentially, generally as some power of time. They may relax, in which case the base state is still asymptotically stable, or grow, in which case it is unstable. This type of situation can be dealt with using the concept of *orbital stability* which is weaker than that of asymptotic stability since it only requires that the perturbed state can permanently depart, but in a controlled way, from the base state, thus remaining in its vicinity. A typical example is that of an elliptic point in a mechanical system, locally equivalent to a harmonic oscillator: trajectories circle around the fixed point without approaching it as would be the case is the eigenvalue had negative real part and the fixed point be asymptotically

stable. So, an elliptic point with purely imaginary eigenvalue, marginal at a linear level, can be orbitally stable or unstable at the nonlinear stage.

The persistent occurrence of purely imaginary eigenvalues often results from symmetry conditions and especially the invariance of the dynamics upon time reversal, which is characteristic of mechanics. One can indeed observe that reversibility implies an exchange ' $s \leftrightarrow -s$ ' upon the change ' $t \mapsto -t$ ', and thus either to complex conjugation in case of purely imaginary roots or an exchange 'stable \leftrightarrow unstable' within a pair of real eigenvalues.⁶

Otherwise the presence of neutral modes must be considered as accidental: since the system usually depends on control parameters, this circumstance only occurs at specific locations in the parameter space. At such points the system is said to be *structurally unstable* since a slight modification of its definition can turn the considered fixed point from stable to unstable. It is ready for a *bifurcation* associated with a qualitative change of its phase portrait.

The breadth of possibilities increases with the dimension of the system but the terminology introduced for two-dimensional systems can be extended straightforwardly. For example, one still speaks of a node when all the eigenvalues are real and have the same sign. In the same way, a fixed point can be called a saddle-focus if it has a pair of complex conjugate eigenvalues with real parts of one sign and its other eigenvalues real with the opposite sign. The case of a three-dimensional system is considered in Exercise 2.5.3.

2.3 Two-dimensional Nonlinear Systems

In this section we study the dynamics of autonomous nonlinear systems that evolve in a two-dimensional phase space. In a first instance, we take an essentially qualitative view point and use what precedes to draw phase portraits. Then we attack the problem of the explicit quantitative determination of the period of oscillators, presenting several methods in a computational perspective that is however indispensable to the classical nonlinear culture.

⁶The existence of quadruples $\pm \sigma \pm i\omega$ is not ruled out in the general case, but the symmetry does not change the fact that the system is unstable since, within a quadruple, there is always a pair of roots with positive real parts, see Exercise 2.5.4.

2.3.1 Two examples of oscillators

2.3.1.1 The rigid pendulum

The first classical example of nonlinear oscillator that we consider is the simple *rigid pendulum* already introduced in Section 1.1, see Figure 1.1 (right) on p. 3. The potential energy from which the external gravity force can be derived, $\mathcal{V}(\theta) = mg(1 - \cos\theta)$, is illustrated in Figure 2.9 (left). After a convenient choice of the time unit (the angular frequency of small oscillations being given by $\omega_0^2 = g/l$, one performs the change $t \mapsto t/\omega_0$), the evolution equation can be written as a two-dimensional first order system:

$$\dot{\theta} = \varphi, \qquad \dot{\varphi} = -\sin\theta.$$
 (2.39)

The corresponding phase space $\{\theta; \varphi\}$ is a cylinder $\mathbb{S}^1 \times \mathbb{R}$, where \mathbb{S}^1 is the unit circle (the one-dimensional sphere, hence the letter 'S') parameterized by the cyclic variable θ , *i.e.* $\theta + 2\pi \equiv \theta$ (the interval $[-\pi, +\pi[$ is made periodic by identifying $-\pi$ with $+\pi$). On the other hand $\varphi \in \mathbb{R}$ can take its values from $-\infty$ to $+\infty$. This cylinder being open along the generatrix $\theta = \pi$, one obtains a reduced representation of the phase space as an band of width 2π in the θ direction and infinite length in the φ direction, with identification of the sides at $\theta = \pm \pi$.

As already noticed, it is here the geometrical constraint fixing the distance of the mass to the rotation axis that generates the nonlinearity. The global character of this constraint is reflected in the topological structure



Fig. 2.9 Left: Potential energy of the rigid pendulum. Right: Phase portrait in reduced representation with indication of bounded states, the separatrix, and passing orbits.

of the phase space. Nonlinearities reveal themselves when large amplitude motions are considered, whereas linearization remains legitimate close to the origin. Classically, the distinction is made between *passing trajectories* at sufficiently high energy and *bounded trajectories* at low energy, see Figure 2.9 (right).

Fixed points of the system are given by $\varphi = 0$ and $\theta = k\pi$, $k \in \mathbb{N}$. The study of the dynamics close to these points is a straightforward application of the linear approach developed in previous sections. So, it appears that point ($\theta = 0, \varphi = 0$) is a center with eigenvalues $\pm i$ and corresponds to the small-oscillation regime. By contrast, the point ($\theta = \pm \pi, \varphi = 0$) is a saddle with eigenvalues ± 1 . Specific trajectories called *separatrices* link one of these points to the other, thus separating the domain of bounded orbits from that of passing orbits.

Breaking the Hamiltonian character of the dynamics, we now consider the effect of viscous friction. The second equation of (2.39) is therefore completed by a term proportional to the angular velocity $\varphi = \dot{\theta}$:

$$\dot{\varphi} = -\eta\varphi - \sin\theta \,.$$

This case is further illustrated in Figure 2.10 using an extended representation that no longer takes advantage of the limitation to the periodized interval $[-\pi, \pi]$. As a matter of fact, this representation makes it easier to understand how the pendulum returns to its rest position after a certain number of complete turns around the axis, a number that depends on the initial energy.



Fig. 2.10 Damped pendulum. Left: Decay of a high energy trajectory. Right: Phase portrait in extended representation.

It can be observed that the centers persist but are converted into stable spiral points. On the other hand unstable points at $\theta = \pm (2k + 1)\pi$ do not change their nature but trajectories that emerge from them along their unstable direction miss the next point and spiral towards the foci, whereas trajectories that arrive to them along their stable direction need a slightly larger energy than in the conservative case, as expected. Other similar systems are proposed to the study in Section 2.5.

2.3.1.2 Van der Pol oscillator

Examples of oscillatory processes are numerous in fields other than mechanics, from electronics (the case we consider now) to ecology (prey-predator systems, Exercise 2.5.12) or economics (expansion-recession cycles, see [Anderson *et al.* (1988)]).

In the RLC circuit described in Figure 2.11, the Joule effect in resistor R, accounting for dissipation, is described by a standard ohmic voltageintensity relation, $U_{\rm A} - U_{\rm B} = RI$. The other elements, the coil with inductance L and the capacitor with capacity C introduce the equivalent of an inertia at the origin of oscillations by setting the intensity and the voltage out of phase: the charge of the capacitor is given by $Q = C(U_{\rm C} - U_{\rm G})$, with $Q = \int I dt$ while, when submitted to a varying intensity, the coil responds by building a voltage difference $U_{\rm B} - U_{\rm C} = L\dot{I}$.



Fig. 2.11 Sketch of the RLC circuit modeling the van der Pol oscillator. Connection (1) is a simple short circuit. Connection (2) is through an active dipole with a voltage-intensity characteristic displaying a range of negative resistance (negative-resistance dipole = nrd, the gray line corresponds to an ordinary ohmic resistor).

For a series circuit we get:

$$U = U_{\rm A} - U_{\rm G} = \frac{Q}{C} + R\dot{Q} + L\ddot{Q}.$$
 (2.40)

The capacitor discharge after closing the loop along (1), *i.e.* $U_{\rm A} = U_{\rm G}$, is then given by:

$$\ddot{Q}+rac{R}{L}\dot{Q}+rac{1}{LC}Q=0$$
 .

This equation governs a damped linear oscillator $\propto \exp(-i\omega t)$ with a complex angular frequency ω solution to:

$$\omega^2 + i\gamma\omega - \omega_0^2 = 0\,,$$

where the resonance angular frequency ω_0 is given by $\omega_0^2 = 1/LC$ and the damping factor γ by $\gamma = R/L$.

The energy initially stored in the capacitor is dissipated in the resistor, at the origin of the damping. If one succeeds in injecting energy in the system so as to compensate the losses, one can obtain self-sustained oscillations. To achieve this aim, the circuit is closed on a negative-resistance dipole [loop (2)]. This "active element" is concretely implemented with an operational amplifier that draws its energy from an external electric supply maintaining the whole system in a permanent out-of-equilibrium state.

The voltage-intensity relation accounting for the active element is supposed to be ohmic but with a negative resistance coefficient. In practice, saturation effects come and limit the validity of the "anomalous" Ohm law so that we may take (cf. Fig. 2.11, right):

$$U_{{
m G}'} - U_{{
m A}'} = -R_0 I + b I^3 \qquad {
m with} \quad R_0 > 0 \quad {
m and} \quad b > 0 \,.$$

Using $I = \dot{Q}$ as a variable rather than Q itself, the equation governing the circuit then reads:

$$\left[L\dot{I} + RI + (1/C) \int I dt\right] + \left[-R_0 I + bI^3\right] = 0,$$

or, upon differentiation with respect to time:

$$L\ddot{I} + \left[(R - R_0) + 3bI^2 \right] \dot{I} + I/C = 0.$$
 (2.41)

As long as $R > R_0$ the coefficient of I is positive and dissipation plays a normal role: oscillations are damped. This is no longer the case when $R < R_0$: a small perturbation (*i.e.* such that $3bI^2$ is negligible when compared to $|R - R_0|$) is amplified and the oscillation develops. As soon as the amplitude of the oscillation is large enough, the nonlinear dissipation term plays a normal role and stops the divergence. Decreasing R, we can therefore control the bifurcation from a time-independent steady state (oscillatory perturbations are damped) towards the regime of self-sustained oscillations.

Performing the changes $I \mapsto X$ and $t \mapsto t/\omega_0$ in equation (2.41) we get one of the forms of the van der Pol model

$$\ddot{X} - (r - gX^2)\dot{X} + X = 0, \qquad (2.42)$$

where $r \propto R_0 - R$ is the control parameter and where g > 0 is a measure of the strength of nonlinearities that could have been suppressed by a rescaling of X, ending with g = 1. Figure 2.12 presents the results of the numerical integration of (2.42) for two different initial conditions in the quasi-harmonic regime (top line) or strongly anharmonic regime (middle line). In both cases one can observe that orbits spirals toward a closed curve called a *limit cycle*, either from the inside or from the outside depending on the initial condition. This special orbit is nearly elliptical in the first case and rather quadrangular in the second. To them correspond nearly sinusoidal or on the contrary highly anharmonic oscillations, which is also illustrated in the Fourier spectra that have a higher level of harmonics in the second case than in the first. The bottom graphs in Figure 2.12 illustrate the deformation of the limit cycles as the parameter r is varied from 0.1 to 2.0.

The van der Pol limit cycle is an example of attractor that is not trivially reduced to a single point (as was the case for the damped pendulum). It should be noted that, by contrast with ideal mechanical oscillators that do not have attractors and for which the amplitude of the motion is fixed by the total energy (kinetic+potential) in the initial condition, here it is the competition between the destabilization by energy injection and the dissipation that fully determines the characteristics (amplitude and period) of the regime achieved beyond the instability threshold.

In Chapter 4 we will come back to the description of this bifurcation when r changes from negative to positive values. For the moment let us determine in an approximate way the amplitude of the cycle in the quasiharmonic regime, close to the threshold.

Inserting $X \simeq X_{\rm m} \cos(t)$ in equation (2.42) we get $-X_{\rm m} \cos(t) + (r - gX_{\rm m}^2 \cos^2(t)) X_{\rm m} \sin(t) + X_{\rm m} \cos(t) = 0$.



Fig. 2.12 Van der Pol oscillator (2.42) for r = 0.1 (quasi-harmonic, first line) and r = 1.0 (strongly anharmonic, second line), both with g = 1. Left: Phase portraits showing the convergence toward a limit cycle indicated by a dashed line. Middle: Corresponding time series of the intensity signal. Right: Corresponding Fourier spectra (lin-log plot of the modulus of the Fourier amplitudes F_{ω} squared). Bottom: Limit cycles for increasing values of control parameter r.

Let us restrict to a first harmonic approximation. It consists in demanding that the equation be identically fulfilled for terms in $\sin(t)$ and $\cos(t)$, without worrying about higher harmonics generated by the nonlinearities. It is easily observed that the compensation is automatic for the cosine terms. For the sine terms, using the classical formulas $\cos^2(t) = \frac{1}{2}(1 + \cos(2t))$ and $\cos(2t)\sin(t) = \frac{1}{2}(\sin(3t) - \sin(t))$, neglecting the $\sin(3t)$ term, we get:

$$\left(r-\frac{1}{4}gX_{\rm m}^2\right)\sin(t)=0\,,$$

which leads to:

$$r - \frac{1}{4}gX_{\rm m}^2 = 0$$
 that is $X_{\rm m} = 2\sqrt{r/g}$. (2.43)

This relation fixes the amplitude of the cycle and it can further be checked that the amplitude of term in $\sin(3t)$ generated by the nonlinearities is of order $X_{\rm m}^3$, hence $\sim r^{3/2} \ll r^{1/2}$ for r small, thus justifying the expression 'quasi-harmonic', see Figure 2.12 (top-right).

Plotting $X_{\rm m}$ as a function of r, we get exactly the same bifurcation diagram as for A_* in Figure 1.4 (right), p. 14, but restricted to its upper branch, which makes sense since the change $t \mapsto t + \pi$ (legitimate since the system is autonomous) brings the branch $X_{\rm m} < 0$ on top of the other one.

Getting the result this way is rather crude. In order to improve the solution, one should try to fulfill the equation harmonic by harmonic (a special case of the so-called *Galerkin method*). This would lead to an infinite nonlinear algebraic system, the lowest order consistent truncation of which is precisely (2.43). Here, an additional implicit assumption has been that the nonlinearities did not change the angular frequency. This property turns out to be correct at lowest order in r for the van der Pol model but not necessarily in other cases, which opens the problem of the general determination of the period of nonlinear oscillators to be examined now.

2.3.2 Amplitude and phase of nonlinear oscillators

2.3.2.1 The Duffing oscillator: period from a direct computation

Let us come back to the special case of the harmonic oscillator and note that the linear relation between the restoring force and the elongation may not stay indefinitely valid. In general the microscopic characteristics of the elastic forces induce nonlinearities. If the spring is "hard" the force necessary to obtain a specific elongation grows faster than just being proportional to it and we can assume $F = -kX(1 + cX^2)$ with c > 0. (The opposite case of a "soft" spring with c < 0 is studied as part of Exercise 2.5.7.) In the absence of friction, choosing the time scale so that the angular frequency of the unperturbed oscillator is equal to 1, we get:

$$\ddot{X} + X + cX^3 = 0, \qquad (2.44)$$

which called the Duffing oscillator.

As long as the amplitude is small, the nonlinear term is negligible and the harmonic approximation is satisfactory. When the amplitude increases, the oscillator "feels" the effects of the nonlinearity and the motion becomes anharmonic. The period and the shape of the orbits change progressively. However, as long as the amplitude remains small enough, corrections to the linear solution may be obtained using *perturbation methods*.

Here the case is particularly simple. Let us multiply (2.44) by $\dot{\mathbf{X}}$ and rewrite the result as

$$\frac{\mathrm{d}}{\mathrm{d}t} \left[\frac{1}{2} (\dot{X})^2 + \frac{1}{2} X^2 + \frac{1}{4} c X^4 \right] = 0.$$
(2.45)

The quantity between the brackets is clearly the total energy, the sum of the kinetic energy and the potential energy $\mathcal{V}(X) = \frac{1}{2}X^2 + \frac{1}{4}cX^4$ from which the elastic restoring force derives. Integrating (2.45) we obtain

$$\frac{1}{2}\dot{X}^2 + \mathcal{V}(X) = E$$

called a *first integral* of (2.44). The differential order of the problem has indeed decreased by one since we can rewrite this equation as

$$\dot{X} = \pm \sqrt{2(E - \mathcal{V}(X))}, \qquad (2.46)$$

provided that the quantity under the root sign be non-negative. The condition $\mathcal{V}(X_t) = E$ defines the *turning points* of the problem, which correspond to points with maximal elongation and zero velocity, whereas the region ' $|X| > X_t$ ' is forbidden⁷ since it corresponds to a negative kinetic energy.

In the case of the nonlinear spring considered here, we get

$$\frac{1}{2}X_{\mathrm{t}}^{2} + \frac{1}{4}cX_{\mathrm{t}}^{4} = E$$

which, for c small enough, yields:

$$X_{ ext{t}} pprox \sqrt{2E} \left(1 - rac{1}{2}cE
ight) \, ,$$

so that the maximum elongation for a hard spring is reduced when compared to that of a harmonic oscillator with the same energy.

The system oscillates between its two turning points. The period can thus be computed by integration of (2.46) between them:

$$\frac{1}{2}T = \int_{-X_t}^{X_t} \frac{\mathrm{d}X}{\sqrt{2(E - \mathcal{V}(X))}} \,. \tag{2.47}$$

The value of this integral, analytically defined as a 'complete elliptic integral of the first kind' can be found in tables or numerically computed. Here it is

⁷(in classical mechanics, not in quantum mechanics)

more interesting to find its expression at low energy when c small, and thus for $X_t \ll 1$. Performing the variable change $X = X_t \sin(\varphi)$ and using the parity of the quantity to be integrated, through an expansion truncated at first order we get:

$$T\simeq 4\int_0^{\pi/2}\left[1-rac{1}{4}cX_{
m t}(1+\sin^2arphi)
ight]darphi\,,$$

hence

$$T \simeq 2\pi \left(1 - \frac{3}{4}cE\right) \quad \text{or} \quad \omega = \frac{2\pi}{T} \simeq 1 + \frac{3}{4}cE.$$
 (2.48)

For a hard spring the period therefore decreases when the energy increases, which is easily understood by writing $X + cX^3 = X(1 + cX^2)$ and observing that $(1 + cX^2)$ plays the role of an effective elastic constant, the average value of which is always larger than that of the reference linear oscillator.

After this example of explicit calculation from (2.47), made possible by the Hamiltonian context (see also Exercise 2.5.10) let us examine three methods for obtaining the value of the period in cases where nonlinear and/or dissipative effects can be considered as perturbations to a harmonic oscillator with intensities scaled by some small parameter.

2.3.2.2 Averaging method

In this first method, it is assumed that the amplitude of the oscillations is modulated on a long time scale that allows one to determine an effective equation for the modulation by *averaging*. Its intuitive simplicity makes it a reasonable first choice but it cannot easily be improved beyond lowest order. As a result we re-obtain the approximate solution to the van der Pol problem previously derived but here with a bonus.

To explain the method let us come back to the weakly damped linear oscillator (2.9) rewritten here with $\varepsilon = 2\eta$:

$$\ddot{X} + X = -\varepsilon \dot{X}, \qquad (2.49)$$

with initial conditions

$$X^{(0)} = 1$$
 and $\dot{X}^{(0)} = 0$.

The solution reads

$$X(t) = A^{(0)} \exp\left(-\frac{1}{2}\varepsilon t\right) \cos\left(\omega t - \varphi^{(0)}\right)$$
(2.50)

with $\omega^2 = 1 - \frac{1}{4}\varepsilon^2$, $A^{(0)} = 1/\omega$, $\varphi^{(0)} = \arctan(\varepsilon/2\omega)$. This expression can be written as $X(t) = A(t)\cos(t + \varphi(t))$ where $A(t) \propto \exp(-\varepsilon t/2)$ appears as the average of the amplitude over a pseudo-period, which varies slowly provided that $\varepsilon \ll 1$. On the other hand, the argument of the cosine, $\omega t - \varphi^{(0)}$, can be written as $t + \varphi(t)$ with $\varphi(t) = (\omega - 1)t - \varphi^{(0)}$, a quantity that is also slowly varying since $\omega - 1 \sim \frac{1}{8}\varepsilon^2$.

The problem is now to generalize these notions of average amplitude A(t) and phase $\varphi(t)$, supposed to be sufficiently *slowly varying*, when the system presents itself as a second order differential equation close to that governing the ideal linear oscillator, *i.e.* when it can be written as:

$$\ddot{X} + X = -\varepsilon f(X, \dot{X}), \qquad \varepsilon \ll 1.$$
 (2.51)

The solution is searched in the form:

$$X(t) = A(t)\cos(t + \varphi(t)), \qquad (2.52)$$

where A(t) and $\varphi(t)$ are two unknown functions of time. Differentiating this expression we get:

$$\dot{X} = -A\sin(t+\varphi) + \left[\dot{A}\cos(t+\varphi) - A\dot{\varphi}\sin(t+\varphi)\right].$$
(2.53)

In the absence of modulation $(\dot{A} \equiv 0, \dot{\varphi} \equiv 0)$, we would have:

$$\dot{X} = -A\sin(t+arphi)$$
,

so that it appears natural to reduce the freedom introduced in replacing the original unknown X by the two unknowns A and φ by forcing the quantity between the brackets to cancel identically. The first equation linking A to φ is therefore:

$$\dot{A}\cos(t+\varphi) - A\dot{\varphi}\sin(t+\varphi) = 0. \qquad (2.54)$$

Differentiating (2.52) once more and taking (2.54) into account we get

$$\ddot{X} = -A\cos(t+\varphi) - \left[\dot{A}\sin(t+\varphi) + A\dot{\varphi}\cos(t+\varphi)\right]$$

that we insert in (2.51) to obtain

$$\dot{A}\sin(t+\varphi) + A\dot{\varphi}\cos(t+\varphi) = \varepsilon g(A,\varphi),$$
 (2.55)

where $g(A,\varphi) \equiv f(X,\dot{X}) = f(A\cos(t+\varphi), -A\sin(t+\varphi))$. In order to isolate \dot{A} and $\dot{\varphi}$ we can combine (2.54) and (2.55) using the usual trigonometric relations: computing $(2.54) \times \cos(t+\varphi) + (2.55) \times \sin(t+\varphi)$ and

 $(2.55) \times \cos(t + \varphi) - (2.54) \times \sin(t + \varphi)$, we obtain:

$$A = \varepsilon f (A \cos(t + \varphi), -A \sin(t + \varphi)) \sin(t + \varphi), \qquad (2.56)$$

$$A\dot{\varphi} = \varepsilon f \left(A\cos(t+\varphi), -A\sin(t+\varphi) \right) \, \cos(t+\varphi) \,. \tag{2.57}$$

Up to now, everything is exact and one can notice that (2.56, 2.57) would result from a change to cylindrical coordinates $X = A\cos(\theta)$, $Y = A\sin(\theta)$, $\theta = -t - \varphi$, in a phase plane where the unperturbed second-order equation $\ddot{X} + X = 0$ would be replaced by a system of two first order equations⁸ $\dot{X} = Y$, $\dot{Y} = -X$.

The approximation comes in as soon as one assume that A(t) and $\varphi(t)$ are slowly variable on the short time scale $T \simeq 2\pi$. If this is the case, we can integrate the above equations over an approximate period while considering A and φ as constants, which leads to

$$\dot{A} = \frac{\varepsilon}{2\pi} \int_0^{2\pi} f(A\cos(t), -A\sin(t))\sin(t) dt, \qquad (2.58)$$

$$A\dot{\varphi} = \frac{\varepsilon}{2\pi} \int_0^{2\pi} f\left(A\cos(t), -A\sin(t)\right)\cos(t) \,\mathrm{d}t \,. \tag{2.59}$$

Let us apply these formulas to the van der Pol oscillator, a nonconservative system here taken in the form

$$\ddot{X} - \varepsilon (1 - X^2) \dot{X} + X = 0.$$
 (2.60)

This equation is slightly different from (2.42) where the scaling of variable X was more adapted to the problem of the bifurcation when r goes through zero. The scaling chosen here is such that the nonlinearity contributes to the solution when $X \sim \mathcal{O}(1)$ and is valid only for $\varepsilon > 0$. With $f = -(1-X^2)\dot{X}$, we get:

$$\dot{A} = \frac{\varepsilon}{2\pi} \int_0^{2\pi} \left(1 - A^2 \cos^2(t)\right) A \sin^2(t) dt = \frac{1}{2} \varepsilon A \left(1 - \frac{1}{4} A^2\right) ,$$

$$A\dot{\varphi} = \frac{\varepsilon}{2\pi} \int_0^{2\pi} \left(1 - A^2 \cos^2(t)\right) A \sin(t) \cos(t) dt = 0 .$$

The second equation shows that there is no correction to the angular frequency (at least at this order) which implicitly justifies the choice made when developing the first harmonic approximation on p. 49. By contrast,

⁸The minus signs in the definition of θ arise from the fact that the so-defined twodimensional vector field generates trajectories that rotate clockwise, opposite to the trigonometric convention.

the first equation governing the amplitude (the bonus alluded at the beginning of the section) is non-trivial and may serve us to study the convergence towards the limit cycle corresponding to $A_* = 2$ which nothing but (2.43) in the scales turning (2.42) in the form (2.60).

2.3.2.3 The Poincaré-Lindstedt method

The second method considered here is called the *Poincaré-Lindstedt* expansion. The solution is now assumed to be periodic with some unknown period close to that of the reference oscillator, and both the period and the solution are searched through an expansion in powers of ε . For simplicity, it is explained in the linear case but can easily be applied when nonlinearities are present (Exercise 2.5.11).

Let us consider two harmonic oscillators with nearly equal angular frequencies, the first one is governed by $\ddot{Y} + Y = 0$, with angular frequency $\omega_Y = 1$, and the second one by $\ddot{X} + (1 - \varepsilon)X = 0$, with angular frequency $\omega_X = (1 - \varepsilon)^{1/2}$. Starting with identical initial conditions are Y = X = 1, $\dot{Y} = \dot{X} = 0$, they will progressively drift out of phase. Choosing the first oscillator as a reference, one can interpret this phase shift as the result of secular terms that appear already in the first order expansion of the solution for the second oscillator $X(t) = \cos(\omega_X t)$. From $\omega_X \approx 1 - \frac{1}{2}\varepsilon$, one gets $\cos(\omega_X t) = \cos(t)\cos(\frac{1}{2}\varepsilon t) - \sin(t)\sin(\frac{1}{2}\varepsilon t)$, so that for sufficiently short times such that $\frac{1}{2}\varepsilon t \ll 1$, with $\cos(\frac{1}{2}\varepsilon t) \sim 1$ and $\sin(\frac{1}{2}\varepsilon t) \sim \frac{1}{2}\varepsilon t$, one finds $\cos(\omega t) = \cos(t) + \frac{1}{2}\varepsilon t \sin(t)$, which points out the secular term correction explicitly (cf. p. 41). It is not difficult to obtain this solution directly from the equation by a perturbation expansion to which we now turn.

For the second oscillator, the problem reads:

$$\ddot{X} + X = \varepsilon X$$

and the solution is searched for as a power expansion in ε :

$$X = X_0 + \varepsilon X_1 + \varepsilon^2 X_2 + \dots$$
 (2.61)

We are led to a series of simple linear problems:

$$(\frac{\mathrm{d}^2}{\mathrm{d}t^2} + 1)X_0 = 0, \qquad (2.62)$$

$$(\frac{\mathrm{d}^2}{\mathrm{d}t^2} + 1)X_1 = X_0, \qquad (2.63)$$

$$(\frac{\mathrm{d}^2}{\mathrm{d}t^2} + 1)X_2 = X_1, \qquad (2.64)$$

The solution to (2.62) with initial conditions X = 1, $\dot{X} = 0$ is nothing but $X_0 = \cos(t)$ so that the inhomogeneity in the problem for X_1 just contains a term that is resonant with the left hand side. A special solution to (2.63) is obtained by identification as $X_1 = \frac{1}{2}t\sin t$, which is indeed responsible for the secular growth of the phase shift at the considered order.

Any expansion obtained in this way has thus a limited validity in time. One says that it is *non-uniform*. The aim of the rest of this section and the next one is to obtain a *uniformly valid* expansion. Here the origin of the discrepancy is obvious: we must correct the clock and pass from the original time t to a new one τ for which secular terms would be absent. This is precisely the essence of the *Poincaré-Lindstedt method*.

Let us set $\tau = \omega t$ and look for the relation between τ and t in the form of an expansion of ω in powers of the small perturbation parameter ε :

$$\omega = 1 + \varepsilon \omega_1 + \varepsilon^2 \omega_2 + \dots \tag{2.65}$$

Denoting differentiation with respect to variable τ as $\frac{d}{d\tau}$, we get $\frac{d}{dt}X \equiv \omega \frac{d}{d\tau^2}X$ and $\frac{d^2}{d\tau^2}X \equiv \omega^2 \frac{d^2}{d\tau^2}X$, and thus

$$\left[(1 + \varepsilon \omega_1 + \varepsilon^2 \omega_2 + \dots)^2 \frac{\mathrm{d}^2}{\mathrm{d}\tau^2} + 1 \right] (X_0 + \varepsilon X_1 + \varepsilon^2 X_2 + \dots)$$
$$= \varepsilon (X_0 + \varepsilon X_1 + \varepsilon^2 X_2 + \dots)$$

leading to a new series of linear problems

$$\left(\frac{d^2}{d\tau^2} + 1\right)X_0 = 0\,,\tag{2.66}$$

$$\left(\frac{d^2}{d\tau^2} + 1\right)X_1 = X_0 - 2\omega_1 \frac{d^2}{d\tau^2} X_0, \qquad (2.67)$$

$$\left(\frac{d^2}{d\tau^2} + 1\right)X_2 = X_1 - \left(\omega_1^2 + 2\omega_2\right)\frac{d^2}{d\tau^2}X_0 - 2\omega_1\frac{d^2}{d\tau^2}X_1, \qquad (2.68)$$

The structure of (2.66, 2.67, ...) is similar to that of (2.62, 2.63, ...), except that inhomogeneities on the right hand side now contain free parameters that can be fixed so as to "kill" all the resonant terms that generate secular terms at the origin of the time non-uniformity in the initial expansion. This operation is an application of the *Fredholm alternative* stipulating that, when the kernel of a linear operator \mathcal{L} is non-trivial, the problem $\mathcal{L}\mathbf{X} = \mathbf{F}$ has solutions only if the right hand side \mathbf{F} is orthogonal to the kernel of the adjoint operator \mathcal{L}^{\dagger} of \mathcal{L} (see appendix A, §A.3.2).

Here the unperturbed problem (2.66) is self-adjoint. Its kernel is generated by the trigonometric lines $\cos(\tau)$ and $\sin(\tau)$. Inserting $X_0 = \cos(\tau)$ in (2.67) we get on the right hand side $F_1 = (1 + 2\omega_1)\cos(\tau)$. The Fredholm alternative imposes us to cancel the coefficient of $\cos(\tau)$, *i.e.* $\omega_1 = -\frac{1}{2}$. At second order we gets $F_2 = (\omega_1^2 + 2\omega_2)\cos(\tau)$ that yields $\omega_2 = -\frac{1}{2}\omega_1^2 = -\frac{1}{8}$, and so on. In this way, by reconstructing ω from its expansion we obtain $\omega = 1 - \frac{1}{2}\varepsilon - \frac{1}{8}\varepsilon^2$, which is the beginning of the Taylor expansion of $\omega = (1 - \varepsilon)^{1/2}$, as expected from the direct calculation. In Exercise 2.5.11 the method is applied to the Duffing oscillator as a typical example of a nonlinear system.

2.3.2.4 The method of multiple scales

The Poincaré-Lindstedt method does not allow for amplitude and phase modulations that were essential in the averaging method. We are thus led to the last and most general approach called the *method of multiple scales* that lifts the restrictions of both previous methods by introducing a hierarchy of time scales.

The natural relevance of the Poincaré-Lindstedt method is to Hamiltonian systems with a single degree of freedom in the mechanical sense, *i.e.* two-dimensional dynamical systems for which energy conservation implies the periodicity of bounded states, a key feature of the direct calculation above, p. 51. Difficulties appear for non-conservative systems since the occurrence of strict periodicity is then a much less trivial matter. In order to understand how to escape this problem, let us consider again the damped oscillator.

Solution (2.50) $\propto \mathcal{R}e\left\{\exp\left[\left(-\frac{1}{2}\varepsilon + i\omega\right)\right]\right\}$ can of course be written as $\propto \mathcal{R}e\left\{\exp\left[i(\omega + i\frac{1}{2}\varepsilon)t\right]\right\}$, where $\omega + i\frac{1}{2}\varepsilon$, though a complex quantity, can be understood as the angular frequency of some oscillation. In much the same way as turning to amplitude and phase in the averaging method comes to a change for cylindrical coordinates, the natural extension of the Poincaré–Lindstedt method thus suggests to change from a Cartesian parameterization of the phase space to complex one. As a matter of fact, using (2.65) and applying the previous procedure to (2.49) without assuming that ω is a real quantity, one gets at first order

$$ig(rac{d^2}{d au^2}+1ig)X_1=-2\omega_1rac{d^2}{d au^2}X_0-rac{\mathrm{d}}{\mathrm{d} au}X_0\,,$$

instead of (2.67). Injecting the complex solution $X_0 = A \exp(it)$ in this equation, one finds the compatibility condition $2\omega_1 - i = 0$, *i.e.* $\omega = 1 + \omega_1 \varepsilon = 1 + \frac{1}{2}i\varepsilon$, which correctly accounts for the expected damping at this order.
However, already in this simple problem, we may note the existence of two different time scales: the damping time of order $1/\varepsilon$, and another time scale, the inverse of the frequency shift, of order $1/\varepsilon^2 \gg 1/\varepsilon$ when $\varepsilon \ll 1$. This suggests to lay the most general method on a hierarchy of time scales, suitable to correct the lack of synchronization between a reference linear oscillator and the nonlinear system at hand. The aim of this strategy is to make the approximation uniformly valid in time, much like the introduction of leap years helps us to adjust the calendar according to a complicated algorithm that is more and more complicated as longer and longer periods of time are considered.

Let us look for the solution of the problem in the form

$$X(t) = X(t_0, t_1, t_2, \dots)$$
 with $t_0 = t$, $t_1 = \varepsilon t$, $t_2 = \varepsilon^2 t$,...

For ε small enough, the time scale measured by t_1 is indeed slow with respect to that by t_0 since when t_0 varies by a quantity $\mathcal{O}(1)$, the arguments of X in t_1, t_2, \ldots vary by $\mathcal{O}(\varepsilon), \mathcal{O}(\varepsilon^2), \ldots$

The differentiation with respect to time is then given by

$$\begin{split} \frac{\mathrm{d}}{\mathrm{d}t} &= \frac{\mathrm{d}}{\mathrm{d}t} t_0 \partial_{t_0} + \frac{\mathrm{d}}{\mathrm{d}t} t_1 \partial_{t_1} + \frac{\mathrm{d}}{\mathrm{d}t} t_2 \partial_{t_2} + \dots \\ &= \partial_{t_0} + \varepsilon \partial_{t_1} + \varepsilon^2 \partial_{t_2} + \dots , \end{split}$$

since $dt_0/dt \equiv 1$, $dt_1/dt \equiv \varepsilon$,.... In the same way the second derivative reads:

$$\frac{\mathrm{d}^2}{\mathrm{d}t^2} = \partial_{t_0^2} + 2\varepsilon \partial_{t_0} \partial_{t_1} + \varepsilon^2 \left(\partial_{t_1^2} + 2\partial_{t_0} \partial_{t_2} \right) + \dots \,.$$

The solution is taken in the form

$$X(t) = X_0(t_0, t_1, t_2, \dots) + \varepsilon X_1(t_0, t_1, t_2, \dots) + \dots,$$

and we have to insert these expansions in the motion equation and isolate the different orders in ε . Here we consider the specific cases of the Duffing and van der Pol oscillators as two complementary illustrations of the method.

• The Duffing Oscillator:

We have:

$$\ddot{X} + X = -\varepsilon X^3 \,,$$

so that at order ε^0 we find

$$(\partial_{t_0^2} + 1)X_0 = 0,$$

the solution of which reads:

$$X_0(t_0, t_1, t_2, \dots) = A_0(t_1, t_2, \dots) \cos (t_0 + \varphi_0(t_1, t_2, \dots)).$$

At order ε^1 we obtain:

$$(\partial_{t_0^2}+1)X_1=-X_0^3-2\partial_{t_0}\partial_{t_1}X_0\,.$$

The strategy is to find conditions on A_0 and φ_0 such that the r.h.s., which expands as:

$$\begin{aligned} -X_0^3 &- 2\partial_{t_0}\partial_{t_1}X_0 \\ &= -A_0^3\cos^3(t_0 + \varphi_0) + 2(\partial_{t_1}A_0)\sin(t_0 + \varphi_0) \\ &+ 2A_0(\partial_{t_1}\varphi_0)\cos(t_0 + \varphi_0) \\ &= \left(-\frac{3}{4}A_0^3 + 2A_0(\partial_{t_1}\varphi_0)\right)\cos(t_0 + \varphi_0) + 2(\partial_{t_1}A_0)\sin(t_0 + \varphi_0) \\ &- \frac{1}{4}A_0^3\cos(3(t_0 + \varphi_0)), \end{aligned}$$

contains no term in resonance with the l.h.s. Clearly only the $\cos(3(t_0 + \varphi_0))$ term on the last line is naturally non-resonant, while the terms on the previous line are. We thus get two conditions, one from cosine, the other from the sine (the kernel is two-dimensional):

$$egin{aligned} &-rac{3}{4}A_0^3+2A_0\partial_{t_1}arphi_0=0\,,\ &\partial_{t_1}A_0=0\,, \end{aligned}$$

that can both be integrated. The second condition means that A_0 does not depend on t_1 and thus, at best, on $t_2, t_3, \ldots, i.e.$ $A_0 = A_0(t_2, \ldots)$. The first equation in turn gives the lowest order correction to the phase:

$$\varphi_0 = \frac{3}{8}A_0^2 t_1 + \bar{\varphi}_0(t_2,\dots)$$
.

The non-resonant part can then be computed, if one wants to stop at this order, the full solution can be reconstructed by setting $t_1 = \varepsilon t$ in the result (compare with the output of the Poincaré-Lindstedt method used in Exercise 2.5.11).

• The van der Pol oscillator

The first steps of the computation do not change. At order ε^1 we get:

$$(\partial_{t_0^2} + 1)X_1 = (1 - X_0^2)\partial_{t_0}X_0 - 2\partial_{t_0}\partial_{t_1}X_0,$$

the right hand side of which expands as

$$\begin{split} \left(1 - A_0^2 \cos^2(t_0 + \varphi_0)\right) \left(-A_0 \sin(t_0 + \varphi_0)\right) + 2(\partial_{t_1} A_0) \sin(t_0 + \varphi_0) \\ + 2A_0(\partial_{t_1} \varphi_0) \cos(t_0 + \varphi_0) \\ = \left(-A_0 \left(1 - \frac{1}{4} A_0^2\right) + 2\partial_{t_1} A_0\right) \sin(t_0 + \varphi_0) + 2A_0(\partial_{t_1} \varphi_0) \cos(t_0 + \varphi_0) \\ + \frac{1}{4} A_0^3 \sin\left(3(t_0 + \varphi_0)\right). \end{split}$$

The conditions that annihilate the resonant terms now read:

$$-A_0(1-A_0^2/4)+2\partial_{t_1}A_0=0\,,\ 2A_0\partial_{t_1}arphi_0=0\,.$$

The second condition, $\partial_{t_1}\varphi_0 = 0$, hence $\varphi_0 = \bar{\varphi}_0(t_2, \ldots)$, shows as before that there is no correction to the angular frequency at this order. On the other hand, the first equation is seen to govern the evolution of A_0 , like in the averaging method, except for the presence of the factor ε that is incorporated in the definition of t_1 and would reappear if we were to come back to the independent variable t.

The method of multiple scales can be pursued at higher order in a completely systematic way. The price to be paid is increasingly heavy computations that are greatly eased by the use of formal algebra softwares such as MAPLE or MATHEMATICA. It can be adapted to treat the problem of periodically forced nonlinear oscillators, consult *e.g.* [Nayfeh and Mook (1979)] for further reference.

2.4 What Next?

The notion of determinism implemented in time-continuous dynamical systems considered up to now implies that a unique trajectory goes through any regular point in phase space (trajectories do not intersect since a given state cannot have several pasts and futures). The consequence is of utmost importance for two-dimensional systems which experience a strong topological constraint, so that their behavior remain "simple" in a sense to be reexamined in Chapter 4. In higher dimensions, by contrast, trajectories have enough space to wind one around each other, which is ultimately at the origin of "complicated" behavior. Before considering this problem, we examine in the next chapter how high dimensional systems, especially continuous media, may behave as effective low dimensional systems so as to render the tools that we have begun to introduce of great practical use.

2.5 Exercises

2.5.1 Evolution of volumes in phase space

Consider a continuously differentiable (\mathcal{C}^1) two-dimensional map

$$egin{aligned} Y_1 &= \mathcal{G}_1\left(X_1,X_2
ight)\,, \ Y_2 &= \mathcal{G}_2\left(X_1,X_2
ight)\,. \end{aligned}$$

1) Determine the transform of a phase space element $[X_1, X_1 + \delta X_1] \times [X_2, X_2 + \delta X_2]$ under map \mathcal{G} and compute its surface at lowest significant order [Hint: Figure 2.13].

2) Assuming that \mathcal{G} is the time- τ map of some continuous-time system

$$\dot{X}_1 = \mathcal{F}_1 \left(X_1, X_2 \right) ,$$

 $\dot{X}_2 = \mathcal{F}_2 \left(X_1, X_2 \right) ,$

over an infinitesimal time interval $\tau = \delta t$, compute \mathcal{G} from \mathcal{F} and infer that the variation of phase space volumes is locally given by the divergence of the vector field \mathcal{F} , formula (2.11).



Fig. 2.13 Transformed domain $[X_1, X_1 + \delta X_1] \times [X_2, X_2 + \delta X_2]$ under map $\boldsymbol{\mathcal{G}}$. Exercise 2.5.1, first part.

2.5.2 The energy method and its application

The energy method is an example of *global* method for studying stability problems without reference to specific perturbations, *e.g.* infinitesimal ones.

1) As a preliminary, consider the linear system $\dot{V} = -\mu V$ and check that the "kinetic energy" $E = \frac{1}{2}V^2$ is governed by:

$$\dot{E}=-\mu V^2\leq 0\,,$$

which implies a monotonic return to equilibrium, illustrating the concept of *asymptotic* stability.

2) Consider next the oscillator defined as

$$\dot{X} = Y + X\sqrt{X^2 + Y^2}, \qquad \dot{Y} = -X + Y\sqrt{X^2 + Y^2},$$

where the nonlinear terms come and modify a marginally stable dynamics in the neighborhood of the origin (a center) and for which the weaker notion of *orbital* stability has been introduced.

2a) Define $E = \frac{1}{2}(X^2 + Y^2)$, compute $\frac{d}{dt}E$ and conclude that the trivial solution is orbitally unstable. [Answer: $\frac{d}{dt}E = (X^2 + Y^2)^{3/2} > 0$ for any $(X,Y) \neq (0,0)$.]

2b) Determine the evolution of $Z = \sqrt{2E}$ (distance to the origin) starting from some initial condition $Z_0 > 0$ at t = 0 and show that that Z diverges at $t_* = 1/Z_0$, *i.e.* the corresponding trajectory spirals away to infinity in a finite time.

3) We now search to extend the concept of energy used up to now, in such a way that the study of its variations allows one to decide about stability or instability of a given state, conveniently taken as the origin of coordinates:

• Definition: Let $\mathcal{G}(\mathbf{X})$ be a function of point **X** in phase space X, taking its values in \mathbb{R}^+ , definite positive, i.e. such that $\mathcal{G}(\mathbf{0}) = 0$ and $\mathcal{G}(\mathbf{X} \neq \mathbf{0}) > 0$ (in practice a definite positive quadratic form). A sufficient condition of asymptotic stability, is that the amplitude of the perturbation, as measured by \mathcal{G} decreases in the course of time and tends to zero as t goes to infinity. This will be the case if $\dot{\mathcal{G}} = \sum_j \partial_{X_j} \mathcal{G} \dot{X}_j = \sum_j \partial_{X_j} \mathcal{G} \mathcal{F}_j$ is negative definite as a function of $\mathbf{X} \in \mathbb{X}$ taking its values in \mathbb{R} , such that $\dot{\mathcal{G}}(\mathbf{0}) = 0$ and $\dot{\mathcal{G}}(\mathbf{X} \neq \mathbf{0}) < 0$. Such a \mathcal{G} is called a Lyapunov function.⁹

⁹This definition can be extended to the case of continuous media described in terms of fields (velocity, temperature, concentration,...), in which case the Lyapunov function becomes a functional of the fields.

• Application: Consider the system defined as:

$$\dot{X} = XY - X^3,$$

 $\dot{Y} = -Y - 2X^2.$

Show that the origin is the unique fixed point and further that it is marginally stable. Since no conclusion can be drawn at the linear stage, by generalizing the energy E to a function in the form:

$$\mathcal{G}(X,Y) = rac{1}{2} \left(lpha X^2 + eta Y^2
ight),$$

determine the conditions under which \mathcal{G} is a Lyapunov function, thus proving the asymptotic stability of this state.

2.5.3 Linear stability and bifurcations in dimension three

As a useful preliminary to Exercise 3.3.3 in Chapter 3, consider a real, three-dimensional, linear dynamical system in the form

$$\dot{X}_i = \sum_{j=1}^3 a_{ij} X_j \,, \quad i=1,2,3 \,.$$

1) Recall the relations between the coefficients and the roots of its characteristic equation written as $s^3 + as^2 + bs + c = 0$. Further identify all the possible non-degenerate or degenerate cases (real or complex nature of the roots, their sign or the sign of their real part, simple and multiple roots). Write down the Jordan normal form of the operator corresponding to each situation (see Appendix A, §A.2 for a reminder) [Answer: see Fig. A.1(b)]. 2) When the control parameters vary, the eigenvalues of the system move in the complex plane. Determine the remarkable relations fulfilled by the coefficients of the characteristic equation in the marginal cases and sketch the spectrum of the operator in the complex plane before, at, and after a stationary bifurcation (s = 0) or an oscillatory bifurcation ($\mathcal{R}e(s) = 0$). [Answer: see Fig. 2.14.]

2.5.4 Coupled linear oscillators

Consider the following system of two coupled oscillators

$$\begin{split} \ddot{X} + X &= Y \,, \\ \ddot{Y} + \Omega^2 Y &= -cX \,. \end{split}$$



Fig. 2.14 Typical bifurcations in the three-dimensional case, Exercise 2.5.3.

Find the equation governing the angular frequencies of the eigenmodes taken as $(X, Y) = (X_0, Y_0) \exp(i\omega t)$, observe that the solutions appear as pairs of opposed eigenvalues. Determine the domain in the plane of parameters Ω^2 and c where the eigenmodes are stable. [Hint: remember that for a mechanical system, stability means orbital stability, with purely imaginary eigenvalues.]

2.5.5 Logistic equation

Reexamine the limited growth model (logistic equation) introduced in Chapter 1, Exercise 1.5.2 about population dynamics. Study the stability of its two fixed points and draw its phase portrait.

2.5.6 Dynamical systems and solitons

1) Consider a dynamical system with a single degree of freedom in the sense of analytical dynamics (a pair of conjugate variables) with a force deriving from the potential

$$V(X) = -\frac{X^2}{2} + \frac{X^3}{3}.$$
 (2.69)



Fig. 2.15 Left: Phase portrait of the system with a force deriving from potential (2.69) Right: Profile of the soliton corresponding to the separatrix on the left part.

1a) Write down the second order differential equation governing X (mass m = 1) and then the first order system for X and $P = \dot{X}$. Find the fixed points, compute their eigenvalues and eigenvectors (if any). Sketch the phase portrait in the phase plane (X, P) [Answer: Figure 2.15, left].

1b) Draw the graph of the potential V(X) and determine the energy corresponding to the separatrix, the special trajectory that limits the domain of orbits bounded around the minimum of the potential. Find the turning point and the equation accounting for this trajectory. Determine its solution by identification with

$$X(t) = X_0 / \cosh^2(t/ au)$$
 .

Why is this form "natural"?

1c) Sketch the phase portrait of the system when perturbed by the introduction of a weak viscous friction, taking the weakly damped pendulum as an example.

2) The Korteweg-de Vries equation reads:

$$\partial_t h + h \partial_x h + \beta \partial_{xxx} h = 0.$$

This is a non-dissipative equation with a nonlinearity of hydrodynamic type, identical to that of the Burgers equation considered in Chapter 1, Exercise 1.5.3. It is integrable and its solution can be expressed in terms of a superposition of interacting solitons. Here we are interested in a solution with a single soliton such that $h \to 0$ when $x \to \pm \infty$ moving without deformation at speed c, and thus only function of the combination $\xi = x - ct$.

2a) Determine the equation in ξ governing such a moving solution (make the substitutions $\partial_x \mapsto d/d\xi$ and $\partial_t \mapsto -c d/d\xi$). Integrate this equation and find the value of the integration constant that corresponds to the soliton solution.

2b) By appropriate variable changes put this equation in the form considered in the first part of the exercise and deduce the analytical expression of the soliton. Discuss the relation between the speed, the height, and the width of the soliton [Hint: Figure 2.15, right].

2.5.7 Variants of the Duffing oscillator

1) Consider the system

$$\ddot{X} + X(a + X^2) = F,$$

where a is a coefficient with unspecified sign for the moment, and F a quantity playing the role of an external force.

1a) Find the potential from which the dynamics can be derived in the sense of mechanics, *i.e.* \mathcal{V} such that $m\ddot{X} = -\partial \mathcal{V}/\partial X$, and the time independent response to a force $0 < F \ll 1$ (observe that the roots of an equation in the form $x^3 - px + q = 0$ with $q \ll 1$ and $p \sim 1$ are approximately given by $px \sim q$ for x "small" and $x^3 \sim px$ for x "large."

1b) Discuss the dynamics in the phase plane $(X, Y \boxminus \dot{X})$ as a function of the sign of parameter *a*. Locate the fixed points and determine their stability properties (when solving the linearized problems, neglect the nonlinearities for the "small" solution and the constant term for the "large" one). Draw a few typical orbits. In which sense can one speak of the system as of an oscillator.

2) Consider now

$$\ddot{X} + X + cX^3 = 0$$
 with $c < 0$ (soft spring).

2a) Find the elastic potential from which the dynamics can be derived and discuss the behavior of the system in its phase plane as above.

2b) For which set of initial conditions is the system a physically well posed one (bounded orbits)? How should one correct the model in order to avoid the divergence of some trajectories? Interpret this limitation by considering the theory leading to the approximate expression (2.48) for the angular frequency of small oscillations when c > 0 and assuming that it is still valid for large amplitude orbits when c < 0.

2.5.8 Carriage with a spring

Consider the mechanical system built with an ideal spring of equilibrium length ℓ_0 fixed by one of its ends at point P as described in Figure 2.16(a). Its other end is attached at M to a small carriage with mass m sliding along a horizontal line forming the x axis. Point P is at a distance ℓ_1 from this axis. The intensity of the restoring force is proportional to the elongation $|\mathbf{F}| = k(\ell - \ell_0)$.

1) Determine the components of the forces exerted on the carriage at abscissa X and the corresponding equation of motion. Taking ℓ_0 as length unit, $\sqrt{m/k}$ as time unit, and setting $\ell_1 = \lambda \ell_0$, rewrite this equation in dimensionless form as a system of two equations for X and $Y = \dot{X}$.

2) Compute the potential $\mathcal{V}(X)$ from which the force F(X) derives, *i.e.* $F(X) = -\partial \mathcal{V}/\partial X$. Find the fixed points as a function of λ , study their stability by linearizing the system.

3) Sketch the phase portraits in the two cases $\lambda < 1$ and $\lambda > 1$. What can be said from the trajectory issued from the vicinity of the origin along the unstable direction in the case $\lambda < 1$. Using the fact that the system is frictionless, determine the corresponding turning point quantitatively.

2.5.9 Carriage sliding on a rotating hoop

Consider now a carriage sliding along a rail in the form of a hoop with radius a, itself rotating around a vertical diameter (cf. Fig. 2.16b). The rotation period if $T = 2\pi/\omega$. The motion is governed by the following



Fig. 2.16 Left: Spring and carriage sliding along a horizontal rail. Right: Carriage sliding along a rail in the form of a hoop further rotating around a vertical diameter.



Fig. 2.17 Phase portrait of the hoop system for $\omega^2 = 0.8$ (left) and $\omega^2 = 1.25$ (right).

equation:

$$a\frac{d^2\theta}{dt^2} = -g\sin(\theta) + a\omega^2\cos(\theta)\sin(\theta)$$

where g is the gravitational acceleration.

1) Justify this equation by taking advantage of the elements of analytical mechanics recalled in §2.1.2. To this aim, compute first the gravitational potential energy and then the kinetic energy resulting from the superposition of the two independent rotation motions, around the axis and within the plane of the hoop. Set the equation in the form of a two-dimensional differential system for θ and $\varphi = \dot{\theta}$.

2) Find the fixed points and study their stability as a function of the angular speed ω . Sketch the phase portraits in the different cases [Answer: Figure 2.17].

3) Examine the effects of a slight viscous friction proportional to φ .

2.5.10 Period of an oscillator in a quartic potential

Consider a strongly anharmonic oscillator with a restoring force deriving from the potential

$$V(X) = V_0 + \frac{1}{4}X^4$$
.

From equation (2.47) giving the period of an oscillator, show without explicit calculation that one gets $T \propto E^{\alpha}$ with an exponent α to be determined (perform the change of variables that applies the interval between the turning points onto [-1, 1]).

2.5.11 Averaging and Poincaré-Lindstedt methods

The Duffing oscillator is taken in the form

$$\left(\frac{\mathrm{d}^2}{\mathrm{d}t^2}+1\right)X=-\varepsilon X^3\,.$$

1) Extract the relation between the amplitude and the angular frequency of the oscillator by making use of equation (2.59) for the phase. What is the meaning of equation (2.58) for the amplitude.

2) Following the Poincaré-Lindstedt method (§2.3.2.3), derive the complete solution at first order in ε starting from the initial condition $X^{(0)} = A$, $\dot{X}^{(0)} = 0$ at t = 0. Using expansions (2.61–2.65), show first that at the relevant order the problem simply reads

$$(\frac{d^2}{d\tau^2}+1)X_1 = 2\omega_1 X_0 - X_0^3,$$

and next that the elimination of resonant terms leads to the same result as the first order Taylor expansion of the solution obtained by a direct calculation (2.48). [Observe that, at lowest order, the averaging method and the Poincaré–Lindstedt method involve exactly the same computations.]

3) Find the complete solution at first order, once the Fredholm alternative is fulfilled.

2.5.12 Prey-predator system

Consider the system derived in Chapter 1, Exercise 1.5.2:

$$\dot{X} = \mathcal{F}_X(X, Y) = (\alpha' Y - \alpha_0) X, \qquad (2.70)$$

$$Y = \mathcal{F}_Y(X, Y) = (\beta - \gamma X)Y, \qquad (2.71)$$

in the physical quadrant $X \ge 0$, $Y \ge 0$ (X and Y are population counts).

1) Draw a qualitative phase portrait by studying the vector field along lines defined by $\mathcal{F}_X(X,Y) = 0$ and $\mathcal{F}_Y(X,Y) = 0$ (vertical and horizontal isoclines respectively). Search and study the characteristics of the fixed points (roots of $\mathcal{F}_X(X,Y) = 0 = \mathcal{F}_Y(X,Y)$).

2) Show that the quantity

$$H(X,Y) = \alpha'Y - \alpha_0 \log(Y) + \gamma X - \beta \log(X)$$

is conserved along a trajectory and that H displays a minimum at the nontrivial fixed point.

3) Consider the successive intersections of a trajectory with line $Y = \alpha_0/\alpha'$

and conclude from the behavior of H that all trajectories are periodic orbits. Notice that they are not limit cycles and explain why the elliptic character of the non-trivial fixed point is a deficiency of the model. Try to correct it by considering the most general model of the form $\mathcal{F}_X = X \tilde{\mathcal{F}}_X(X,Y)$, $\mathcal{F}_Y = Y \tilde{\mathcal{F}}_Y(X,Y)$, at most quadratic in X and Y. In particular find the conditions to be fulfilled in order to reproduce the qualitative features characteristic of a prey-predator system.

Chapter 3

Life and Death of Dissipative Structures

In this transition chapter we focus on the emergence of convection and how patterns that have developed further disaggregate. This rather intuitive example helps us to introduce a few general ideas and techniques to analyze instabilities ($\S3.1$). The theory that allows us to interpret their disorganization, presented here from a purely phenomenological perspective ($\S3.2$) will be reexamined later (Chaps. 4 and 5), in continuity what has begun to be formalized in Chapter 2.

3.1 Emergence of Dissipative Structures

3.1.1 Qualitative analysis of the instability mechanism

Let us go back in more detail on the idea previously introduced (Fig. 1.3, p. 11) to explain the onset of convection. The two parts of the mechanism, instability due to differential buoyancy and stability through dissipation (viscous relaxation and heat diffusion) will first be qualitatively analyzed through an argument in terms of *characteristic times*.

Let us consider a horizontal layer (height h) of fluid heated from below (Fig. 3.1):

$$T_{\rm b} = T_{\rm t} + \Delta T > T_{\rm t}$$

('b' for 'bottom' and 't' for 'top'). The fluid is initially at rest in a regime of *pure conduction*. The temperature profile is linear

$$T_0(z) = T_{
m b} - eta z \qquad ext{with} \quad eta = \Delta T/h \, ,$$

and the notations imply that the temperature gradient β is positive in the case of heating from below.



Fig. 3.1 Left: Geometry of the convection experiment. Right: Profiles of temperature T(z), density $\rho(z)$ and gravitational potential energy $U_g(z)$ for a fluid particle at altitude z, indicating the tendency to restore a stable density stratification with heavy fluid at the bottom.

The corresponding density distribution is given by the equation of state that, in first approximation, reads:

$$\rho(T) = \rho_{\text{ref}} \left(1 - \alpha \left(T - T_{\text{ref}} \right) \right) \tag{3.1}$$

where T_{ref} is a reference temperature, $\rho_{\text{ref}} = \rho(T_{\text{ref}})$, and α is the thermal expansion coefficient (1/273 for an ideal gas), hence $\rho_0(z) = \rho(T_0(z))$.

A first characteristic time, a transport time $\tau_{\rm b}$, can be defined from the buoyancy (hence subscript 'b'). Assume a fluid particle experiencing a temperature fluctuation θ at some height z, *i.e.* $T(z) = T_0(z) + \theta$, from (3.1) the differential force to which it is submitted is $\rho g \alpha \theta$. The quantity $g \alpha \theta$ is thus an acceleration, homogeneous to a length divided by the square of a time. Natural scalings are h for lengths and ΔT for temperatures. We can thus define the time $\tau_{\rm b}$ through:

$$rac{h}{ au_{
m b}^2} = g rac{\Delta
ho}{
ho} \sim g lpha \Delta T$$
 .

Physically, $\tau_{\rm b}$ is the typical time a hot (cold) bubble would take to move up (down) over a distance h with a constant acceleration due to thermal expansion.

Dissipative processes, viscous friction (Stokes law, kinematic viscosity $\nu = \mu/\rho$) and thermal conduction (Fourier law, thermal diffusivity $\kappa = \chi/C$ where χ is the thermal conductivity and C the specific heat) are diffusive in essence. The relaxation times associated with these processes can be deduced from the form of a diffusion equation, $\partial_t q \propto \nabla^2 q$, in which the proportionality coefficient is the diffusivity, homogeneous to $[\ell]^2[t]^{-1}$, hence

here:

$$u = h^2/ au_v\,, \qquad \kappa = h^2/ au_ heta\,.$$

(See also Exercise 1.5.3.)

The result of the competition between the destabilizing mechanism and the stabilizing processes can be estimated by forming the ratio:

$$R = \frac{\tau_v \tau_\theta}{\tau_c^2} = \frac{\alpha g \Delta T h^3}{\kappa \nu}$$
(3.2)

called the *Rayleigh number*. By construction, it is a dimensionless number. Convection develops when the buoyancy is more effective (τ_b short) than the dissipative processes (τ_v and τ_θ long) and thus when R is large. This strictly dimensional analysis is qualitative and cannot help us to determine the value of ΔT necessary to induce convection. It is the reason why we go one step further and develop a more quantitative model of the instability. The detailed analysis is the subject of Exercise 3.3.2.

3.1.2 Simplified model

In order to study the stability of the base flow, here the fluid at rest $(\mathbf{v}_0 \equiv 0)$ in a regime of pure conduction with linear temperature profile $(T_0(z) \sim -\beta z, \beta = \Delta T/h)$, we must derive the equations governing small perturbations around this state, a temperature fluctuation θ defined by $T = T_0(z) + \theta$ and a velocity fluctuation \mathbf{v} . We thus insert the full solution into the primitive equations (here: Navier-Stokes + continuity + Fourier in a fluid), expand these equations in powers of the perturbations, and finally keep only the first order terms (linear stability theory).

The analysis of the mechanism (Chap. 1, §1.3.1) points out a direct coupling of the horizontal modulations of the temperature fluctuation with the vertical velocity component. Accordingly, we assume that a model involving just θ and v_z , depending only on the horizontal coordinate x and time t, will capture the physics.

Equation for the vertical velocity

The unperturbed temperature field $T_0(z)$ induces a density distribution $\rho_0(z) = \rho(T_0(z))$ through (3.1). The differential buoyancy

$$-g(
ho-
ho_0) = -g[
ho(T_0+ heta)-
ho(T_0)] pprox
ho_0 lpha g heta$$

generated by a temperature fluctuation, where the minus sign comes from

the fact that the vertical unit vector is oriented up while the force is directed down, then appears as an external force term in the z component of the Navier-Stokes equation:

$$\partial_t v_z = \nu \partial_{x^2} v_z + \alpha g \theta \,, \tag{3.3}$$

the first term on the r.h.s. corresponding to viscous diffusion (only along x). The term $\mathbf{v} \cdot \nabla \mathbf{v}$ is of higher order since there is no velocity at order zero. Also, one can notice that $\theta > 0$ implies $\partial_t v_z > 0$, *i.e.* an upward acceleration as guessed intuitively.

Heat equation

The Fourier equation must be written for a fluid particle (cf. p. 7) since, as already mentioned, it is its advection in a spatially varying temperature field that plays the essential role in the feedback loop, hence:

$$\frac{\mathrm{d}}{\mathrm{d}t}T = \partial_t T + \mathbf{v} \cdot \boldsymbol{\nabla} T = \kappa \partial_{x^2} T \,.$$

Expanding the term $\mathbf{v} \cdot \nabla T$ to first order (linearization) we find:

$$v_z \partial_z [T_0(z) + heta] = v_z \partial_z T_0(z) = - v_z eta \, ,$$

which comes from the temperature field at order zero. This leads to:

$$\partial_t \theta = \kappa \partial_{x^2} \theta + \beta v_z \,. \tag{3.4}$$

Remarks

Lateral boundary conditions have not yet been specified. Here, we tacitly assume that we deal with a horizontally unbounded layer, or at least that the horizontal dimensions are large when compared to the sole characteristic length in the problem: the height h of the layer. Moreover, the horizontal velocity component and the pressure are absent from the problem at this stage. In fact they are only indirectly coupled to v_z and θ by the need to insure the continuity of the fluid and the closing of flow lines. The model is thus highly simplified. This deficiency will impede us to determine the critical wave-length that will thus be fixed by dimensional considerations. Anyway, we shall now illustrate the extension to continuous media of the stability analysis introduced at the beginning of the previous chapter using the simplified model (3.3, 3.4).

3.1.3 Normal mode analysis, general perspective

The system formed by Eqs. (3.3, 3.4) is typical of linear stability problems in continuous media, *i.e.* a system of linear partial differential equations (with constant coefficients in the simplest case). It presents itself as an *initial value problem* for the perturbations that we write formally as:

$$\partial_t \mathbf{V} = \mathcal{L}_r(\partial_x, \dots) \mathbf{V}, \qquad (3.5)$$

where **V** represents the set of perturbations. The linear operator \mathcal{L}_r , here solved for ∂_t and first order in time, contains spatial partial derivatives (∂_x, \ldots) and also depends on a set of *control parameters* denoted as r. In general, the instability can be controlled using a single quantity that can be varied from the outside (for convection, it is simply the applied temperature gradient β), all other parameters of the system being kept fixed.

Problem (3.5) is linear. Its solution can therefore be searched by means of a *superposition*:

$$\mathbf{V}(\mathbf{x},t) = \sum_{n} A_n \mathbf{X}_n(\mathbf{x},t)$$
(3.6)

that is further introduced in the differential problem. Setting:

$$\mathbf{X}(\mathbf{x},t) = \exp(st)\hat{\mathbf{X}}(\mathbf{x}), \qquad (3.7)$$

and inserting this assumption in (3.5), we get

$$s\hat{\mathbf{X}}(\mathbf{x}) = \mathcal{L}_{\tau}\hat{\mathbf{X}}(\mathbf{x}).$$
 (3.8)

The stability study then comes to an *eigenvalue problem*. The states $\hat{\mathbf{X}}(\mathbf{x})$ are called the *normal modes* of the problem. These modes have spatial structures that are the mathematical expression of the physical *coherence* of the processes at work in the system.

The nature of the spectrum of \mathcal{L}_r depends on the applied boundary conditions (Exercise 3.3.1). When the system is unbounded in some directions of space, the spectrum is formed with continuous branches, indexed by as many continuous "separation" parameters as unbounded directions. For example if the mechanism singles out a specific direction ('vertical' in the case of convection), and if the system is invariant under translations in two complementary directions ('horizontal'), performing a Fourier transform, one looks for normal modes in the form:

$$\mathbf{X}_n(x, y, z) = \exp(i(k_x x + k_y y)) \mathbf{X}_n(z), \qquad (3.9)$$

which defines the wavevector $\mathbf{k}_{\perp} = (k_x, k_y)$ as a separation parameter. Under the substitution $\nabla_{\perp} = (\partial_x, \partial_y) \mapsto i\mathbf{k}_{\perp}$, the operator $\mathcal{L}_r = \mathcal{L}_r(\partial_x, \ldots)$ is transformed into an ordinary differential operator in z, the only remaining independent variable, *i.e.* $\mathcal{L}_r(ik_x, ik_y, d/dz)$. The wavevector \mathbf{k}_{\perp} thus serves to label the eigenmode branches in addition to a discrete index related to the confinement direction z. Equation (3.8) has nontrivial solutions provided that the growth-rate and the wavevector \mathbf{k}_{\perp} , introduced in (3.7) and (3.9) fulfill a compatibility condition

$$s = s_n \left(r, \mathbf{k}_\perp \right) \tag{3.10}$$

called the dispersion relation.

Before developing this approach on the simplified model of convection in Section 3.1.4, let us note a few points relative to confinement effects to be re-examined at the beginning of Chapter 4.

• When the system is rotationally invariant in the plane orthogonal to the direction in which the mechanism is operating, then s_n can only depend on $k = |\mathbf{k}_{\perp}|$ and not on its orientation.

• When the system is bounded in two space directions, e.g. y and z, translationally invariant in the last one x, the continuous component k_y is replaced by a discrete index.

• When the system is bounded in all three directions, *i.e.* when its size is of the order of the scale over which the instability mechanism is operating in all directions, the spectrum loses its last continuous dependence on k and becomes fully discrete. The eigenvalues are all distinct, except for degeneracies linked to physical symmetries.¹ A situation close to that considered in the previous chapter is recovered but with infinite series of eigenvalues and normal modes to acknowledge the fact that we deal with continuous media with infinitely many degrees of freedom.

Assumption (3.6) brings us back to the formulation initially introduced at the end of Section 2.2, p. 35. Considering the evolution of the solution corresponding to a "pure" mode with index n and amplitude A_n (supposed to be infinitesimal) we have

$$\mathbf{V} = A_n \mathbf{X}_n(x,t) = A_n^{(0)} \exp(s_n t) \hat{\mathbf{X}}_n(x) \,,$$

where $A_n^{(0)}$ is the initial condition for the amplitude of mode *n*. Let us recall that the eigenvalue is *a priori* complex since the spectrum is entirely

¹In this case the spatial structure of the normal modes is specific to the geometry considered and mirrors spatial resonance properties of the mechanism with the shape of the set-up.

real only when the operator can be made self-adjoint for some well-chosen scalar product (see Appendix A).

Let us consider the case of an unbounded system with a single continuous separation parameter k and write $s_n(k,r) = \sigma_n(k,r) - i\omega_n(k,r)$. Given r and k, we can order the modes by decreasing values of their real part $\sigma_n(k,r)$, which allows us from definition (3.7) to distinguish between *stable* normal modes with $\sigma_n < 0$ (damped) and *unstable* modes with $\sigma_n > 0$ (amplified). The dissipative character of the medium implies that modes with the shortest wave-lengths are strongly damped, *i.e.* $\sigma_n(k) \to -\infty$ when $k \to \infty$.

The system bifurcates against a given mode n with wavevector k when, upon variation of the control parameter r, this mode goes from stable to unstable. The marginal conditions for this modes, superscript '(m)', are defined by the condition $\sigma_n(k, r) = 0$, that makes it neutral, which can be written as:

$$r = r_n^{(\mathrm{m})}(k)$$

when solved for r. Let us suppose that, as in convection, increasing the stress corresponds to increasing the control parameter r, the marginal curve for some mode n usually reaches its minimum for some $k = k_n^{(c)}$ called the *critical wavevector* for that mode. The corresponding value of r, $r_n^{(c)} = r_n^{(m)}(k_n^{(c)})$ is the corresponding *threshold*.

Now, according to the general discussion about stability in the previous chapter, linear instability takes place as soon as one normal mode becomes unstable. Accordingly, the linear instability threshold r_c is the minimum over n of all the so-defined $r_n^{(c)}$, achieved for, say, $n = n_c$. The wavevector of that mode called the critical wavevector of the instability, hence characterized by the set n_c , $r_c = r_{n_c}^{(c)}$, $k_c = k_{n_c}^{(c)}$.

Apart from its growth properties, the rest of the time dependence of a mode depends on the value of $\omega_n(k,r)$. When $\omega_n = 0$, the mode is said to be *stationary*, whereas when $\omega_n \neq 0$, one speaks of an *oscillatory* mode. The value of the angular frequency at threshold, $\omega_c = \omega_{n_c}(k_c, r_c)$, thus allows one to distinguish stationary from oscillatory instabilities. The classification of instabilities according to the spatio-temporal structure of their critical mode will be reexamined in §3.1.6.

3.1.4 Back to the model

Let us come back to the simplified model (3.3, 3.4). We assume that the fluid layer is unbounded in the x direction so that, according to (3.7-3.9), solutions are searched in the form $\{v_z, \theta\} = \{V, \Theta\} \exp(st) \exp(ikx)$. We obtain:

$$egin{aligned} sV &= -
u k^2 V + lpha g \Theta \,, \ s\Theta &= -\kappa k^2 \Theta + eta V \,, \end{aligned}$$

which is in fact a homogeneous algebraic system of two equations for two unknowns:

$$(s + \nu k^2)V - \alpha g\Theta = 0,$$

$$-\beta V + (s + \kappa k^2)\Theta = 0.$$

The system has non-trivial solutions only if its determinant cancels:

$$(s + \nu k^2)(s + \kappa k^2) - \alpha g\beta = s^2 + (\nu k^2 + \kappa k^2)s + \kappa \nu k^4 - \alpha g\beta = 0.$$
(3.11)

This compatibility condition linking the growth-rate s and the wavevector k of the perturbation is here the expression taken by the dispersion relation (3.10). We get a single branch (and thus no discrete index as alluded to above) since the differential problem in z has been replaced by an algebraic system, due to our neglect of the z-dependence of the fluctuations.

If the real part of s(k) is negative, the mode is damped and the layer is *stable* against a perturbation with wavelength $\lambda = 2\pi/k$. Otherwise the fluctuation is amplified and the mode k is *unstable*.

Let us try to determine the threshold from (3.11). Here it is a quadratic equation in s that can have two real or complex roots.² The discriminant

$$\Delta = \left(
u k^2 + \kappa k^2
ight)^2 + 4\left[lpha geta - (
u k^2)(\kappa k^2)
ight] = \left(
u k^2 - \kappa k^2
ight)^2 + 4lpha geta$$

can be negative, and the corresponding solutions of (3.11) have non-zero imaginary parts, only when β is sufficiently large and negative, *i.e.* according to our conventions, in case of strong heating from above, but in this case the modes are always damped since the sum of the roots

$$S = -rac{1}{2} \left(
u k^2 + \kappa k^2
ight)$$
 (3.12)

²Let $s_{1,2}$ be the two roots, one has $0 = (s - s_1)(s - s_2) = s^2 - (s_1 + s_2)s + s_1s_2 = s^2 - Ss + P$ where S is the sum of the roots and P their product. The discriminant is $\Delta = S^2 - 4P$ and the roots are $s_{1,2} = s_{\pm} = \frac{1}{2}(S \pm \sqrt{\Delta})$.

is then negative. This analysis thus confirms the intuition according to which heating must be from below in order to have an instability. Moreover, if there is an instability, it can only be stationary, with two real roots, one positive, the other negative since their sum is negative. In order to determine the sign of the roots we have just to consider their product

$$P = (\nu k^2)(\kappa k^2) - \alpha g\beta. \qquad (3.13)$$

The change of sign takes place at $\beta = \beta^{(m)}(k)$ with

$$\beta^{(\mathrm{m})}(k) = \frac{\nu \kappa k^4}{\alpha g} \,. \tag{3.14}$$

As long as $\beta < \beta^{(m)}$, the product is positive and the two roots negative, mode k is stable. When $\beta > \beta^{(m)}$, it becomes negative and one of the roots is positive, the mode is unstable, convection sets in. The value $\beta = \beta^{(m)}(k)$ of the applied temperature gradient thus here defines the marginal stability condition that makes mode k neutral. One can observe that the negativity of the sum and the positivity of the product, the two stability factors, come from the stabilizing dissipative processes and that the instability factor involves a term in β arising from the advection of the temperature field.

Let us come back to the marginal stability curve described by (3.14). It accounts for an increase of the marginal temperature gradient as k^4 for klarge, which expresses the growing efficiency of the stabilizing mechanisms as the scale of the fluctuations decreases (Fig. 3.2, left). If we trust in this relation, the longer the wavelength, the lower the threshold. However we should not conclude that the fluid layer is unstable at k = 0 for $\beta = 0$, *i.e.* $\Delta T = 0$. As a matter of fact, this low-k behavior is an artifact of the one-dimensional model which neglects the z-dependence of the fluctuations and the associated dissipation processes: viscous damping by the horizontal component of the flow that closes the streamlines can no longer be neglected as $k \to 0$ (Fig. 3.2, right), hence the corrected argument valid for k small.

Let us keep v_z as a reference since it is directly involved in the instability mechanism. We can estimate the order of magnitude of v_x from the continuity equation

$$\partial_x v_x + \partial_z v_z = 0.$$

Boundary conditions on v_z are at the horizontal plates, a distance h apart, and imply a z-dependence such that $\partial_z v_z \sim v_z/h$ and therefore $kv_x \sim v_z/h$ or $v_x \sim v_z/kh$. But the presence of v_x imposes us to take the x-component



Fig. 3.2 Left: For $\lambda = 2\pi/k \ll h$, viscous dissipation associated with the horizontal shear $\partial_x v_z$ (black arrows) and thermal diffusion (undulated arrows) combine their effects to prevent convection, while the vertical shear $\partial_z v_z$ (gray arrows) can be neglected. Right: When $\lambda \gg h$ the horizontal shear (gray arrows) becomes negligible while the vertical shear (black arrows) becomes dominant.

of the Navier-Stokes equation into account. We can simplify it as:

$$-kp/
ho-
u v_x/h^2\simeq 0\,,$$

and introduce the so-evaluated pressure in the equation for v_z , which yields:

$$\partial_t v_z = -\partial_z p / \rho + \nu \left(\partial_{x^2} + \partial_{z^2} \right) v_z + \alpha g \theta \,. \tag{3.15}$$

A sketchy analysis of the space dependence of the different perturbations then shows that, once expressed in terms of v_z using the continuity equation, $\partial_z p/\rho$ goes as $\nu v_z/k^2h^4$, so that the pressure term dominates those involving v_z on the r.h.s. of (3.15). As a matter of fact, it diverges as k^{-2} when $k \to 0$, while the second term tends to zero as k^2 and the third one does not vary with k. We thus arrive at an effective equation for v_z replacing (3.3):

$$\partial_t v_z = -
u (1/h^4 k^2) v_z + lpha g heta \,,$$

only valid in the limit $k \ll 1/h$ (the minus sign expresses the fact that it is indeed a damping term). In this limit, it suffices to replace $-k^2$ by $-1/h^2$ in the heat equation (3.4) to account for the dominant dissipative process. A stability analysis parallel to that leading to (3.14) yields:

$$\kappa
u/(h^6k^2)-lpha geta^{({
m m})}=0\,,$$

so that the marginal stability condition for small wavevectors reads:

$$\beta^{(m)}(k) = \frac{\nu\kappa}{\alpha g h^6 k^2} \tag{3.16}$$

thus showing a divergence as $1/k^2$ for $k \to 0$ that can be demonstrated through a detailed calculation (Exercise 3.3.2, see Fig. 3.16, p. 109).



Fig. 3.3 Marginal stability curve from the semi-quantitative argument.

Inbetween the divergence as k^4 for $k \gg 1/h$ and as $1/k^2$ for $k \ll 1/h$, we must find a minimum corresponding to some optimum between the stabilizing effects of different origins and the destabilizing buoyancy force. This optimum is achieved for some intermediate value of the wavevector that, for dimensional reasons, can only be related to the thickness of the layer (Fig. 3.3). Assuming that the diameter of the convection cells is, at threshold, of the order of h, *i.e.*

$$k_{
m c}=rac{2\pi}{\lambda_{
m c}}=rac{\pi}{h}\,,$$

and inserting this value of the *critical wavevector* in the expression of $\beta_{\rm m}$ we get the *instability threshold* beyond which the convection regime develops, in the form of regular structures with typical wavelength $\lambda_{\rm c} = 2\pi/k_{\rm c} \simeq 2h$. The estimation of the threshold from (3.14) gives

$$R_{
m c}\sim\pi^4$$
 ;

where we have used the expression (3.2) of the Rayleigh number.

The semi-quantitative argument developed so far stresses on the physics of the processes at stake. The threshold value turns out to be grossly underestimated because a large part of the dissipating processes is badly evaluated, but it remains reasonable as an order of magnitude. We should however notice that the model, how simplified it could be, reproduces the two main characteristics of the instability: its stationary character and the general shape of the marginal stability curve with a correct asymptotic behavior for $k \ll k_c$ and $k \gg k_c$. Similar simplified analyses will be developed as exercises to study the effects of molecular diffusion on convection in binary mixtures (Exercise 3.3.3), the stability of an angular momentum stratification in a cylindrical shear experiment (Taylor-Couette instability, Exercise 3.3.5), or the emergence of spatial structures in reaction-diffusion systems (Turing instability, Exercise 3.3.4).

3.1.5 Vicinity of the threshold: linear stage

The study of linear dynamics of the fluctuations in the neighborhood of the threshold is best developed by first turning equations (3.3, 3.4) into dimensionless form. In order to do this, we have to choose length, time, and temperature scales. The thickness h of the layer is the obvious natural length scale. The thermal diffusion time over the distance h, $\tau_{\theta} = h^2/\kappa$, being retained as the time scale (the alternate possibility would be the viscous time $\tau_v = h^2/\nu$), the velocity scale then reads $h/\tau_{\theta} = \kappa/h$. Since it is preferable to keep ΔT as the control parameter, the composite quantity³ $\kappa \nu/\alpha g h^3$ is taken as the temperature scale. Performing the changes $x \mapsto hx$, $t \mapsto \tau_{\theta} t, \ldots$ in (3.3, 3.4) we obtain

$$\partial_t v_z = P \left(\partial_{xx} v_z + \theta \right) \,, \tag{3.17}$$

$$\partial_t \theta = \partial_{xx} \theta + R v_z \,. \tag{3.18}$$

A second dimensionless number has been introduced:

$$P = \frac{\nu}{\kappa} = \frac{\ell^2/\kappa}{\ell^2/\nu} = \frac{\tau_\theta}{\tau_v}.$$
(3.19)

It is called the *Prandtl number* and characterizes the physical properties of the fluid, specifying which of the viscous diffusion $(\tau_v = h^2/\nu)$ or the thermal diffusion $(\tau_{\theta}$ defined above) is the dominant relaxation process.

In gases P is of the order of unity and varies little with the nature of the gas since momentum (τ_v) is transported by the molecules themselves at the same rate as energy (τ_{θ}) . In condensed fluids this number can largely vary. For example, in liquid metals (*e.g.* mercury) it is very small, typically $< 10^{-2}$, since energy is efficiently transported by conduction electrons while atoms must be moved to smooth out velocity fluctuations, hence $\tau_{\theta} \ll \tau_v$. In isolating fluids, thermal diffusion mainly involves molecular vibrations

³From expression (3.2) for the Rayleigh number, it is easily checked that it is homogeneous to a temperature.

that keep the same order of magnitude whatever the fluid, while the viscosity can vary by large amounts. P is of the order of 2–10 in water or alcohol, 10^2-10^4 in silicon oils depending on the polymerization degree (length of molecules), and essentially infinite for the Earth mantle which is extraordinarily viscous and in which convection develops on geological times only.

In the limit $P \gg 1$, the flow adjusts itself to the temperature field instantaneously, which can be understood from the consideration of equation (3.17) written as

$$P^{-1}\partial_t v_z \simeq 0 = \partial_{xx} v_z + \theta \,,$$

showing that v_z is merely obtained by integrating θ over space. The dynamics is therefore simplified since we have just one relevant scalar field. On the contrary, when P is small, the inertia of the fluid cannot be neglected and a full hydrodynamic problem is recovered, with the vector nature of the velocity field and the incompressibility condition playing a crucial role.

Let us stay in the limit $P \gg 1$ and consider the critical mode $(v_z, \theta) \sim \sin(k_c x)$ with $k_c \simeq \pi \ (\pi/h$ if the physical dimension is restored). Within the framework of the simplified model we get:

$$v_z = heta/\pi^2$$
,

and upon insertion in (3.18):

$$\partial_t \theta = -\pi^2 \theta + R v_z = \left(-\pi^2 + R/\pi^2\right) \theta$$
.

Dividing both members of this equation by π^2 and defining:

$$au_0 = rac{1}{\pi^2} \qquad {
m and} \qquad r = rac{R-R_c}{R_c}\,, ag{3.20}$$

with here $R_{\rm c} = \pi^4$ (but this value is only anecdotal) we simply get:

$$\tau_0 \partial_t \theta = r \,\theta \,. \tag{3.21}$$

The coefficient τ_0 therefore presents itself as a characteristic evolution time $(\tau_{\theta}/\pi^2 \text{ in physical units})$ for convection, while r measures the relative distance to the threshold and is of course our control parameter.

Defining A as the amplitude of the most unstable convection mode and setting:

$$\theta \propto A(t)\sin(k_{\rm c}x),$$
(3.22)

we get from (3.21) the linear evolution equation for A

$$\dot{A} = \sigma A \,, \tag{3.23}$$

where $\sigma = r/\tau_0$ is its effective growth-rate. The corresponding time $\tau = 1/\sigma = \tau_0/r$ therefore diverges as r^{-1} close to the threshold $(r \ll 1)$, a phenomenon called the *critical slowing-down*. Amplitude A plays the role of an *effective degree of freedom* for the fluid layer as a whole.

Result (3.23) is valid much more generally than suggested by the derivation above on the special case $P \to \infty$, and indeed holds in the vicinity of any linear instability. This is a consequence of the fact that, a priori, σ is a non-singular function of the parameters, and thus can be expanded in Taylor series. Since the condition that defines the threshold $r_c = 0$ is precisely $\sigma = 0$, generically the expansion begins with its first order term $\sigma = r \partial_r \sigma|_c$, hence the observed behavior of σ as a function of r.

The argument just produced can be repeated for a value of k different from the critical value k_c provided that we replace the threshold R_c by the corresponding marginal value $R^{(m)}(k)$. As long as k stays sufficiently close to k_c , the natural characteristic evolution time has no reason to be very different from τ_0 , so that we can write at lowest order

$$\tau_0 \sigma(k) = \frac{R - R^{(m)}(k)}{R^{(m)}(k)} \,. \tag{3.24}$$

On the other hand, any curve in the vicinity of an *extremum* is generically equivalent to a parabola. The marginal curve close to its minimum at (k_c, R_c) is not an exception so that, for $k = k_c + \delta k$ and $\delta k/k_c \ll 1$, we can write

$$\frac{R^{(\mathrm{m})}(k) - R_{\mathrm{c}}}{R_{\mathrm{c}}} = \xi_0^2 \,\delta k^2 \,, \tag{3.25}$$

where ξ_0^2 presents itself as the square of a characteristic length, the *coherence length*, which accounts for the curvature of the marginal stability curve at threshold.

3.1.6 Classification of unstable modes

One can arrange (3.24) and (3.25) together to write down the real part of the dispersion relation in the condensed form:

$$\tau_0 \sigma(k) \simeq r - \xi_0^2 \left(k - k_c\right)^2. \tag{3.26}$$

When the minimum of the marginal stability curve is reached for $k_c \neq 0$, case considered up to now of convection in a simple fluid, one says that the instability is *cellular*. Otherwise, it may happen that the most unstable mode is for $k_c = 0$, the instability is then termed *homogeneous*. This situation, which occurs for example when convection takes place between horizontal plates that are bad thermal conductors, is often difficult to treat since the system is then sensitive to lateral boundary conditions and/or any kind of slowly varying perturbations, while when $k_c \neq 0$, each cell with width $\lambda_c/2$ plays its own game, without worrying about lateral boundaries as soon as they are sufficiently far apart, say three or four wavelengths.

From a temporal viewpoint, Rayleigh-Bénard convection in a simple fluid is a stationary instability, the imaginary part $\omega(k)$ of the dispersion relation is identically zero. In other cases, the instability may be oscillatory with $\omega_c \neq 0$, where ω_c is the angular frequency at threshold. When the instability sets in with $k_c \neq 0$ and $\omega_c \neq 0$, the critical mode is in fact a wave propagating at some phase velocity c since, factoring out k_c one can write $\exp(i(k_c x - \omega_c t) \boxminus \exp(ik_c(x - ct))$, as systematically done in Chapter 6.

The real part σ of the eigenvalue of the marginal mode is well approximated by (3.26) in the neighborhood of the threshold (r_c, k_c) . In the same way, its imaginary part $\omega(k)$ can be expanded as:

$$\omega(k) = \omega_{\rm c} + r \,\partial_r \omega|_{\rm c} + \delta k \,\partial_k \omega|_{\rm c} + \frac{1}{2} \delta k^2 \,\partial_{kk} \omega|_{\rm c} \,, \tag{3.27}$$

where derivatives with respect to r or k are computed at threshold.

Considering this expansion in more detail, let us notice first that the coefficient of δk (third term on the r.h.s.) corresponds to the group velocity of the waves. This can be seen by looking at a wave packet formed by superposition of elementary waves written as $V(x,t) = \int A(k) \exp(i(kx - \omega t) dk)$, where A(k) is the amplitude of mode k presenting a peak around some wavevector $k = k_0$. Setting $\omega_0 = \omega(k_0)$, we get:

$$V(x,t) = \exp(i(k_0x-\omega_0t){\int}A(k_0+\delta k)\exp[i\delta k(x-\partial_k\omega|_{k_0}t)+\mathcal{O}(\delta k^2))]\,\mathrm{d}\delta k.$$

In the long time limit $(t \gg 1/\omega_0)$, V is negligible everywhere except where the argument of the exponential is zero ('stationary phase' approximation) since elsewhere the rapid oscillations of the complex exponential "kill" the signal. This happens when $x/t = \partial_k \omega|_{k_0}$ which shows that this quantity is precisely the velocity of the wavepacket's peak. In the same way, the coefficient of δk^2 in (3.27) accounts for the *dispersion* of wavepackets, *i.e.* their smearing out due to changes in phase velocity. In a non-dispersive medium, the phase velocity is independent of the wavevector, *i.e.* $\partial_k(\omega/k) = 0$, so that $c_g \equiv \partial_k \omega = \omega/k \equiv c$ and of course $\partial_{kk} \omega \equiv 0$.

The Taylor-Couette instability of a fluid sheared between two coaxial cylinders rotating at different angular speeds (Exercise 3.3.5) is also cellular and stationary. In chemistry, the Belousov-Zhabotinsky reaction is an example of oscillatory homogeneous instability. Finally, in some circumstances, convection in binary fluid mixture develops in the form of dissipative waves (Exercise 3.3.3).

3.2 Disintegration of Dissipative Structures

The study of the transition to turbulence of structures generated by an instability mechanism consists of several steps. The first one is the determination of equilibria achieved beyond threshold. The next relates to the destabilization of such equilibria, and so on. The game is then repeated up to a point where the regime obtained is completely irregular. In this section we begin with a simple modeling of nonlinear effects in convection, $\S3.2.1$. A brief account of experimental observations about the transition is then given in $\S3.2.2$, where we point out the role of geometrical effects. This leads to a fundamental distinction between confined systems for which the concept of temporal chaos is relevant, §3.2.3, and extended systems for which the disorganization in space is as important as the irregularity in time, *i.e. spatio-temporal chaos*, §3.2.4. We conclude the chapter by a brief presentation of convection in the post-transitional regime where the concept of developed turbulence begins to make sense, §3.2.5. Here we mostly stay at a phenomenological level, deferring the introduction of theoretical tools to subsequent chapters.

3.2.1 Simplified model of nonlinear convection

Relation (3.22) defines a variable A measuring the intensity of the perturbation to the base state. At steady state, we thus expect $A \equiv 0$ below threshold and $A \neq 0$ above. In the theory of thermodynamic *phase transitions* A would be called an *order parameter* [Stanley (1988)]. However, Eq. (3.23) is valid only as long as A stays infinitesimal and must be completed to account for the range r > 0. In order to get Eq. (1.19), p. 13, we just postulated heuristically that convection was a self-limiting process and we replaced σ in (3.23) by an effective value,⁴

$$\sigma_{\mathrm{eff}} = au_0^{-1}(r - gA^2) \qquad \mathrm{with} \qquad g > 0 \, .$$

Beyond threshold, for r > 0 $(R > R_c)$, several processes indeed come and limit the growth of A. First the dissipation increases, and second the destabilizing force decreases since part of the heat is transported by the flow, so that the bulk effective temperature gradient that governs the conductive part of the heat flux decreases below its nominal value β .

It is of course possible to derive an accurate model of nonlinear evolution from the primitive equations in a systematic way. Here we rather continue to develop a heuristic formulation, guided by the result to be obtained. We no longer assume that $P \gg 1$ but restrict ourselves to the consideration of the most unstable linear mode. On more general grounds than for (3.22), we then take $\{v_z, \theta\} = \{V(t), \Theta(t)\} \sin(k_c x)$, where V et Θ are two amplitudes functions of time. Injecting this assumption in (3.17, 3.18) we obtain:

$$\dot{V} = P(\Theta - \pi^2 V), \qquad (3.28)$$

$$\dot{\Theta} = RV - \pi^2 \Theta \,. \tag{3.29}$$

Let us now try to complete (3.28, 3.29) with nonlinear terms arising from the advection of the fluctuations $\mathbf{v}\nabla\mathbf{v}$ and $\mathbf{v}\nabla\theta$. In the spirit of a first harmonic approximation now developed in space and not in time as for the van der Pol oscillator, p. 49, we guess that the terms that contribute are those resonating with the postulated dependence in $\sin(k_c x)$. From the continuity equation $\partial_x v_x + \partial_z v_z = 0$, assuming $v_z \propto \sin(k_c x)$ one gets $v_x \propto \cos(k_c x)$, so that, in the equation for v_z , the advection term $v_x \partial_x v_z +$ $v_z \partial_z v_z$ varies as $\sin^2(k_c x) = \frac{1}{2}(1 - \cos(2k_c x))$, *i.e.* produces nothing in resonance with $\sin(k_c x)$. Averaging over the thickness of the layer and over a wavelength, we thus expect a negligible contribution from these terms to (3.28) that remains unchanged at this order.

The problem is different for (3.29). As a matter of fact, a parallel argument would also imply no complementary term, but this would not reflect the fact that, as indicated above, part of the heat is transported by the convection motion. The corresponding flux is easily identified to the product $v_z \theta$. As discussed in Sec. 3.1.2, below threshold, the destabilizing part of the convection mechanism relies on the advection, due to differential

⁴Notation g introduced to measure the intensity of nonlinear couplings is traditional. In the context of convection it should not be mistaken with the gravitational acceleration, but the risk is limited.

buoyancy, of temperature fluctuations in a purely conductive temperature gradient. Above threshold, since part of the heat is transported by convection, differential buoyancy has to be appreciated with respect to a conductive temperature gradient β_{eff} which is decreased by the contribution of convection from its nominal value β evaluated from the applied temperature. That contribution, $v_z\theta \propto \sin^2(k_c x)$, produces (i) a second harmonic component $\propto \cos(2k_c x)$ that averages to zero over a fluctuation wavelength but also (ii) a term at k = 0 that precisely corresponds to the correction to the averaged temperature profile.

Let us consider this correction as a variable in itself, call it Ψ and look for its governing equation. A simple calculation then yields:

$$\dot{\Psi} = V\Theta - b\Psi, \qquad (3.30)$$

where the first term on the right hand side is the source term issued from space-independent part of $v_z \theta$ and the second term accounts for its diffusive relaxation according to the Fourier law, at a decay rate b that could be computed explicitly.

On the other hand, the argument sketched above is implemented by subtracting the convective contribution Ψ from the nominal Rayleigh number R to form an effective Rayleigh number R_{eff} replacing it in (3.29). This yields:

$$\dot{\Theta} = (R - \Psi)V - \pi^2 \Theta. \qquad (3.31)$$

Equations (3.28, 3.31, 3.30) generalize the linear model derived previously. They form the celebrated *Lorenz model* that played such an important role in the development of ideas about chaos since 1963, when Lorenz first derived it⁵ by truncating a Galerkin expansion of the hydrodynamic equations on a trigonometric function basis, and further pointed out its "non-periodic" properties (see Appendix B, §B.4.2, p. 373).

Let us first show how this model allows us to recover the effective Landau equation (1.19) introduced in Chapter 1, extending (3.23) to the nonlinear regime. As noticed earlier, close to the threshold, the dynamics of the system is very slow. Its evolution rate is proportional to $r = (R - R_c)/R_c \ll 1$. But equation (3.30) shows that the natural relaxation time of correction Ψ remains $\mathcal{O}(1)$. We can thus assume that Ψ rapidly relaxes towards a value $V\Theta/b$, itself slowly varying at a rate $\mathcal{O}(r)$. Let us insert this value in (3.31)

⁵E.N. Lorenz, "Deterministic non-periodic flow," J. Atm. Sc. **20** (1963) 130-141. The original expression of the system is recovered by appropriate rescalings of time, variables V, Θ , and Ψ , and parameter R.

and admit that $V = \Theta/\pi^2$ for all times, which, from (3.28), is true only in the limit $P \to \infty$. We get:

$$\dot{\Theta} = [(R/\pi^2) - \pi^2]\Theta - \Theta^3/\pi^4 b \,,$$

that we rewrite as:

$$\frac{1}{\pi^2} \dot{\Theta} = \frac{R - \pi^4}{\pi^4} \Theta - \frac{1}{b\pi^6} \Theta^3 \,. \tag{3.32}$$

This equation is therefore exactly (1.19), p. 13, i.e.:

$$\tau_0 \dot{A} = rA - gA^3 \,, \tag{3.33}$$

with Θ , the amplitude of the temperature modulation playing the role of the effective variable A, $\tau_0 = 1/\pi^2$, $r = (R - R_c)/R_c$, $R_c = \pi^4$, and $g = 1/b\pi^6$.

Obtaining the effective model (3.33) is an example of reduction by *adia*batic elimination of enslaved variables. Here variables V and Ψ are enslaved to Θ : At every instant their values are fixed by that of $\Theta \equiv A$ according to relations $V = \Theta/\pi^2$ and $\Psi = \Theta^2/b\pi^2$. This step plays an essential role in the study of nonlinear dissipative systems.

Equation (3.33) accounts for the *bifurcation* from the conduction branch corresponding to the trivial solution $A \equiv 0$ towards the convection branch associated to the pair of (here time-independent) nontrivial solutions. These *bifurcated solutions*, $A_{(\pm)} = \pm \sqrt{r/g}$, are given by the condition $\dot{A} = 0$, and therefore correspond to *fixed points* of (3.33). Quantity g being positive, they appear for r > 0 so that the bifurcation is *supercritical*. All this has already been presented in the introductory chapter, see Figure 1.4 p. 14, and will be extended to general systems involving a single stationary mode in Chapter 4, especially in Exercise 4.6.3, p. 169.

We have previously stated without justification that the bifurcated solutions corresponding to convection beyond threshold are stable. Let us show how this arises from (3.33) using tools introduced in Chapter 2. Setting $A = +\sqrt{r/g} + A'$, we easily get the equation for perturbation A' by substitution. After simplification we get

$$\tau_0 \dot{A}' = -2rA' \,, \tag{3.34}$$

so that A' decays for r > 0. In fact this is valid only close enough to the threshold, before new instabilities have any chance to set in.

The previous argument indeed only holds as long as $R \approx R_c$. When $R \gg R_c$, the relaxation time of Θ and V towards their equilibrium values, derived from (3.34), infinite at threshold, shortens as r increases and rapidly

becomes of the order of magnitude of Ψ 's relaxation time. Adiabatic elimination of the latter is then no longer legitimate: Ψ is less and less enslaved to Θ and V but on the contrary gains a status of genuine degree of freedom. Secondary instabilities and the transition to chaos could be studied from the Lorenz system but, as far as physical observations are concerned, results turn out to be unrealistic, due to the simplifications made. Having recognized that this model remains an excellent test-bed for chaos ideas (see Appendix B, §B.4.2, for a first numerical approach), we now consider nonlinear convection from a concrete experimental point of view.

3.2.2 Transition to turbulence of convection cells

The actual situation is indeed less transparent than that described by means of the simplified Lorenz model. But one fact remains: the effective dimension of the problem increases with R. Unfortunately, the physical mechanisms that destabilize the cellular structure to produce of the *secondary* modes are much less intuitive than the primary mechanism.

The method is the same in principle but, at steady state, the base flow beyond threshold is now made of finite-amplitude time-independent convection cells. The study is considerably more complicated than when we had to deal with the uniform conducting state since the new base flow is periodic along one horizontal direction. Accordingly, the operator obtained through linearization now explicitly depends on space, which forbids the direct recourse to Fourier transforms to solve the problem. This will be reexamined theoretically later. For the moment, let us describe the cascade towards turbulence from a phenomenological point of view.

The convection threshold was independent of the Prandtl number P, whose value just played some role in the nature of the primary mode, thermal when $P \gg 1$, hydrodynamic when $P \ll 1$. This simple fact has profound consequences on the shape of the secondary modes and the subsequent cascade of bifurcations towards turbulence. A compilation of early results adapted from Krishnamurti⁶ is displayed in 3.4.

Upon increasing R, the fluid layer first experiences a transition from pure conduction (fluid uniformly at rest) to two-dimensional time independent convection (2D: fluctuations depend locally on two coordinates, say x and z). At sufficiently large Prandtl number a three-dimensional regime sets in (3D: fluctuations now depend on x, y, z), at first time-independent,

⁶R. Krishnamurti, "Some further studies on the transition to turbulent convection," J. Fluid Mech. **60** (1973) 285.



Fig. 3.4 Transition toward turbulence in convection, after Krishnamurti, Note 6. The Prandtl number is varied by changing the fluid. The transition lines are intentionally made thick to indicate orders of magnitude rather that precise thresholds.

next periodic, and eventually turbulent. At small P the 'time-independent two-dimensional' domain is very narrow and an irregular time dependence rapidly sets in, here called 'turbulent convection'.

3.2.2.1 Large Prandtl number fluids

When $P \gg 1$ (e.g., with highly viscous oils), the temperature field drives everything, inducing the vertical velocity component directly and the horizontal component indirectly through the continuity condition.

Secondary instabilities specific to this case remain localized within thermal boundary layers close to the horizontal plates. These boundary layers get thinner and thinner as the Rayleigh number is increased and, at some point, they become unstable against the plain Rayleigh mechanism. A stationary secondary instability called *bimodal* sets in, with rolls oriented at right angles with the primary rolls and located in the thermal boundary layers. Since the fluctuations are now modulated in the three directions of space, the regime is labelled 'time-independent 3D' in Figure 3.5. Time dependence next manifests itself as a periodic break-down and reformation of thermal boundary layers first analyzed by Howard. Strict periodicity is then lost and an irregular dynamics sets in.

3.2.2.2 Intermediate and low Prandtl number fluids

At smaller P, the situation is more confused. Busse and his collaborators have identified a large number of possible secondary modes leading to a complicated picture in the (R, P, k) parameter space called the *Busse* balloon⁷ owing to the global shape of the region where straight rolls are stable. When $P \sim 1$ (water, gases) or smaller (liquid metals), the velocity field becomes dominant through specific contributions of the advection term $\mathbf{v} \cdot \nabla \mathbf{v}$. The unstable secondary modes appear close to the convection threshold and occupy the whole thickness of the layer. Cells enter a kind of free-wheel regime where friction on the plates is dealt with inside thin viscous boundary layers. Time dependence enters very early in the form of 'Busse oscillations' that are sorts of waves propagating along the convection rolls and due to an inertial call-back of roll axis undulations. Most often it turns out to be difficult to identify a range of Rayleigh numbers over which the periodic behavior is strictly regular, and the flow is often considered turbulent right at the onset of oscillations.

3.2.2.3 Transition towards turbulence, conceptual problems

At least for $P \gg 1$ there seems to be a small number of well defined steps on the way between the conduction regime and turbulence. This apparently supports the viewpoint advanced by Ruelle et Takens in 1971⁸ according to whom the stochastic behavior, a fundamental property of turbulence, generically appears at the end of a short cascade of three or four bifurcations. Previously, Landau⁹ explained his understanding of turbulence as the result of an indefinite superposition of modes, each with its own time-space scale, *i.e.* quasi-periodicity with an infinite number of incommensurate frequencies. These two interpretations are sketched in Figure 3.5.

In the context of experiments reported above (Fig. 3.4), neither the Ruelle-Takens interpretation nor *a fortiori* that of Landau, are satisfactory. All observations were made in containers that were very wide (in order to check theories developed for a laterally unbounded system). The

⁷For an early review, see, *e.g.* F.H. Busse, "Transition to turbulence in thermal convection" in Convective transport and instability phenomena, J. Zierep & H. Oertel Jr., eds. (Braun, Karlsruhe, 1982).

⁸D. Ruelle and F. Takens, "On the nature of turbulence," Commun. Math. Phys. **20** (1971) 167-192. Addendum **23** (1971) 343-344.

⁹L.D. Landau, "On the problem of turbulence," Akad. Nauk. Doklady 44 (1944) 339, translation in *Collected Papers of L.D. Landau*, D. ter Haar ed. (Pergamon Press, 1965), pp. 387-391.



Fig. 3.5 The "nature of turbulence" according to Landau (top) and Ruelle & Takens (bottom). Ellipses attached one to the next in a string feature quasi-periodic behavior as results from the superposition of periodic motions with incommensurate frequencies. The chaotic end of the Ruelle-Takens cascade is represented by a miniature of the attractor of the Curry & Yorke model that illustrates the disintegration of the ellipse born at the previous step (see Chapter 4, Exercise 4.6.13).

so-obtained convection patterns were rarely regularly organized but on the contrary presented lots of defects and inhomogeneities, so that the transition thresholds were not as sharply defined as a bifurcation point. Moreover a slow residual time dependence was often observed.

Having recognized that these interpretation problems were mostly due to spatial disorder, which in turn resulted from the presence of a large number of cells, and that lateral boundaries at large distances were ineffective in maintaining long range order in the patterns, experimentalists have tried to better control the situation by turning to systems with a small number of cells, hence lateral dimensions of containers of basically the same order of magnitude as their heights.

Confinement effects can be appreciated through aspect ratios defined as

$$\Gamma = \ell/h \,, \tag{3.35}$$

where ℓ represents the typical lateral extension of the system, see Figure 3.6. We shall reexamine their physical role later in Chapter 4, §4.1.

Early experiments reported above were performed in the limit $\Gamma \gg 1$ that characterize *extended systems* and for which the concept of *spatio-temporal chaos* to be introduced in Chapter 5 seems more appropriate.


Fig. 3.6 Aspect ratio for closed systems, either confined (left) or extended (right).

By contrast, strongly *confined* systems, characterized by $\Gamma \sim 1$, can be expected to better fit the framework proposed by Ruelle and Takens and their concept of *temporal chaos*. As a matter of fact, guaranteeing strong spatial coherence among a small number of convection cells, confinement effects should be instrumental in restricting the dynamics to couplings within a small set of effective variables.

3.2.3 Transition toward chaos in confined systems

The literature about the transition from regular to chaotic time behavior is sufficiently rich that we can limit ourselves to the presentation of few experimental results obtained at the beginning of the eighties as typical examples of the main scenarios. This sketchy description is given mainly as an illustration of the kind of phenomena to be understood theoretically rather than as a review that would rather be premature at this stage. Consult the general bibliography for more detailed information, especially [Hao (1990); Cvitanović (1989)].

3.2.3.1 Subharmonic cascade

The first experiment to be reported here has been performed by Libchaber et Maurer.¹⁰ Liquid helium with $P \sim 1$ is placed in a parallelepipedic container with aspect ratios $\Gamma_x = 2.4$, $\Gamma_y = 1.2$. At the beginning, stationary convection sets in beyond some threshold R_c . At $R \simeq 30R_c$ the system experiences a bifurcation toward an oscillatory regime. Then, at $R \simeq 39.5R_c$, a second mode with an incommensurate period sets in. This two-periodic

¹⁰A. Libchaber and J. Maurer, "Une expérience de Rayleigh-Bénard en géométrie réduite; multiplication, accrochage et démultiplication de fréquences," J. Physique Colloques **41-C3** (1980) 51.



Fig. 3.7 Time series of the temperature signal measured at a given point during the first steps of a subharmonic cascade after Libchaber and Maurer, Note 10.

regime persists up to $R \simeq 40.5R_c$ when, while shifting, the second period gets locked to twice the first one, the system is then periodic with a period 2T. The scenario under study now begins: a second period doubling (period 4T) at $R \simeq 42.7R_c$ (Figure 3.7). After several supplementary period doublings (period 8T, 16T,... the system enters a chaotic regime for $R > 43R_c$. A complementary study of Fourier spectra would show first fine lines at one frequency and its harmonics, then a second family of lines and many combinations (two-periodic regime), then, after the locking, a return to a simpler spectrum with one fundamental line at $\omega = 2\pi/T$ and its harmonics. The period doubling cascade manifests itself by the growth of subharmonics at $\omega/2$, next $\omega/4$, etc. As long as the system is periodic, no matter how long the period, the spectral lines remain narrow but when it becomes chaotic, they get measurably enlarged at their foot.

3.2.3.2 Chaos on a two-periodic background

This transition, closely reminiscent of the scenario originally proposed by Ruelle and Takens, has been observed roughly at the same epoch by Dubois and Bergé¹¹ again in parallelepipedic geometry with similar aspect ratios, $\Gamma_x = 2$, $\Gamma_y = 1.2$, but this time with silicon oil ($P \simeq 130$). By contrast with the previous experiment, visualization of the structure was possible by means of differential interferometry, Fig. 3.8 (top), which made easier the understanding of motions at the origin of the observed fluctuations and the choice of points where to measure the velocity by LASER Doppler anemometry.

Here is the (simplified) sequence observed: 1) conduction regime up to R_c . 2) Time-independent convection from R_c to $R \simeq 215R_c$. 3) Bifurcation towards a periodic regime with period T_1 . 4) Two-periodic dynamics with a second period T_2 from $R \simeq 250R_c$ up. Geographically well separated

 $^{^{11}\}mathrm{M.}$ Dubois and P. Bergé, "Instabilités de couche limite dans un fluide en convection: évolution vers la turbulence," J. Physique **42** (1981) 167.



Fig. 3.8 Top: Isotherms in silicon oil can be visualized by differential interferometry; the fringes originate from the variations of the refraction index induced by the local temperature gradients (picture kindly provided by M. Dubois). Bottom: Fourier spectra (left) and time series (right) of a velocity component at a given point in the experimental cell. In the two-periodic regime, the spectrum is essentially composed of two principal lines, each of which can be attributed to a definite fluid perturbation well localized in space. The chaotic regime is characterized by spectral lines with enlarged foot and a large amount of power at low frequency that corresponds to the slow fluctuations in the corresponding time series. After Dubois and Bergé, Note 11.

the two oscillation modes are weakly coupled, which explains the relative robustness of the two-periodic regime and a characteristic alternation of locking/unlockings when the ratio of the periods, that slightly shifts with R, passes from incommensurate to commensurate values and vice versa. 5) For $R > 305R_c$, temporal chaos enters as an irregular slow modulation of a locked periodic behavior that gives a series of widened spectral lines and low frequency power in the Fourier spectrum, Fig. 3.8 (bottom).



Fig. 3.9 The intermittency scenario: A periodic regime ("laminar" intermissions) is irregularly interrupted by chaotic bursts that become more frequent as R increases beyond the *intermittency threshold*. Upper trace: regular signal observed below threshold. Time is running from left to right. After Bergé *et al.*, Note 12.

3.2.3.3 Intermittency

The third scenario to be described here has also been observed by Bergé and Dubois¹² with the same fluid and the same experimental set-up but with a slightly different initial convection structure. Accordingly, a different transition scenario developed after a single step involving a secondary instability mechanism with a hot droplet transported by the general convection and playing the role of a pacemaker. Convection was time-independent up to $R \simeq 250R_{\rm c}$, then periodic with period T. Not far above a subharmonic bifurcation (hence period 2T), for $R = 290R_{\rm c}$, the system experienced a transition to chaos with irregularly distributed "turbulent" bursts interrupting the previously observed regular periodic behavior forming "laminar" intermissions. When the Rayleigh number was increased, the frequency of the bursts was seen to increase. Time series of the velocity signal before and after the transition are displayed in Fig. 3.9, upper trace and lower trace, respectively.

3.2.4 Dynamics of "textures" in extended systems

By contrast with what has just been described, before 1975 the transition to turbulence was studied in extended systems and focused more on the occurrence of a developed turbulent regime where most of the spatial structure was lost and the time dependence strongly irregular. They did not recognize the often observed low frequency noise as an interesting phenomenon related to the transition process. The study of the emergence of chaos in confined systems has also led to reconsidering the situation for extended

¹²P. Bergé, M. Dubois, P. M., Y. Pomeau, "Intermittency in Rayleigh-Bénard convection," J. Physique Lettres **40** (1979) L505.



Fig. 3.10 Texture observed in convection at large Prandtl number as seen from above. Picture kindly provided by V. Croquette, Note 13.

systems and introducing the notion of spatio-temporal chaos as an element of interpretation of the transition process. In practice, the instability mechanisms preserve coherence at the *local* scale (few convection cells) but are unable to maintain it on a global scale (the set-up). This can be understood as the result of the possible interference of a large number of neighboring modes easily excited immediately beyond threshold (§3.1.5 and Chapter. 5). In the absence of any induction process forcing the growth of regularly oriented "clean" convection structures, disordered patterns with many defects of all sorts, called *textures*, are generally observed. Here again we have to distinguish between fluids according to their Prandtl number.

3.2.4.1 Textures in fluids with high Prandtl number

When P is large, quasi-stationary convection structures are obtained, which relates to the fact that the dominant field is the temperature, a scalar, and that the fluid layer mostly behaves as a gradient system (see Chapter 2, §2.1.3, p. 33). It is then possible to interpret the principal features of the textures observed in terms of a single amplitude field. An example is given in Fig. 3.10, where one can easily identify grains of convection rolls with nearly uniform orientations, grain boundaries along which two grains with different orientations meet, dislocations¹³ where a pair of rolls suddenly ends. It should also be noted that rolls arrive mostly perpendicular to the lateral boundaries and that the frustration implied by this topological constraint is partly resolved by the presence of a large scale curvature of the rolls. The evolution of such textures is very slow, when compared to the

¹³A. Pocheau & V. Croquette, "Dislocation motion: a wavenumber selection mechanism in Rayleigh-Bénard convection," J. Physique **45** (1984) 35-48.



Fig. 3.11 Weak turbulence observed in a cylindrical cell with $\Gamma = D/h = 12$ in liquid helium at low Prandtl number. Typical background noise is the gray line in each figure. For $R = 3.35R_c$ low frequency noise develops first. At $R = 3.70R_c$ noisy Busse oscillations at finite frequency around 0.5 Hz have settled in the system. As seen for $R = 4.50R_c$ and beyond, the level of noise continues to increase gently. After Libchaber and Maurer, Note 14.

velocity of the convecting fluid. The transition from 2D to 3D convection is mediated by the defects and the emergence of (weak) turbulence can be understood as a kind of "melting" of the global pattern, with progressive loss of local order.

3.2.4.2 Transition at small Prandtl number and large aspect ratios

All experimental observations show that turbulence occurs early when P is small. This feature has to be attributed to the fact that the viscosity is low and that Reynolds numbers constructed from the velocity induced by convection and the size of the cells rapidly become large. Like for low-dimensional systems, the complexity of the dynamics is considerably enriched by inertial effects that favor oscillatory behavior (here mainly the Busse oscillations). This explains that mode interactions, even close to the convection threshold, generate a much more "active" behavior than what is observed at higher Prandtl numbers. Another source of complexity comes from the existence of large scale flows directly generated by curvature and defects in the global texture as now shown.

Figure 3.11 illustrates the scenario observed again by Libchaber and Maurer¹⁴ in liquid helium but in a cylindrical container with diameter D and aspect ratio $\Gamma = D/h = 24$. Remarkably enough, a low frequency noise sets in before the trace of any secondary instability in the form of Busse oscillations is apparent, and when the latter develops the system is already disordered so that the corresponding frequency is not sharply

¹⁴A. Libchaber and J. Maurer, "Local probe in a Rayleigh-Bénard experiment in liquid helium," J. Physique Lettres **39** (1978) L-69.



Fig. 3.12 Cyclic evolution from state $(1) \rightarrow (2) \rightarrow \ldots \rightarrow (6) \rightarrow (1)$: Mechanism of nucleation-dissociation-migration-annihilation of dislocations pointed by Pocheau *et al*, Note 15, and here illustrated with pictures kindly provided by V. Croquette from a similar experiment with $\Gamma = D/h = 40$).

defined. It took some time before this behavior could be understood.

In an experiment where helium was replaced by argon under pressure at room temperature, making visualizations possible, Pocheau *et al.*¹⁵ later showed that the noise developing slightly above threshold was due to the synchronization loss of an initially periodic process of nucleation, migration, annihilation of dislocations, as illustrated in Figure 3.12. The dislocation migration was driven by a secondary flow at the scale of the container, the existence of which was predicted by earlier theoretical studies¹⁶ and explicitly confirmed by specific experiments later.¹⁷

Still for low Prandtl numbers, somewhat above the range of Rayleigh numbers where the convection pattern is made of possibly slowly evolving but essentially straight rolls, a much more disorganized active state is observed in the form of rotating spirals as illustrated in Figure 3.13. This

¹⁷V. Croquette, P. Le Gal, A. Pocheau, P. Guglielmetti, "Large-scale flow characterization in a Rayleigh-Bénard convective pattern," Europhys. Lett. **1** (1986) 393-399.

¹⁵A. Pocheau, V. Croquette, and P. Le Gal, "Turbulence in a cylindrical container of Argon near threshold of convection," Phys. Rev. Lett. **55** (1985) 1094–1097, later reviewed by V. Croquette, "Convective Pattern Dynamics at Low Prandtl Number. Part I, II," Contemporary Physics **30** (1989) 113-133, 153–171.

¹⁶(a) E.D. Siggia, A. Zippelius, "Pattern selection in Rayleigh-Bénard convection near threshold," Phys. Rev. Lett. **47** (1981) 835-838. (b) P. M. and J.M. Piquemal, "Transverse phase diffusion in Rayleigh-Bénard convection," J. Physique Lettres **43** (1982) L253-L258. (c) M.C. Cross, A.C. Newell, "Convection patterns in large aspect ratio systems," Physica D **10** (1984) 299-328.



Fig. 3.13 Convection in CO₂ with $\Gamma = D/h \simeq 150$, spiral defect chaos observed at r = 0.536 (left) and r = 0.894 (right). After Morris *et al.*, Note 18, courtesy G. Ahlers (UCSB).

spiral defect chaos, observed in particular by Morris et al.¹⁸ permanently evolves in both space and time. It takes place only at large aspect ratios but then does not depend on the lateral shape of the container. Furthermore, it is extensive in the sense that it can be characterized by a surface density of spiral cores seen to increase with increasing Rayleigh numbers beyond a threshold that depends on the Prandtl number. Curvature-induced secondary flows seem essential to its occurrence.

3.2.5 Turbulent convection

The local study of convection structures has, since its very beginning, been completed by measurements of the heat flux through the whole experimental container, characterizing the global behavior. Results are usually expressed in terms of the dimensionless *Nusselt number*:

$$N = \frac{\text{total heat flux}}{\text{conduction heat flux}}$$

where the 'total heat flux' is the quantity actually measured and the 'conduction heat flux' is the flux that would be computed from the temperature difference upon assuming that the fluid is at rest in the pure conduction state. Hence one gets $N \equiv 1$ when $R < R_c$, while N - 1 measures the

¹⁸S.W. Morris, E. Bodenschatz, D.S. Cannell, G. Ahlers, "The spatio-temporal structure of spiral-defect chaos," Physica D 97 (1996) 164–179.

contribution of convection. Close to threshold, one expects:

$$N - 1 \propto v_z \theta \propto \frac{R - R_c}{R_c}$$
, (3.36)

since both θ and v_z vary as $[(R - R_c)/R_c]^{1/2}$, which is indeed well observed experimentally, see Figure 3.14.

Far beyond threshold, from scaling arguments familiar in the theory of turbulence (Chapter 7), a power law behavior is expected instead:

$$N \sim R^{\gamma}$$
 .

Early experiments seemed to support a theory by Malkus predicting $\gamma = 1/3$ but the range of Rayleigh number studied was too narrow, while other studies for $P \ll 1$ suggested rather $\gamma = 1/4$.

At the end of the eighties, the problem became a topic of renewed interest, experimental and theoretical. The heat flux was studied over *R*-ranges extending up to 10^6 then 10^{12} , and even 10^{17} , in fluids with various Prandtl numbers and in containers with aspect ratio of order 1/2 or 1. Exponents γ ranging from 1/2 to 1/4, through 1/3, 0.3 or 2/7, have been measured over (sometimes very) limited ranges of Rayleigh numbers. Experimental results and their theoretical understanding are still the subject of debate.¹⁹

In Figure 3.14 drawn after the results of Chavanne *et al.* in liquid helium, Note 19(c), one can identify the linear behavior close to threshold expected from (3.36), a 'soft turbulence' regime where chaos is still mostly temporal as discussed in §3.2.3, then 'hard turbulence' with an exponent $\gamma \simeq 2/7$ explained by a theory involving thermal transfer through turbulent layers sheared by the general circulation blowing as "wind" along the horizontal walls, and an "ultimate" regime with exponent tending to 1/2. The discussion bears on the existence of asymptotic regimes with a single exponent or rather on superpositions of power laws in the form

¹⁹We quote here only few references, first the general presentation by (a) E.D. Siggia, "High Rayleigh number convection," Annu. Rev. Fluid Mech. **26** (1994) 137-168, and next specific results by (b) J.J. Niemela *et al.*, "Turbulent convection at very high Rayleigh numbers," Nature **404** (2000) 837-840, err. **406** (2000) 439; (c) X. Chavanne *et al.*, "Turbulent Rayleigh-Bénard convection in gaseous and liquid He," Phys. Fluids **13** (2001) 1300-1320; a theory by (d) S. Grossmann and D. Lohse, "Scaling in thermal convection: a unifying theory," J. Fluid Mech. **407** (2000) 27-56 and "Thermal convection for large Prandtl numbers," Phys. Rev. Lett. **86** (2001) 3316-3319; corresponding experimental work by (e) G. Ahlers and X.-c. Xu, "Prandtl-number dependence of heat transport in turbulent Rayleigh-Bénard convection," Phys. Rev. Lett. **86** (2001) 3320-3323; and all references quoted by these authors.



Fig. 3.14 Total heat flux as measured in terms of the Nusselt number as a function of the Rayleigh number (in log-log scale), after Chavanne *et al*, Note 19(c).

 $N = C_1 R^{\gamma_1} + C_2 R^{\gamma_2}$ and the role of P (Note 19 d, e), of the geometry and nature of lateral walls, *etc.*

To conclude, convection presents itself as the prototype of stationary cellular instabilities. In this chapter we have described its particularly intuitive mechanism, and its subsequent destabilization up to turbulence. We have also noted the role of confinement effects on the nature of these steps. In the next two chapters we shall examine in more detail some mathematical aspects of the theory that allows us to interpret these phenomena and, at the same time, to tackle a large class of instabilities in continuous media. We shall not come back to turbulent convection, owing to the limited scope of Chapter 7 devoted to the simpler case of turbulent shear flows.

3.3 Exercises

3.3.1 Simple model of cellular instability

Consider the linear part of the Swift-Hohenberg $model^{20}$ that reads:

$$\partial_t v = rv - (\boldsymbol{\nabla}_{\perp}^2 + 1)^2 v. \qquad (3.37)$$

 $^{^{20}}$ J. Swift & P.C. Hohenberg, "Hydrodynamic fluctuations at the convective instability," Phys. Rev. A **15** (1977) 319-328.

This model accounts for the emergence of convection cells in a simplified but physically meaningful way and will be used further in Appendix B, Secs. B.4.4 and B.4.5.

Variable v may represent the vertical velocity component in the fluid or the departure from the base temperature profile, while r is the control parameter measuring the relative distance to the threshold: $r \propto (T-T_c)/T_c$. In (3.37), $\nabla_{\perp}^2 \equiv \partial_{xx} + \partial_{yy}$ is the Laplacian operator acting on the space dependence of the fluctuations in the plane of the layer.

1) Laterally unbounded medium.

a) Determine the dispersion relation $s = s(\mathbf{k})$ for Fourier modes taken in the form $\exp(i\mathbf{k} \cdot \mathbf{x})$ where $\mathbf{k} = (k_x, k_y)$ and $\mathbf{x} = (x, y)$. Check that it depends only on $k = |\mathbf{k}_h|$.

b) Draw the graph of s(k) for r < 0, r = 0 and r > 0 and conclude that the system bifurcates towards a cellular structure at r = 0. Show in particular that the most dangerous modes correspond to $|\mathbf{k}| = 1$.

c) Sketch the marginal stability surface $(s(\mathbf{k}r) = 0)$ in the threedimensional space (k_x, k_y, r) ; determine the set of marginal modes and the domain of unstable wavevectors $s(\mathbf{k}, r) > 0$ when r > 0. [Answer: Figure 3.15 (left)].

d) System (3.37) is isotropic in the (x-y) plane. Consider now the modified anisotropic model

$$\partial_t v = rv - \left[(\partial_{xx} + 1)^2 - \partial_{yy} \right] v. \qquad (3.38)$$

Sketch the marginal stability surface for this case and conclude that, by contrast with the isotropic case, the linear stability operator selects a non degenerated mode. [Answer: Figure 3.15 (right)].



Fig. 3.15 Marginal stability surfaces for the linearized Swift-Hohenberg model in the isotropic (left) and anisotropic (right) cases.

2) Consider now the same system but restricted to one space dimension:

$$\partial_t v = rv - (\partial_{xx} + 1)^2 v, \qquad (3.39)$$

for a function v(x,t) defined on a finite interval of length ℓ with boundary conditions $v = \partial_{xx}v = 0$ at x = 0 and $x = \ell$.

Check that the eigenmodes can be taken in the form $V = A \sin(k_n x)$ with $k_n = n\pi/\ell$. Determine the marginal stability condition $r = r_n^{(m)}(\ell)$ for mode *n* at given ℓ , and next the instability threshold $r_c(\ell) = \inf_n r_n^{(m)}(\ell)$, as a function of ℓ .

How does the spatial resonance between the intrinsic length-scale $\lambda_c = 2\pi/k_c = 2\pi$ and the size ℓ of the system manifest itself? Find the condition for two neighboring modes being simultaneously marginal.

The student is encouraged to perform the same (but much more difficult) study for boundary conditions $v = \partial_x v = 0$ at $x = \pm \ell/2$. This is more typical of the general case since simple trigonometric lines are no longer appropriate. The eigenmodes will be searched for as superpositions of elementary solutions to the 4th order differential equation

$$sv = \left[r - \left(\frac{\mathrm{d}^2}{\mathrm{d}x^2} + 1\right)^2\right]v$$

that fulfill the boundary conditions (the scalar s is the eigenvalue). Separate odd from even solutions and find corresponding marginal conditions (given by transcendental equations to be solved numerically by some root-finder program.)

This model, completed by appropriate nonlinear terms, will be used again in Chapter 4, Exercise 4.6.2 and in hands-on numerical experiments, Appendix B.

3.3.2 Rayleigh-Bénard convection: detailed study

The purpose of the exercise is to go beyond the semi-quantitative approach developed in §3.1.4 and determine the marginal stability condition from the full primitive equations. The theory rests on the *Boussinesq approximation* of moderate heating which supports the idea that the fluid's physical parameters are independent of the temperature, except the density in the term responsible for the differential buoyancy force. The linearized thermo-hydrodynamic equations governing two-dimensional (x, z) pertur-

bations then read²¹:

$$\partial_t (\partial_{xx} + \partial_{zz}) v_z = P \left((\partial_{xx} + \partial_{zz})^2 v_z + \partial_{xx} \theta \right), \tag{3.40}$$

$$\partial_t \theta = R \, v_z + (\partial_{xx} + \partial_{zz}) \theta \,. \tag{3.41}$$

These equations are written here in dimensionless form after elimination of the pressure and horizontal velocity component. The scales are the same as those leading to (3.17, 3.18). As far as the (x, t)-dependence is considered, their structure is also the same as that of the simplified system, but they now retain the additional z-dependence of the fluctuations explicitly.

Boundary conditions are set at the position of the plates z_p , at top, $z_t = 1$ or +1/2, and bottom $z_b = 0$ or -1/2, one or the other choice making computations more transparent depending on the cases considered.

We consider here infinitely good heat-conducting horizontal plates. Accordingly the temperature is strictly fixed by heat baths so that the fluctuations are zero there:

$$\theta|_{z_{\mathbf{p}}} = 0. \tag{3.42}$$

For the velocity components, rigid plates imply a no slip condition

$$v_z|_{z_p} = \partial_z v_z|_{z_p} = 0. \tag{3.43}$$

In practice, calculations are easier with the somewhat artificial stress-free conditions considered initially by Rayleigh, which leads to the replacement of (3.43) by

$$v_{z}|_{z_{p}} = \partial_{zz} v_{z}|_{z_{p}} = 0.$$
 (3.44)

In the following we consider symmetrical cases where top and bottom conditions are identical but any non-symmetrical condition (good/bad conductor, no-slip/stress-free) can be considered in the same way at the expense of a (much) more cumbersome analysis since parity considerations are no longer useful to classify the solutions.

1) Normal mode analysis in the stress-free case (Rayleigh solution, 1916). Boundary conditions are set at $z_p = 0$ and 1.

a) For a laterally unbounded layer, check that solutions to (3.40, 3.41) can be taken in the form:

$$(v_z, \theta) = (\overline{V}, \overline{\Theta}) \sin(n\pi z) \exp(ikx) \exp(s_n t).$$

²¹See p. 108 for a derivation of equations and boundary conditions.

b) Determine the marginal stability condition for the mode n that first bifurcates when R is increased. Find the corresponding critical wavevector and threshold.

[Answer:

$$R_n^{(m)}(k) = (n^2 \pi^2 + k^2)^3 / k^2, \qquad (3.45)$$

 $k_{\rm c} = \pi/\sqrt{2}, R_{\rm c} = 27\pi^4/4$. See Fig. 3.16, curve A.]

c) Show that the linear dynamics close to the threshold is governed by:

$$au_0 s = r - \xi_0^2 (k - k_c)^2$$
,

with $r \equiv (R - R_c)/R_c$ and coefficients τ_0 and ξ_0 to be computed.

2) Normal mode analysis in the no-slip case. The exact solution, obtained by Pellew and Southwell (1940) is presented in [Chandrasekhar (1961)]. The corresponding marginal stability curve is given in Figure 3.16, as line B' (fine, solid) with critical conditions $k_c \approx 3.11632$, $R_c \approx 1707.76$.

Here we look for an approximate solution by a so-called *Galerkin* method, a special case of weighted-residual approximation introduced, e.g. in [Finlayson (1972)].

In a few words, the solution, expanded on a complete basis of functions, is further injected in the equations that are projected on a complementary basis using some scalar product. The Galerkin method comes in when the functions in the original and complementary bases are identical and fulfill the boundary conditions of the problem.

The approximate solution is seen to converge to the exact solution as the number of functions is increased, especially when the problem has an underlying variational structure, which is the case for Rayleigh-Bénard convection, as discussed at length in [Chandrasekhar (1961)].

Here, the solution of (3.40, 3.41) is searched for in the form:

$$\{v_z(x,z,t), heta(x,z,t)\}=\{ar{V}(z),ar{\Theta}(z)\}\sin(k_xx)\exp(st)$$
 .

The z dependence of \bar{V}_z and $\bar{\Theta}$ is taken as polynomials. Boundary conditions are set at $z = \pm 1/2$.

a) Show that in order to fulfill boundary conditions automatically, one must take:

$$ar{V}(z) = (rac{1}{4} - z^2)^2 P_v(z) \,, \ ar{\Theta}(z) = (rac{1}{4} - z^2) P_ heta(z) \,,$$

where $P_v(z)$ and $P_{\theta}(z)$ are polynomials in z, $P_v(z) = \sum_{n=0}^{\infty} V_n z^n$, $P_{\theta}(z) = \sum_{n=0}^{\infty} \Theta_n z^n$. The approximation enters when polynomials are truncated beyond some given maximal degree N.

b) The differential problem being formally written as $\mathcal{L}\mathbf{U} = 0$, where **U** has two components $\bar{\Theta}$ et \bar{V}_z , the projection onto the basis is defined by integrals

$$\begin{split} \int_{1/2}^{1/2} z^n (\tfrac{1}{4} - z^2)^2 (\mathcal{L} \mathbf{U})_{v_z} \, \mathrm{d}z &= 0 \,, \\ \int_{1/2}^{1/2} z^n (\tfrac{1}{4} - z^2) (\mathcal{L} \mathbf{U})_{\theta} \, \mathrm{d}z &= 0 \,, \end{split}$$

for n = 0, 1, ..., N. This leads to a system of 2(N + 1) linear equations for the 2(N + 1) unknown coefficients introduced in the polynomials.

Considering only the stationary case at marginality, *i.e.* s = 0, derive the system at lowest significant order, *i.e.* N = 0 (2 equations for 2 unknowns, Θ_0 and V_0) and the corresponding marginal stability condition; compare the result to the Rayleigh solution (3.45). Then compute the critical wavevector and the threshold; further compare them to the exact result given above.

[Answer:

$$R^{(m)}(k) = \frac{28}{27} \frac{(k^4 + 24k^2 + 504)(k^2 + 10)}{k^2}, \qquad (3.46)$$

threshold: $k_c \approx 3.1165 \simeq$ exact value, $R_c \approx 1750\ 2.5\%$ too high only. See Fig. 3.16, curve B (dot-dashed line), but so close an agreement is somehow accidental!]

Derivation of system (3.40-3.44).

In the two-dimensional case (x, z), the continuity, Navier–Stokes, and Fourier equations read:

$$egin{aligned} &\partial_x v_x + \partial_z v_z = 0\,, \ &
ho(\partial_t v_x + v_x \partial_x v_x + v_z \partial_z v_x) = -\partial_x p + \eta(\partial_{xx} + \partial_{zz}) v_x\,, \ &
ho(\partial_t v_z + v_x \partial_x v_z + v_z \partial_z v_z) = -\partial_z p + \eta(\partial_{xx} + \partial_{zz}) v_z + g lpha heta\,, \ &\partial_t heta + v_x \partial_x heta + v_z \partial_z heta = \kappa(\partial_{xx} + \partial_{zz}) heta + eta v_z\,. \end{aligned}$$

The formally quadratic terms have been dropped owing to the linearization step. Following the usual procedure, pressure is eliminated by differentiating the equation for v_z with respect to x, the equation for v_x with respect



Fig. 3.16 Marginal stability curves of the most dangerous modes in the stress-free case (curve A) and no-slip case (curves B). Thin solid line \rightarrow exact result. Dot-dashed line \rightarrow Galerkin approximation at lowest significant order.

to z and subtracting the two. In order to eliminate v_x , the result is further differentiated with respect to x and the so-obtained $\partial_x v_x$ replaced with $-\partial_z v_z$ using the continuity equation. This leads to:

$$\partial_t (\partial_{xx} + \partial_{zz}) v_z =
u (\partial_{xx} + \partial_{zz})^2 v_z + \alpha g \partial_{xx} \theta ,$$

 $\partial_t \theta = \kappa (\partial_{xx} + \partial_{zz}) \theta + \beta v_z ,$

which is finally cast into (3.40, 3.41) by scaling v_z and θ appropriately.

No-slip and stress-free boundary conditions respectively read $v_x(z_p) = v_z(z_p) = 0$ and $\partial_z v_x(z_p) = v_z(z_p) = 0$. Differentiating the conditions on v_x with respect to x and replacing $\partial_x v_x$ with $-\partial_z v_z$ yields the boundary conditions (3.43, 3.44), expressed in terms of v_z exclusively.

3.3.3 Simplified model of convection in a binary mixture

Thermal convection in the presence of additional molecular diffusion processes is called *thermohaline*, by reference to the diffusion of salt in water. Here we consider the emergence of convection in a binary mixture. The local state of the fluid is thus characterized by the concentration C of a solute, in addition to its temperature T and its velocity \mathbf{v} . The temperature gradient is still generated by the temperature difference ΔT , while a concentration gradient is applied by putting the fluid layer in contact with two reservoirs at different concentration through appropriate porous membranes. The solute concentration difference between top and bottom is a second control parameter and molecular diffusion (Fick law, coefficient D) is a supplementary stabilizing mechanism, while solute advection may contribute to the destabilization of the layer. We develop a simplified onedimensional model in the spirit of Section 3.1.2, *i.e.* assuming fluctuations that are functions of x and t only, while the driving gradients are imposed along the z direction.

1) Construction of the model:

The fluid layer is supposed to be at rest ($\mathbf{v} \equiv 0$) in contact with two baths at temperatures T_t and $T_b = T_t + \Delta T$ (heating from below implies $\Delta T > 0$) and concentrations C_t and $C_b = C_t + \Delta C$ (ΔC positive or negative). The purely diffusive temperature and concentration profiles read:

$$T(z)=T_{
m b}-eta z \qquad {
m and} \qquad C(z)=C_{
m b}-eta' z\,,$$

where $\beta = \Delta T/h$ et $\beta' = \Delta C/h$ are the applied gradients. The differential buoyancy force is still induced by variations of the density ρ but it now has two origins, thermal expansion and composition change. Accordingly, the state equation can be taken as

$$\rho = \rho_{\rm b}(1 - \alpha\theta - \alpha'c)\,,$$

where α (> 0) is the same coefficient as that introduced previously and where the sign of α' depends on the composition of the mixture. Justify the equations governing the linearized model:

$$egin{aligned} \partial_t v_z &=
u \partial_{x^2} v_z + g \left(lpha heta + lpha' c
ight), \ \partial_t heta &= \kappa \partial_{x^2} heta + eta v_z \,, \ \partial_t c &= D \partial_{x^2} c + eta' v_z \,, \end{aligned}$$

and explain the origin of terms βv_z and $\beta' v_z$ and discuss in simple terms the possible instability mechanisms involving each fluctuation.

Take h as length unit, $\tau_{\theta} = h^2/\kappa$ as time unit, introduce the Lewis number $L = D/\kappa$ in addition to the Prandtl number $P = \kappa/\nu$, define the control parameters

$$R = rac{lpha g \Delta T h^3}{\kappa
u}$$
 et $R' = rac{lpha' g \Delta C h^3}{D
u}$

(thermal and chemical Rayleigh numbers), and finally cast these equations

in the form

$$egin{aligned} \partial_t v_z &= P(\partial_{x^2} v_z + heta + Lc)\,, \ \partial_t \theta &= \partial_{x^2} heta + R v_z\,, \ \partial_t c &= L \partial_{x^2} c + R' v_z\,. \end{aligned}$$

Interpret the Lewis number from a physical viewpoint; what can be its order of magnitude in a liquid, in a gas?

2) Normal mode analysis:

a) Introducing $\{v_z, \theta, c\} = \{V, \Theta, C\} \cos(kx) \exp(st)$, write down the algebraic linear system fulfilled by the amplitudes $\{V, \Theta, C\}$. (Here we assume directly and without justification that $k \sim \pi$ (*i.e.*, in dimensional units $\lambda = 2h$ where λ is the wavelength of the unstable mode).

b) Show that for highly viscous fluids $(P \to \infty)$, the resulting system of three equations for three unknowns can be reduced to a system of two equations for two unknowns by eliminating the velocity component v_z . In the following we keep this supplementary assumption but the general case can be treated in the same way using Exercise 2.5.3, Chapter 2, p. 63.

c) Derive the compatibility condition of the simplified two-dimensional system and show that, by contrast with ordinary convection, complex roots are possible in the marginal case.

3) Different instability modes:

a) From a discussion of the sign of coefficients of the quadratic equation expressing the compatibility condition above, find the threshold of the stationary instability mode (the product of roots change its sign), then that of the oscillatory mode (the sum of roots change its sign).

b) In the parameter plane of parameters R' (horizontal axis) and R, draw the graph of these threshold conditions. Discuss the nature of the regime expected in each of the regions bound by these lines. Find the coordinates of the point where the system is simultaneously marginal against the two modes. Try to explain the physical origin of the oscillations by returning to the different contributions to the density changes and their respective relaxation times.

[Answer: Figure 3.17, next page. The mechanism result from an interplay of differential buoyancy with two competing dissipation processes, one (thermal diffusion) being much faster than the other (molecular diffusion). This induces delays and phase shifts between the different fluctuations, ending in overshoots and oscillations.]



Fig. 3.17 Stability diagram for convection in a binary fluid mixture. When R' > 0, the temperature and concentration fluctuations cooperate, whereas when R' < 0, they play antagonistic roles. Oscillations occur as their evolutions get sufficiently out of phase. At the intersection of the two lines (called a *codimension-two* point, where two parameters have to be tuned) the two mechanisms are equally efficient to destabilize the layer.

3.3.4 Turing patterns and reaction-diffusion systems.

In chemistry, mechanisms combining reaction and diffusion may produce dissipative structures called *Turing patterns*. Here we consider a simplified reaction-diffusion system where two species U and V with diffusion coefficients D_U and D_V also react with each other.

In dimension one, with coordinate x, the model reads:

$$\partial_t U = F_U(U, V) + D_U \partial_{xx} U, \qquad (3.47)$$

$$\partial_t V = F_V(U, V) + D_V \partial_{xx} V, \qquad (3.48)$$

where reaction terms F_U and F_V need not be specified at this stage. 1) Neglect diffusion $(D_U = D_V \boxminus 0)$ and assume that a fixed point solution (U_0, V_0) exists. Linearize the governing equations around that point, set

$$a = \partial_U F_U|_0$$
, $b = \partial_V F_U|_0$, $c = \partial_U F_V|_0$, $d = \partial_V F_V|_0$,

'|₀' meaning computed at (U_0, V_0) , next determine the conditions on a, b, c, d that guarantee the stability of solution (U_0, V_0) .

[Answer: the linear stability matrix must have eigenvalues either real and negative or complex with negative real parts, hence negative sum (a + d < 0) and positive product (ad - bc > 0)].

2) Assume that these conditions are fulfilled and add the effect of diffusion $(D_U \neq 0, D_V \neq 0)$. Determine the dispersion relation of fluctuations around the state (U_0, V_0) supposed uniform in space.

Write down the linearized system governing the amplitudes $\overline{U}(t)$ and $\overline{V}(t)$ of Fourier normal modes in $\exp(ikx)$ and show that the instability, if any, is necessarily cellular and stationary $(k \neq 0, \omega = 0)$.

Show that the occurrence of the instability requires:

$$aD_V+dD_U>0 \qquad ext{and} \qquad 4(ad-bc)D_UD_V\leq (aD_V+dD_U)^2,$$

in addition to the conditions already found. When this is the case, determine the range of unstable wavevectors k.

3.3.5 Taylor-Couette instability

We study the stability of the flow between infinite coaxial cylinders rotating at different angular speeds (Couette flow). This problem was studied by Rayleigh (1916) at the limit of zero viscosity. Taylor (1923) developed the first theoretical analysis of the viscous case and performed the corresponding experiments.

In the geometry of Figure 3.18 (left), assume that the base flow is purely azimuthal, show that $v_{\phi} = r\omega_0(r)$ with

$$\omega_0(r) = a + b/r^2, \qquad (3.49)$$

obtain a and b from the no-slip condition $\omega(r_i) = \omega_i$ at r_i , i = 1, 2.

[Answer:
$$a = (\omega_2 r_2^2 - \omega_1 r_1^2)/(r_2^2 - r_1^2), \quad b = (\omega_1 - \omega_2)r_1^2 r_2^2)/(r_2^2 - r_1^2).$$
]

1) Rayleigh instability mechanism (see also [Chandrasekhar (1961)] or [Drazin and Reid (1981)]). The base flow is characterized by the fact that the centrifugal force at distance r from the axis, $\rho r \omega^2(r)$, is compensated by a centripetal pressure gradient.

Consider a rotating fluid particle displaced from distance r to distance $r + \delta r > r$. In the absence of viscous friction, angular momentum $\rho r v_{\phi}$ is a conserved quantity. From this conservation law, derive its angular speed at the new position and compare it with that of a fluid particle originally at the same place.

2) If the speed of the displaced particle is smaller than that of the surrounding fluid, the local pressure gradient is larger and pushes the particle back to its original position, the purely azimuthal flow is stable. In the opposite



Fig. 3.18 Left: Geometry of the Taylor-Couette experiment. Right: Examples of curved flows.

situation it is unstable. Show that this stability condition reads

$$rac{\mathrm{d}\omega}{\mathrm{d}r}\,\delta r\geq -2rac{\omega}{r}\,\delta r\,,$$

then turn it to the form:

 $\delta\left(r^2\omega\right) \ge 0$

and express this Rayleigh stability criterion with words.

3) Application to the Couette profile and other curved flows.

a) Coming back to the base flow profile (3.49) identify the different possible cases (rotation direction identical or different, and when the rotation directions are identical, which cylinder is rotating faster) and find the situations that are stable according to the Rayleigh criterion.

b) When cylinders rotate in opposite directions, determine the region which is stable according to Rayleigh.

c) By comparison with the case of convection, and by anticipation of Chapter 6 about shear flows, guess the role of viscosity, especially when the flow is mechanically stable.

d) The same instability mechanism is expected to work when flow lines are curved. Identify the regions of the flow where the centrifugal instability can develop according to Rayleigh, in the boundary layer flow along a concave wall (Görtler instability; what about a convex wall?) or in the flow along a curved channel (Dean instability) depicted in Figure. 3.18 (right).

Chapter 4

Nonlinear Dynamics: from Simple to Complex

We now examine in more detail the theoretical context where the convection experiments described in the previous chapter can be best situated. The interest of this study stems from the fact that most instabilities that develop in bounded continuous media enter the framework of dynamical systems theory. Its foundation rests on an analysis of the effects of lateral confinement, §4.1. In a strongly confined system, even far enough from the instability threshold, the space dependence of unstable modes remains frozen and the state of the system is sufficiently well characterized by few amplitudes playing the role of effective degrees of freedom, which permits the analysis of the transition to turbulence in the spirit of Chapter 2. Here we limit ourselves to a heuristic presentation of the recent developments of nonlinear dynamics without insisting much on mathematical aspects.

We are thus mainly interested in characterizing sustained regimes of dissipative dynamical systems obtained after the decay of transients. These asymptotic regimes reached in the limit $t \to \pm \infty$ are represented in phase space by objects called limit sets. Stable limit sets are reached in the normal course of time, *i.e.* for $t \to +\infty$ and unstable limit set in the reverse case $t \to -\infty$. We have already seen examples of limit sets, fixed points and limit cycles for time-independent and periodic regimes, respectively. More complicated, multi-periodic and chaotic, behaviors are accounted for by objects called limit tori and strange attractors, which remain to be introduced. The second main purpose of this chapter is the description of the growth of complexity observed as the number of instability modes increases when control parameters are varied. The transitions from time-independent states to regular periodic or multi-periodic regimes, next to chaos, take place according to universal scenarios that are independent of the particular system studied. Section 4.2 is devoted to the simplest ones.

By contrast with regular regimes, chaotic states are characterized by their *unpredictability*, generally associated with a *fractal* occupation of phase space. These different facets are introduced at a theoretical level in Section 4.3. Next we turn to *empirical* approaches useful to interpret experimental results in Section 4.4.

However, all this is valid only in so far as the projection of the dynamics onto well isolated spatially frozen modes makes sense. In practice, unfreezing takes place when lateral confinement effects are not strong enough to maintain coherence much beyond the cell size. In systems with many cells chaos then develops both in time and in space, but this topic is defered to Chapter 5.

4.1 Reduction of the Number of Degrees of Freedom

4.1.1 Role of aspect ratios

Let us come back to eigenmodes obtained from the linear stability analysis of the instability studied, *e.g.* convection. The problem has been formalized in Section 3.1.3, p. 75. We assume that confinement is specified by *lateral* boundary conditions at a distance ℓ , whereas the instability generates cells with typical scale λ_c . The aspect ratio can be defined as $\Gamma = \ell/\lambda_c$. The difference between the cases " $\Gamma \sim 1$ " and " $\Gamma \gg 1$ " is illustrated in Figure 4.1 and 4.2. Exercises 4.6.1 and 4.6.2 should be worked out to gain a more analytical understanding of the problem.

We first consider the case when the horizontal dimensions are of the order of the wavelength λ_c so that all modes are physically distinct (Fig. 4.1). Only the value of the corresponding thresholds R_1, R_2, \ldots matters, the two-dimensional presentation of the different modes is given just in order to rationalize the classification in terms of the number of substructures in x and z, which is supported by the idea that these states should stay close to the branches that would exist in the absence of lateral boundary conditions (fine solid lines). They account for a kind of spatial resonance between the lateral extension of the system and the length scale originating from the instability mechanism, ending in the formation of an integer number of roughly circular cells, as already illustrated in Figure 3.8 (top), p. 96 (visualization of convection in silicon oil). The lower branch would then correspond to the "fundamental" states with one layer of cells and the upper one to "excited" states with two layers of cells stacked one above the other as could be obtained at higher values of the control parameter.



Fig. 4.1 Spectrum of the linearized stability problem for a confined system, $\Gamma \sim 1$; for $R_1 < R < R_2$, mode 1 is unstable, the others are stable.

In practice, the actual structure of the modes can be obtained at the price of a full three-dimensional calculation, in general difficult and with little supplementary insight into their physical significance. Since the spectrum of the linear stability problem is entirely made of isolated values far from one another, it is in principle easy to limit the number of linearly excited modes $(R > R_n \text{ hence } \sigma_n > 0)$ from those that are still damped $(R < R_m, \sigma_m < 0, |\sigma_m| \gg 1)$.

By contrast, when $\Gamma \gg 1$, that is $\ell \gg \lambda_c$ (Figure 4.2), the spectrum is quasi-continuous. It is obtained from the dispersion relation for a laterally unbounded system by adding a quantization condition on the wavevector, typically $k_n \sim \pi n/\ell$ for n cells of width $\lambda/2$ over a interval of width ℓ . The difference between neighboring wavevectors for patterns with ± 1 cell is $k_{n+1} - k_n = \pi/\ell \ll k_c$, which is small when ℓ is large. Close to k_c , owing to the quadratic shape of the dispersion relation (3.26), p. 84, one expects $(R_{n_c\pm 1}-R_{n_c})/R_c = \xi_0^2 \pi^2/\ell^2$, where n_c is here the number of cells expected at threshold and given by $n_c \lambda_c / 2 \simeq \ell$. Corresponding modes are thus quasidegenerate. Apart from the precise value of the wavevector, all modes of the lowest branch are expected to have similar spatial structures. The interactions between modes can then be understood as interferences yielding modulated patterns, also called *textures*, as illustrated in Figure 3.10, p. 98. The present chapter is entirely devoted to the case $\Gamma \sim 1$ and to the development of temporal chaos phenomenologically introduced in Chapter 3, Section 3.2.3, p. 94, and is thus placed in direct continuity with Chapter 2.



Fig. 4.2 Spectrum of the linearized stability problem for an extended system, $\Gamma \gg 1$; for $R > R_c$ the modes belonging to a range $\mathcal{O}(\sqrt{R-R_c})$ centered at k_c are unstable and may serve to build the nonlinear pattern.

4.1.2 Low dimensional effective dynamics

4.1.2.1 Framework

From a general viewpoint, it may seem reasonable to take advantage of the work done in solving the linear stability problem to treat the nonlinear problem. This is done by expanding the solution onto the basis of eigenmodes:

$$\mathbf{V} = \sum_{n} A_n \mathbf{X}_n \,, \tag{4.1}$$

and then searching the equations to be fulfilled by the amplitudes A_n of the modes \mathbf{X}_n , in order to extend the linear dynamics simply governed by $\dot{A}_n = s_n A_n$ to the nonlinear range. From the qualitative arguments developed above, we are led to think that the unstable and neutral modes play the most active role and that the stable modes more passively evolve under the forcing action of the others.

Let us incorporate this feature in the formalism itself and assume that the problem is initially posed in the general form:

$$\dot{\mathbf{V}} = \mathcal{L}\mathbf{V} + \mathcal{N}(\mathbf{V}), \qquad (4.2)$$

where \mathcal{L} accounts for the linear interactions and where $\mathcal{N}(\mathbf{V})$ is at least

quadratic in V. An appropriate projection of the equations onto the eigenbasis yields a dynamical system for the amplitudes A_n introduced through (4.1), which formally reads:

$$\dot{A}_n = s_n A_n + \sum_{mp} g_{n,mp} A_m A_p + \dots , \qquad (4.3)$$

where the coefficient $g_{n,mp}$ describes how the interaction of modes \mathbf{X}_m and \mathbf{X}_p contributes to the dynamics of mode \mathbf{X}_n . In fact this projection procedure is rarely achievable without approximation (and further modeling of the physical situation). A favorable case is when the analytical expression of eigenmodes can be handled by hand (or using formal algebra software), for example trigonometric functions as illustrated in Exercises 4.6.1 and 4.6.2. Otherwise one can attempt to extend to the nonlinear domain the *Galerkin method* already used at the linear stage, *e.g.* in the second part of Exercise 3.3.2, p. 107.

Progress made by replacing the primitive problem expressed in terms of partial differential equations by (4.3), *i.e.* a differential system of infinite order, may seem slight. However, it is very powerful when combined with the idea of separating *driving* modes and *enslaved* modes to truncate the system, eliminate the latter and obtain an effective system involving only the former. This is the *adiabatic elimination procedure* advanced as the founding principle of *synergetics*, the science of nonlinear cooperative phenomena in out-of-equilibrium systems [Haken (1983)].

Let us consider a system on the verge of bifurcating or having just bifurcated $(R \approx R_c)$. This comes to the assumption that, among all linear eigenmodes, we can isolate a small subgroup of those that are "dangerous," nearly neutral, either unstable or stable but slightly, *i.e.* with the real parts σ of their growth rates either positive or negative but small. Strictly speaking, the *center subspace*, subscript 'c', is the space spanned by eigenmodes having eigenvalues with $\sigma = 0$ exactly, and we assume that we have n_c such modes. On general grounds, having n_c strictly critical modes is a problem of codimension n_c , since n_c conditions $\sigma_n(\mathbf{r}) = 0$ have to be fulfilled, \mathbf{r} being the set of control parameters. In practice, we can vary only one control parameter while remaining in the vicinity of a manifold of codimension n_c in parameter space, hence $\sigma_c \approx 0$ for n_c driving modes that by extension will continue to be labelled with the same subscript.

On the contrary, all other modes are supposed to remain *stable*, hence with subscript¹ 's' and a related assumption $\sigma_s < 0$, meaning more precisely

¹So, 'c' may equally well mean 'central' or 'critical' and 's' either 'stable or 'slave'!



Fig. 4.3 Left: Spectrum of a linear operator with a group of *center modes* separated from *stable modes* by a wide 'spectral gap'. Right: Adiabatic elimination of stable modes. The whole sets of center-mode and stable-mode amplitudes are featured by axes A_c and A_s , respectively. A trajectory starting anywhere in phase space rapidly migrates towards a *center manifold* with equation (4.6) which is the nonlinear extrapolation of the center subspace at the origin ($A_s = 0$), then slowly evolves along it according to the effective dynamics (4.7).

that $|\sigma_c| \ll |\sigma_s|$, so that the driving modes have relaxation rates well separated from those of the enslaved modes. The linear spectrum is therefore supposed to display a wide *spectral gap*, see Figure 4.3 (left).

The effective dynamics is obtained by elimination of the enslaved modes. For simplicity we consider a *d*-dimensional case, $d < \infty$, with n_c driving modes and $n_s = d - n_c$ stable modes. The considerations developed now can be, at least at a heuristic level, easily extended to the infinite dimensional case corresponding to instabilities in continuous media, with an infinite number of stable modes and the series of eigenvalues with real parts extending down to $-\infty$. The adaptation of this elimination procedure to the case of extended systems will be sketched in Chapter 5.

4.1.2.2 Heuristic approach

Let us explicitly separate driving modes with amplitudes collectively denoted as A_c from enslaved ones with amplitudes A_s in (4.3). For the driving modes we get:

$$\dot{\mathbf{A}}_{c} = \mathcal{L}_{c}\mathbf{A}_{c} + \mathcal{N}_{c}(\mathbf{A}_{c}, \mathbf{A}_{s}), \qquad (4.4)$$

where \mathcal{L}_c is the restriction of the linearized operator \mathcal{L}_r to the space spanned by the \mathbf{A}_c and \mathcal{N}_c accounts for the nonlinear interactions between

driving and enslaved modes. Similarly we write:

$$\dot{\mathbf{A}}_{s} = \mathcal{L}_{s}\mathbf{A}_{s} + \mathcal{N}_{s}(\mathbf{A}_{c}, \mathbf{A}_{s})$$
(4.5)

for the amplitudes of stable modes. The idea of the reduction is to solve the problem for the stable modes as if they were submitted to a forcing from the (slowly evolving) driving modes, then to insert the solution back into (4.4) to obtain the effective dynamical system we are looking for.

In order to simplify the application of this strategy, let us assume that the s_s are real and non degenerate, and that \mathcal{N}_s involves only the \mathbf{A}_c so that we can rewrite (4.5) in a basis such that \mathcal{L}_s is in diagonal form, which leads to:

$$\mathbf{\dot{A}_{s}}+|s_{s}|\mathbf{A}_{s}=\boldsymbol{\mathcal{N}}_{s}\left(\mathbf{A}_{c}(t)
ight)$$
 .

It is easily checked that the complete solution of this non-homogeneous linear problem reads

$$\mathbf{A}_{s}(t) = \exp(-|s_{s}|t)\mathbf{A}_{s}(0) + \int_{0}^{t} \exp\left(-|s_{s}|(t-t')\right) \mathcal{N}_{s}\left(\mathbf{A}_{c}(t')\right) dt'$$

where $\mathbf{A}_{s}(0)$ is specified by the initial conditions at t = 0. According to the assumption made about the stable part of the spectrum, we see that the first term on the right hand side contributes to the solution only during a brief transient of duration $\mathcal{O}(1/|s_{s}|)$ and that, in the integral, the exponential kernel is short-range so that we may approximate the solution as

$$\mathbf{A}_{s}(t) \approx \frac{1}{|s_{s}|} \mathcal{N}_{s}(\mathbf{A}_{c}(t))$$
 (4.6)

In the *d*-dimensional phase space, these $n_{\rm s} = d - n_{\rm c}$ relations define a $n_{\rm c}$ -dimensional manifold parameterized by the $n_{\rm c}$ amplitudes of the driving modes. Inserting them into the evolution equations for the $A_{\rm c}$, we get:

$$\dot{\mathbf{A}}_{c} = \mathcal{L}_{c}\mathbf{A}_{c} + \mathcal{N}_{c}(\mathbf{A}_{c}, \mathcal{N}_{s}(\mathbf{A}_{c})) = \mathcal{L}_{c}\mathbf{A}_{c} + \mathcal{N}_{eff}(\mathbf{A}_{c}), \qquad (4.7)$$

which defines an *effective dynamics* on this manifold. An example is treated in Exercise 4.6.2.

4.1.3 Center manifolds and normal forms

4.1.3.1 Reduction to the center manifold

The result illustrated in Figure 4.3 (right) can be derived in a mathematically rigorous way, asymptotically valid at the long time limit, *i.e.* $t \gg 1/\min(|\sigma_s|)$. This approach is called the *center manifold reduction*.² Assuming that, as suggested by the heuristic approach, the stable modes "live" on a slow manifold defined through

$$\mathbf{A}_{s} = \mathcal{H}(\mathbf{A}_{c}), \qquad (4.8)$$

one can determine the functional equation governing \mathcal{H} by inserting (4.8) in (4.5) and replacing $\dot{\mathbf{A}}_{c}$ by its expression from (4.4). This yields

$$\partial \mathcal{H}(\mathbf{A}_{c}) \cdot \left[\mathcal{L}_{c}\mathbf{A}_{c} + \mathcal{N}_{c} \Big(\mathbf{A}_{c}, \mathcal{H}(\mathbf{A}_{c}) \Big) \right] = \mathcal{L}_{s} \mathcal{H}(\mathbf{A}_{c}) + \mathcal{N}_{s} \big(\mathbf{A}_{c}, \mathcal{H}(\mathbf{A}_{c}) \big),$$

where $\partial \mathcal{H}$ denotes the Jacobian matrix of \mathcal{H} with elements $\mathcal{H}_{ij} = \partial \mathcal{H}_i / \partial A_{cj}$. This functional equation for \mathcal{H} is then solved by representing $\mathcal{H}(\mathbf{A}_c)$ as a formal series in powers of the components of \mathbf{A}_c :

$$\mathcal{H}(\mathbf{A}_{\mathrm{c}})\equiv\sum_{n\geq2}\mathcal{H}_{n}(\mathbf{A}_{\mathrm{c}})\,,$$

where $\mathcal{H}_n(\mathbf{A}_c)$ is a polynomial formed with a series of homogeneous increasing-degree monomials in the form:

$$\prod_{m=1}^{d_{\mathrm{c}}} A_{\mathrm{c}\,m}^{n_m}, \qquad n_m \geq 0\,, \qquad \sum_m n_m = n\,,$$

with coefficients to be determined by identification. The so-obtained expression is further inserted in (4.4), which leads to the effective dynamics one is looking for.

4.1.3.2 Reduction to the normal form

In practice, the physical contents of the expression just obtained for the effective dynamics is obscured by the presence of a large number of terms that can be eliminated by nonlinear changes of variables, in much the same way as linear changes of variables allow one to represent a linear operator in its (much more transparent) Jordan normal form. This supplementary step, called *normal form reduction* leads to the elimination of all non-resonant terms in the effective dynamical system. The term "resonant" is here understood in the same way as in the study of the improper node, Chapter 2, p. 40, or more generally in Appendix A, §A.2, p. 334.

²See for example: J.D. Crawford, "Introduction to bifurcation theory" Rev. Mod. Phys. **63** (1991) 991-1037.

For example, let us consider the case of a two-dimensional center manifold parameterized with a pair of complex conjugate modes (Z, Z^*) with eigenvalues $\pm i\omega_c$, $\omega_c \neq 0$,³ exactly marginal at some specific value $\mathbf{r} = \mathbf{r}_c$ of the set of control parameters, *i.e.* $\sigma(\mathbf{r}_c) = 0$. Consider one of these variables, say Z, the previous reduction procedure generically leads to an effective dynamics

$$\dot{Z} = -i\omega_{\rm c}Z + \sum_{m=0}^{2} g_m^{(2)} Z^{2-m} Z^{*m} + \sum_{m=0}^{3} g_m^{(3)} Z^{3-m} Z^{*m} + \dots, \qquad (4.9)$$

where the coefficients $g_m^{(n)}$ are complex a priori. The equation governing the other variable Z^* is of course the complex conjugate of (4.9). It is however easily seen that none of the quadratic terms present in these equations is resonant:

Looking for a solution of (4.9) as an expansion in powers of Z, at first order we indeed get $Z^{(1)} \propto \exp(-i\omega_c t)$. The correction brought by the first quadratic term is then a solution to:

$$\dot{Z}^{(2)} + i \omega_{
m c} Z^{(2)} \propto (Z^{(1)})^2 \sim \exp(-2i \omega_{
m c} t) \, ,$$

that admits $\exp(-2i\omega_{\rm c}t)$ as special solution with a non-singular coefficient. Terms

$$ZZ^* \sim 1$$
 and $(Z^*)^2 \sim \exp(2i\omega_c t)$

are non-resonant for the same reason and, among cubic terms only the monomial

$$Z^*Z^2 = |Z|^2 Z \sim \exp(-i\omega_{
m c} t)$$

turns out to be resonant. It is just a little long and tedious to find the change of variables:

$$Z = \bar{Z} + \sum_{m=0}^{2} a_m^{(2)} \bar{Z}^{2-m} (\bar{Z}^*)^m + \sum_{m=0}^{3} a_m^{(3)} \bar{Z}^{3-m} (\bar{Z}^*)^m + \dots$$
(4.10)

that leads to the so-call normal form:

$$\dot{\bar{Z}} = -i\omega_{\rm c}\bar{Z} + \bar{g}_3|\bar{Z}|^2\bar{Z} + \dots$$
 (4.11)

appropriate to the case at hand.

 $^{{}^{3}\}omega_{c} \neq 0$ means $\omega_{c} \sim \mathcal{O}(1)$ since $\omega_{c} \ll 1$ would rather correspond to a reference situation with a double root s = 0, perturbed so as to display real and imaginary parts that would be simultaneously small.

4.1.3.3 Slightly off the critical conditions

One last step remains to be performed. Up to now we have assumed that the control parameters had precisely the values that make the relevant modes marginal. This defines a *critical surface* in parameter space, $\mathbf{r} = \mathbf{r}_c$. Close to this surface, the condition is no longer fulfilled but the real parts of the growth rates remain small.

A first-order Taylor expansion of these growth rates in a direction of the parameter space that is transverse (*i.e.* not tangent) to the critical surface allows one to get off the critical conditions. Let us simple denote the coordinate in that direction by r. For the pair of complex modes considered above, this comes to take:

$$s=\sigma\pm i\omega \quad ext{with} \quad \sigma\propto r\,, \quad \omegapprox\omega_{ ext{c}}\,, \quad g_3pproxar{g}_3\,,$$

and to "unfold" the critical dynamics (4.11) by correcting the growth rate, hence $-i\omega_c \mapsto s = \sigma - i\omega_c$. This corrected form⁴ will be used below to account for the emergence of periodicity. A rewarding exercise is to derive it explicitly from Equation (2.42) governing the van der Pol oscillator, p. 48.

4.1.3.4 Universality and modeling

More generally, one may remark that the normal form and the perturbations that describe the departure from criticality depend in an essential way on resonance relations existing among the eigenvalues of the linearized stability problem at criticality. In the previous example we had:

$$s_1 = s_2 + 2 s_1 \quad {
m for} \quad s_{1,2} = s_\pm = \pm i \omega_{
m c} \, ,$$

which renders the term Z^*Z^2 resonant. Here, the precise reason is:

$$\exp(s_2 t) \exp(s_1 t)^2 \boxminus \exp[(s_2 + 2s_1)t] = \exp(s_1 t)$$

with $s_2 = i\omega_c$ and $s_1 = -i\omega_c$, but the generalization is immediate to any resonance and leads to the identification of nonlinear resonant terms.

This plain observation underlies the notion of *universality*:

All systems that bifurcate in the same way, have the same symmetries, and are controlled by the same number of parameters, have qualitatively similar dynamics.

⁴In line with the remark in Note 3, p. 123, it is assumed that $\sigma \ll \omega_c \sim \mathcal{O}(1)$ because admitting $\sigma \sim \omega_c$ is not compatible with the idea of a near-marginal situation.

To set this similarity on a quantitative footing requires complicated and tedious changes of variables that can be avoided by having recourse to *phenomenological modeling* of the considered system close to its critical conditions, that is:

1) Determine the linear normal form governing the driving modes, identify the corresponding resonance conditions and add the most general linear perturbations corresponding the departure from criticality (see Appendix A, \S A.2, and more specifically \S A.2.3).

2) Introduce all possible nonlinear resonant terms —-different equivalent forms are admissible— with appropriate phenomenological coefficients. (Notice that these coefficients could be fitted from well-designed on-purpose experiments.)

3) Takes full advantage of a preliminary abstract mathematical study (with few generic parameters) to account for the behavior of the specific system considered as its control parameters are varied.

4.2 Transition to Chaos

The approach in terms of low dimensional *dynamical systems* now being established, we now briefly illustrate the theoretical approach to the growth of complexity until chaos is obtained. At steady state, a chaotic regime is characterized by an evolution that, in spite of being governed by a deterministic system, has lost its long-term predictability.

As empirically shown in Chapter 3, complex evolution is observed when the system is driven ever farther from equilibrium at the end of an instability cascade. The understanding of this cascade relies on the progressive increase of the dimension of the manifold on which the effective dynamics develops. By varying the control parameter, one increases the number of potentially unstable modes (see Figure 4.1) and thus the dimension of the effective dynamical system.

While one variable is enough to account for transitions between timeindependent regimes, we need two for periodic motion, and even more for more complicated behavior, regular (multi-periodic in practice) or irregular.

4.2.1 First steps: time-independent and periodic regimes

4.2.1.1 Bifurcation between time-independent regimes

The simplest case is of course that of time-independent regimes accounted for by fixed points in phase space. Accordingly, the simplest bifurcations are also those that correspond to transitions from one time-independent regime to another. In practice, they can be accounted for by a single variable through appropriate generalizations of the Landau model (1.19), p. 13, see Exercise 4.6.3 for further relevant considerations.

Let us start with:

$$\dot{A} = rA - gA^3 + H, \qquad (4.12)$$

describing the evolution of a system at a perturbed supercritical bifurcation. The dynamics described by (4.12) is typical of convection which, as we have seen, is a stationary instability that saturates beyond threshold (r > 0, g > 0). Here time t has been rescaled with the natural evolution time τ_0 of the unstable mode.

Quantity H measures the intensity of some perturbation coupled to the instability mode. In convection this could be an induction process producing a modulation of the background fluid density through non-uniform heating, which encourages the rising or sinking of the fluid already in the absence of temperature gradient. See also the remark on the facing page.

The bifurcation diagram,⁵ corresponding to (4.12), is displayed in Figure 4.4. The diagram corresponding to perfect bifurcation, with H = 0, features:

$$A_* = \pm (r/g)^{1/2} \tag{4.13}$$

and has already been displayed in Figure 1.4, p. 14. It is recalled here as a fine solid line. In the general case, the fixed points A_* of (4.12), obtained by solving it for A when $\dot{A} = 0$, now depend on r and H. The introduction of a small field $H \neq 0$ induces a non-trivial response $A_* \neq 0$ below the theoretical bifurcation point at r = 0. This response remains small as long as r is large and negative, so that

$$A_* \approx (-1/r)H, \qquad (4.14)$$

but closer to the threshold, as |r| decreases, the amplitude of the response

 $^{^5{\}rm The}$ bifurcation diagram is a graph locating the fixed points, and more generally the limit sets, as functions of the control parameters.



Fig. 4.4 Imperfect fork bifurcation accounted for by (4.12) with g = 1 for H = 0.054. Thick solid line (dashed line) corresponds to stable (unstable) fixed points as functions of r. The thin solid line is for the perfect bifurcation when H = 0 already displayed in Figure 1.4, p. 14.

increases up to a point where nonlinearities can no longer be neglected. For r = 0 this yields

$$A_* = (H/g)^{1/3} . (4.15)$$

When r varies from $-\infty$ to $+\infty$, A_* follows a first branch of stable solutions, the thick line that continuously joins the axis A = 0 for $r \to -\infty$ to the arc of parabola $A_* = +(r/g)^{1/2}$ for $r \to +\infty$. Sufficiently above threshold (r > 0), a second branch appears, disconnected from the first, that joins the arc of parabola $A_* = -(r/g)^{1/2}$ in its stable part (continuous thick line) to the axis A = 0 for $r \to +\infty$ in its unstable part (dashed line). The singularity at r = A = 0 has disappeared and one speaks of an *imperfect bifurcation*. In convection, a tiny forcing by thermal impression of a given pattern would produce this effect. The detailed study of this bifurcation diagram and other related ones is the subject of Exercise 4.6.3.

Remark

This description is in fact in close correspondence with the Landau theory of 'second order' thermodynamic phase transitions.⁶

An example of such phenomena is the paramagnetic-ferromagnetic transition in a magnet close to its Curie point T_c . A_* then plays the role of

⁶An old but nice reference that exactly fits the present purpose is Section II of a review article by L.P. Kadanoff *et al.*, "Static phenomena near critical point: Theory and Experiment," Rev. Mod. Phys. **39** (1967) 395-431. See also [Stanley (1988)].

the magnetization, the order parameter for that transition, and H is the magnetic field. Exponent 1/2 in (4.13) and 1/3 in (4.15) are the "classical" values of critical exponents β and $1/\delta$ linking the magnetization to the departure from T_c on the one hand, and the response to an applied field at $T = T_c$ on the other. Relation (4.14) defines a susceptibility that diverges as the inverse of the departure to criticality (critical exponent γ with classical value -1). Thermodynamic fluctuations slightly change the classical values that are easily derived from the Landau theory.

The notion of *universality* is central to the theory of so-called 'critical phenomena' associated to 'macroscopic ordering' measured by the order parameter. The values of the critical exponents involve only very general features of the system, especially the number of dimensions of physical space (usually 1, 2, or 3, for linear, planar, or bulk systems) and the symmetries of the order parameter. To stay with the magnetic example, the magnetization behaves as a vector with 1, 2, or 3 components, according to the existence of an easy direction (Ising model), an easy plane (XY model) or full isotropy (Heisenberg model), respectively.

The Landau theory can also deal with 'first order' transitions that are discontinuous, provided that an order parameter can be defined. For the liquid-gas transition, the order parameter is the density difference between the liquid and the gaseous phases.

The mathematical counterpart of Landau theory is Thom's catastrophe theory, see e.g. [Poston and Stewart (1978)].

4.2.1.2 Emergence of periodicity

Increasing the complexity of the dynamics, we now consider the emergence of temporal oscillations, which requires a two-dimensional effective phase space, as already learned in Chapter 2. This phenomenon, called *Hopf* bifurcation, (or *Landau-Hopf*, *Hopf-Andronov*, *Poincaré-Andronov*) is best described through a complex representation of the relevant subspace where it develops. The generic model reads:

$$\dot{A} = sA - g|A|^2 A$$
, (4.16)

with $A \in \mathbb{C}$, $s = \sigma - i\omega$, and g = g' + ig''. Setting $A = \rho \exp(i\varphi)$, upon substitution and after simplification by $\exp(i\varphi)$ we get:

$$\dot{
ho}+i
ho\dot{arphi}=(\sigma-i\omega)
ho-(g'+ig'')
ho^3$$

and, separating real and imaginary parts:

$$\dot{\rho} = \sigma \rho - g' \rho^3, \qquad (4.17)$$

$$\dot{\varphi} = -\omega - g'' \rho^2. \tag{4.18}$$

As already discussed, the coefficients in (4.16) are a priori functions of the control parameter r, but at the dominant order we can assume that ω and g are constant, and introduce the dependence on r only where it is indispensable, *i.e.* to control the growth rate. Since it must be negative below threshold (damping) and positive above (amplification), we can simply take $\sigma \propto r$.

The study of the problem for the modulus ρ is similar to that for the order parameter A in the previous section. Solving (4.17) for ρ at steady state (*i.e.* after elimination of the transient behavior), we get:

$$\rho = \rho_* = (\sigma/g')^{1/2} \tag{4.19}$$

which, once inserted in (4.18), yields:

$$\varphi(t) = -\omega_* t + \varphi_0 \quad \text{with} \quad \omega_* = \omega + g'' r/g' \,, \tag{4.20}$$

where φ_0 is a constant that depends on initial conditions. The so-obtained *limit cycle* is stable when the bifurcation is supercritical, which implies g' > 0, as illustrated in Figure 4.5. On the other hand, as understood from (4.20), coefficient g''/g' determines the change in angular frequency due to nonlinear couplings, *i.e.* the oscillation's nonlinear dispersion.



Fig. 4.5 Supercritical Hopf bifurcation (emergence of a stable limit cycle). The oscillations relax towards the origin below the threshold (left) and saturate at a finite amplitude above (right).
4.2.1.3 Dynamics in dimension two, general case

In fact one can show in full generality that the behavior of a system with a two-dimensional phase space which is isomorphic⁷ to \mathbb{R}^2 cannot have regimes that are more complicated than periodic according to the Poincaré-Bendixson theorem. A qualitative idea of the reason why complex behavior (chaos) is excluded for two-dimensional time-continuous systems stems from the observation that, at a regular point (*i.e.* $\mathcal{F}(\mathbf{X}) \neq 0$) the vector field defines only one tangent direction, so that two trajectories cannot cross at this point. In dimension 2, the consequence of this fact is particularly drastic: a trajectory is a line (dimension 1) that splits the space (dimension 2) in two disconnected parts. Once transported everywhere by the vector field, this topological constraint implies that trajectories corresponding to steady states can be either fixed points or closed curves, *i.e.* time-independent or time-periodic states. The rigorous mathematical proof of this result is not as easy as it seems; accessible references are [Hirsch and Smale (1974)] or [Lefschetz (1977)], for example. By contrast, many things become possible in dimensions strictly larger than two since trajectories have then enough room to wind in complicated ways without crossing by "escaping" in the supplementary dimensions.⁸

In what follows, we examine first the case of periodically forced twodimensional systems, which are in fact three-dimensional systems with a particular structure. The stroboscopic analysis of trajectories (concretely illustrated in the numerical experiments of appendix B) is used to introduce discrete-time dynamical systems essential to the understanding of the subsequent transition steps toward chaos. Before doing this we examine first the respective roles of linear resonance and nonlinearity on the particularly simple case of the forced Duffing oscillator.

4.2.2 Quasi-periodicity and resonances

4.2.2.1 Forced systems

Up to now we have considered *autonomous systems* in which time t does not appear explicitly, so that the trajectory is independent of the instant chosen to specify the initial condition in phase space, which is no longer

⁷The surface of a torus is not isomorphic to \mathbb{R}^2 though tangent planes can be defined everywhere.

⁸The same will hold for two-dimensional manifolds with nontrivial topology such as the Mœbius band, which cannot be applied on \mathbb{R}^2 without tear but can indeed be embedded in \mathbb{R}^3 , hence the third dimension.

the case of *forced systems* introduced on p. 27. It is easily seen that the effective dimension of a forced system formally written as:

$$\dot{\mathbf{X}} = \mathcal{F}(\mathbf{X}; t) \,, \tag{4.21}$$

is increased by one with respect to the dimension of the corresponding unforced system. In order to specify a trajectory completely, we have indeed to choose not only the initial condition $\mathbf{X}^{(0)}$ but, since the system is no longer time-translationally invariant, to tell also the time $t^{(0)}$ at which we start the system. This very need for a supplementary initial condition already points to an increase of the effective dimension.

More formally, one can pass to an *extended phase space* in which the system is autonomous by introducing an auxiliary state variable U trivially governed by $\dot{U} \equiv 1$. Setting $\mathbf{Y} = \{\mathbf{X}, U\}$ we then obtain:

$$\mathbf{Y} = \mathbf{\mathcal{G}}(\mathbf{Y}) \quad \text{with} \quad \mathbf{\mathcal{G}} = \{\mathbf{\mathcal{F}}, 1\}$$

$$(4.22)$$

which is indeed autonomous. If the initial condition was $\mathbf{X} = \mathbf{X}^{(0)}$ at $t = t^{(0)}$ for (4.21), then for system (4.22), $U^{(0)} = t^{(0)}$ presents itself as the initial condition for U. The extended phase space is thus the product of the original phase space \mathbb{X} by \mathbb{R} which accounts for the time variable.

In practice, the most familiar case of non-autonomous system corresponds to *periodic forcing* with period T (angular frequency $\omega = 2\pi/T$), *i.e.* $\mathcal{F}(\mathbf{X}; t + T) = \mathcal{F}(\mathbf{X}; t)$. From a mechanical viewpoint, one can understand the supplementary variable U as characterizing the rotation of a wheel with a moment of inertia so large that its angular speed ω is independent of the state of the system to which it is coupled, acting on it like the connecting rod of a steam engine.

Stroboscopic analysis of the trajectories at the period of the forcing then comes to take pictures of the system at a series of times $t_k = kT$, $k \in \mathbb{N}$. Geometrically, this operation corresponds to a series of "sections" of the extended phase space (Figure 4.6, left). During the time interval between two sections, the trajectory can be computed with as much precision as desired by integration of (4.21).

Owing to the periodicity of the forcing, we can represent the full trajectory by registering the state of the system in its (reduced) phase space Xwhile keeping track of the instant it passes through this state using label k. In this perspective the full phase space of the system in restricted representation is the product of X by the periodized interval [0, T]; the periodized interval [0; 1] is usually called the one-dimensional sphere S^1 , so that one



Fig. 4.6 Extended representation (left) and restricted representation (right) of the phase space of a periodically forced system. The stroboscopic analysis corresponds to a series of "sections" in the extended phase space, further identified modulo T in the restricted representation.

can denote it as $\mathbb{X} \oplus T \mathbb{S}^1$ (Figure 4.6, right). The generalization of this operation to the case of autonomous dynamical systems (*Poincaré section*) is a particularly efficient tool for the understanding of the emergence of chaos.

4.2.2.2 Steady state of a periodically forced oscillator

Let us consider a Duffing oscillator with natural period $\omega_0 = 1$, damped but externally forced:

$$\ddot{X} + 2\eta \dot{X} + X + \varepsilon X^3 = f \cos(\omega t). \qquad (4.23)$$

We consider here only the case of a saturating nonlinearity $\varepsilon \ge 0$ (see also Exercise 2.5.7).

• Linear preliminary. When $\varepsilon = 0$, after the damping of transients, the steady state response is obtained by inserting $X = A_f \cos(\omega t - \varphi)$ in (4.23), which yields:

$$[(1-\omega^2)\cos(\omega t-\varphi)-2\eta\sin(\omega t-\varphi)]A_{
m f}=f\cos(\omega t)$$

Expanding $\cos(\omega t - \varphi)$ and $\sin(\omega t - \varphi)$, by identification of terms in $\cos(\omega t)$ and $\sin(\omega t)$, we get:

$$egin{aligned} & \left[\left(1 - \omega^2
ight) \cos arphi + 2\eta \omega \sin arphi
ight] A_{
m f} = f \,, \ & -2\eta \omega \cos arphi + \left(1 - \omega^2
ight) \sin arphi = 0 \,. \end{aligned}$$

We derive immediately from the second equation that

$$\tan\varphi = \frac{2\eta\omega}{1-\omega^2}\,,\tag{4.24}$$

and, upon elimination of $\cos \varphi$ and $\sin \varphi$

$$\left\{ \left(1 - \omega^2\right)^2 + 4\eta^2 \omega^2 \right\} A_{\rm f}^2 = f^2 \,, \tag{4.25}$$

which leads to

$$A_{\rm f} = \frac{f}{\sqrt{(1-\omega^2)^2 + 4\eta^2 \omega^2}} \,. \tag{4.26}$$

In Figure 4.7, this solution of the linear problem for f = 1, labelled $\varepsilon = 0$, is represented as a thick solid line for $\eta = 0.1$ and as a thin solid line in the limit $\eta = 0$.

• Effect of the nonlinearity. Let us assume that the forcing is sufficiently weak $(f \ll 1)$ and that the response is not much distorted so that we can restrict ourselves to a first harmonic approximation, even close to the linear resonance. At steady state, we can still keep $X = A_f \cos(\omega t - \varphi)$ which, once inserted in (4.23), adds to the terms already obtained the quantity $\varepsilon A_f^3 \cos^3(\omega t - \varphi) = \varepsilon A_f^3 \left[\frac{3}{4}\cos(\omega t - \varphi) + \frac{1}{4}\cos(3(\omega t - \varphi))\right]$. Only the first term of this sum is of interest to us now. Expanding it as before in $\cos(\omega t)$ and $\sin(\omega t)$, we immediately see that the solution can be obtained by replacing $(1 - \omega^2)$ by $\left[(1 - \omega^2) + \frac{3}{4}\varepsilon A_f^2\right]$ everywhere.



Fig. 4.7 Response $A_{\rm f}$ of the periodically forced Duffing oscillator upon sweeping the frequency around its resonance value. Thick and thin lines correspond to $\eta = 0.1$ and $\eta = 0$, respectively in the linear ($\varepsilon = 0$) and nonlinear ($\varepsilon = 0.3$) cases. Stable solutions are represented by solid lines and unstable ones by dashed lines.

Equation (4.25) now reads

$$\left\{ \left[\left(1 - \omega^2 \right) + \frac{3}{4} \varepsilon A_{\rm f}^2 \right]^2 + 4\eta^2 \omega^2 \right\} A_{\rm f}^2 = f^2 \,. \tag{4.27}$$

This equation is cubic in $A_{\rm f}^2$ and can thus generically have one or three solutions, by contrast with the linear case ($\varepsilon = 0$) for which the solution in $A_{\rm f}^2$ is unique.

Expanding (4.27) and considering the sign of its coefficients, one can easily show that there is always at least one positive root in $A_{\rm f}^2$. In order to convince ourselves that it can also have other positive roots, we now examine the limiting case $\eta = 0$ which reads:

$$\left[\left(1-\omega^2\right)+\frac{3}{4}\varepsilon A_{\rm f}^2\right]^2 A_{\rm f}^2=f^2\,.$$

In the absence of forcing (f = 0), in addition to the trivial solution $A_f = 0$, this equation has the double root:

$$A_{\rm f}^2 = A_0^2 = \frac{4}{3\varepsilon} \left(\omega^2 - 1 \right), \qquad (4.28)$$

which is acceptable as soon as $\omega \geq 1$ (for $\varepsilon > 0$, which we assume). This solution is perturbed by the introduction of a small forcing that suppresses the degeneracy. For ω sufficiently large, the pair of solutions that derives from it is given with a good approximation by

$$A_{
m f}^2 = A_0^2 \pm ilde{A} \qquad {
m with} \qquad ilde{A} = fig[4ig(\omega^2-1ig)/3arepsilonig]^{-1/2} \,.$$

where the correction \tilde{A} remains small when compared to A_0^2 , itself large in the considered limit $\omega \gg 1$. These two solution branches are traced for f = 1 and $\varepsilon = 0.3$ as thin lines in Figure 4.7. The solution for f = 0 derived from (4.28) is an arc of hyperbola (not drawn) that starts at $\omega = 1$ and serves as a common asymptote inbetween these two branches. The thick line corresponds to the solution of the equation (4.27) including the viscous friction term with $\eta = 0.1$. As usual, unstable solutions are indicated by dashes.

In applications, one can fix f and change ω , or the reverse. In both cases one observes that there exist ranges of parameters over which multiple solutions exist. Then there are three solutions, one can show by methods adapted from those developed in Chapter 2 that the solution with intermediate amplitude is unstable and thus cannot be observed. By sweeping the control parameter (angular frequency or amplitude of the forcing) one may drive the system to perform hysteresis cycles during which the oscillator jumps from a large amplitude to a small one or the reverse. One can also notice that when the nonlinearity is saturating ($\varepsilon > 0$), the curves bend in the direction of high frequencies, which is easily understood from the relation between the amplitude and the angular frequency for the free oscillator, Equation (2.48) p. 52 (increased average amplitude implies increased effective stiffness and thus shortening of the period).

• Secondary resonances. The modification of the response curve in the neighborhood of the natural period of the oscillator, the so-called primary resonance is the most immediate effect of nonlinearity. The existence of secondary resonances at angular frequencies close to multiples or sub-multiples of this angular frequency is slightly less intuitive, though they still result from elementary trigonometric relations. The subharmonic resonance is most easily understood. As a matter of fact, if the forcing period $\omega = 1/3$ is apparently far from the resonance, the cubic nonlinearity induces some response at $\omega = 1$ since $\cos^3(t/3) = \frac{3}{4}\cos(t/3) + \frac{1}{4}\cos(t)$. In turn, this response can be seen as an external forcing of the fundamental mode, inducing its own resonant response. Considering a superposition $X = [A\cos(t+\psi) + A_f\cos(\omega t+\varphi)]$ and computing X^3 one finds terms in

$$\cos^{3}(t + \psi), \quad \cos^{2}(t + \psi)\cos(\omega t + \varphi), \\ \cos(t + \psi)\cos^{2}(\omega t + \varphi), \quad \cos^{3}(\omega t + \varphi),$$

which generate terms with angular frequencies ± 1 , ± 3 , $\pm 2 \pm \omega$, $\pm 1 \pm 2\omega$, $\pm \omega$ et $\pm 3\omega$, where the \pm signs come from the fact that the resonance does not depend on the sign of the arguments of the sines and cosines. A resonance occurs every time a combined angular frequency is equal to the natural angular frequency (± 1) .⁹ In addition to $\omega \approx 1$ corresponding to the primary resonance, this rule adds $\omega \approx 1/3$ ($3\omega = 1$), $\omega \approx 3$ ($2 - \omega = -1$), and of course $\omega \approx 0$. However the orders of magnitude of these different perturbations to the initial problem (the free, linear, frictionless oscillator) are different in each case, which imposes separate studies of lesser interest than that of the primary resonance.

• Simple or complex response? As in Chapter 2, if one wants to go beyond the first harmonic approximation, one has to develop rigorous perturbation approaches by multiple scale methods. In the present context this would however lead us too far with little compensation as far as the insight into

⁹Expanding the solution in terms of complex variables would lead to two conjugate resonance conditions, and at the end the same combinations, hence the ± 1 .

nonlinear phenomena is concerned. (Asymptotic approaches to nonlinear oscillations are examined in greater detail in [Nayfeh and Mook (1979)].)

In practice, the relative simplicity of the response of the Duffing oscillator with a saturating nonlinearity ($\varepsilon > 0$) submitted to a harmonic forcing stems from the fact that its phase space contains just one center at the origin and no unstable elements. This would no longer be the case if, instead of $\ddot{X} + X + X^3 = 0$, we had taken $\ddot{X} - X + X^3 = 0$. The study of its phase portrait similar to that in Exercises 2.5.6 and 2.5.8 shows that the center at the origin is replaced by a saddle and it is precisely the instability inherent in this type of fixed point which is at the origin of the chaotic behavior observed under external periodic forcing when the forcing increases beyond some threshold. More on this subject can be found in [Guckenheimer and Holmes (1983)], whereas the numerical study proposed in Appendix B, Section B.4.1, shows that the same result holds in the case of the periodically forced damped pendulum for the same reason: the presence of saddles in the phase portrait of the unperturbed system, Figure 2.9, p. 44).

Before entering the realm of chaos, let us consider the (still regular) case of self-sustained oscillations submitted to a periodic external forcing.

4.2.3 Quasi-periodicity and lockings

The phase portrait of the (unforced) van der Pol oscillator is less trivial than that of the standard damped Duffing oscillator since it displays an unstable focus at the origin and an attracting limit cycle surrounding it at some distance. We take it here in the form (2.60), p. 54 which, when externally forced, reads:

$$\ddot{X} - \epsilon (1 - X^2) \dot{X} + X = f \cos(\omega t). \qquad (4.29)$$

Its different regimes will systematically be studied by means of stroboscopic analysis at the forcing period $T = 2\pi/\omega$.

In the absence of forcing (f = 0), the temporal evolution of some observable $\mathcal{W}(\mathbf{X})$ function of the state $\mathbf{X} \equiv \{X, \dot{X}\}$ of the oscillator is a periodic function of time with period $T_0 = 2\pi/\omega_0$ ($\omega_0 \approx 1$ for $\varepsilon \ll 1$, see. §2.3.2.2, p. 52).

Assuming that the forcing is weak $(f \ll 1)$ and that the imposed angular frequency is not commensurate with ω_0 , *i.e.* $\omega/\omega_0 = \alpha$ irrational, after transients have decayed we expect $\mathcal{W}(\mathbf{X}(t)) = G(\omega_0 t + \varphi, \omega t + \psi)$ with $G(u + 2\pi, v) \equiv G(u, v + 2\pi) \equiv G(u, v)$, *i.e.* two-periodic behavior.



Fig. 4.8 Bifurcation diagram of the forced van der Pol oscillator. Left: $\omega \approx 1$, $\varepsilon = 0.1$, and f = 0.5. Right: $\omega \approx 3$, $\varepsilon = 0.1$, and a larger forcing, f = 1.0, to make its effects more visible. Notice also the expanded scale for ω .

By contrast, when ω and ω_0 are commensurate, *i.e.* $\omega/\omega_0 = \alpha \in \mathbb{Q}$, $\alpha = p/q$, integers p and q being relative primes, $\mathcal{W}(\mathbf{X})$ is a periodic function of time with period equal to the smallest common multiple of T and T_0 , $\tilde{T} = pT = qT_0$.

• Frequency locking. When the intensity of the forcing increases, the oscillator can "feel better" if it leaves its own frequency and adopts that of the forcing. It is the *locking* phenomenon: over a full range of values of ω , around a condition such that $qT_0 = pT$, the oscillator adjusts its motion so that its effective period \bar{T}_0 , a priori function of f and $\omega = 2\pi/T$, still fulfills the resonance condition, $q\bar{T}_0 = \tilde{T} = pT$ so that the period of the forced system remains \tilde{T} .

Figure 4.8 displays the set of stroboscopic recordings $X_{\rm str}$ as a function of the forcing frequency ω (bifurcation diagram) in a given numerical simulation of (4.29), thus illustrating this spectacular persistence phenomenon of *locking windows* for $\omega \approx 1$ with $\varepsilon = 0.1$ and f = 0.5 (left), and for $\omega \approx 3$ with $\varepsilon = 0.1$ and f = 1.0 (right).

Let us consider first the window at $\omega \approx 1$. When the locking takes place, the system is periodic with period T so that, when stroboscopically sampled at period T, it takes one and a single value $X_{\rm str}(\omega)$. On the contrary, as soon as ω gets out of the window, though the oscillator is still sensitive to the forcing, it recovers its independence and the signal becomes two-periodic. The stroboscopic analysis then produces a picture such as that for $\omega = 0.75$ in Figure 4.9 (left): the state of the oscillator (X and \dot{X}) is registered at increasing multiples of the forcing period, $t_k = kT$,



Fig. 4.9 Left: Stroboscopic analysis of a trajectory of the forced van der Pol oscillator for $\omega = 0.75$ and f = 0.5 (two-periodic unlocked regime). Right: Main resonance tongue $\omega \approx 1$: region of the (ω, f) parameter plane where locking is observed.

and it can be observed that the corresponding points arrange themselves along a closed curve that is continuously covered. Considering successive points shows that they jump from one position to the next with a slight shift, which explains the continuous coverage due to incommensurability of the two periods (while only a finite set of points would be reached in case of locking). The projection of this curve on the X axis produces the full vertical segments shown at values of ω corresponding to unlocked behavior on the bifurcation diagram of Figure 4.8. Notice that with $\omega = 0.75$ one has $4T_0 \simeq 3T$, *i.e.* close return every three samplings, which explains the proximity of points labelled 1, 4, and 7, etc.

The case of the window at $\omega \approx 3$ (Fig. 4.8, right) is analogous but, inside, we now have three values of $X_{\rm srt}$ for each ω . This is easily understood from the fact that the trajectory is sampled at period T which is about 1/3 of the natural period of the oscillator, so that it is regularly sampled three times during one of its own turns. In the locked regime, it returns exactly at the same places (modulo 3) whereas outside the window, in the unlocked regime, return points shift slightly from one sampling to the next, which again gives continuous segments on the bifurcation diagram. Notice that, in spite of a much stronger forcing, the window is extremely narrow.

• Emergence of complex behavior. The variation of the range in ω where the van der Pol oscillator is locked to the forcing, the so-called *resonance* tongue, is displayed in Figure 4.9 (right) as a function of the intensity fof the forcing (on the vertical axis). For this resonance, it widens linearly with f at least as long as it is sufficiently weak. Higher order resonances are narrower and widen more slowly. Frequency locking is a profoundly original manifestation of nonlinear effects. In the case of the van der Pol oscillator in a quasi-harmonic regime ($\varepsilon = 0.1$) it seems that one cannot observe phenomena more complicated than a decay of two-periodic behavior into simply periodic behavior, as long as one stays with reasonable values of the parameters. Things are different when the oscillator is more anharmonic (*e.g.* $\varepsilon = 1.0$). A wide variety of behaviors can be observed, with whole ranges of angular frequency where the response to forcing is chaotic, especially when the periodic driving is slow.

• Mutual locking. Up to now we have considered forcing in a strict sense, that is to say without feedback of the forced oscillator on the driving system. Besides, it is the presence of this immutable clock that makes the stroboscopic analysis easy to perform. Relaxing this condition, we can consider a system composed of two weakly coupled oscillators. Transposing the observations above, we may expect mutual locking in the neighborhood of resonance conditions fulfilled by the angular frequencies of these oscillators, for example Huygens' twin pendulum clocks.¹⁰ A system of two coupled oscillators is, as a whole, a four-dimensional system, each subsystem being two-dimensional. We are thus lead back to the case of autonomous systems in a dimension greater than two, to which we devote the end of this section.

4.2.3.1 Stability of a limit cycle: general case

To reach a higher complexity level, we must study the stability of selfsustained oscillations corresponding to limit cycles in phase space. This study must be performed not only on the surface containing the cycle, locally represented by the complex plane ($A \in \mathbb{C}$), which allows only perturbations in the radial direction, see Eq. (4.17), but in the full space in which the cycle is embedded, which permits "escape" in the supplementary dimensions. At this stage, it is useful to recall that we have arrived to an effective two-dimensional dynamics by adiabatic elimination of all stable modes, leaving us with a single relevant pair of center modes. Upon increasing the stress applied to the system, we have to consider the possibility that among stable modes, some become "dangerous." Mathematically speaking, the situation is more complex than before when we had to deal with a fixed point, since now we have to perturb a well established periodic dynamics.

¹⁰See the article by J. Whitfield, "Synchronized swinging," Nature Science Update, http://www.nature.com/nsu/020218/020218-16.html



Fig. 4.10 Stability of a limit cycle from a geometrical viewpoint. Top-left: Poincaré section and first return (Poincaré) map $\Phi(\mathbf{M})$. Top-right: Spectrum of the linearized map Λ . Bottom: Stability/instability of the fixed point of the map.

Let us keep the geometrical approach sketched in Figure 4.6 illustrating the case of a periodic forcing, but now consider the intersections of the trajectories in a d-dimensional phase space with a (d-1)-dimensional surface Σ , called the *Poincaré surface of section*, see Figure 4.10 (top-left). The oscillation that has settled can be viewed as a forcing for all other perturbations, and in the vicinity of the cycle, the phase space has indeed this product structure of the sphere S^1 times a relevant space in which "transverse" perturbations live, as extrapolated from Fig. 4.6 (right). A correspondence between successive intersections of trajectories with Σ is thus established in the form of a map of Σ onto itself $\mathbf{M}' = \mathbf{\Phi}(\mathbf{M})$, called the first return map or the Poincaré map. This transition from a continuous-time dynamics to a discrete-time one, from a d-dimensional differential system to a (d-1)dimensional map, is therefore basically similar to the stroboscopic analysis used for periodically forced systems. The difference just comes from the fact that the time interval between two sections is no longer the external forcing period but varies slightly with the position in the neighborhood of the cycle and tends to the period of the cycle as it is approached. Saying

this simply assumes that the differential dynamical system is smooth, so that the properties of the trajectories vary continuously, and that the surface of section is transverse (*i.e.* not tangent) to the cycle, so that the trajectories are correctly sampled.

Obviously, the cycle is associated with a fixed point **P** of the map Φ (*i.e.* $\mathbf{P} = \Phi(\mathbf{P})$) and will be stable is **P** is stable for Φ . In the neighborhood of **P**, the map Φ can be linearized into an operator represented by a square $(d-1) \times (d-1)$ matrix Λ , and the stability properties of the cycle can be derived from its spectrum, as illustrated in Figure 4.10 (top-right).

In full generality, one expects non-degenerate complex eigenvalues and corresponding two-dimensional invariant subspaces that are easier to parameterize using complex variables. In such a subspace, the linearized dynamics then reads:

$$Z_{k+1} = \lambda Z_k \,, \tag{4.30}$$

 $Z \in \mathbb{C}$ measuring the departure from the fixed point in the eigenspace of eigenvalue λ .

According to the definition, we have stability when the successive iterates of some initial condition $Z^{(0)}$ approach the origin as k increases and instability when they get away, Figure 4.10 (bottom). A given complex eigenmode is thus stable when $|\lambda| < 1$, marginal when $|\lambda| = 1$, and unstable when $|\lambda| > 1$. Writing λ as

$$\lambda = \rho \exp(2\pi i\alpha) \,,$$

leads to a clear separation of the modulus ρ from the phase $\pm 2\pi\alpha$ of the eigenvalues.

The quantity α specifies the angular frequency of the mode that makes the cycle possibly unstable: $\alpha = \omega/\omega_0$. This can be irrational or rational like in the case of periodic forcing. Cases when $\lambda \in \mathbb{R}$ are special. Equation (4.30) then turns into a real iteration:

$$X_{k+1} = \lambda X_k , \qquad (4.31)$$

and we must distinguish the dynamics corresponding to $\lambda > 0$ (*i.e.* $\alpha = 0$) for which iterates tend to or depart from the origin in a monotonic fashion from that observed when $\lambda < 0$ (*i.e.* $\alpha = 1/2$) for which they evolve by jumping alternatively from one side to the other.

Eigenvalue λ can be complex or not, but the corresponding stability condition remains $|\lambda| < 1$ and a bifurcation takes place when at least one

(pair of) eigenvalue(s) leaves the interior of the unit disc as the control parameters are varied. Different possibilities are indicated by arrows in Figure 4.10 (top-right).

Though the main purpose of the course is not a mathematical study of the different scenarios of transition to chaos in the perspective opened by Ruelle and Takens (see Figure 3.5, p. 93), let us mention that one can account for the simplest cases by completing (4.30) or (4.31) with appropriate terms. In the original Ruelle-Takens scenario, the destabilization of the limit cycle still leads to a regular but two-periodic regime ($\lambda \in \mathbb{C}$). The iteration that accounts for it reads

$$Z_{k+1} = \lambda Z_k - \gamma |Z_k|^2 Z_k, \qquad \gamma = \gamma' + i\gamma''. \tag{4.32}$$

After the damping of transients, it yields iterates evolving around a circle with radius $\sqrt{\rho-1}$, where $\rho = |\lambda|$, Fig. 4.11 (top). Trajectories in phase space wound on an invariant *limit torus* which is the Cartesian product of two circles, one corresponding to the unstable limit cycle (longitude) and



Fig. 4.11 Hopf bifurcation for a map (also called Neimark-Sacker bifurcation). Top: Iteration (4.32) for $|\lambda| < 1$ (left) and $|\lambda| > 1$ (right). Bottom: Perspective sketch of the skeleton of the *torus* over which the two-periodic dynamics develops asymptotically in time beyond bifurcation.

one for the new mode (meridian section), Fig. 4.11 (bottom), correspond to this description on the Poincaré surface.

The picture obtained in this case is very similar to that for the forced van der Pol oscillator. The difference is just that the two angular frequencies in the system now have intrinsic origin, the first one, which plays the role of the forcing, is that of the original cycle and the second one is that of its most dangerous instability mode. They can easily be identified on the records of physical observables, *e.g.* the two-periodic regime observed in convection and illustrated in Figure 3.8. As before, two-periodicity is observed only in the absence of resonant interaction between the two angular frequencies, that is to say as long as the ratio α remains irrational. Nonlinear couplings are therefore expected to be responsible for lockings when α approaches a rational value p/q as the control parameters are varied, lockings all the more visible when the denominator q is a small integer.

Two-periodic states, and a fortiori locked one-periodic states, are regular asymptotic regimes. In order to observe a chaotic behavior, it seems natural to think of an effective dimension still increased by one. This was precisely the context of the Ruelle and Takens approach, introducing strange attractors with properties accounting for irregularity in spite of determinism. More precisely, these authors showed (Note 8, p. 92) that when the system is smooth (\mathcal{C}^{∞}), chaos is generic after supplementary bifurcations introducing two new angular frequencies in the system. The analyticity condition was later made milder (*i.e.* \mathcal{C}^2 instead of \mathcal{C}^{∞}) with chaos possibly occurring in three-frequency quasi-periodic systems.¹¹ However, concrete scenarios from two-periodic regimes to chaos generally rely on different explanatory models which would lead us on mathematical grounds too far away from our main purpose. On the other hand, it is more interesting to spend some time to consider resonant cases.

When $\alpha = p/q$ the two periods are resonant. Little is changed if q is large, *i.e.* $q \geq 5$. The resonance is *weak*) and the system behaves roughly as in the non-resonant case: typically, the dynamics still takes place on the torus that has emerged from the bifurcation. Genuinely two-periodic unlocked regimes alternate with locked regimes inside narrow tongues as the control parameters are varied. In a quasi-periodic state the trajectory winds on the torus without closing (the torus is densely covered with it), whereas in case of locking the trajectory is a closed cycle wound on the torus, making p turns along the meridian and q turns in the longitudinal

¹¹S. Newhouse, D. Ruelle, and F. Takens, "Occurrence of strange axiom A attractors near quasi-periodic flows on T^m , $m \leq 3$," Commun. Math. Phys **64** (1978) 35.



Fig. 4.12 Trajectories on the torus are represented by their phases: Longitude $\theta = \omega_0 t + \theta^{(0)}$ (horizontal axis) and meridian coordinate $\varphi = \omega t + \varphi^{(0)}$ (vertical axis) defined modulo 2π . Up to a factor 2π , the torus is isomorphic to the square $\theta \in [0; 1]$, $\varphi \in [0; 1]$ with opposite sides identified. Left: For a two-periodic state, $\alpha \notin \mathbb{Q}$, the whole (θ, φ) square is covered by any given trajectory as $t \to \infty$, here $\alpha = (32/71)^{1/2} = 0.671345...$. Right: When $\alpha = p/q$, the same state is reached after a time $qT_0 = pT$, here $\alpha = 2/3 = 0.666666...$ (Notice that $\alpha = 2/3$ is in principle a case of strong resonance; this value has been chosen here only because the figure is easy to read.)

direction (Fig. 4.12). Chaos may enter after a new instability mode sets in.

By contrast, when $q \leq 4$ the resonance is *strong* and has marked effects, destroying the torus over which the motion takes place. Most extreme cases correspond to q = 2 and q = 1 that bring back the map from a complex iteration (4.30) to a real iteration (4.31) since for p/q = 1/2 one has $\lambda = \exp(i\pi) = -1$ and for p/q = 1/1, $\lambda = \exp(2i\pi) = 1$. Each of these two cases govern a specific scenario.

• The subharmonic cascade observed in convection (Fig. 3.7) corresponds to the strong resonance $\alpha = 1/2$, whose normal form reads:

$$X_{k+1} = f_r (X_k) = -(1+r)X_k - X_k^3.$$
(4.33)

The corresponding scenario is illustrated here using the so-called *logistic* map^{12}

$$X_{k+1} = 4rX_k (1 - X_k) , \qquad (4.34)$$

with $0 \le r \le 1$. The full bifurcation diagram in Figure 4.13 displays how the attractor (values of X reached as the iteration proceeds, after elimination

 $^{^{12}}$ For an introduction, see R.M. May, "Simple Mathematical models with very complicated dynamics," Nature **261** (1976) 459-467.



Fig. 4.13 Subharmonic cascade for (4.34).

of transients) changes as the control parameter r is varied.

For r < 3/4, a single point is obtained, corresponding to a periodic trajectory with period one (period T_0 if one considers the underlying limit cycle). At $r = r_1 = 3/4$ a first period-doubling takes place: a period-two regime sets in with two values X_1 et X_2 alternatively visited as the system evolves. (Working out the nonlinear variable change that turns (4.34) into the normal form (4.33) close to the bifurcation point is an interesting exercise.)

At $r = r_2 = 0.86237...$, this period-two attractor is destabilized against a period-four regime, *etc.*, as observed in convection experiments (Fig. 3.7, p. 95). The bifurcation cascade ends up at $r = r_{\infty} = 0.89248...$, where the system becomes periodic with infinite period in the form 2^{2n} , $n \to \infty$. At this value, the so-called *margin of chaos*, the system is in fact aperiodic since it never returns exactly in the same state.

Beyond r_{∞} , continuous vertical segments, in fact densely covered with points, are observed on the bifurcation diagram for each value of r. They are the trace of chaotic attractors for which infinite sets of unequally distributed points are visited all along a trajectory.

In fact, the situation is also very complicated in the parameter space since the return to periodic attractors is clearly visible in some parameter windows, the corresponding periodic states themselves decaying to chaos through subharmonic cascades. Programming (4.34) or its variant (4.69), p. 175, is so easy that one should not miss exploring their bifurcation diagram numerically by oneself.

• Intermittency (type I, that featured in Figure 3.9, because several other types exist) takes place at a 1/1-resonance generically accounted for by the map:

$$X_{k+1} = r + X_k + X_k^2 \,. \tag{4.35}$$

Before the transition, as long as r < 0, it has two fixed points solutions of:

$$X_{\pm} = r + X_{\pm} + X_{\pm}^2 \quad \Rightarrow \quad X_{\pm} = \pm \sqrt{-r} \,.$$

All trajectories starting with $X < X_{(+)}$ (and small enough) converge towards the fixed point at $X_{(-)}$. Other trajectories leave this limited region of phase space, Figure 4.14. Beyond the transition, for r > 0, the fixed points have disappeared but the local structure of phase space keeps a track of their presence as "ghosts": iterates go through the kind of tunnel opened between the graph of the map and the line $X_{k+1} = X_k$. They move very slowly since $X_{k+1} = X_k$ plus some tiny correction, approach the origin and then go away.

Now the *global* structure of phase space may provide the opportunity for trajectories visiting remote regions of phase space to come in the neigh-



Fig. 4.14 Type-I intermittency according to (4.35). Left: r < 0, two fixed points, $X_{(-)}$ stable, $X_{(+)}$ stable. Right: r > 0, no fixed points but a narrow tunnel between the graph of the map and the line $X_{n+1} = X_n$.

borhood of the origin. When this is the case, for r < 0 the system coming close to the stable fixed point eventually converge to it. By contrast for r > 0 fixed points no longer exist, iterates sent in the vicinity of the origin, travel slowly through the tunnel, then get away and come back later, hence the intermittent behavior observed in Figure 3.9, p. 97.

Laminar intermissions are uncorrelated and their durations are unpredictable. This is due to the more or less uncontrolled evolution far from the previously existing limit cycle which generates randomness in the reinjection process.

While interpreting the experiments, one must remember that the map describes what happens after an appropriate section of the dynamics has been performed. At the bifurcation, twin limit cycles, one stable and the other unstable, collide and disappear. Laminar intermissions are made of slowly evolving regular periodic oscillations corresponding to the "ghost" limit cycles. They are interrupted by chaotic bursts during which the system explores remote regions in phase space, until it is sent back in the laminar region, ready for a new intermission.

4.3 Characterization of Chaotic Regimes

As conceived by Ruelle, *chaos* is a dynamical regime characterized by a specific *sensitivity* of trajectories to *initial conditions* and small perturbations. In phase space, it is accounted for by *strange attractors* that are robust attracting limit sets on which small departures between two trajectories are indefinitely amplified (the general meaning of instability) as a result of a stretching-and-folding process analogous to the making of Danish pastry (baker map, see later Figure 4.17).

The main interest of these concepts is to reconcile *determinism* and *stochasticity* in dissipative dynamical systems, which was initially supposed to further shed some light on the problem of the "nature of turbulence" as discussed in Chapter 3, $\S3.2.2.3$, p. 92.

In this section we thus focus on a quantitative estimate of the two main facets of chaotic dynamics: the "longitudinal" instability measured by Lya-punov exponents and the fractal dimensions characterizing the "transverse" foliated structure.

4.3.1 Instability of trajectories and Lyapunov exponents

The instability of trajectories on a strange attractor is illustrated here by means of an everywhere expanding iteration called the *dyadic map* (Fig. 4.15, top-left). It reads:

$$X_{k+1} = 2X_k \pmod{1}$$
. (4.36)

It is convenient to visualize the trajectory $\{X_k, k = 0, 1, ...\}$ as a walk in the complex plane $Z_{k+1} = Z_k + \exp(2\pi i X_k)$. (Z has no dynamical significance here by contrast to X which is governed by the map.) This simple



Fig. 4.15 Interpretation of chaos. Top left: Dyadic map. Top right: Divergence of neighboring trajectories. Bottom left: Dyadic walk. Bottom right: Random walk.

representation helps us to clearly illustrate the divergence of trajectories starting at neighboring points (Fig. 4.15, top-right) and the long-term evolution (Fig. 4.15, bottom-left) that leads one to think of the random walk that would be obtained by drawing the successive X_k uniformly at random over the unit interval (Fig. 4.15, bottom-right).

The divergence rate of trajectories is a good measure of chaotic behavior. Let us show how the analysis proceeds in a simple example and consider a map f of a single real variable X:

$$X_{k+1}=f(X_k)\,,$$

more general than (4.36). To begin with, a given trajectory $\{X_k; k = 0, 1, 2...\}$ starting at $X_0 = X^{(0)}$ is taken as a reference and we consider a neighboring trajectory $\{\tilde{X}_k; k = 0, 1, 2...\}$ starting at $\tilde{X}_0 = X^{(0)} + \delta X_0$. Denoting f' = df/dX we have:

$$\bar{X}_1 = X_1 + \delta X_1 = f(X_0 + \delta X_0) = f(X_0) + f'(X_0) \, \delta X_0$$

$$\Rightarrow \qquad \delta X_1 = f'(X_0) \, \delta X_0$$

and using the chain rule:

$$\delta X_k = \left(\prod_{m=0}^{k-1} f'(X_m)\right) \delta X_0 \,.$$

Assuming a geometrical growth/decay as $|\delta X_k| \sim \gamma^k |\delta X_0| \equiv \exp(k\lambda) |\delta X_0|$, which defines both γ and $\lambda = \ln \gamma$, we get:

$$\gamma = \left(\left| \frac{\delta X_k}{\delta X_0} \right| \right)^{1/k} = \left(\prod_{m=0}^{k-1} |f'(X_m)| \right)^{1/k}$$

At this stage, γ still depends on k. To get rid of this dependence we take the limit $k \to \infty$ which leads to the definition of the Lyapunov exponent:

$$\lambda = \ln(\gamma) = \lim_{k \to \infty} \frac{1}{k} \sum_{m=0}^{k-1} \ln(|f'(X_m)|).$$
 (4.37)

In other words, the Lyapunov exponent is thus the time average of the local divergence rate $\ln(|f'|)$, *i.e.* a measure of long term *unpredictability*. Trajectories diverge in the mean when $\lambda > 0$, which can be taken as the definition of a chaotic dynamics. For the dyadic map (4.36), one gets $\lambda = \ln 2 > 0$.

The extension to maps with several variables leads to the definition of the Lyapunov spectrum, which rests on the analysis of the asymptotic behavior of the product of Jacobian matrices obtained from the chain rule. This generalizes the eigenspectrum of the stability matrix at a fixed point to the case of arbitrary trajectories. Technically, the matter is difficult but, in the limited context of this course, it is sufficient to know that this can be done and that, by successive generalizations, one can determine Lyapunov spectra for differential systems (from time- τ maps, see p. 28) and for continuous media governed by partial differential equations (after spectral approximation or discretization leading to finite-order ordinary differential systems, see Appendix B).

The Lyapunov spectrum can be ordered by decreasing values of the individual exponents. The system is then declared to be chaotic when the largest exponent is positive. The empirical determination of Lyapunov exponents from time records will be examined later, p. 165.

Let us stress the fact that the whole approach in terms of dynamical systems is a progressive extension of linear instability concepts from timeindependent regimes (fixed points of continuous-time systems) to periodic regimes (limit cycles then seen as fixed points of discrete-time systems) and finally to irregular aperiodic regimes.

4.3.2 Fractal aspects

Up to now we have been interested in expansion properties "along" the attractor. However the systems of interest are supposed to be dissipative, which implies overall contraction of volumes in phase space. As we have seen, chaotic behavior is marked by instability, which means expansion in some directions. This expansion has to be more than compensated for by stronger contraction in other directions. Some folding must then take place in order to maintain trajectories in a bounded region of the phase space, as sketched in Figure 4.16. (The case of regular regimes would be much less anisotropic with, at most, neutral directions on average.)

What happens in the "transverse directions" can be concretely illustrated using a celebrated simple two-dimensional map called the *baker map* (Fig. 4.17, top-left). This map of the unit square $[0,1] \times [0,1]$ expands it onto itself by a factor $\kappa = 2$ in the direction of the first coordinate and by a factor $\kappa' = \alpha/\kappa$ in the other direction ($\kappa' < 1 \Rightarrow$ contraction).

When $\alpha = 1$, that is $\kappa' = 1/2$, the baker's map is conservative (areas are preserved). In order to obtain a dissipative system, contraction must



Fig. 4.16 Expansion in unstable direction, contraction due to dissipation, folding by nonlinearities, all in combination, produce *horseshoes* typical of chaotic attractors.

be larger than expansion, *i.e.* $\alpha < 1$, so that $\kappa \kappa' = \alpha < 1$. The strange attractor obtained in that case displays a characteristic *fractal* transverse structure (Fig. 4.17, top-right). This fractal structure is a triadic *Cantor* set classically obtained by removing the middle third of a segment and repeating indefinitely the operation on the two segments left apart in that operation as shown in Figure 4.17 (bottom), yielding a self-similar set that is invariant upon magnification by a factor of three.



Fig. 4.17 Top-left: Conservative baker map (1st step: expand along X_1 ($\kappa = 2$) and contract along X_2 ($\kappa' = 1/2$); 2nd step: cut right half of rectangle and place it back in square). Top-right: Strange attractor for the dissipative map with $\kappa' = 1/3$ (expansion rate $\kappa\kappa' = 2/3 < 1$). Bottom: Few steps of the construction rule of the triadic Cantor set, to be read from right to left. (Notice that, to build the dissipative baker attractor, we used a map that translates part (2) above part (1) and puts it upside down so that the fractal structure follows from the conventional Cantor middle-third rule.)



Fig. 4.18 Hénon attractor obtained by iterating map (4.38, 4.39) for a = 1.4 and b = 0.3 from some initial condition after elimination of the transient. Left: The full attractor. Right: Zoom on the boxed region.

In practice, nonlinearities usually fold the trajectories as seen with the Hénon map: 13

$$X_{k+1} = 1 - aX_k^2 + bY_k, (4.38)$$

$$Y_{k+1} = X_k \,. \tag{4.39}$$

The attractor corresponding to a = 1.4 and b = 0.3 is presented in Figure 4.18. The folding originates from the nonlinearity in (4.38), which is of "logistic" type (see Exercise 4.6.9). Coefficient b in (4.39) guarantees the dissipative character of the map provided that |b| < 1, as shown by performing Exercise 2.5.1 with this map. The fractal character of the attractor is particularly obvious from the magnification of the boxed region displayed in the right part of the figure.

The fractal dimension gives a good idea of the way the attractor occupies the phase space. In a *d*-dimensional space, it is obtained by covering the considered set, here the attractor, by elementary volume elements of size ε^d , next counting the number $\mathcal{N}(\varepsilon)$ of such elements necessary to cover it, and finally study how that number grows as the linear size ε of the volume elements tends to zero. The fractal dimension $d_{\rm f}$ is thus defined through

 $^{^{13}}$ M. Hénon, "A two-dimensional mapping with a strange attractor," Commun. math. Phys. **50** (1976) 69.

the formula:

$$d_{\rm f} = \lim_{\varepsilon \to 0} \frac{\log \left(\mathcal{N}(\varepsilon) \right)}{\log(1/\varepsilon)} \,. \tag{4.40}$$

For an ordinary, connected, continuous set, this definition yields the usual topological dimension (0 for a point, 1 for a line, 2 for an ordinary surface, etc.). Let us see how it works for an indefinitely fragmented object like the triadic Cantor set by considering it at step k of the construction process. It can be covered by segments of length $\varepsilon = 1/3^k$, and 2^k such segments are needed at that step, hence:

$$d_{
m f}(k) = rac{\log(2^k)}{\log\left(1/(1/3)^k
ight)} = rac{\log(2)}{\log(3)} pprox 0.63092975\,,$$

somewhere between the dimension of a countable set of isolated points (0) and that of a continuous segment (1).

Here the evaluation does not depend on k because the set is strictly self-similar but more generally the limit $\varepsilon \to 0$ must really be taken. The computation of the fractal dimensions of other classical self-similar sets from a direct application of (4.40) is the subject of exercise 4.6.12.

Turning to the dissipative baker map, let us focus on what happens in the transverse direction X_2 . We take for granted that, owing to the expanding character of the map along X_1 , the attractor is continuous in that direction, which just adds 1 to the dimension found for the transverse part. Following the same idea as for the triadic Cantor set, we can observe that, after one iteration, the length of a segment along X_2 is multiplied by $\kappa' < 1$, which suggests us to take $\varepsilon = (\kappa')^k$ after k iterations. Each application of the map brings about κ reduced copies of the full set, so that we have $\mathcal{N}(\epsilon) = \kappa^n$. From the formula, we get $d_f = \log(\kappa)/\log(1/\kappa')$ along the contracting direction and thus $1 + \log(\kappa)/\log(1/\kappa')$ for the full attractor displayed in Figure 4.17 (middle part). With $\kappa = 2$ and $\kappa' = \alpha/\kappa$ with $\alpha = 2/3$, this yields $d_f \approx 1.63$.

As an exercise one can look at the aspect of dissipative generalized baker maps with definitions more complex than that illustrated in Figure 4.17, *e.g.* with $\kappa > 2$ and different values of α for different pieces, and then try to determine the dimension of the corresponding (transverse) Cantor sets, which might be less simple than it seems.

4.4 Empirical Approach of Chaotic Systems

Let us consider a chaotic regime observed in a given experiment, *e.g.* in convection. The need of an empirical approach becomes obvious when it appears impossible from a practical point of view to get an *ab intio* understanding of its nature. Most often, even something as qualitative as the type of transition scenario that develops under specific conditions cannot be predicted. For example, in Chapter 3, §3.2.3, we have seen that not only the physical properties of the fluid matters, but the experimental configuration and, in case of attractor coexistence, the history of the experiment also play a role in the transition (though the system, when engaged in a given scenario, follows all its steps at a quantitative level).

Sufficiently far from the threshold of the primary instability, a multiplicity of different possible permanent regimes can be reached by following specific experimental procedures. The effective phase space is thus already very complicated and poorly understood, even when confinement effects select a small number of driving modes. It is therefore fully justified to spend some time to seek a representation of the dynamics in some *reconstruction space* obtained from the experimental records, a space in which the evolution can be described, the amount of chaos can be measured and, hopefully, techniques of control can be developed.

The output of experiments, either in the lab or using a computer, generally presents itself in the form of *time series* of some observable, *i.e.* a series of numerical values taken by some function $W = \mathcal{W}(\mathbf{X})$ of the system's state $\mathbf{X} = (X_1, \ldots, X_d)$ in its phase space X with dimension d.¹⁴ Let us assume a regularly sampled time series:

 $\{W_k, k=0,1,\ldots\}, \text{ with } W_k = \mathcal{W}(\mathbf{X}(t_k)), t_k = k\tau,$

where τ is the inverse of the sampling frequency.

Just having a look at the plotted time series is a step that should never be skipped since this gives one valuable information on the recurrent or intermittent character of the signal. But clearly more objective analyses are required, especially when the system has reached some sort of attractor, *i.e.* a permanent regime characterized by statistically stationary signals, with distributions that do not change significantly as times goes on, and for

¹⁴For systems evolving in space, the output may also be pictures. This opens the vast field of image processing, basically extending to two-dimensional quasi-continuous arrays the viewpoint of one-dimensional, discrete scalar series on which we focus here exclusively.

which finite-length series are typical of the dynamics, provided that they are not too short (*i.e.* correspond to several turns around the attractor).

The first operations to perform on a time series of length K, $\{W_0, W_1, \ldots, W_{K-1}\}$, are of course to compute its average:

$$\overline{W} = rac{1}{K}\sum_{k=0}^{K-1} W_k \, ,$$

and its variance:15

$$\sigma_W^2 = \frac{1}{K} \sum_{k=0}^{K-1} \left(W - \overline{W} \right)^2,$$

and then to rescale it by making the changes:

$$rac{\left(W_k-\overline{W}
ight)}{\sigma_W}\mapsto W_k\,,\qquad k=0,1,\ldots,K-1\,.$$

For simplicity, we assume in the following that this preprocessing has been performed, *i.e.* that we work with a signal supposed to be stationary, with zero mean and unit variance.

In a second instance, one usually considers the autocorrelation of the signal defined as:

$$C(\Delta t) = \lim_{T \to \infty} \frac{1}{T} \int_0^T W(t) W(t + \Delta t) \,\mathrm{d}t \,. \tag{4.41}$$

In the case of a finite series of discrete records, we get:

$$C(\kappa) = \frac{1}{K-\kappa} \sum_{k=0}^{K-\kappa-1} W_k W_{k+\kappa}, \qquad (4.42)$$

but when the series is very long and as long as the shifts κ are small enough $(\kappa \ll K)$, any difference with the result of the discrete evaluation of the integral (4.41) for $\Delta t = \kappa \tau$ is not perceptible.

The autocorrelation is the simplest device to identify periodicities in the signal. It is usually not computed from its definition but rather through Fourier transforms. The next section thus begins with a brief reminder about them and continues by introducing the Hilbert transform, a useful tool to perform the demodulation of periodic signals with superimposed slow amplitude and phase variations.

¹⁵The "unbiased" variance should be defined by dividing with K-1 instead of K but we are always interested in K large so that this makes no difference.

4.4.1 Standard analysis by means of Fourier transform

On general grounds the direct Fourier transform is defined by:

$$ilde{f}(\omega) = rac{1}{2\pi} \int_{-\infty}^{+\infty} f(t) \exp(-i\omega t) \, \mathrm{d}t$$

and the inverse transform by:

$$f(t) = \int_{-\infty}^{+\infty} \tilde{f}(\omega) \exp(i\omega t) \,\mathrm{d}\omega$$

The Fourier transform of the autocorrelation function then reads:

$$S(\omega) = rac{1}{2\pi} \int_{-\infty}^{\infty} C(au) \exp(-i\omega au) \,\mathrm{d} au$$

and a straightforward computation shows that

$$S(\omega) = | ilde W(\omega)|^2 \,,$$

where $\tilde{W}(\omega)$ is the Fourier transform of W(t). $S(\omega)$ is called the *Fourier* spectrum of the signal.¹⁶

Direct inspection of the signal sometimes suggests that one has to deal with a periodic process that is modulated in amplitude and/or phase. An elegant way to perform the *demodulation* consists in constructing a complex signal Z(t), whose real part is the primitive signal W(t) and whose imaginary part is constructed so as to make a $\pi/2$ phase angle with it. This is easily understood from the consideration of a strictly periodic signal: let $W(t) = A\cos(\omega t)$, the signal at $\pi/2$ is then $A\cos(\omega t - \pi/2) = A\sin(\omega t)$, so that $Z(t) = A\exp(i\varphi(t))$, with amplitude A = |Z| and phase φ , such that $d\varphi/dt = \omega$ is the angular frequency.

In the general case, the instantaneous amplitude of the modulated signal is then given by the modulus of Z and its instantaneous period is derived from its argument by differentiating it with respect to time, provided that one knows how to obtain the signal at $\pi/2$. This is done from the Fourier transform of W(t), by copying the case of the periodic signal taken as an example above, frequency by frequency. Starting with:

$$W(t) = \int_{-\infty}^{+\infty} \tilde{W}(\omega) \exp(i\omega t) d\omega, \qquad (4.43)$$

¹⁶This result is called the Wiener-Kintchine theorem. The derivation uses a few tricks, among which the fact that the Fourier transform of a constant $f \equiv 1$ is a Dirac distribution $\delta_{\rm D}(\omega)$.

and setting:¹⁷

$$\tilde{W}(\omega) = \frac{1}{2}[A(\omega) + iB(\omega)] = \tilde{W}(-\omega)^*, \qquad (4.44)$$

one can observe that $A(\omega)$ and $B(\omega)$ are even and odd functions of ω , respectively. In full generality one gets B(0) = 0 and, further also A(0) = 0, since W(t) is assumed to have zero mean. Expression (4.43) can then be rewritten as:

$$W(t) = \int_0^{+\infty} [A(\omega)\cos(\omega t) - B(\omega)\sin(\omega t)] \,\mathrm{d}\omega,$$

which allows one to get the imaginary part, here denoted W':

$$W'(t) = \int_0^{+\infty} [A(\omega)\cos(\omega t - \pi/2) - B(\omega)\sin(\omega t - \pi/2)] d\omega,$$

which more simply reads:

$$W'(t) = \int_0^{+\infty} [B(\omega)\cos(\omega t) + A(\omega)\sin(\omega t)] d\omega$$

By definition of Z = W + iW', one then obtains:

$$Z(t) = \int_0^{+\infty} \exp(i\omega t) [A(\omega) + iB(\omega)] \,\mathrm{d}\omega, \qquad (4.45)$$

which involves a sum over the positive angular frequencies (the so-called 'analytical signal'). Comparing (4.45) and (4.43, 4.44), one obtains the Fourier transform $\tilde{Z}(\omega)$ by setting to zero all components of the Fourier transform of W corresponding to negative angular frequencies and by doubling all the others. The analytic signal Z(t) itself is then recovered by computing the inverse transform of \tilde{Z} . Mathematically W and W' are Hilbert transform of each other¹⁸ hence the expression 'Hilbert transform demodulation' (the full procedure is implemented by hilbert.m in the MATLAB software).

4.4.2 Reconstruction by the method of delays

Let us now take a point of view more in line with the theory of dynamical systems. A difficulty arises immediately from the fact no specific assumption can be made, except that a deterministic framework is relevant, so that

 $^{^{17}}$ The complex conjugation property arises from the fact that the observable W is supposed to be a real function of **X**.

¹⁸Formally, $\tilde{W}'(\omega) = -(1/\pi) \mathcal{P} \int_{-\infty}^{\infty} [\tilde{W}(\omega')/(\omega-\omega')] d\omega'$, where \mathcal{P} denote the Cauchy principal part of the integral.

one can just write formally $\dot{\mathbf{X}} = \mathcal{F}(\mathbf{X})$ for unspecified states living in some space X. In particular, the nature and the dimension d_{eff} of the manifold supporting the dynamics,¹⁹ and the explicit relation between states and the observable $W = \mathcal{W}(\mathbf{X})$ are not known a priori.

Practically all reconstruction techniques derive from the *method of delays* mathematically formalized by Takens in 1981,²⁰ which is numerically more robust than a previous approach based on the evaluation of successive time derivatives of the experimental signal in terms of finite differences.²¹

Let us illustrate the reconstruction approach using a discrete-time system for simplicity. At the beginning we assume that

$$\mathbf{X}_{k+1} = \mathcal{F}(\mathbf{X}_k)$$

and that the time series of some scalar observable \mathcal{W} is available (the method is, at least conceptually, easy to extend to the case of several observables). Reconstructing the dynamics means determining an empirical relation between the \mathbf{X}_k in their phase space only from the knowledge of the W_k , $k = 0, 1, \ldots$

A single measure $W_0 = \mathcal{W}(\mathbf{X}_0)$ is not sufficient to determine the state \mathbf{X}_0 since we surely need more than one coordinate to define it. But we assume that the next value W_1 corresponds to a point \mathbf{X}_1 that evolves from \mathbf{X}_0 under the map \mathcal{F} , unknown but existing. The second measurement thus adds a piece of information about the coordinates of \mathbf{X}_0 through $W_1 = \mathcal{W}(\mathbf{X}_1) = \mathcal{W}(\mathcal{F}(\mathbf{X}_0))$. The third one, $W_2 = \mathcal{W}(\mathbf{X}_2) = \mathcal{W}(\mathcal{F}(\mathbf{X}_1)) = \mathcal{W}(\mathcal{F}(\mathcal{F}(\mathbf{X}_0)))$ adds another piece, *etc.* In principle, a sufficiently long series of d_{test} successive measurements, $\{W_0, \ldots, W_{d_{\text{test}}-1}\}$, should serve us to specify \mathbf{X}_0 . In the same way, $\{W_1, \ldots, W_{d_{\text{test}}}\}$ would do so for \mathbf{X}_1 , *etc.* Eventually a whole trajectory would then be reconstructed from the series of vectors in $\mathbb{R}^{d_{\text{test}}}$:

$$\mathbf{V}_k = [W_k; \ldots; W_{k+d_{\text{test}}-1}] \; .$$

The concrete implementation of the method is by increasing the dimension of the reconstruction space until a consistent quantitative assessment of the observations can be given. In practice, the problem can be reformu-

¹⁹The dimension of the physical system itself is usually infinite; think of continuous media.

 $^{^{20}}$ F. Takens, "Detecting strange attractors in turbulence," Lect. Notes Math. 898 (1981) 366-381.

²¹N.H. Packard, J.P. Crutchfield, J.P., J.D. Farmer, R.S. Shaw, "Geometry from a time series," Phys. Rev. Lett. **45** (1980) 712-716.

lated in terms of the *reliability* of the reconstruction. The space of vectors \mathbf{V}_k has to be in correspondence with the region of phase space visited by the system when the permanent regime is reached. Whereas we can accept redundancy, *i.e.* a dimension d_{test} that is too large, we must not lose useful dynamical information. In mathematical terms the representation of the system must be *injective*, so that different states have different reconstructions:

$$\mathbf{X}_k \neq \mathbf{X}_{k'} \quad \Rightarrow \quad \mathbf{V}_k \neq \mathbf{V}_{k'}. \tag{4.46}$$

What was defined as a tentative number of component d_{test} is thus more mathematically understood as the dimension of the space in which the effective phase space can be *embedded* by means of some injective map. To stick with this abstract point of view more closely, we now define d_{test} as the *embedding dimension*, denote it d_e , and thus specify states in the embedding space as:

$$\mathbf{V}_{k} = [W_{k}; \dots; W_{k+d_{e}-1}], \qquad (4.47)$$

Takens' method of delays is sketched in Figure 4.19. His theorem states that the \mathbf{V}_k defined above, where the observable W is defined by a differentiable functional on phase space \mathcal{W} , achieve a reliable reconstruction provided that the dimension d_e is large enough: $d_e \geq 2d_{eff} + 1$, where d_{eff} is the topological dimension of the manifold supporting the effective dynamics.

This theorem can be understood as an extension to the present context of Witney's theorem that states that a compact *d*-dimensional differentiable manifold \mathbb{M} can be embedded in an Euclidean space \mathbb{R}^{2d+1} . As an illus-



Fig. 4.19 Illustration of Takens' theorem. \mathbb{M} is the manifold in the space \mathbb{X} over which the effective dynamics takes place. Coordinates of points representing the system in \mathbb{R}^{d_e} are obtained from the series of measurements W_k through (4.47).

tration of the latter, one can consider a loop (dimension 1) that would be projected on a plane as the figure eight. In order to resolve its structure, and in particular to check that the intersection is a fake, one must be able to look at the loop from another side, *i.e.* to stand in a $2 \times 1 + 1 = 3$ dimensional space.

In fact the reconstruction proposed by Takens is more general than that proposed by (4.47) since the \mathbf{V}_k can be any series of d_e measurements, $[W_k; W_{k+\kappa_1}; ...; W_{k+\kappa_{d_e-1}}]$, and nothing forbids it to take irregularly distributed intervals κ_q . It is however natural to take κ_q as the successive multiples of some basic κ , *i.e.* $\kappa_q = q\kappa$, $q = 1, 2, ..., d_e - 1$. When the signal is obtained from the time sampling of a continuous-time system with period τ this corresponds to a sub-sampling at period $\kappa\tau$.

The mathematical viewpoint developed so far is apparently strong. However our enthusiasm must be somewhat tempered and a pragmatic perspective has to be taken, for it is not clear that the physical systems we are interested in fulfill the theoretical conditions underlying the theorem, and first of all that the d_{eff} -dimensional manifold over which the dynamics takes place is sufficiently smooth. As it is usually the case for chaotic nonlinear systems, the attractor has a fractal dimension $d_{\rm f} > d_{\rm eff}$ and, following Mañe (1981), one can replace d_{eff} by d_{f} in the inequality for d_{e} . This still does not lead to any concrete estimate, in part because measurements are always polluted by "noise" that comes and hinders the reconstruction. Strategies have thus been developed to get around these difficulties and determine more or less optimally the two basic ingredients of any reconstruction: the base delay κ and the embedding dimension $d_{\rm e}$. This is what we now briefly discuss, inviting the interested reader to consult, e.g. [Abarbanel (1996); Kanz and Schreiber (1997); Weigend and Gershenfeld (1993)] for more detailed approaches.

4.4.3 Sampling frequency and embedding dimension

For more specificity we use a synthetic signal obtained by numerical simulation of a noisy limit cycle governed in the complex plane Z = X + iY by a Langevin equation

$$\dot{Z} = (1+i)Z - (1-i)|Z|^2 Z + \zeta(t),$$

where $\zeta(t)$ is a Gaussian noise. We assume that the trajectories are computed by a second order Runge-Kutta scheme (B.9-B.10) with time step $\delta t = 0.01$. A noise with amplitude 0.05 is added to X and Y at each time



Fig. 4.20 Top: Original noisy periodic signal in its own phase space. Bottom: Reconstruction with sampling time $\tau = 0.03$ and delays $\kappa = 5$, 25 and 45 of observable $W(t) \equiv X(t)$.

step (for the generation of Gaussian noise see [Press *et al.* (1986)]). The trajectory used for this example is displayed in Figure 4.20 (top). The period of the (deterministic) signal is $T = \pi$. The signal W(t) that we take is just the X component of the trajectory and we mimic the sampling process by recording its value every $\tau = 3\delta t$.

The first problem is that of the sampling time, and thus that of κ . An exaggerately high frequency is not an advantage since a huge volume of data is produced, the most part of which is redundant owing to the noise that blurs the information. It is then believed that the delay between two successive records must be sufficient to bring novel information. A practical rule is to take the delay corresponding to the first zero of the autocorrelation as the effective sampling time $\kappa\tau$. For the present (deterministic) sinusoidal signal, $W(t) = \sin(2t)$, the autocorrelation defined through (4.41) is $C(\Delta t) = \cos(2\Delta t)$ and this rule prescribes $\Delta t = \pi/4$, *i.e.* a quarter of a period: two measurements out of phase by such a shift are indeed fully discriminated since one is maximum when the other is zero.

The discrete estimate of the autocorrelation function (4.42) is displayed in the left part of Figure 4.21 where one can observe that the noise adds



Fig. 4.21 Autocorrelation $C(\kappa)$ and mutual information $I_{\text{mut}}(\kappa)$ of signal W in Figure 4.20 as a function of the delay κ .

a slight damping to the behavior expected for a periodic signal. Applying the rule leads to $\kappa \simeq 25$, which indeed corresponds to a quarter of a period once recalled that $\tau = 3\delta t = 0.03$. Two-dimensional reconstructions in the plane $(W_k, W_{k+\kappa})$ for $\kappa = 5$, 25 and 45 are displayed in Figure 4.20 (bottom row) which shows that the representation with $\kappa = 25$ is the most similar to the original signal, whereas for $\kappa = 5$ and $\kappa = 45$, corresponding to measurements too close in time or practically in phase opposition, the reconstructions align along directions $W_{k+\kappa} = +W_k$ and $W_{k+\kappa} = -W_k$, respectively.

The study of the autocorrelation function, of linear essence, does not always lead to a satisfactory choice of κ . Let us now mention a more general criterion resting on a similar philosophy but with a better nonlinear flavor. This criterion bears on the *mutual information* contained in two records shifted by some time amount when compared to that contained in a single record. Again, stationarity of the signal is assumed. First the probability distribution $\mathcal{P}(W)$ of $W_k = W$ is obtained by an appropriate normalization of the histogram of the values of W_k . Next, the joint probability $\mathcal{P}_{\kappa}(W', W'')$ is determined in the same way but for couples $(W_k = W'; W_{k+\kappa} = W'')$, with $\kappa = 1, 2, \ldots$ From these, the *mutual information* is defined as:

$$I_{\rm mut}(\kappa) = \sum_{W',W''} \mathcal{P}_{\kappa}(W',W'') \ln\left(\frac{\mathcal{P}_{\kappa}(W',W'')}{\mathcal{P}(W')\mathcal{P}(W'')}\right).$$
(4.48)

This quantity is a measure of the redundancy in the signal: when κ is small, points in phase space are highly correlated so that learning about

 $W_{k+\kappa}$ when W_k is known does not bring much novel information; by contrast when $\kappa \gg 1$, the points become uncorrelated and $\mathcal{P}_{\kappa}(W_k, W_{k+\kappa})$ is essentially $\mathcal{P}(W_k)\mathcal{P}(W_{k+\kappa})$, so that the mutual information is nearly zero, the information gained is the same as from independent drawings W_k and $W_{k+\kappa}$ using $\mathcal{P}(W)$. As can be seen from the right part of Figure 4.21 that displays the graph of $I_{\text{mut}}(\kappa)$ obtained by applying (4.48) to our signal, the mutual information does not decrease monotonically but in general presents a first minimum at some intermediate κ that defines an optimal value κ_{opt} corresponding to a minimum of the redundancy, before ultimately decaying owing to statistical decorrelation: taking simultaneously W_k and $W_{k+\kappa_{\text{opt}}}$ should thus give the best information about the dynamical evolution. Here, without surprise κ_{opt} is the same as that given by the previous rule, but more generally this is not the case.

It should be noted that, when no clear minimum of $I_{\text{mut}}(\kappa)$ is obtained, this can mean either the presence of a very large noise, or that the observable has been under-sampled, or that too many degrees of freedom are involved, all cases where methods of the theory of low dimensional deterministic dynamical systems are of little help.

Once the parameter κ has been determined optimally, one can re-sample the time series with the new step and discard redundant information contained in the intermediate values that are too closely correlated to the retained ones. In fact, the κ parallel time series obtained by changing the phase of the reconstruction modulo κ are not independent but yield κ equivalent reconstructions of one and the same trajectory. Comparing these reconstructions may serve to appreciate the amount of noise quantitatively.

Renumbering the time series, we can assume $\kappa = 1$ and focus on our second problem: finding the most appropriate embedding dimension $d_{\rm e}$.

Keeping in mind that we want to obtain an injective representation, we reverse the implication (4.46), now stated:

$$\mathbf{V}_k = \mathbf{V}_{k'} \quad \Rightarrow \quad \mathbf{X}_k = \mathbf{X}_{k'} ,$$

which suggests to analyze the reliability of the state identification (r.h.s.) from comparisons of different reconstructions (l.h.s) especially in spaces with different dimensions: neighbors in some space are "true neighbors" if they remain so for all reconstructions. This property can be checked by increasing $d_{\rm e}$, *i.e.* enlarging the width of the window dragged on the data.

The method of false neighbors is an efficient strategy to decide when to stop adding coordinates. Consider a trial dimension d', *i.e.* $[W_k; W_{k+1}; \ldots; W_{k+d'-1}]$ and the (d'+1)-dimensional reconstruction ob-



Fig. 4.22 Comparison of the reconstruction in a d'-dimensional space featured by the horizontal axis (coordinates $\{W_k; \ldots; W_{k+d'-1}\}$, and a reconstruction in the (d'+1)-dimensional space obtained by adding coordinate $W_{k+d'}$.

tained by adding a component $W_{k+d'}$. Choose a distance in reconstruction space and a criterion to decide which is neighbor and which is not (depending on the noise amplitude). Next determine the number of false neighbors, *i.e.* the number of pairs of points that were neighbors in d' dimensions and are no longer neighbors in d' + 1 dimension, as illustrated in Figure 4.22. Then increase d' up to the point when the fraction of false neighbors decreases significantly and choose that value as optimal embedding dimension.

The Euclidean distance derived from the L^2 norm is usually not a convenient choice for neighborhood evaluations, since it requires a lot of computations. By contrast, the distance derived from the L^{∞} -norm, *i.e.*

dist
$$(\mathbf{V}^{(1)}, \mathbf{V}^{(2)}) = \sup_{k} |V_{k}^{(1)} - V_{k}^{(2)}|.$$

only requires comparisons and is most economical.

Choosing the embedding dimension as given by the false-neighbor method should appropriately unfold the structure of the attractor. The final presentation of the result can be improved by changing from the canonical basis in the reconstruction space to a basis correlated to the data in the least-square sense.²² The method known as *proper orthogonal decomposi*-

 $^{^{22}}$ (a) The idea was introduced by D.S. Broomhead and G.P. King, "Extracting qualitative dynamics from experimental data," Physica D 20 (1986) 217–236. (b) For a concrete implementation, see R. Vautard, P. Yiou, and M. Ghil, "Singular spectrum analysis: a toolkit for short, noisy chaotic signals," Physica D 58 (1992) 95–126.



Fig. 4.23 Empirical determination of the largest Lyapunov exponent of a system in its reconstructed phase space. (a) Reconstructed attractor. (b) Rescaling of the distance between the reference portion of trajectory and its neighbor.

tion or singular value decomposition is implemented in numerical softwares such as MATLAB and is also used in the field of pattern recognition, which may be of interest in the analysis of space-time chaos to be introduced in Chapter 5.

4.4.4 Application

From a d_{e} -dimensional reconstruction one can next extract quantitative information of the system, and in particular the amount of chaos present, using quantities such as Lyapunov exponents or fractal dimensions of the attractor.

By contrast with the theoretical approach, in which the expression of the dynamical system is known, here we have just a (very long) time series of some observable at given control parameter. The attractor is then first reconstructed by the method of delays from this time series, Figure 4.23(a), and we must assume at this stage that the permanent regime is reached and that the system explores its attractor repeatedly and satisfactorily in a statistical sense. If this is the case, when the system comes in a given region of phase space, it never comes exactly at the same place but in some neighborhood. Taking a portion of the trajectory as a reference, the vector field in its neighborhood can be estimated from the set of trajectory pieces followed in close approaches to that reference portion, *e.g.* through a least-square adjustment of the coefficients of the local Jacobian matrix. This is in general a hard matter. A quantity that can however be determined more easily


Fig. 4.24 Determination of the correlation dimension in a convection experiment. Left: Distribution of distances in phase space for several embedding dimensions. Right: Variation of the slope at small R as d_e is increased. Exponent ν is seen to saturate at 2.8 which is the correlation dimension of the considered attractor. When d_e is too small, trajectories fill the reconstruction space homogeneously, as if the signal was a white noise, hence the corresponding line $\nu = d_e$. After Malraison *et al.*, Note 25.

is the largest Lyapunov exponent. The procedure used in early attempts²³ reflects particularly clearly its nature as an average trajectory divergence rate: since Lyapunov exponents are quantities relative to the tangent evolution, the distance between pieces of trajectories serving to compute the divergence rate must remain small enough; when following the reference trajectory, one is therefore obliged to look for pieces of trajectories as close as possible to the current point, in the direction of fastest divergence, as suggested in Figure 4.23(b).

A quantity that can serve to characterize the fractal properties of strange attractors is the *correlation dimension* introduced by Grassberger and Procaccia.²⁴ It is extracted from the distribution of distances between pairs of points on the reconstructed attractor $\Delta(\mathbf{V}_i, \mathbf{V}_j)$, obtained by computing

$$C(R) = \lim_{N \to \infty} \frac{1}{N^2} \sum_{\{\mathbf{X}_i, \mathbf{X}_j\}} \Upsilon \left(R - \Delta(\mathbf{V}_i, \mathbf{V}_j) \right), \qquad (4.49)$$

where $\Upsilon = 0$ for u < 0, and $\Upsilon(u) = 1$ for u > 0 (Heaviside distribution). Provided that the statistic is sufficient, this quantity measures the number of points in volume elements of radius R around each point, roughly speaking its mass, as it varies with R. For a compact *d*-dimensional object, one would have $C(R) \sim R^d$ as $R \to 0$. The correlation dimension is thus

²³A. Wolf, J.B. Swift, H.L. Swinney, J.A. Vastano, "Determining Lyapunov exponents from a time series," Physica D **16** (1985) 285-317.

 $^{^{24}}$ P. Grassberger, I. Procaccia, "Measuring the strangeness of strange attractors," Physica D 9 (1983) 189-208.

defined as

$$\nu = \lim_{R \to 0} \frac{\log\left(C(R)\right)}{\log(R)} \,. \tag{4.50}$$

The way it is defined makes exponent ν similar to $d_{\rm f}$. As such, it gives an easily determined measure of the fractal character of the attractor from the statistics of a long time record in the embedding space.

This approach has been developed many times. The example given here is the first application to a concrete convection experiment in confined geometry, by Malraison *et al.*²⁵ It is seen in Figure 4.24 (left) that the correlation integral (4.49) plotted in log-log scale as a function of R indeed display a linear part at small R from which an exponent ν can be derived. At the time of the experiment, the determination of the optimal embedding dimension was made by just increasing d_e progressively.

If the embedding dimension is chosen too small, one observes that ν is close to d_e which is easily understood from the fact that the reconstruction does not contain enough information to evidence the deterministic character of the dynamics. Points fill the V-space homogeneously as a random signal would do. On the contrary, if upon increasing d_e , it happens that ν saturates at some finite value, as here, it is a good indication that some deterministic dynamics is at work producing a low dimensional chaos, see Figure 4.24 (right). If unfortunately, ν continues to increase with d_e , either confinement effects are too weak and the effective signal is not low dimensional, or the noise level is too high.

To conclude this section, let us remark that a reliable quantitative determination of the amount of chaos requires a lot of data but that useful information can anyway be obtained from the reconstruction technique, *e.g.* to attempt controlling chaos.²⁶ Software packages are available for an automatic treatment of experimental data, *e.g.* Note 22(b).

²⁵B. Malraison, P. Atten, P. Bergé, M. Dubois, "Dimension d'attracteurs étranges: une détermination expérimentale en régime chaotique de deux systèmes convectifs," C.R. Acad. Sc. Paris **297** Série II (1983) 209-214.

²⁶Chaos control and its applications have received considerable attention in the 1990s. See, *e.g.* T. Shinbrot *et al.*, "Using small perturbations to control chaos," Nature **363** (1993) 411-417; S. Hayes *et al.*, "Communicating with chaos," Phys. Rev. Lett. **70** (1993) 3031; for another approach, see K. Pyragas, "Continuous control of chaos by self-controlling feedback," Phys. Lett. A **170** (1992) 421-428, and W. Just, *et al.*, "Mechanism of time-delayed feedback control," Phys. Rev. Lett. **78** 203-206. A more complete presentation can be found in [Schuster (1999)].

4.5 Conclusion

As a whole, one can say that the problem of the transition to turbulence in strongly confined systems is, at least at a conceptual level but also most often at a practical quantitative level, well understood in terms of temporal chaos and a small set of universal scenarios.

This framework is, by construction, that of discrete systems introduced in Chapter 2. It can also legitimately be applied to continuous systems in case of strong confinement effects, by virtue of the distinction between driving and enslaved modes and the adiabatic elimination of the latter, thus offering a vast field where a detailed comparison of theory, tools and concepts, and laboratory or numerical experiments is meaningful.

The next chapter is devoted to the study of the case when confinement effects are no longer sufficiently strong to restrict the effective dynamics to the interaction of such a small number of discrete modes.

4.6 Exercises

4.6.1 Homogeneous instability in a confined context

Consider a spatially extended system governed by

$$\partial_t v = rv + \partial_{xx}v - v^3$$
.

1) Linearized problem: determine the growth rate of Fourier modes $\propto \exp(ikx)$ with infinitesimal amplitude in a laterally unbounded system and obtain the marginal stability condition of the trivial solution $v \equiv 0$.

2) Adding boundary conditions v(x) = 0 at both ends of the interval $[0, \pi]$, check that one can take eigenmodes in the form $X_n = \sin(nx)$ and determine their growth rate as a function of n.

3) Find the condition on r such that only the first mode is excited, while others remain strongly stable. When this is the case, assuming $v(x,t) = A_1(t)X_1(x)$, determine the equation governing A_1 within a first-harmonic approximation. Show that the system experiences a supercritical fork bifurcation.

4) The cubic nonlinearity generates modes X_n , n > 1. Find the equation governing A_3 , its order of magnitude at steady state when $r \ll 1$, and justify the approximation made.

4.6.2 Cellular instability in a confined context

Go back to model (3.39) and complete it by a nonlinear advection-like term to get:

$$\partial_t v + v \partial_x v = rv - (\partial_{xx} + 1)^2 v.$$
(4.51)

Then consider a confined system with boundary conditions $v = \partial_{xx}v = 0$ at x = 0 and $x = \ell$. Check that modes in the form $\sin(k_n x)$ are appropriate and insert expansion:

$$v(x,t) = \sum_{n=1}^{\infty} A_n(t) \sin(k_n x), \qquad k_n = n\pi/\ell, \qquad (4.52)$$

in (4.51). Further separate the different harmonics and obtain the set of ordinary differential equations governing the amplitudes A_n . Write down explicitly the equations for A_1 , A_2 and A_3 when $\ell = \pi$ for $r \ll 1$. Show that in this limit A_2, A_3, \ldots are enslaved to A_1 . Propose a coherent truncation of the system above some order $N \geq 2$ fixed in advance.

When truncated at order 2 the system reads:

$$\dot{A}_1 = rA_1 + \frac{1}{2}A_1A_2 \tag{4.53}$$

$$\dot{A}_2 = -9A_2 - \frac{1}{2}A_1^2 \tag{4.54}$$

Sketch its phase portrait in the reduced phase space (A_1, A_2) . Determine the effective dynamics of A_1 obtained by adiabatic elimination of A_2 and the nature of the bifurcation as r increase from negative to positive values.

4.6.3 Landau model of bifurcation for one real mode

Let A be a real variable governed by a first-order differential equation:

$$\dot{A} = \mathcal{F}(A) \,. \tag{4.55}$$

In Chapter 2, §2.1.3 it was shown that this equation can be written in gradient form $\dot{A} = -\partial \mathcal{G}/\partial A$, so that \mathcal{G} decreases as A evolves. Consider now the neighborhood of a time-independent base state A_0 such that \mathcal{G} is stationary, $F(A_0) = -\partial_A \mathcal{G}(A_0) \equiv 0$.

1) Rewrite the equation governing the dynamics of a "perturbation" A'

defined by $A = A_0 + A'$ as a Taylor expansion

$$\dot{A}' = \sum_{n} a_n (A')^n , \qquad (4.56)$$

express the linear stability condition for A_0 in terms of the derivatives of \mathcal{F} at A_0 and interpret this condition in terms of \mathcal{G} . By appropriate notational changes and rescalings we assume that $A_0 = 0$ and that the linear growth-rate σ of A is simply given by $\sigma = r$, which define r = 0 as the bifurcation threshold. Equation (4.56) then reads

$$\dot{A} = rA + \sum_{n \ge 2} a_n A^n \,. \tag{4.57}$$

Assuming that the system can be truncated beyond some order N, using either \mathcal{F} or \mathcal{G} , show that the dynamics described by (4.57) is meaningful in the sense that A remains finite for all times, provided that N is an odd integer and a_N is negative. (When N is even, the model is only locally valid and should be completed by terms of higher degree to acquire a more global validity). Find the conditions fulfilled by coefficients a_n in (4.57) when the system is invariant through the symmetry $A \mapsto -A$.

2) When N = 2, consider the systems:

$$\dot{A} = r_0 - A^2 \tag{4.58}$$

 \mathbf{and}

$$\dot{A} = r_1 A - A^2 \,. \tag{4.59}$$

Draw the graphs giving their fixed points A_* (bifurcation diagrams) against the control parameters r_0 or r_1 , and indicate their stability properties, using solid (dashed) lines for stable (unstable) fixed points. Justify the terms saddle-node and trans-critical for these two bifurcations.

3) Same exercise when N = 3 with systems:

$$\dot{A} = rA - A^3 - H, (4.60)$$

producing the imperfect bifurcation (Fig. 4.4, p. 127) and

$$\dot{A} = rA - A^2 - A^3 \,. \tag{4.61}$$

accounting for a perfect bifurcation perturbed by a breaking of the symmetry $A \mapsto -A$. Observe that, in both cases, elements of the case N = 2 are recovered locally in the plane (r, A_*) .

6) Consider the case N = 5 with symmetry $A \mapsto -A$:

$$\dot{A} = rA + aA^3 - A^5 , \qquad (4.62)$$

coefficient a being positive or negative.

a) Determine the position of the fixed points at given (r, a).

b) When a is positive, show that the bifurcation at r = 0 is subcritical and that the system experiences hysteresis cycles as r is varied. Draw the bifurcation diagram and observe the presence of situations already studied for smaller maximum degree N. Determine the shape of potential \mathcal{G} corresponding to the different cases.

c) consider briefly (4.62) with a < 0 and identify the different nonlinear regimes according to which of the nonlinear terms controls the position of the fixed point.

Remark. Equation (4.58) is said to be the generic unfolding of the quadratic nonlinearity. It is indeed more general than (4.59) that additionally supposes the persistence of the fixed point at A = 0 [incidentally, this is also the case of (4.57)]. On more general grounds, one can always find the translation $A \mapsto A + \alpha$ that suppresses the coefficient a_{N-1} , and rescale A so that $a_N = \pm 1$. The generic unfolding of $\dot{A} = \pm A^N$ then reads:

$$\dot{A} = r_0 + r_1 A + \dots + r_{N-2} A^{N-2} \pm A^N$$
,

where the unfolding parameters r_n are perturbations, *i.e.* must remain small when compared to the absolute value of the coefficient of the highest degree term (±1 after rescaling). This system may have no fixed points, as seen above for N = 2 with (4.58) when $r_0 < 0$.

4.6.4 Variation on the theme of Exercise 4.6.3

Consider the system

$$\dot{A} = \mathcal{F}_r(A) = -A/r + 2A^2 - A^3, \quad r > 0.$$

1) Determine explicitly the potential $\mathcal{G}_r(A)$ from which it derives, up to an arbitrary constant that will be taken as zero.

2) Find the fixed points as functions of the control parameter r. Show that a pair of fixed points appear through a saddle-node bifurcation at some value of r to be determined. Study the stability of the different fixed points and compute the value of the corresponding potentials.

3) Find the asymptotic behavior of the non-trivial fixed points as $r \to +\infty$ and draw the bifurcation diagram.

4) In the Landau theory of phase transition \mathcal{G} would be a good model of free energy for a first-order transition, *e.g.* the liquid-gas transition. Compute the value $r_{\rm M}$ of parameter r corresponding to the Maxwell plateau at which the free energies of the two competing stable states are equal. Draw the graph of \mathcal{G}_r for $r < r_{\rm M}$, $r = r_{\rm M}$, and $r > r_{\rm M}$.

4.6.5 Excitable system

Consider a dynamical system in the form

$$\dot{X} = F_Y(X) = X - X^3 - Y,$$
 (4.63)

where Y is, for the moment, a parameter.

Determine the fixed points of (4.63), their number and nature (stable or unstable) as Y is varied. Determine the potential from which this equation derives (see Exercise 4.6.3, above).

Suppose now that Y is a dynamical variable of its own, governed by

$$\dot{Y} = \epsilon(X - r),$$
 (4.64)

where $0 < \epsilon$ is a constant and r the control parameter.

Determine the character of the fixed point of system (4.63, 4.64) as a function of r, assuming that $1 - 3r^2 \neq 0$ and $\epsilon \ll |1 - 3r^2|$.

Interpret the phase portrait of the system in the two cases $|r| < 1/\sqrt{3}$ and $|r| > 1/\sqrt{3}$. In the first case, explain the shape of the relaxation oscillations observed after damping of the transient. See Fig. 4.25.

In the second case, the system is said to describe an excitable medium such that, for initial conditions X > r, the relaxation of the system towards its fixed point occurs after a large pulse. [In a reaction-diffusion system where the reaction part is accounted for by (4.63, 4.64), the system develops fronts separating excited regions from relaxed one].

4.6.6 Stability and attractors of a two-dimensional system

Consider the system

$$\dot{X} = -aX + Y + XY, \qquad (4.65)$$

$$\dot{Y} = -bY - X^2 \,, \tag{4.66}$$



Fig. 4.25 Excitable system. Top: Phase portraits. Bottom: Time series of variable X

where a and b are constants. By computing the eigenvalues and eigenvectors of the system when linearized around the origin X = Y = 0, show that it remains locally stable (stable with respect to infinitesimal perturbations) as long as a > 0 and b > 0. Sketch the phase portrait of the linearized system for a < b.

Consider now the system with its nonlinear terms and determine the fixed points as functions of $\Delta = 1 - 4ab$. This quantity will serve as a control parameter in the following. Show that a saddle-node bifurcation takes place when Δ changes its sign. Find the eigenvalues and eigenvectors of the vector field linearized around each non-trivial fixed point.

Sketch the phase portrait of the system for $\Delta < 0$, $\Delta = 0$, and $\Delta > 0$ (take *e.g.* b = 1 and values of *a* that lead to simple numerical applications, in particular a = 1/4, 3/16). Observe that the stable manifold of the unstable fixed point that extrapolates its stable eigendirection is the boundary

between the attraction basins of the two stable fixed points when they exist. Notice that the nonlinearities profoundly modify the aspect of the phase portrait expected from the linear analysis close to the origin. Numerical simulation of (4.65, 4.66) is recommended.²⁷

4.6.7 Bruxellator

The system obtained in Exercise 1.5.1 reads

$$\dot{X} = A - (B+1)X + X^2Y,$$
 (4.67)

$$\dot{Y} = BX - X^2 Y \,. \tag{4.68}$$

Find its fixed points as functions of parameters A and B and discuss their linear stability properties. Show that the reaction can bifurcate from a time-independent state towards an oscillatory regime beyond some critical value $B_{\rm c}$ to be determined as a function of A.

Compute the eigenvalues at lowest order in $\varepsilon = B - B_c$ Adapt the first harmonic approximation of Chapter 2, §2.3.1.2, p. 49 to derive the amplitude equation governing the system beyond threshold.

4.6.8 Locking

In a permanent regime, an oscillator can be described using a phase variable θ defined modulo 2π and governed by $\dot{\theta} = \omega$ where ω is the angular frequency of the oscillator. Consider two such oscillators weakly but non-linearly coupled, governed by:

$$\dot{ heta}_1 = \omega_1 - f(heta_2 - heta_1),$$

 $\dot{ heta}_2 = \omega_2 - f(heta_1 - heta_2).$

Justify the fact that function f has to be periodic with period 2π . In the following, take $f(\varphi) = \frac{1}{2}K\sin(\varphi)$ and write down the system for $\bar{\theta} = \frac{1}{2}(\theta_1 + \theta_2)$ and $\varphi = \frac{1}{2}(\theta_1 - \theta_2)$ (define $\delta\omega = \frac{1}{2}(\omega_1 - \omega_2)$).

Determine the condition on K ensuring the existence of fixed points to the equation for φ . Discuss the nature of the asymptotic regime when this condition is (is not) fulfilled. Qualitatively describe the phase intermittency regime that takes place when the existence condition just misses being fulfilled.

 $^{^{27}}$ For a discussion of the physical relevance of this exercise, to be reconsidered in chapter 6, see: O. Dauchot et P.M., "Local versus global concepts in hydrodynamic stability theory," J. Phys. II France 7 (1997) 371-389.

4.6.9 Logistic map (variants)

Consider the map

$$X_{k+1} = f_a(X_k) = a - X_k^2$$
(4.69)

where a > 0 is the control parameter. Draw the graph of f_a , find its fixed points and study their stability properties.

In the following, consider the dynamics of a trajectory starting in the neighborhood of $X_*^{(+)} > 0$ for a close to a_0 where the fixed point bifurcates. Setting $a = a_0 + \epsilon$, rewrite the nonlinear map for $X' = X - X_*^{(+)}$ in terms of $\epsilon \ll 1$ at first order in ϵ .

Check that the map for X' has no fixed point in the neighborhood of X' = 0 and that the system bifurcates towards a period-2 cycle by studying the iterated map $X'_{n+2} = g_{\epsilon}(X'_n)$ as ϵ crosses zero. By an expansion to third order in X'_n conveniently simplified thanks to the assumption $\epsilon \ll 1$. Show that the bifurcated cycle is stable for ϵ small enough.

Derive the change of variables that put (4.69) in the equivalent forms $Y_{k+1} = rY_k(1-Y_k)$ and $Z_{k+1} = 1 - bZ_k^2$.

4.6.10 Delayed logistic map

Consider the map

$$X_{k+1} = rX_k \left(1 - X_{k-1}\right) \,. \tag{4.70}$$

Two initial conditions are necessary to start this recurrence which is thus a second-order discrete-time dynamical system. Its non-delayed version is one of the equivalent variants of the logistic map used in the previous exercise. 1) Convert (4.70) into a first-order system by setting $Y_k = X_{k-1}$, compute the Jacobian of the map and determine its inverse when it is possible.

2) Find the fixed points of the map $X_{k+1} = X_k = Y_{k+1} = Y_k$ and study the stability of the non-trivial fixed point $X_* \neq 0$ as a function of $r \in [1, 4]$. In order to do this, find the eigenvalues of the linearized system and their nature —real or complex— as a function of r. Show that the system displays a Hopf bifurcation in the sense of (4.32) for some value of r to be determined.

4.6.11 Arnold's cat

Consider the discrete-time two-dimensional dynamical system $(X, Y) \mapsto (U, V) = \mathcal{S}(X, Y)$ defined on the torus \mathbb{T}^2 , the unit square (modulo 1),



Fig. 4.26 von Koch snow flake.

with opposite sides identified, called "Arnold's cat map":

$$U = 2X + Y$$
$$V = X + Y$$

This system is a classical example of chaotic iteration. Determine the inverse map $(U, V) \mapsto (X, Y) = \mathcal{T}(U, V)$ and check that $\mathcal{S} \circ \mathcal{T} = \mathcal{I}$ where \mathcal{I} is the identity map. Compute the determinant of its Jacobian matrix and shows that the system is conservative (Chapter 2, Exercice 2.5.1, p. 61). Illustrate this property by considering the transform of the square (0, 0), (0, 1/3), (1/3, 0), (1/3, 1/3). Show that all trajectories are unstable. Compute the second iterate $\mathcal{S}^2 \equiv \mathcal{S} \circ \mathcal{S}$.

4.6.12 Fractals

Apply formula (4.40) for a topologically connected set, usual line, surface, volume, and observe that it yields its topological dimension as expected. Then compute the fractal dimension of the following objects:

1) von Koch snow flake (Fig. 4.26): starting with a triangle with sides of unit length, replace each side by a broken line formed with 4 segments of length 1/3 as shown in Figure 4.26 and repeat indefinitely the construction on each so obtained segment.

2) Sierpinski sets (Fig. 4.27): On the plane, start with the unit square and suppress the central square with side 1/3; repeat on each of the eight remaining squares; and so on. Same rule but starting with a triangle and dropping the center part. In three-dimensional space, start with a cube cut it into 27 cubes of side 1/3, suppress seven cubes, the six in the middle of the sides and the center one, repeat indefinitely to get a fractal sponge.

3) Cantor dust (Fig. 4.28): Take the square and keep the elements shown,



Fig. 4.27 Two-dimensional Sierpinski sets.



Fig. 4.28 Cantor dust.

repeat indefinitely. Observe that the fractal dimension can be an integer and that here the result could have been obtained by noticing that, at every step, the whole set is in one-to-one correspondence with a continuous interval (by projection along the indicated direction).

4.6.13 Curry-Yorke model (transition $QP \rightarrow chaos$)

Consider a two-dimensional map²⁸ expressed in terms of the two components (X, Y) components of a complex number Z = X + iY, and defined

²⁸ J. Curry, J.A. Yorke, "A transition from Hopf bifurcation to chaos: computer experiment with maps on \mathbb{R}^2 ," Springer Notes in Mathematics **668** (1977) 48ff.

in two steps:

(1) a rotation of $Z_k = X_k + iY_k = |Z|_k \exp(i\varphi_k)$ by an angle φ :

 $\varphi_{k+1/2} = \varphi_k + \bar{\varphi} \pmod{2\pi}$

and a stretching of the modulus $|Z|_{k+1/2}$:

$$|Z|_{k+1/2} = (1+\eta)\log(1+|Z|_k),$$

where η is a control parameter controlling the stability of the fixed point at the origin. (2) a nonlinear transformation is expressed in Cartesian coordinates through:

$$X_{k+1} = X_{k+1/2}\,, \qquad Y_{k+1} = Y_{k+1/2} + X_{k+1/2}^2\,.$$

The model can be understood as the Poincaré map of a time-continuous system with a limit cycle bifurcating towards a two-periodic regime at $\eta = 0$ and may serve to illustrate the breakdown of a torus into a chaotic attractor.

Draw the attractors obtained numerically for $\bar{\varphi} = 2$ and various values of $\eta > 0$.

[Answer: Fig. 4.29. For $\eta = 0.27$ the attractor is a smooth loop (section of a smooth torus corresponding to a two-periodic regime for the time continuous system). For $\eta = 0.48$, the attractor is a singular curve with fractal structure generated by stretchings and foldings in a regime of developed chaos. These corrugations appear around $\eta = 0.40$ which is thus close to the border of chaos.]

4.6.14 Permanent vs. transient chaos

Consider one-dimensional maps in the form

$$X_{k+1} = \mathcal{F}(X_k)$$

and more particularly the tent map defined by

$$egin{aligned} 0 &\leq X_k \leq 1/2 & X_{k+1} = lpha X_k \,, \ 1/2 &\leq X_k \leq 1 & X_{k+1} = lpha (1-X_k) \,. \end{aligned}$$

1) Suppose first $\alpha = 2$ and draw the graph of \mathcal{F} and check that interval $\mathbb{I}_0 \equiv [0,1]$ is invariant in the sense that it contains its image by \mathcal{F} . Then find the fixed points of the map and show that they are unstable. Draw the graph of the iterated map $\mathcal{F} \circ \mathcal{F}$, *i.e.* $X \mapsto \mathcal{F}(\mathcal{F}(X))$. Find its fixed points and obtain from them a periodic orbit of \mathcal{F} with period 2. Show that this



Fig. 4.29 The Curry-Yorke map $\bar{\varphi} = 2$ and several values of η .

orbit is also unstable. The orbit of an initial condition chosen at random is usually (with probability 1) chaotic. Compute its Lyapunov exponent using (4.37).

2) Suppose now that $1 < \alpha < 2$ and draw the graph of \mathcal{F} . Find the image \mathbb{I}_1 of \mathbb{I}_0 and the image \mathbb{I}_2 of \mathbb{I}_1 , so that the definition interval of the map can be restricted to \mathbb{I}_1 . Show that all trajectories are unstable by computing the Lyapunov exponent.

3) Consider the case $2 < \alpha$. The interval \mathbb{I}_0 is no longer invariant. Draw the corresponding graph of \mathcal{F} and find the set \mathbb{J}_1 of initial conditions that escape at the first iteration. Construct the pre-image \mathbb{J}_2 of this set, *i.e.* such that $\mathcal{F}(\mathbb{J}_2) = \mathbb{J}_1$, to find initial conditions that escape in two iterations, and notice that it is made of two disconnected parts \mathbb{J}_2^1 and \mathbb{J}_2^2 . Continue and observe that the initial conditions that do not escape belong to a Cantor set. When $\alpha = 3$, the construction rule is exactly that of the triadic Cantor set. From the formula, compute the Lyapunov exponent of a trajectory starting right on this set (and remains in it). Such an unstable invariant set is called a *chaotic repellor*. This page intentionally left blank

Chapter 5

Nonlinear Dynamics of Patterns

The theory of deterministic chaos represents an important conceptual advance and offers an appropriate framework to account for experiments in confined geometry. However, it becomes rapidly inapplicable when the effective dimension of the dynamics increases, which is the case for extended systems (large aspect ratio). As already indicated, eigenmodes are then quasi-degenerate with wavevectors typically such that

$$k_{n+1}-k_n=\pi/\ell\ll k_c\,,$$

 ℓ being the lateral extension (see p. 117). Many modes may be unstable close to the threshold. A large part of the interaction between them can be understood in terms of *linear* interferences accounting for spatial modulations brought to a regular uniform reference pattern (at the limit of a laterally unbounded system), e.g. parallel straight rolls for convection. At the *nonlinear* stage, focusing on the modulations directly leads one to the *envelope* formalism that gives a satisfactory account of confinement effects and defects at lowest order, a first step toward understanding *spatiotemporal chaos*.

For simplicity, we mostly limit this presentation to the case of cellular stationary instabilities. The case of dissipative waves, just briefly introduced in §5.4.4, is left to more advanced studies. We consider first, in §5.1, the determination of uniform quasi-one-dimensional structures, periodic in a single space direction. Then we generalize the theory to the case of quasi-two-dimensional patterns, with square or triangular/hexagonal plan-forms, in §5.2 and §5.3. Their most universal instability modes in terms of long wavelength perturbations are next investigated in §5.4. We conclude the chapter by §5.5 where we present some other modeling approaches that may help one analyze space-time chaotic regimes in extended systems.

5.1 Quasi-one-dimensional Cellular Structures

5.1.1 Steady states

Consider an instability mechanism which, like plain Rayleigh-Bénard convection, generates stationary structures locally periodic in a single space direction, say x. At the linear stage we have:

$$\mathcal{L}(\partial_x, \partial_t; R) \mathbf{V} = 0, \qquad (5.1)$$

the solutions of which are searched for in the form:

$$\mathbf{V} = \bar{\mathbf{V}} \exp(ikx + st) \,, \tag{5.2}$$

which leads to the dispersion relation:

$$\mathcal{L}(ik,s;R) = 0. \tag{5.3}$$

We are particularly interested in the neighborhood of the critical conditions $R \approx R_c$, corresponding to a marginal mode $k = k_c$ for which $\mathcal{R}e(s) = \sigma = 0$. $(\mathcal{I}m(s) \equiv 0 \text{ along a stationary instability branch.})$ At threshold, we have $\mathcal{L}(ik_c, 0; R_c) = 0$, so that the linear problem (5.1) has a non-trivial solution:

$$\mathbf{V}_{\rm c} = \bar{\mathbf{V}}_{\rm c} \exp(ik_{\rm c}x) + {\rm c.c.} , \qquad (5.4)$$

where $\bar{\mathbf{V}}_{c}$ accounts for the structure of the critical normal mode and c.c. means 'complex conjugate'.

The nonlinear problem extending (5.1) reads:

$$\mathcal{L}(\partial_x, \partial_t; R) \mathbf{V} = \mathcal{N}(\mathbf{V}, \mathbf{V}), \qquad (5.5)$$

where $\mathcal{N}(\mathbf{V}, \mathbf{V})$ represents the higher order terms that were neglected in the linearization procedure. The notation suggests formally quadratic nonlinearities, as in hydrodynamics. The solution is searched for as an expansion in powers of a small parameter ϵ :

$$\mathbf{V} = \epsilon \mathbf{V}_1 + \epsilon^2 \mathbf{V}_2 + \epsilon^3 \mathbf{V}_3 \dots, \tag{5.6}$$

and, as in the calculation of the period of nonlinear oscillators (Chapter 2, especially §2.3.2.3), the control parameter is also expanded:

$$R = R_{\rm c} + \epsilon R_1 + \epsilon^2 R_2 + \dots \tag{5.7}$$

Isolating the distance to threshold in the expression of the linear part on the l.h.s. of (5.5) we can write:

$$\mathcal{L} = \mathcal{L}_{\rm c} - (R - R_{\rm c})\mathcal{M}, \qquad (5.8)$$

where \mathcal{M} is the opposite of the formal derivative of \mathcal{L} with respect to R evaluated at threshold. Inserting expansions (5.6, 5.7) into (5.5) and taking (5.8) into account, we get a series of linear problems:

$$\mathcal{L}\mathbf{V}_1 = 0 \tag{5.9}$$

$$\mathcal{L}\mathbf{V}_2 = R_1 \mathcal{M} \mathbf{V}_1 + \mathcal{N}(\mathbf{V}_1, \mathbf{V}_1), \qquad (5.10)$$

$$\mathcal{C}\mathbf{V}_3 = R_2\mathcal{M}\mathbf{V}_1 + R_1\mathcal{M}\mathbf{V}_2 + \mathcal{N}(\mathbf{V}_2,\mathbf{V}_1) + \mathcal{N}(\mathbf{V}_1,\mathbf{V}_2), \quad (5.11)$$

of the general form:

$$\mathcal{L}\mathbf{V}_k = \mathbf{F}_k \,. \tag{5.12}$$

The first problem is homogeneous. Since the critical conditions are fulfilled, it has a non-trivial solution $\mathbf{V}_1 \propto \mathbf{V}_c$. The higher order problems (k > 1)are all inhomogeneous and depend on the solutions computed at previous orders (k' < k). Unknown free quantities R_k introduced through (5.7) are fixed by the condition that the r.h.s. of (5.12) do not contain resonant terms, exactly like in the Poincaré-Lindstedt calculation, p. 55ff.

At a formal level, let the relevant scalar product¹ be denoted by $\langle ... | ... \rangle$ and the adjoint \mathcal{L}^{\dagger} to \mathcal{L} be defined by:

$$\langle {f W} | {\cal L} {f V}
angle = \langle {f V} | {\cal L}^\dagger {f W}
angle^st$$
 ,

the conditions (Fredholm alternative) fixing the unknown parameters in (5.7) read:

$$\langle ilde{\mathbf{V}} | \mathbf{F}_k
angle = 0$$
 .

where $\tilde{\mathbf{V}}$ is the solution generating the kernel of \mathcal{L}^{\dagger} ($\mathcal{L}^{\dagger}\tilde{\mathbf{V}}=0$).

From (5.9) one gets $\mathbf{V}_1 \propto \mathbf{V}_c$ and when applied to (5.10), the condition for k = 2 yields R_1 . Once this condition is fulfilled, a particular solution can be found, to which one may add an arbitrary solution of the homogeneous problem to obtain the most general solution. To fix this solution uniquely, one may ask it to be orthogonal to $\tilde{\mathbf{V}}_c$, so that it appears as a true correction to the first order solution (in the sense of the scalar product):

$$\langle \mathbf{V}_{\mathrm{c}} | \mathbf{V}_{2} \rangle = 0$$
.

¹See Appendix A, §A.3.2 for a reminder.

Once V_2 is determined, it is reported in (5.11), where the sole unknown on the r.h.s. is R_2 , and so on.

When $R_1 \neq 0$, one can truncate the expansion at lowest order, which gives:

$$R = R_{\rm c} + \epsilon R_1, \qquad \mathbf{V} = \epsilon \mathbf{V}_{\rm c},$$

and, after elimination of ϵ between the two equations

$$\mathbf{V} = \frac{(R-R_{\rm c})}{R_1} \mathbf{V}_{\rm c} \,,$$

so that the bifurcation is in fact two-sided, the solution exists for both $R < R_c$ and $R > R_c$, and its amplitude varies linearly. This situation was encountered in Exercise 4.6.3, Eq. (4.59) describing what was called a *trans-critical* bifurcation where two solutions exchanged their stability. Though non-generic, this situation was shown to occur in the absence of $A \rightarrow -A'$ symmetry.

In fact, it often happens that $R_1 = 0$ for symmetry reasons. This is the case of Rayleigh-Bénard convection with symmetric top/bottom boundary conditions within the Boussinesq approximation. The ' $A \mapsto -A$ ' symmetry then results from the translation invariance by $\lambda_c/2$ in the direction perpendicular to the roll axis. When $R_1 = 0$, the lowest non-trivial truncation of the expansion is one order higher:

$$R=R_{
m c}+\epsilon^2R_2\,, \qquad \qquad {f V}=\epsilon{f V}_{
m c}+\epsilon^2{f V}_2\,,$$

and thus, neglecting $\epsilon^2 \mathbf{V}_2$ when compared to $\epsilon \mathbf{V}_c$ for ϵ sufficiently small, we obtain:

$$\mathbf{V}\simeq\pm\sqrt{(R-R_{
m c})/R_2}\,\mathbf{V}_{
m c}$$

The bifurcation is now one-sided. Bifurcated states are to be found either for $R > R_c$ when $R_2 > 0$ (supercritical bifurcation) or for $R < R_c$ when $R_2 < 0$ (subcritical).

Explicit calculation shows that Rayleigh-Bénard convection between good-conducting plates is supercritical. It is of course possible to continue the expansion and determine a more accurate solution by going to next order k+1 since everything is known at order k (R_{k-1}) or can be determined (\mathbf{V}_{k-1}) and that the compatibility condition contains only R_k as unknown.

5.1.2 Amplitude equation

We now turn to a variant of the same calculation that brings back a problem similar to that studied in Chapter 4, $\S4.1$, reintroducing time in a way similar to the method of multiple scales, Chapter 2, $\S2.3.2.4$.

We consider the emergence stationary dissipative structure and make it explicit that the bifurcation is supercritical by defining a new small parameter ε through:

$$R = R_{\rm c} + \varepsilon^2 \,. \tag{5.13}$$

We again search the solution as an expansion:

$$\mathbf{V} = \varepsilon \mathbf{V}_1 + \varepsilon^2 \mathbf{V}_2 + \dots, \tag{5.14}$$

but we no longer assume that it is time-independent. In order to account for this new feature we introduce a *slow* time scale t_1 in addition to the natural time scale that we now denote as t_0 by setting:

$$\partial_t = \partial_{t_0} + \varepsilon^2 \partial_{t_1} \,. \tag{5.15}$$

The order of t_1 in ε results from the choice (5.13) and anticipates the fact that, close to the threshold, the growth rate of perturbations varies as $R - R_c$. Here, since the instability is stationary, the action of ∂_{t_0} is trivial.² Let us come back to (5.5) and expand also the ∂_t present in \mathcal{L} . With respect to system (5.9,...), in addition to the assumptions $R_1 = 0$ and $R_2 = 1$ inherent in (5.13) the first important modification enters Eq. (5.11) where a term ∂_{t_1} appears. We can thus rewrite it as:

$$\mathcal{L}\mathbf{V}_3 = \mathcal{M}\mathbf{V}_1 + \mathcal{N}(\mathbf{V}_2, \mathbf{V}_1) + \mathcal{N}(\mathbf{V}_1, \mathbf{V}_2) - \mathcal{Q}\,\partial_{t_1}\mathbf{V}_1\,, \qquad (5.16)$$

where \mathcal{Q} is the opposite of the operator obtained by differentiating \mathcal{L} with respect to ∂_t formally.

The solution at order ε reads

$$\mathbf{V}_1 = A_1 \bar{\mathbf{V}}_{\rm c} \exp(ik_{\rm c} x) + {\rm c.c.}$$

where A_1 is now a function of the slow variable t_1 .

At order ε^2 , Eq. (5.10) is left unchanged with the introduction of time. The compatibility condition that determined R_1 in the previous approach, is now trivially fulfilled by assumption. Eq. (5.10) contains non-resonant

²Things would be different for an oscillatory instability, in which case we would have $\partial_{t_0} = -i\omega_c$, but the approach can easily be extended owing to what we learned in Chapter 2.

terms of the form $\exp(ink_c x)$ with n = 0 and $n = \pm 2$ issued from the evaluation of $\mathcal{N}(A_1 \exp(ik_c x) + \text{c.c.}, A_1 \exp(ik_c x) + \text{c.c.})$. A particular solution can thus be found in the form:

$$\mathbf{V}_{2,\mathrm{part}} = |A_1|^2 \mathbf{V}_{20} + (A_1^2 \exp(2ik_\mathrm{c} x) \mathbf{V}_{22} + \mathrm{c.c.}),$$

where the first subscript indicates the order in ε and the second one the harmonic generated by the nonlinear couplings. A solution of the homogeneous problem, $A_2 \bar{\mathbf{V}}_c \exp(ik_c x) + \text{c.c.}$, must be added to this particular solution in order to obtain the full solution at order ε^2 .

At order ε^3 , it is easily seen that Eq. (5.16) contains a certain number of resonant terms coming from the evaluation of $\mathcal{N}(\mathbf{V}_2, \mathbf{V}_1) + \mathcal{N}(\mathbf{V}_1, \mathbf{V}_2)$. Because $\mathbf{V}_{2,\text{part}}$ contains harmonics 0 and ± 2 and \mathbf{V}_1 harmonics ± 1 , these terms contain harmonics 0 ± 1 , *i.e.* ± 1 , and $\pm 2 \pm 1$, producing ± 3 and ± 1 . Other resonant terms come from $R_2\mathbf{V}_1$ (with $R_2 = 1$ by definition), and $\mathcal{Q} \partial_{t_1}\mathbf{V}_1$. Instead of giving R_2 as in the previous approach, the compatibility condition now reads:

$$0 = A_1 \langle \tilde{\mathbf{V}} | \mathcal{M} \tilde{\mathbf{V}}_c \rangle + \langle \tilde{\mathbf{V}} | \mathcal{N}(\mathbf{V}_2, \tilde{\mathbf{V}}_c) + \mathcal{N}(\tilde{\mathbf{V}}_c, \mathbf{V}_2) \rangle - \partial_{t_1} A_1 \langle \tilde{\mathbf{V}} | \mathcal{Q} \tilde{\mathbf{V}}_c \rangle.$$
(5.17)

One gets easily convinced that the second term is of the form $|A_1|^2 A_1$ by counting the powers of $\exp(ik_c x)$, so that (5.17) effectively reads:

$$\tau_0 \partial_{t_1} A_1 = a A_1 - g |A_1|^2 A_1 , \qquad (5.18)$$

where τ_0 , a, and g are constants that can be evaluated by computing the scalar products in (5.17). Returning to notations introduced in Chapter 3, especially through (3.23), p. 84, coefficient a can be identified with R_c^{-1} since the linear growth rate was defined there as $\sigma = \tau_0^{-1} (R - R_c)/R_c$.

For a stationary instability, it is easily shown that if the system is symmetrical under the change $x \mapsto -x$, then (5.18) has real coefficients since this symmetry implies symmetry under complex conjugation: $A \exp(ik_c x) \mapsto A^* \exp(-ik_c x)$. This is no longer the case for oscillatory instabilities and dissipative waves which are much more complicated in this respect, see §5.4.4 below.

The expansion can be continued. It is then observed that the freedom introduced by the R_k in (5.6) for the computation of the time-independent solutions (5.7) is now replaced by the introduction at each order of the successive amplitudes A_k . The Fredholm alternative now governs these A_k as functions of the slow time t_1 .

At lowest order we have just $A = \varepsilon A_1$. Coming back to the natural time t using (5.15), observing that all terms of (5.18) are of order ε^3 , we get:

$$\tau_0 \partial_t A = a(R - R_c)A - g|A|^2 A \tag{5.19}$$

called the *amplitude equation*. As announced, this result is in line with the argument developed at the beginning of Chapter 4, and especially with that leading to (4.7).

5.2 Dissipative Crystals

We now generalize the previous calculation for the case of systems that are isotropic in the (x, y) plane perpendicular to direction z singled out by the instability mechanism. The linearized problem now reads:

$$\mathcal{L}(\nabla_{\perp}, \partial_t; R) \mathbf{V} = 0, \qquad \nabla_{\perp} \equiv (\partial_x, \partial_y) \tag{5.20}$$

and, owing to the orientation degeneracy, its solution can be searched for as a superposition of plane waves with wavevectors in different directions

$$\mathbf{V}_{1} = \sum_{\mathbf{k}_{j}} \bar{\mathbf{V}}_{\mathbf{k}_{j}} \exp(i\mathbf{k}_{j} \cdot \mathbf{x}_{h}) + \text{c.c.} , \qquad (5.21)$$

with $|\mathbf{k}_j| = k_c$ for all j.

The expansion in powers of ϵ is performed as in §5.1.1 and again leads to (5.9, ...). Let us work sequentially as before. Though the equation at order ϵ^2 remains formally the same as in the one-dimensional case, the twodimensional character of the problem implies novelty in the determination of resonant terms: nonlinearities $\mathcal{N}(\mathbf{V}_1, \mathbf{V}_1)$ that were then only able to generate the non-resonant harmonics 0 and 2, can now produce resonant combinations as shown in Figure 5.1(b).

We can now rewrite (5.10) in the form

$$\mathcal{L}\mathbf{V}_{2} = R_{1} \sum_{\mathbf{k}_{j}} \mathcal{M}\bar{\mathbf{V}}_{\mathbf{k}_{j}} \exp(i\mathbf{k}_{j} \cdot \mathbf{x}_{h}) + \sum_{\mathbf{k}_{j^{\prime}},\mathbf{k}_{j^{\prime\prime}}} \mathcal{N}(\bar{\mathbf{V}}_{\mathbf{k}_{j^{\prime}}},\bar{\mathbf{V}}_{k_{j^{\prime\prime}}}) \exp(i(\mathbf{k}_{j^{\prime}}+\mathbf{k}_{j^{\prime\prime}}) \cdot \mathbf{x}_{h}), \quad (5.22)$$

so that when $\mathbf{k}_{j'} + \mathbf{k}_{j''}$ falls right on the critical circle, we must compensate this term with a term in $R_1 \neq 0$ for some well chosen \mathbf{k}_j . In the case of formally quadratic nonlinearities, it is thus generically expected that solutions at order ϵ^2 exist in the form of regular superpositions of three wavevectors at angles $2\pi/3$ (and their opposites). Such solutions bifurcate



Fig. 5.1 Spatial resonance at second order: (a) Non-resonant combination $\mathbf{k} = \mathbf{k}_1 + \mathbf{k}_2$ with $|\mathbf{k}| \neq k_c$. (b) Resonant combination $\mathbf{k} = \mathbf{k}_1 + \mathbf{k}_2$ with $|\mathbf{k}| = k_c$. (c) Regular superposition of three wavevectors making a three-branch star at $2\pi/3$ forming a resonant set with their opposites (dashed arrows).

trans-critically as already shown and only special circumstances can suppress them by canceling the scalar products involving the last terms on the r.h.s. of (5.22) for symmetry reasons (e.g. the top-bottom symmetry in convection).

Let us suppose that the nonlinearities do not generate resonant terms at second order. The solution is still *a priori* made of a superposition of linear modes (5.21). So, let us consider two pairs of wavevectors $\pm \mathbf{k}_1$ and $\pm \mathbf{k}_2$ and start with:

$$\mathbf{V}_1 = \bar{\mathbf{V}}_{\mathbf{k}_1} \exp(i\mathbf{k}_1 \cdot \mathbf{x}_h) + \bar{\mathbf{V}}_{\mathbf{k}_2} \exp(i\mathbf{k}_2 \cdot \mathbf{x}_h) + \text{c.c.}$$

The special solution at second order then formally reads:

$$\mathbf{V}_2 = \sum_{\pm} \mathbf{V}_{\mathbf{k}_1,\mathbf{k}_2}^{(\pm)} \exp(i(\mathbf{k}_1 \pm \mathbf{k}_2) \cdot \mathbf{x}_h) + \mathrm{c.c.} \; .$$

Inserting these expressions in (5.11), we obtain terms with space dependences in the form:

$$\exp(i(\mathbf{k}_{j'} + \mathbf{k}_{j''} + \mathbf{k}_{j'''}) \cdot \mathbf{x}_{\mathrm{h}}), \quad \text{with} \quad \mathbf{k}_{j'}, \mathbf{k}_{j'}, \mathbf{k}_{j'''} = \pm \mathbf{k}_1 \text{ or } \pm \mathbf{k}_2,$$

so that we cannot avoid the generation of resonant terms through the relation:

$$\mathbf{k}_{j'} + \mathbf{k}_{j''} + \mathbf{k}_{j'''} = \mathbf{k} \,,$$

with **k** lying right on the critical circle, as shown in Figure 5.2.

The difficulty in the computation only comes from the fact that this relation can be fulfilled in many ways. Among all possible combinations, those involving a single pair of wavevectors are immediately identified,



Fig. 5.2 Spatial resonance at third order: (a) An example of resonant superposition; (b) N pairs of wavevectors making an angle π/N between nearest neighbors, here with N = 3. (c) Combinations with a single pair of wavevectors $\pm k$ (rolls). (d, e) Combinations with two pairs of wavevector forming a parallelogram, degenerate (d) or not (e).

which we already had in the quasi-one-dimensional case. But other combinations with nontrivial contribution to R_2 are readily discovered, Fig. 5.2(a), whose contributions further depend on the angles made by the wavevectors pairs in a quantitative way.

In general one is first interested in the coefficient $R_2^{(N)}$ associated to superpositions of N pairs of wavevectors forming angles of π/N between neighbors, Fig. 5.2(b). Apart from stability considerations to be examined in §5.3 below, superpositions with an arbitrary number of pairs can be considered. However, obtaining *simple* periodic patterns implies either N =1 for *rolls*, N = 2 for *squares*, or N = 3 for *hexagons* as demonstrated for standard two-dimensional crystals.³

Superpositions with more than three regularly disposed pairs of wavevectors in general form *multi-periodic* patterns called *quasi-crystals*. The latter may degenerate into periodic *super-lattices* with large periods when certain commensurability relations are fulfilled. See Chapters 5 and 6 of [Rabinovich *et al.* (2000)]. The parallel is complete with periodic locking

³See, Chapter 13 of L.D. Landau & E.M. Lifshitz, *Statistical Physics* (Butterworth-Heinemann, 1980).

of quasi-periodic regimes studied in Chapter 4, 4.2.3. A natural extension of the temporal setting would suggest the existence of *chaotic crystals*.⁴

5.3 Short Term Selection of Patterns

What has just been said relates only to the existence of stationary nonlinear dissipative structures but the approach leading to the amplitude equation (5.19) for a roll pattern, characterized by a single complex amplitude A, can be reproduced for more complex patterns with superpositions of plane waves $\exp(i\mathbf{k}_j \cdot \mathbf{x}_h)$, with amplitudes A_j and their complex conjugates. Symmetry and resonance considerations (see Exercise 5.6.1) lead to phenomenological amplitude equations generalizing (5.19). Assuming formally cubic nonlinearities (or formally quadratic but such that relevant resonances at second order are killed for symmetry reasons), one obtains:

$$\tau_0 \partial_t A_j = a(R - R_c) A_j - g_0 |A_j|^2 A_j - \sum_{j'} g_{jj'} |A_{j'}|^2 A_j$$
(5.23)

where coefficient g_0 and $g_{jj'}$ respectively accounts for interactions of type (c) and (d, e) displayed in Figure 5.2.

When formally quadratic interactions do not kill resonances at second order, sets of three pairs of wavevectors, \mathbf{k}_j , j = 1, 2, 3, such as in Figure 5.1(b) have to be considered with the corresponding amplitudes A_j , j = 1, 2, 3. The resonance condition $\mathbf{k}_1 = -\mathbf{k}_2 - \mathbf{k}_3$ then implies the presence of a term $A_2^*A_3^*$ in the equation for A_1 , which leads to:

$$\tau_0 \partial_t A_1 = a(R - R_c) A_1 - \tilde{g} A_2^* A_3^* - g[|A_1|^2 + \mu(|A_2|^2 + |A_3|^2)] A_1 \quad (5.24)$$

and two other equations obtained by circular permutation of the subscripts. Coefficients g, μ and \tilde{g} in (5.24) are to be determined from a detailed nonlinear calculation or just introduced through a phenomenological argument.

Regular configurations correspond to specific fixed points of the amplitude equations (5.23) or (5.24) with $|A_j| = A$ for all j. Patterns selected by nonlinearities can be discussed from the stability of these fixed points in the strictly temporal setting of Chapters 2 and 4. The case of squares is examined in Exercise 5.6.1.

⁴A.C. Newell and Y. Pomeau, "Turbulent crystals in macroscopic systems," J. Phys. A: Math. Gen. **26** (1993) L429-L434.

5.4 Modulations and Envelope Equations

The previous analyses all referred to uniform cellular structures with wavelengths equal to the critical wavelength λ_c . By assumption, confinement effects are weak in extended geometry and thus may prove unable to control the development of such "ideal" structures. "Natural" patterns that develop are therefore usually disordered, with local vectors **k** such that $|\mathbf{k}| \approx k_c$ but with slowly variable lengths and/or orientations not everywhere perfectly aligned with the directions defining a unique underlying reference pattern. Defects can also perturb the regular ordering of the individual cells. An example from convection in a large Prandtl number fluid was displayed in Figure 3.10, p. 98. Imperfect patterns are usually called *textures*.

The problem is approached within the framework of so-called *envelope* equations, which adds a spatial meaning to the temporal dimension of the amplitude equations introduced above, thus making possible the description of slow modulations to regular patterns and universal instabilities attached to them.

The derivation again starts with a solution to the nonlinear problem as a superposition of modes but with amplitudes that can be *slowly* varying in time and space, now called *envelopes*. Through the derivation process, most specificities of the primary instability are rubbed out, so that the result is expected to bear a universal content: all patterns with the same symmetries behave in the same way. Here we restrict ourselves to a heuristic approach⁵ mostly using symmetry arguments and we rest the discussion on properties of the linear *dispersion relation* in the neighborhood of the threshold as discussed earlier in Chapter 3, §3.1.6, p. 84. We consider first the case of a one-dimensional cellular instability before proceeding to several extensions.

5.4.1 Quasi-one-dimensional cellular patterns

At distance $r = (R - R_c)/R_c$ from threshold R_c , the growth rate of a normal mode with wavevector $k = k_c + \delta k$ is generically given by (3.26), *i.e.*:

$$\tau_0 \sigma = r - \xi_0^2 \delta k^2 \,. \tag{5.25}$$

⁵For technical details consult the seminal paper by A.C. Newell, "Envelope equations," Lectures in Appl. Math. **15** (1974) 157–163; or else: S. Fauve, "Pattern forming instabilities" in [Godrèche and Manneville (1998)]; A.C. Newell, Th. Passot, and J. Lega, "Order parameter equations for patterns," Annu. Rev. Fluid Mech. **25** (1993) 399–453; [Cross and Hohenberg (1993)] or else [Rabinovich *et al.* (2000)].

Slightly above threshold $(0 < r \ll 1)$ the wavepackets serving to build the modulated pattern are made of unstable wavevectors (with $\sigma > 0$) in the band $k \in [k_c - \Delta k, k_c - \Delta k]$ with $\Delta k = \xi_0^{-1}\sqrt{r}$. The space modulations are thus slow when compared to variations at the scale of the wavelength λ_c ($\Delta k/k_c \ll 1$). Two time variables, t_0 and t_1 , were introduced in order to obtain the amplitude equations. In the same way, two space variables are defined, a fast one and a slow one, x_0 and x_1 , respectively. Like the slow time variable t_1 , the slow space variable is linked to the distance to threshold r through the growth rate. From (5.25), we guess:

$$\partial_{x_1x_1} \sim \delta k^2 \sim r\sigma \sim \partial_{t_1}$$

A systematic expansion in powers of a small parameter ε should therefore rest on the assumptions:

$$r = \varepsilon^2, \quad \partial_t \mapsto \partial_{t_0} + \varepsilon^2 \partial_{t_1}, \quad \partial_x \mapsto \partial_{x_0} + \varepsilon^2 \partial_{x_1},$$

 \mathbf{and}

$$V = \varepsilon A_1(x_1, t_1) \exp(ik_{
m c}x_0) + {
m c.c.} + \dots$$

where the distinction between the *carrier wave* at k_c and the *modulation* is made explicit.

The translation in physical space (slow x_1) is easily obtained by formally performing an inverse Fourier transform $i\delta k \mapsto \partial_{x_1}$ which leads to

$$\tau_0 \partial_{t_1} A_1 = A_1 + \xi_0^2 \partial_{x_1 x_1} A_1 \,. \tag{5.26}$$

After having unfolded the space dependence we have to add the contribution of the nonlinearities previously computed and accounted for in (5.18). Back to the natural variables with $A = \varepsilon A_1$, (5.26) reads:⁶

$$\tau_0 \partial_t A = rA + \xi_0^2 \partial_{xx} A - g|A|^2 A \tag{5.27}$$

where it can be seen that each term is of order $r^{3/2}$, which is in fact the lowest significant order.

In the case of convection, Segel⁷ has shown that one could account for lateral boundary effects inhibiting the instability mechanism by imposing:

$$4(x_{\rm b}, t) = 0 \tag{5.28}$$

⁶As already noticed, coefficient a in (5.18) is absorbed in the definition of r.

⁷L.A. Segel, "Distant side-walls cause slow amplitude modulation of cellular convection," J. Fluid Mech. **38** (1969) 203-224.

at the position of the lateral wall x_b . In a semi-infinite medium, setting the origin at the wall, one easily determines the profile of the modulation by integrating the second order differential equation

$$\xi_0^2 rac{{
m d}^2}{{
m d} x^2} A + r A - g A^3 = 0$$
 .

(Since the equation is invariant under the change $A \mapsto A \exp(i\varphi)$ one can choose the phase A that makes it real.) By identification one finds

$$A=A_0 anh(x/\xi\sqrt{2})$$

with $A_0 = \sqrt{r/g}$ and where $\xi = \xi_0/\sqrt{r}$, often called the *coherence length*, diverges in the vicinity of the threshold.

5.4.2 2D modulations of quasi-1D cellular patterns

Let us now proceed to several extensions of (5.27) and first consider the case of an instability that still favors rolls but in an effective two-dimensional medium which is rotationally invariant in its plane. An argument due to Newell and Whitehead⁸ shows that, x being the direction of the local wavevector and y the perpendicular direction, modulations along y are slow but faster than along x ($\partial_y \sim \mathcal{O}(r^{1/4}) \gg \partial_x \sim \mathcal{O}(r^{1/2})$ for $r \ll 1$) and that rotational invariance is accounted for by the replacement of ∂_x by $\partial_x - (i/2k_c)\partial_{yy}$ in (5.27). This substitution leads to what is known as the Newell-Whitehead-Segel (NWS) equation that reads:

$$\tau_0 \partial_t A = rA + \xi_0^2 \left[\partial_x + \frac{1}{2ik_c} \partial_{yy} \right]^2 A - g|A|^2 A, \qquad (5.29)$$

in the original variables x, y, and t, and where each term is again easily seen to be of order $r^{3/2}$.

The origin of the substitution can be understood by considering the operator $(\nabla_{\perp}^2 + k_c^2)^2$ from which periodic structures with space periodicity close to $\lambda_c = 2\pi/k_c$ emerge in media with rotational invariance in the (x, y) plane. Under Fourier transform, for a wavevector $\mathbf{k} = (k_c + \delta k_x)\hat{\mathbf{x}} + \delta k_y\hat{\mathbf{y}}$, at lowest significant order in $(\delta k_x, \delta k_y)$ one obtains:

$$\begin{split} \left[-(k_{\rm c}+\delta k_x)^2 - \delta k_y^2 + k_{\rm c}^2 \right]^2 &\simeq \left[-2k_{\rm c}\delta k_x - \delta k_y^2 \right]^2 \\ &= \left[(2ik_{\rm c})(i\delta k_x) + (i\delta k_y)^2 \right]^2 \end{split}$$

⁸A.C. Newell and J.A. Whitehead, "Finite bandwidth, finite amplitude convection," J. Fluid Mech. **38** (1969) 279-303.



Fig. 5.3 Left: Pattern with rolls arriving perpendicular to a lateral wall. Right: Corresponding solution of (5.31).

from which the indicated replacement derives by just performing the inverse Fourier transform $(i\delta k_x \mapsto \partial_x, i\delta k_y \mapsto \partial_y)$.

Brown and Stewartson⁹ have shown that the boundary condition at a wall inhibiting the instability and perpendicular to the y direction reads:

$$A(x, y_{\mathbf{b}}, t) = 0 \qquad \text{and} \qquad \partial_y A(x, y_{\mathbf{b}}, t) = 0 \tag{5.30}$$

The y-dependence of the envelope in the vicinity of the lateral wall, depicted in Figure 5.3, has been obtained by numerical integration of the fourth order differential equation

$$(\xi_0^2/4k_c^2)\frac{d^4}{dy^4}A = rA - gA^3$$
(5.31)

with boundary conditions (5.30) at y = 0 and $A \to A_0 = \sqrt{r/g}$ for $y \to \infty$. The value of the second derivative at y = 0 is obtained by multiplying (5.31) with $\frac{d}{dy}A$ which can be integrated. The so-obtained first integral is then evaluated at $y \to \infty$ where all derivatives of A are zero, which fixes $\frac{d^2}{dy^2}A$ at y = 0. Writing (5.31) as a four-dimensional first-order system, one finally obtains the solution by integrating it as an initial value problem, by means of a shooting method in which the value of the third derivative of A at y = 0 is the sole unknown initial condition to be adjusted so that $A(y) \to A_0$ as $y \to \infty$.

⁹S.N. Brown and K.S. Stewartson, "On finite amplitude Bénard convection in a cylindrical container," Proc. R. Soc. Lond. A **360** (1978) 455-469.

5.4.3 Quasi-two-dimensional cellular patterns

It is not difficult to extend the formalism to treat patterns with several pairs of wavevectors at a phenomenological level. We shall consider here only the case of a square pattern simply obtained by noticing that the x direction for one of the wavevectors is the transverse direction to the other and reciprocally. Combining results already obtained we get

$$\tau_0 \partial_t A_1 = rA_1 + \xi_0^2 \left[\partial_x + \frac{1}{2ik_c} \partial_{yy} \right]^2 A_1 - g(|A_1|^2 + \mu |A_2|^2) A_1, \quad (5.32)$$

$$\tau_0 \partial_t A_2 = rA_2 + \xi_0^2 \left[\partial_y + \frac{1}{2ik_c} \partial_{xx} \right]^2 A_2 - g(|A_2|^2 + \mu |A_1|^2) A_2. \quad (5.33)$$

When $\mu > 1$, the calculation developed in Exercise 5.6.1 shows that rolls are preferred locally.¹⁰ Exercise 5.6.3 then shows that a system of rolls with a wavevector too far from k_c also becomes unstable against the formation of rectangles owing to the growth of a system of rolls with wavevector k_c but at right angles with it. This is the *cross-roll* instability, and the way it is shown to exist implies its *universal character*: all roll patterns in rotationally invariant media may experience it. On the other hand, owing to their local stability properties, roll systems at right angles may coexist in different but contiguous regions of space if their wavevectors are sufficiently close to k_c , forming a stationary grain boundary, see Figure 5.4 (left). A grain boundary parallel to the y axis is obtained by solving (5.32, 5.33) in



Fig. 5.4 Left: Roll system with a grain boundary. Right: Solution for the corresponding amplitudes.

¹⁰In essence, when $\mu < 1$, everything occurs as if $\mu = 0$, the two systems of rolls ignore each other, and grow, so that squares are obtained everywhere in space; when $\mu > 1$, they cannot but feel each other and one kills the other, at least locally.

the special case $\partial_y \equiv 0$ which leaves one with a differential system in x. For example if both underlying wavevectors are equal to k_c , one gets:

$$\begin{split} 0 &= rA_1 + \xi_0^2 A_1'' - g(|A_1|^2 + \mu |A_2|^2)A_1, \\ 0 &= rA_2 - \frac{\xi_0^2}{4k_c^4} A_2''' - g(|A_2|^2 + \mu |A_1|^2)A_2, \end{split}$$

where the primes indicate differentiation with respect to x. The solution illustrated in Figure 5.4 (right) has here been rapidly and accurately obtained by searching it as the asymptotic solution of a partial differential problem in x and t using a simple numerical scheme of the kind described in Appendix B, §B.2.1. The mathematical solution can however be obtained by analytical means (matched asymptotic expansions).¹¹

The envelope formalism just introduced can account for the essentials of the dynamics of *scalar* textures such as those observed in convection at large Prandtl numbers. In particular, it explains the orientation of rolls at lateral boundaries, the presence of curvature and well ordered *grains* and other *defects* such as those present in Figure 3.10, p. 98. All this follows from the fact that the NWS equation (5.29) derives from a potential in the sense of Chapter 2, §2.1.3, p. 33, *i.e.* can be written as

$$au_0 \partial_t A = -rac{\delta \mathcal{G}}{\delta A^*}$$
 (5.34)

where the right hand side is a notation representing the functional differentiation of

$$\mathcal{G}(A,A^*) = -r|A|^2 + \xi_0^2 \left| \left[\partial_x + \frac{1}{2ik_c} \partial_{yy} \right] A \right|^2 + \frac{1}{2}g|A|^4 , \qquad (5.35)$$

with respect to A^* , with $g \in \mathbb{R}$, hence $\mathcal{G} \in \mathbb{R}$.

Functional differentiation in (5.34) is understood in the sense of variation calculus. In the change $A \mapsto A + \delta A$:

• variations of $|A|^2$ and $|A|^4$ expand as $A \,\delta A^* + A^* \delta A$ and $2|A|^2 (A \,\delta A^* + A^* \delta A)$, which immediately gives the corresponding terms in (5.29) upon isolation of terms in δA^* ;

• terms involving derivatives need a slightly more complicated treatment; for example, the variation of $\partial_x A \partial_x A^*$ gives $\partial_x A \partial_x \delta A^* + \partial_x \delta A \partial_x A^*$ but $\partial_x \delta A^*$ is not independent of δA^* so that an integration by part has to be performed to isolate the latter, which yields $-\delta A^* \partial_{xx} A$ plus boundary

 $^{^{11}{\}rm P.}$ M. and Y. Pomeau, "A grain boundary in cellular structures near the onset of convection," Phil. Mag. A 48 (1983) 607-621.

terms; the other terms can be treated in the same way (two integrations by part for the terms arising from $|\partial_{yy}A|^2$) so that all the partial derivatives of (5.29) with respect to space are also recovered;

• boundary conditions (5.28, 5.30) for rolls parallel or perpendicular to the lateral boundaries may be used to cancel boundary terms arising from integrations by parts in case of a rectangular domain; otherwise contributions from the boundary term can simply be neglected when compared to bulk contributions in the general case.

The system then evolves so as to minimize the potential

$$G(t) = \int \mathcal{G}ig(A(x,y,t),A^*(x,y,t)ig)\mathrm{d}x\,\mathrm{d}y$$

over the domain considered since $\frac{d}{dt}G \leq 0$. Using the same argument as in §2.1.3, one obtains that the solution that achieves this local minimum is time independent. On this basis, selected patterns are those corresponding to stationary solutions of (5.29) at local minima of \mathcal{G} (local maxima or saddles are unstable solutions). Favored textures can be found by comparing the contributions of the different causes of inhomogeneity (lateral walls, defects) to the potential increase with respect to the uniform solution.

In this context, the slow residual time dependence often observed in experiments can be attributed to higher order terms omitted in (5.29) which only holds at lowest significant order, at least in the scalar case. However, the interpretation of space-time chaos (weak turbulence) is at any rate more complex as soon as one leaves this simple framework, in particular for convection at low Prandtl numbers for which the most relevant field is no longer the temperature but the velocity.

5.4.4 Oscillatory patterns and dissipative waves

Up to now, we have considered only stationary instabilities. The approach in terms of envelope equations extends also in the more difficult case of oscillatory instabilities, $\omega_c \neq 0$, in particular for waves when $k_c \neq 0$. Difficulties arise from the fast time dependence, drastically destroying the relaxational property by introducing complex coefficients in the evolution equations, as in the strictly temporal setting of standard Hopf bifurcation, Chapter 4, §4.2.1.2, p. 128, and not mildly by higher order corrections as above.

Here, we just give a few results practically without derivation, inviting the reader to consult references mentioned in Note 5. As in the stationary case, the primary role is taken by the linear dispersion relation introduced in Chapter 3, $\S3.1.6$. Considerations about resonances developed in Chapter 4, $\S4.1.3$, are then used to deal with nonlinearities directly written as *normal* forms relevant to the case at hand.

For an oscillatory instability with $k_c = 0$ or for a dissipative wave with $k_c \neq 0$ but in a reference frame moving at the group velocity (see p. 85), we have:

$$au_0 \partial_t A = (r-i ildelpha)A + \xi_0^2(1+ilpha)\partial_{xx}A - g(1+ieta)|A|^2A$$
 .

This equation accounts for the space unfolding of the Hopf bifurcation (4.16) as derived from the dispersion relation of the corresponding instability (3.26, 3.27), p. 84. Coefficient $\tilde{\alpha}$ is the critical pulsation in units of τ_0 , *i.e.* $\omega_c = \tilde{\alpha}/\tau_0$; it can be eliminated by changing to a 'rotating frame' $A(x,t) \mapsto A(x,t) \exp(-i\omega_0 t)$.

The next coefficient, α , is a measure of linear dispersion in units of the natural coherence length ξ_0^2 . Finally, β is the nonlinear dispersion coefficient rescaled by g, which accounts for saturation effects by nonlinearities.

Upon further rescaling of time, length, and amplitude, one obtains the universal form of the so-called *complex Ginzburg-Landau equation* (CGL):¹²

$$\partial_t A = A + (1+i\alpha)\partial_{xx}A - (1+i\beta)|A|^2A.$$
(5.36)

This formulation is not restricted to the one-dimensional case introduced here but can be extended to higher dimensions, ∂_{xx} simply being replaced by the Laplacian when the physical system is isotropic in space.

The structure of the equations governing one-dimensional waves propagating in opposite directions (complex envelopes $A_{1,2}$) can be obtained in the same way by symmetry and resonance considerations which lead to:

$$\begin{split} \partial_t A_1 + v_g \partial_x A_1 &= A_1 + (1 + i\alpha) \partial_{xx} A_1 \\ &\quad - [(1 + i\beta)|A_1|^2 + (\mu' + i\mu'')|A_2|^2]A_1 \,, \\ \partial_t A_2 - v_g \partial_x A_2 &= A_2 + (1 + i\alpha) \partial_{xx} A_2 \\ &\quad - [(1 + i\beta)|A_2|^2 + (\mu' + i\mu'')|A_1|^2]A_2 \,, \end{split}$$

through appropriate rescalings. Notice however that, since the first order partial derivative ∂_x is expected to be of order $r^{1/2}$ before rescaling, the unscaled group velocity must also be of order $r^{1/2}$ (*i.e.* small enough), to insure the consistency of the set of equations at order $r^{3/2}$.

¹²W. van Saarloos, "The complex Ginzburg-Landau equation for beginners," in [Cladis and Palffy-Muhoray (1995)].

Solutions with $A_2 = 0$ and $A_1 \neq 0$ (or the reverse) account for propagating waves moving to the right (or left), *i.e.* $\equiv A_1(x - vt)$ (or $\equiv A_2(x + vt)$). By contrast condition $|A_1| = |A_2|$ describes a superposition of right and left waves forming a standing wave. A calculation analogous to that in Exercise 5.6.1 shows that the system prefers propagating waves when $|\mu'| > 1$ (one wave "feels" the presence of the other and "kills" it) and standing waves when $|\mu'| < 1$ (each wave does not "feel" the presence of the other much and therefore "accept" cohabitation).

When propagating waves are preferred, defects similar to grain boundaries may separate homogeneous domains with opposite kinds of wave in each. These defects are called *sources* or *sinks* according to whether the waves travel away from the defect or toward it.

5.4.5 Universal long-wavelength instabilities

One of the interests of the envelope formalism is to offer a framework for the study of universal secondary instabilities of dissipative structures linked to the symmetry properties of patterns at the limit of a laterally unbounded system. For stationary cellular structures, two such secondary modes are the *Eckhaus instability* against local compression/expansion and the *zigzag instability* against local torsion of the rolls, as sketched in Figure 5.5.

These two universal modes are called *phase instabilities* because they relate to modulations of the position of the cells. Within the envelope approach, the solution is searched for in the form $A = |A| \exp(i\phi)$, and while the modulus |A| describes the intensity of the response to the primary instability mechanism, the phase serves to specify the absolute position of the pattern in the laboratory frame. The universal instability linked to the



Fig. 5.5 Initial aspect of perturbation associated to the Eckhaus instability (left) and the zigzag instability (right).

amplitude is the cross roll instability mentioned earlier, p. 195, and further studied in Exercise 5.6.3. Contrasting with the amplitude that has a finite relaxation time, the phase is dynamically neutral as long as the solution is uniform.

A slightly irregular pattern can be described by a phase modulation which decays when the pattern is stable or gets amplified when it is unstable. Typically, at least for scalar dissipative structures, the phase of the envelope is governed by a diffusion equation:

$$\partial_t \phi = D_{\parallel} \partial_{xx} \phi + D_{\perp} \partial_{yy} \phi \,, \tag{5.37}$$

where D_{\parallel} and D_{\perp} are diffusion coefficients along the wavevector of the structure, or perpendicular to it. As long as these coefficients are both positive, the phase perturbation relaxes and the roll pattern is stable against (infinitesimal) phase perturbations. The instability is observed when one of these coefficients changes its sign. The Eckhaus instability corresponds to D_{\parallel} becoming negative, and the zigzag when it is D_{\perp} .

The phase diffusion coefficients depend on the wavelength of the underlying pattern. Exercise 5.6.2 is a first approach to the Eckhaus instability resting on (5.27), the zigzag instability would be studied in the same way using (5.29). The result is generally presented as a stability diagram in the parameter plane $(\delta k, r)$ where δk measures how the wavevector of the underlying pattern departs from the critical wavevector and r is the control parameter. Figure 5.6 displays the results at lowest significant order.

The Eckhaus instability is a side-band instability that develops "far from" k_c and close to the marginal stability curve, in a region where the instability mechanism is not very efficient and the amplitude of the solution, $A_0 = \sqrt{(r - \xi_0^2 \delta k^2)/g}$, is small. "Preferring" a solution with a more optimal wavevector, the system amplifies the phase modulation by compressing certain regions and expanding others (Fig. 5.5, left). Supposing for example that the initial wavelength is too short, expanded regions will have a more favorable local wavevector while in compressed regions the local wavelength will be even shorter. The primary mode completely disappears at the location of the most compressed places since the amplitude of the primary mode drops to zero at the marginal stability curve. The pattern then loses a pair of rolls at such places and the wavelength further relaxes toward an increased more optimal value. If the obtained wavevector is still in the unstable domain, the process repeats. If not, it stops and the system evolves through diffusive relaxation toward a uniform pattern with a wavelength inside the stable domain.



Fig. 5.6 Stability diagram of roll patterns against universal phase modes of compression/expansion (Eckhaus) and torsion (zigzag). As obtained from equation (5.29) valid at lowest order, the marginal stability curve is given by $r = \xi_0^2 \delta k^2$. Rolls are unstable against the Eckhaus mode in the region $\xi_0^2 \delta k^2 < r < 3\xi_0^3 \delta k^2$. The zigzag instability domain is for $\delta k < 0$. Stable rolls have wavevectors in the domain $0 < \delta k$, $r > 3\xi_0^3 \delta k^2$.

The zigzag instability sets in when $\delta k < 0$, *i.e.* for wavelengths larger than critical. This can also be easily interpreted by noticing (Fig. 5.5, right) that, when measured perpendicularly to the local axis of the rolls, the wavelength is shorter than that measured along the x axis by a factor equal to the cosine of the angle between the local periodicity direction and the x axis. The instability again develops so as to make the (too long) initial wavelength shorter and thus closer to the critical value. By contrast with the case of the Eckhaus instability which changes the number of rolls, the zigzag instability saturates as it amplifies. The process ends in wide regularly spaced domains of 'zig' and 'zag' straight rolls connected by barrow bands of strong bending.

Up to now, we have only considered the initial development of these universal phase instabilities and just sketched the ultimate fate of the unstable state. Analytically, one can go a little further and complete the phase diffusion equation (5.37) by appropriate nonlinearities. In this context, let us mention the *Kuramoto-Sivashinsky equation*¹³ (KS) to be used in the numerical experiments described at the end of Appendix B. This equation describes the dynamics of the phase of nearly uniform solutions to the complex Ginzburg-Landau equation (5.36) in a narrow neighborhood of the threshold of the *Benjamin-Feir* instability.

¹³Y. Kuramoto, "Phase dynamics of weakly unstable periodic structures," Prog. Theor. Phys. **71** (1984) 1182-1196.
On general grounds this instability, which closely corresponds to the Eckhaus instability of one-dimensional stationary patterns, develops when the phase diffusion coefficient $D = 1 + \alpha\beta$ becomes negative (*Newell's criterion*, see Note 5. By symmetry considerations one can check that the equation governing the phase, at most quadratic in ϕ and up to order four in ∂_x , reads:

$$\partial_t \phi = D \partial_{xx} \phi - K \partial_{xxxx} \phi + g_0 (\partial_x \phi)^2 + g_1 (\partial_x \phi) (\partial_{xxx} \phi) + g_2 (\partial_{xx} \phi)^2 + g_3 (\partial_x \phi)^2 (\partial_{xx} \phi) , \qquad (5.38)$$

where each coefficient can be derived from those in (5.36). In addition to $D = 1 + \alpha\beta$, one gets¹⁴ $K = \frac{1}{2}\alpha^2(1 + \beta^2) > 0$, $g_0 = \beta - \alpha$, $g_1 = 2g_2 = \alpha g_3 = -2\alpha(1 + \beta^2)$.

In practice, for $D \approx 0$ but negative, the order of magnitude of the space gradient is fixed by the competition between the two linear terms and all nonlinear terms except the first can be neglected. After appropriate rescalings, Equation (5.38) can be reduced to the KS equation which reads in universal form:¹⁵

$$\partial_t \phi + \partial_{xx} \phi + \partial_{xxxx} \phi + \frac{1}{2} (\partial_x \phi)^2 = 0$$
(5.39)

or, after differentiating it with respect to x and setting $\psi = \partial_x \phi$:

$$\partial_t \psi + \psi \partial_x \psi + \partial_{xx} \psi + \partial_{xxxx} \psi = 0.$$
 (5.40)

The nonlinearity $\psi \partial_x \psi$ present in this last expression is reminiscent of the advection term of hydrodynamics. It already appeared in the *Burgers* equation (Chapter 1, Exercise 1.5.3) producing shocks (Exercise 5.6.4, part 1) and the *Korteweg-de Vries* equation producing solitons (Chapter 2, Exercise 2.5.6). The KS equation in one or the other form, (5.39) or (5.40) is a particularly simple model of *phase turbulence*.¹⁶

As long as D remains small, the modulus of the envelope is enslaved to the phase gradient, stays close to its nominal value A_0 , and therefore remains bounded away from zero. This is no longer the case when the phase instability is more intense. The field |A| then revolts and recovers a

¹⁴J. Lega, Défauts topologiques associés à la brisure d'invariance de translation dans le temps, PhD Dissertation, Nice University, 1989 (in French).

 $^{^{15}}$ The equation also appears in problems of front propagation as shown by G.I. Sivashinsky, "On self-turbulization of a laminar flame," Acta Astronautica **6** (1979) 569-591; hence the joint names for the equation.

¹⁶For a brief review with references, see: H. Chaté and P. M., "Phase turbulence," in [Tabeling and Cardoso (1994)].

dynamics of its own. It then explores a larger range of values that extends down to zero. At places where |A| = 0, the phase is no longer defined and phase defects nucleate, with 2π -jumps of ϕ . The CGL equation then enters new regimes whose precise nature depends on the value of α and β in (5.36) and the space dimension. Strong space-time chaos called *defect turbulence* then sets in. In two dimensions, it is characterized by the permanent birth of defects in pairs, that further dissociate and move around before merging.¹⁷

The theory of the transition to chaos is rooted in the idea of dimensional reduction leading to effective dynamical systems in terms of ordinary differential equations. To summarize this section, one can say that the envelope formalism is the required adaptation of this idea when confinement effects are too weak to legitimate the approach in terms of isolated modes and corresponding discrete amplitudes. Pattern selection strictly relies on this reduction for uniform solutions.

Defects and universal secondary instabilities involve modulations to some ideal reference situation. As long as the system remains sufficiently close to a stable regularly ordered pattern, the relevant instability modes relates to the phase of envelope and a supplementary reduction is possible by adiabatic elimination of the envelope modulus. Localized defects also play a role, either because they are present in the initial conditions (when the pattern emerges) or as the result of secondary instabilities that do not saturate. Sometimes tools borrowed from the theory of dynamical systems can still be used, e.g. to determine special solutions to the envelope equations.

It should also be noticed that, while the envelope formalism can be made rigorous within the framework of multiple scale methods, an interesting alternative is the derivation of generic models by phenomenological arguments resting on resonance and symmetry considerations. Further, the numerical simulation of such models has proved crucial to the understanding of space-time chaos. Such simulations usually do not require considerable investment and the reader is encouraged to practice them using the simple numerical methods presented in Appendix B, in order to get a personal intuition of the problem while taking advantage of the published material listed in, e.g. [Rabinovich *et al.* (2000)] or to be found on the Internet.

¹⁷P. Coullet, L. Gil, J. Lega, "Defect-mediated turbulence," Phys. Rev. Lett. **62** (1989) 1619–1622. A complete phase diagram is given in the one-dimensional case by H. Chaté, "Disordered regimes of the one-dimensional complex Ginzburg-Landau equation," in [Cladis and Palffy-Muhoray (1995)].

5.5 What Lies Beyond?

Before closing this chapter, let us point out the interest of fully analogical approaches to the modeling of extended systems, useful when the minimum space-time coherence necessary to apply the envelope formalism is absent from the system. As a matter of fact, when the space-time coherence is limited, there is some advantage to consider the continuous system as an aggregate of subsystems coupled to each other. The local dynamics is accounted for at the scale of the subsystem while the space extension arises through the coupling between the subsystems usually arranged at the node of a lattice.

A system can thus be discretized at several levels. At first, one can just discretize the physical space and get lattices of differential systems, e.g. lattices of Hopf oscillators or Lorenz systems. These systems are then coupled by some rule. For example, nearest-neighbor diffusive coupling of identical Hopf oscillators in one dimension would yield

$$\frac{\mathrm{d}}{\mathrm{d}t}Z_n = (r - i\omega)Z_n - g|Z_n|^2 Z_n + D(Z_{n+1} - 2Z_n + Z_{n-1}),$$

where subscript n indicates the space position of an individual oscillator and the discrete version of the diffusion operator, here $D\partial_{xx}$ is easily recognized.

Next step, time can also be discretized, which gives *coupled map lattices*. Time and space being indicated by superscript k and subscript n, respectively, in one dimension and again for diffusive coupling, one will start with

$$\mathbf{X}_{n}^{k+1} = (1-2D)\mathcal{F}\left(\mathbf{X}_{n}^{k}\right) + D\left[\mathcal{F}\left(\mathbf{X}_{n+1}^{k}\right) + \mathcal{F}\left(\mathbf{X}_{n-1}^{k}\right)\right],$$

which is preferable to the seemingly more straightforward formulation:

$$\mathbf{X}_{n}^{k+1} = \mathcal{F}\left(\mathbf{X}_{n}^{k}\right) + D\left[\mathbf{X}_{n+1}^{k} - 2\mathbf{X}_{n}^{k} + \mathbf{X}_{n-1}^{k}\right],$$

since it is immediately checked that the latter may not preserve the invariant domain of each subsystem while the former does (if $\mathbf{X} \in \mathcal{D}$ implies $\mathcal{F}(\mathbf{X}) \in \mathcal{D}$, then $\mathbf{X}_n^k \in \mathcal{D}$ implies $\mathbf{X}_n^{k+1} \in \mathcal{D}$ for all n). The appropriateness of the model relies entirely on the skill of the modeler while choosing the local map and the type of coupling.¹⁸

Up to now, the local phase space was still a continuous set. One can make a last step by considering *cellular automata* where each subsystem

¹⁸See for example [Kaneko (1993)] and for a specific application H. Chaté and P. M., "Spatiotemporal intermittency" in [Tabeling and Cardoso (1994)].

can be in one of a finite set of states. Boolean automata have states labelled by bits 0 and 1, with evolution rules functions of the configuration of the neighborhood of each cell. Such systems can already have complicated dynamics in spite of very simple definitions [Wolfram (1986)].

Lattice gas automata are automata specially adapted to hydrodynamics. They describe fluids at the level of individual molecules but governed by simplified evolution rules. "Living" at the nodes of a regular lattice, the molecules can be in one of a discrete set of motion state (speed and orientation), jump from node to node and change their state of motion when they meet at some given node according to rules given in collision tables. At first, when massively parallel computers began to appear, they were presented as an alternative to the direct numerical simulation of hydrodynamics equations. Nowadays, they are rather considered as useful models when the local dynamics is rich or complicated (chemical reactions), or when the boundaries have a complex topology or special physical properties (surface catalysis). An interesting review of lattice gas automata as applied to complex hydrodynamics is [Rothman and Zaleski (1997)].

The present chapter was mostly devoted to stationary patterns in weakly confined systems, with only a few words about waves. The latter will be central to the study of open flow instabilities in the next chapter.

5.6 Exercises

5.6.1 Pattern selection

Consider patterns described by superpositions of plane waves:

$$V = \sum_j rac{1}{2} [A_j \exp(i \mathbf{k_j} \cdot \mathbf{x_h}) + \mathrm{c.c.}] \, ,$$

namely, rolls with a single pair of wavevectors and squares with two pairs at right angles. The wavelength of the participating modes is $\lambda_c = 2\pi/k_c$. 1) Show that translational invariance along wavevector $\mathbf{k}_j = k_c \hat{\mathbf{x}}_j$ (unit vector $\hat{\mathbf{x}}_j$), $x_j \mapsto x_j + x_j^{(0)}$, j = 1, 2, implies a (gauge) invariance for the corresponding complex amplitude $A_j \mapsto A_j \exp(i\phi_j^{(0)})$.

Taking these symmetries into account, show that the supercritical bifurcation towards a uniform pattern is governed at lowest order by

$$\frac{\mathrm{d}}{\mathrm{d}t}A_1 = rA_1 - g_{11}|A_1|^2A_1 - g_{12}|A_2|^2A_1 \tag{5.41}$$

$$\frac{\mathrm{d}}{\mathrm{d}t}A_2 = rA_2 - g_{22}|A_2|^2A_2 - g_{21}|A_1|^2A_2 \tag{5.42}$$

with $g_{ij} \in \mathbb{R}$. Use the rotational invariance to reduce the number of interaction coefficients to g (self-interaction) and $g' = \mu g$ (cross-interaction). Show that g can be eliminated by appropriate rescaling of A_1 and A_2 .

2) Determine the potential $G(A_1, A_1^*, A_2, A_2^*)$ from which (5.41, 5.42) derives in the sense of (5.34), here simply

$$\frac{\mathrm{d}}{\mathrm{d}t}A_j = -\partial G/\partial A_j^* \,.$$

and conclude that permanent regimes are time-independent.

3) Find all fixed points (A_{1*}, A_{2*}) of system (5.41, 5.42) and, apart from the trivial solution, give their explicit expression and physical significance (take advantage of the gauge invariance to specify the phase of the envelopes so as to have real amplitudes).

4) Compute the value of the potential and study its curvature at the different fixed points to guess their stability properties. Then perform the explicit stability analysis by inserting $A_j = A_{j*} + a_j$ in the system and linearizing. Even though the A_{j*} are real quantities, the most general perturbations are not, thus take $a_j = u_j + iv_j$. Show that the fourth order linear system obtained splits into two subsystems, producing two neutral modes (explain their origin) and two non-trivial modes that can be stable or unstable, according to the value of μ and the considered fixed point. Use the result to interpret phase portraits displayed in Figure 5.7. What happens when $\mu = 1$?



Fig. 5.7 Phase portrait of system (5.41, 5.42) for $\mu = 0.80$ (left) and $\mu = 1.25$ (right).

5.6.2 Eckhaus instability

Consider a roll pattern described by an envelope evolving according to:

$$\partial_t A = rA + \partial_{xx} A - |A|^2 A.$$
(5.43)

The Eckhaus instability relates to the phase ϕ of the complex field A.

1) Determine amplitude $\tilde{A}^{(0)}$ of a *phase-winding* solution corresponding to uniform rolls with wavevector $k = k_c + \delta k$, $A(x, t) = \tilde{A}^{(0)} \exp(i \, \delta k \, x)$.

The stability of this solution against long wavelength perturbations (phase modes) is studied by assuming that the prefactor of $\exp(i \, \delta k \, x)$ can be a function of space and time. From (5.43), derive the equation governing $\tilde{A}(x,t)$ defined through $A(x,t) = \tilde{A}(x,t) \exp(i \, \delta k \, x)$.

2) Insert $\tilde{A}(x,t) = \tilde{A}^{(0)} + a(x,t)$ in this new equation, and show that the linearized equations governing the real and imaginary parts of a(x,t) = u(x,t) + iv(x,t) read:

$$egin{aligned} \partial_t u &= -2(r-\delta k^2)u + \partial_{xx}u - 2\delta k\partial_x v\,,\ \partial_t v &= 2\delta k\partial_x u + \partial_{xx}v\,. \end{aligned}$$

Derive the dispersion relation for Fourier normal modes with growth rate s and wavevector q taken in the form $u = \bar{u} \exp(st) \cos(qx)$ and $v = \bar{v} \exp(st) \sin(qx)$.

3) Show that the roots are real and that the phase-winding solution is unstable when

$$2(r-3\delta k^2)+q^2\leq 0\,,$$

derive from this that the instability occurs first for $q \to 0$ (long wavelength) and recover the result displayed in Figure 5.6.

5.6.3 Cross roll instability

In the conditions of Exercise 5.6.1, assume $\mu > 1$ (rolls preferred to squares), consider a phase winding solution with $k_x = k_c + \delta k$, and study its stability against rolls at right angles with $k_y = k_c$.

From (5.32, 5.33), show that:

is an appropriate starting point.

Show that the condition for instability has the same form as the Eckhaus condition but with a different prefactor to be determined. Compute the value of μ that makes the cross-roll instability more dangerous than the Eckhaus instability and interpret the limit $\mu \rightarrow 1$.

5.6.4 Dynamical systems and nonlinear waves

5.6.4.1 The Burgers equation

Consider the Burgers equation:

$$\partial_t v + v \partial_x v = \nu \partial_{xx} v \,.$$

1) Show that it is invariant through a Galilean change of frame (*i.e.* that the equation for $\tilde{v}(\tilde{x},t)$ with $\tilde{x} = x - Vt$ and $\tilde{v} = v - V$ is the same as that governing v(x,t)). (The corresponding symmetry of the NS equation is discussed in Note 9, p. 239.)

2) Derive the differential equation governing a solution that propagates at speed V without deformation ($\tilde{x} = x - Vt$ is the sole independent variable) and integrate it once with respect to x. Consider a solution such that v_{\pm} for $\tilde{x} \to \pm \infty$, derive its speed V from the first integral just obtained, and derive the analytical shape of the corresponding solution. [Solution: a hyperbolic tangent.]

5.6.4.2 Kuramoto-Sivashinsky equation

Consider variant (5.40) of the KS equation:

$$\partial_t v + v \partial_x v + \partial_{xx} v + \partial_{xxxx} v = 0.$$

1) Derive the linear dispersion relation governing infinitesimal perturbations $\delta v = \bar{\delta v} \exp(st + iqx)$ to the trivial solution $v \equiv 0$. Show the instability of those belonging to a range of q to be determined.

2) Write down the ordinary differential equation in $\tilde{x} = x - Vt$ governing a solitary wave solution propagating without deformation in a frame moving at speed V and derive a first integral. By identification, find a solution in the form

$$ilde{v} = lpha anh(\kappa ilde{x}) + eta anh^3(\kappa ilde{x}) \,.$$

and compare the value of κ with the wavevector q_{\max} of the perturbation with maximum growth rate s_{\max} .

5.6.4.3 Flow down an inclined plane

The Benney equation is a partial differential equation governing the long wavelength perturbations to a fluid film of uniform thickness flowing down an inclined plane. After appropriate rescalings of length, time, and film thickness, one obtains¹⁹

$$\partial_t h + rac{2}{3}\partial_x \left[h^3 + \left(lpha h^6 - h^3
ight)\partial_x h + h^3\partial_{xxx}h
ight] = 0$$

where α is the control parameter.

1) Derive the linearized equation governing infinitesimal perturbations to a time-independent uniform solution $h = h_0$. Show that solution $h \equiv h_0 = 1$ is linearly unstable for $\alpha > \alpha_c = 1$.

2) A hydraulic jump is a solution such that $h \to 1$ for $x \to -\infty$ and $h \to h_{\infty}$ for $x \to +\infty$. A solitary wave precisely corresponds to $h_{\infty} = 1$. Write down the ordinary differential equation in $\tilde{x} = x - Vt$ governing a solution that propagates without deformation in a frame moving at speed V, integrate this equation to find a first integral and determine the value K of this first integral as a function of c when $x \to -\infty$. Then derive the relation between V and $h_{\infty} < 1$.

¹⁹See: A. Pumir, P. M., Y. Pomeau, "On solitary waves running down an inclined plane," J. Fluid Mech. **135** (1983) 27-50, and cited references.

This page intentionally left blank

Chapter 6

Open Flows: Instability and Transition

Convection studied in Chapter 3 is the prototype of systems becoming turbulent by steps when progressively driven far from equilibrium upon increasing the temperature gradient. All along this progression, the physical problem is clearly ruled by a mechanical cause, buoyancy-induced advection, counteracted by thermodynamic dissipation processes, viscous friction and thermal diffusion. Turning to open flows, one could thus try to stay within the same framework and start with the no-flow equilibrium situation, then consider weakly out-of-equilibrium *laminar* regimes mostly controlled by viscous dissipation, and further increase the shear¹ to observe the transition to *turbulence* after some cascade of instabilities. This approach can indeed be followed in some cases (*e.g.* wakes) but, in most situations, open flows turn out to be representative of strongly out-of-equilibrium systems in which mechanical evolution largely preempts the relaxation trends of thermodynamic origin.

This has been recognized early since the historical tradition gives the first place to the study of inviscid fluids, for which velocity gradients are not rubbed out by viscous friction. A simple illustration of this change of perspective is given by the damped linear oscillator $m\ddot{X} + \eta\dot{X} + kX = 0$: the high-viscosity limit $\eta^2 \gg 4km$, yielding over-damped behavior (stable node at $X = \dot{X} = 0$) typical of thermodynamic relaxation, is not a good starting point to understand weakly damped oscillator $\eta = 0$ (elliptic point) is more relevant.

Here, we restrict our attention to one-directional and time-independent base flows (§6.1). Inertial effects due to rotation or time dependence (e.g.

¹The Reynolds number introduced in Chapter 1, p. 9, compares mechanical advection to supposedly stabilizing viscous effects.

alternating with $\partial_t \equiv i\omega$) will thus practically not be considered. Linear stability theory (§6.2) was first developed for inviscid fluids in the second half of the Nineteenth Century, pointing out the possibility of purely mechanical instabilities (§6.2.2), namely the *Kelvin-Helmholtz instability* of inflexional flow profiles, and a contrario the existence of mechanically stable flow profiles of more delicate study.

The inviscid case then served as a natural starting point for the study of the effects of viscous dissipation considered as a perturbation $(R < \infty)$ but large). While viscosity tends to stabilize purely mechanical instabilities as expected, it can destabilize mechanically stable flows according to mechanisms that are not much intuitive, producing the so-called Tollmien-Schlichting waves. During the second third of the Twentieth Century, these mechanisms were elucidated by analytical means resting on asymptotic analyses in the limit $R \to \infty$, which gave the behavior of the asymptotes of marginal stability curves and, to a limited extent, estimates of instability thresholds. Precise numerical results were obtained later using computers. We will not consider this theoretical and numerical issue in detail but just present some results (§6.2.3) and redirect the interested reader toward specialized literature, the chapter by Huerre and Rossi in [Godrèche and Manneville (1998)] to begin with, and [Drazin (2002); Schmid and Henningson (2001); Drazin and Reid (1981); Betchov and Criminale (1967)], among others, for more detailed presentations.

Another source of difficulty is linked to the presence of the upstreamdownstream flow of matter which, by the way, also transports the perturbations. The latter can thus be carried away (so-called *convective* instability) or instead develop *in situ* in spite of global transport (*absolute* instability). The technical approach to this problem is highly delicate and relies on a mastery of complex variable analysis that goes much beyond the prerequisites for this course. Accordingly we will only introduce the subject at a qualitative level (§6.2.4).

After the description of primary instabilities we turn to the situation beyond threshold and consider the main physical scenarios of transition to turbulence, the understanding of which is still incomplete owing to specific difficulties. We begin with an abstract presentation of the characteristics of secondary instability modes (§6.3.1). Then we focus on the most typical cases representing the types of flow pointed out at the linear stages: the *wake* of a blunt body for mechanically unstable flows, and the *boundary layer* along a flat plate for mechanically stable flows possibly unstable due to viscous effects. The case of the wake, which bifurcates toward the classical Bénard-von Kármán vortex street (see Figure 1.6, p. 16) through a supercritical Hopf bifurcation, is examined first (§6.3.2) since it can be approached with the same tools as those described earlier. By contrast, wall flows such as the boundary layer considered next (§6.3.3) are more difficult to understand, leaving us with still partially open problems. One known source of difficulties is that, at the Reynolds numbers relevant to the transition, the base flows may remain stable against unlocalized infinitesimal disturbances, but become unstable against localized finite-amplitude perturbations. *Turbulent spots* arising from the breakdown of such perturbations can therefore coexist (in physical space) with laminar flow, see Figure 1.7, p. 17. The global stability of wall flows, the role of transient structures and the spatio-temporal competition between laminar and turbulent domains are thus examined in a separate section (§6.3.4) with special emphasis on the case of plane Couette flow known to be linearly stable for all Reynolds numbers.

6.1 Base Flow Profiles

6.1.1 Strictly one-dimensional flows

The Navier-Stokes (NS) equation reads:

$$\rho(\partial_t \mathbf{v} + \mathbf{v} \cdot \nabla \mathbf{v}) = -\nabla p + \mu \nabla^2 \mathbf{v}, \qquad (6.1)$$

where ρ is the density, p the pressure and μ the dynamical viscosity. It is completed by the *continuity* equation for an incompressible flow:

$$\boldsymbol{\nabla} \cdot \mathbf{v} = 0. \tag{6.2}$$

We consider only the case of constant density fluids so that we can divide (6.1) by ρ and introduce the kinematic viscosity $\nu = \mu/\rho$, which is the diffusivity of velocity fluctuations (homogeneous to $[L]^2[T]^{-1}$). The quantity p/ρ does not play a dynamical role and can be eliminated by appropriate analytical manipulations, *i.e.* taking the curl of (6.1) while taking (6.2) into account.

Let us examine the case of a time-independent one-directional flow whose direction defines the x axis. Here we denote u, v, and w the velocity components along x, y, and z, respectively. Introducing $\mathbf{v} \equiv u(x, y, z) \hat{\mathbf{x}}$ in the continuity equation, we simply get:

$$\partial_x u = 0, \qquad (6.3)$$

i.e. u is independent of x but still a function of the transverse coordinates (y, z). The advection term $\mathbf{v} \cdot \nabla \mathbf{v}$ then cancels identically, while the equations for v and w simply read $\partial_z p = \partial_y p \equiv 0$, *i.e.* the pressure is a function of x only. Differentiating the x component of the NS equation with respect to x yields:

$$-\partial_{xx}p + \mu(\partial_{xx} + \partial_{yy} + \partial_{zz})\partial_x u = 0$$

and, using (6.3),

 $\partial_{xx}p=0$,

so that, for a straight tube with length L, we obtain:

$$p(x) = p_0 - Gx$$
 with $G = \Delta p/L$,

where p_0 is the pressure at x = 0 and $\Delta p = p_{up} - p_{down}$ is the pressure head. The velocity component u(y, z) is then a solution to the Poisson equation

$$\left(\partial_{y^2} + \partial_{z^2}\right)u = -G/\mu\,,\tag{6.4}$$

with the usual no-slip boundary condition at the wall. Of course u(y, z) depends on the shape of the tube's section. The case of a tube with circular cross-section is standard (Hagen-Poiseuille flow). The base flow is the axisymmetric solution of (6.4) in cylindrical coordinates:

$$\mu \left(\partial_{\varrho \varrho} + \varrho^{-1} \partial_{\varrho}\right) u + G = 0,$$

where ρ denotes the radial coordinate. Integrating this equation, requiring the regularity of the solution on the tube axis $\rho = 0$, and using the no-slip condition at $\rho = \bar{\rho}$, the interior radius of the tube, we get

$$u = U_{\max} \left(1 - \left(\varrho/\bar{\varrho} \right)^2 \right)$$
 with $U_{\max} = \frac{G\bar{\varrho}^2}{4\mu}$. (6.5)

The classical expression of the flow rate is obtained by integrating over ϱ :

$$Q = \int_0^{ar arrho} u(arrho) 2 \pi arrho \, \mathrm{d} arrho = rac{\pi G ar arrho^4}{8 \mu} \, .$$

For simplicity we further restrict ourselves to the consideration of plane base flows, *i.e.* flows that develop in channels such that the velocity is uniform in along one of the transverse coordinates, called *spanwise* and traditionally labelled z. The velocity profile u of course remains a function of the *cross-stream* coordinate y, *i.e.* u = U(y). This configuration is depicted in Figure 6.1 (left). The profile of the flow driven by a pressure



Fig. 6.1 One-directional flow between parallel planes. Left: Geometry and notation conventions for the axes. Right: Plane Couette-Poiseuille profile.

gradient G between two parallel plates moving along the x axis at speeds U_1 and U_2 is then given by the classical plane Couette-Poiseuille expression:

$$U(y) = -\frac{1}{2}Gy^2 + by + c\,,$$

where constants b and c are determined from the *no-slip* conditions:

$$U(y_i) = U_i, \qquad i = 1, 2.$$

This profile is displayed in Figure 6.1 (right).

6.1.2 More general velocity profiles

The class of base profiles to be studied can be enlarged by relaxing the condition of strict parallelism. Let us, for example, consider a flow that is slowly evolving downstream. The continuity equation (6.2), *i.e.* $\partial_x u = -\partial_y v$, implies a slight divergence of the stream lines: $v \neq 0$ but small as long as ∂_x remains small in a region where $\partial_y = \mathcal{O}(1)$. The boundary layer close to a wall, the wake downstream a body in a stream, the plane jet illustrated in Figure 6.2 are examples of nearly-parallel flows.



Fig. 6.2 Example of flows developing downstream. From left to right: The boundary layer, the jet, the wake.

Taking into account the orders of magnitude of the fields u, v, and their x and y derivatives, the time-independent ($\partial_t \equiv 0$) two-dimensional ($\partial_z \equiv 0$) NS equation and continuity equations at lowest order read:

$$u\partial_x u + v\partial_y u = -\rho^{-1} \mathrm{d}p_0/\mathrm{d}x + \nu\partial_{xx} u\,, \tag{6.6}$$

$$\partial_x u + \partial_y v = 0, (6.7)$$

where dp_0/dx is a possible applied pressure gradient (so-called *boundary* layer approximation). The solution is then searched by standard similarity variable methods (see [Tritton (1990)], Chap. 11) in the form:

$$u = u_{\max}(X)\bar{u}(y/\delta(X)) \tag{6.8}$$

where $u_{\max}(X)$ and $\delta(X)$ are the relevant scales for the streamwise speed and the transverse distances at the point with x-coordinate X.

Let us consider first the case of the boundary layer that develops along a wall in the absence of pressure gradient, *i.e.* the region where the stream with speed U_{∞} far from the wall accommodates the no-slip condition. The thickness δ of the layer varies with the distance X from the leading edge. Considering the equation for u, it is immediately observed that the advection term, which is no longer strictly zero, must be compensated by the dominant dissipative term, hence:

$$u\partial_x u \sim \nu \partial_{yy} u \,. \tag{6.9}$$

The distance X from the leading edge is the only streamwise length scale available, while δ is obviously the relevant cross-stream scale at that position. Dimensionally, this yields:

$$\partial_x u \propto U_\infty/X\,, \qquad \partial_{yy} u \propto U_\infty/\delta^2\,,$$

and, upon insertion of these relations in (6.9):

$$\delta(x) = \sqrt{\nu x / U_{\infty}} \,. \tag{6.10}$$

The local velocity profile $u = U_{\infty}f'(\zeta)$, where $\zeta = y/\delta(X)$ is the similarity variable and the prime denote differentiation with respect to ζ , is obtained as the solution of a third-order nonlinear differential equation in ζ for the stream function f:

$$ff'' + 2f''' = 0, (6.11)$$

with boundary conditions f = f' = 0 at $\zeta = 0$, $f' \to 1$ for $\zeta \to \infty$. The corresponding *Blasius velocity profile* is displayed in Fig. 6.3 (left). It starts



Fig. 6.3 Left: Blasius profile. Right: Sketch illustrating the definition of the displacement thickness δ_1 (shaded domains have the same surface).

with very weak curvature for $\zeta < 1$, and quickly tends to its asymptote for $\zeta \to \infty$.

The thickness Δ of the boundary layer is usually defined as the distance at which the speed equals a given fraction of U_{∞} , for example $\Delta_{0.99}$ defined by $u(\Delta_{0.99}) = 0.99 U_{\infty}$ is about $5\delta(X)$, where $\delta(X)$ is defined by (6.10). Since 0.99 is somewhat arbitrary, other quantities with more physical significance have been defined in order to characterize measured boundary layer velocity profile. The quantities:

$$\delta_1=rac{1}{U_\infty}\int_0^\infty (U_\infty-u(y))\mathrm{d}y \quad ext{and} \quad \delta_2=rac{1}{U_\infty^2}\int_0^\infty (U_\infty-u(y))u(y)\mathrm{d}y\,,$$

respectively called the displacement thickness and the momentum thickness are classical examples. They represent the thicknesses of the layers over which one could condense the loss of flow rate and the loss of momentum due to the slowing down of the flow at the wall. Figure 6.3 (right) illustrates the definition of the displacement thickness. For the Blasius profile, one finds $\delta_1 = 1.721 \,\delta(X)$ and $\delta_2 = 0.664 \,\delta(X)$.

In the case of a plane jet (Fig. 6.2, middle), the maximum streamwise speed decreases as X increases due to spreading and flow-rate conservation, so that $u_{\max}(X)$ introduced in (6.8) now depends on X. The similarity argument leads to:

$$u_{
m max} \propto M^{2/3}/(
u X)^{1/3} ~~{
m and}~~ \delta \propto (
u X)^{2/3}/M^{1/3} ~,$$

where $M = \int_{-\infty}^{+\infty} u^2 dy$. Equation (6.11) is replaced by another slightly different equation that can be explicitly integrated yielding a stream-function

 $f \propto \tanh(\zeta)$. The base flow speed u = f' accordingly reads:

$$u_{\rm jet}(X) = \left(\frac{3M^2}{32\nu X}\right)^{1/3} \left\{ 1 - \tanh^2 \left[y \left(\frac{M}{48\nu^2 X^2}\right)^{1/3} \right] \right\} .$$
 (6.12)

The far wake of a blunt obstacle (Fig. 6.2, right) can be modeled using a similar profile by considering the slowing-down behind it as a jet in the reversed direction (velocity *defect*), that is to say a profile in the form:

$$u_{ ext{wake}}(X) = U_{ ext{min}} + (U_{\infty} - U_{ ext{min}}) anh^2(y/\delta),$$

where U_{\min} is the minimal speed in the plane of the wake passing through the obstacle.

A similar approach could have been developed for flows that are slowly time-dependent (e.g. growing of a boundary layer along an impulsively started plate).

6.1.3 Extension to arbitrary profiles

Velocity profiles considered up to now were all controlled by viscous effects. However, on physical grounds, viscosity is expected to be dominant at low Reynolds numbers R only. Given a typical shear $\Delta U/\Delta \ell$ measured by a Reynolds number $R = \Delta U \Delta \ell / \nu$, the viscosity can impose its law only as long as the viscous time $\tau_v = (\Delta \ell)^2 / \nu$ stays much shorter than the advection time $\Delta \ell / \Delta U$, which precisely makes the inviscid limit $\nu \to 0$ relevant, as already pointed out. As a first approximation, any velocity profile U(y)can be thus accepted provided the Reynolds number is sufficiently large.

The stability theory that we are now ready to develop focuses on strictly parallel and time-independent flows. As applied to the extended class of flows considered, its results are thus expected to give hints on what happens provided that the critical wavelengths and angular frequencies of the unstable modes be sufficiently large when compared to their imposed evolution scales in space or time.

6.2 Linear Stability

6.2.1 General framework

6.2.1.1 Setting

Let us consider a base flow $\mathbf{V}_0 \equiv \{(U(y), 0, 0); p_0(x)\}$ to which we add a perturbation $\mathbf{V} \equiv \{(u, v, w); p\}$, function of (x, y, z, t). The NS equations governing the perturbation are easily obtained by inserting $\mathbf{V}_0 + \mathbf{V}$ into (6.1, 6.2). These equations are written here directly in dimensionless form using the scales characterizing the unperturbed profile U(y), typically $\Delta U = U_{\max} - U_{\min}$ where $U_{\max} = \max_y U(y)$ and $U_{\min} = \min_y U(y)$ and $\Delta \ell = y_2 - y_1$ for a channel flow, or some appropriately defined reference thickness δ in other cases. The corresponding time and pressure scales are $\Delta t = \Delta \ell / \Delta U$ and $\Delta p = \rho (\Delta U)^2$. The Reynolds number is defined as $R = \Delta U \Delta \ell / \nu$ accordingly. The NS equations for the perturbation then read:

$$\partial_t u + \left[(U+u)\partial_x + v\partial_y + w\partial_z \right] (U+u) = -\partial_x p + R^{-1} \nabla^2 u \,, \quad (6.13)$$

$$\partial_t v + \left[(U+u)\partial_x + v\partial_y + w\partial_z \right] v = -\partial_y p + R^{-1} \nabla^2 v \,, \quad (6.14)$$

$$\partial_t w + \left[(U+u)\partial_x + v\partial_y + w\partial_z \right] w = -\partial_z p + R^{-1} \nabla^2 w \,, \quad (6.15)$$

$$\partial_x u + \partial_v v + \partial_z w = 0. ag{6.16}$$

Neglecting all terms quadratic in u, v, and w, we get

$$\partial_t u + U \partial_x u + v \partial_y U = -\partial_x p + R^{-1} \nabla^2 u , \qquad (6.17)$$

$$\partial_t v + U \partial_x v = -\partial_v p + R^{-1} \nabla^2 v,$$
 (6.18)

$$\partial_t w + U \partial_x w = -\partial_z p + R^{-1} \nabla^2 w, \qquad (6.19)$$

$$\partial_x u + \partial_y v + \partial_z w = 0, \qquad (6.20)$$

that can be formally written as

$$\mathcal{L}(\partial_t, \nabla; r) \mathbf{V} = 0.$$
(6.21)

At this stage, it can be useful to eliminate the pressure field and reduce (6.17-6.20) to a system of two equations for two unknowns. This reduction is the subject of Exercise 6.4.1 in which the perturbation is expressed in terms of the cross-stream velocity component v and the component of the vorticity normal to the boundary, $\Omega_y = (\nabla \times \mathbf{v})_y = \partial_x w - \partial_z u$.

6.2.1.2 Reduction to two dimensions

The solution of the linearized problem (6.21) is, as usual, sought by superposition of elementary modes in the form

$$\mathbf{V} = \tilde{\mathbf{V}}(y) \exp[i(\mathbf{k}_{\rm h} \cdot \mathbf{x}_{\rm h} - \omega t)], \qquad (6.22)$$

where subscript "h" means "horizontal," *i.e.* in the plane of the flow. A satisfactory representation of the physical perturbations would be obtained by taking the real part of these complex modes.

The relevance of the Fourier modes $\exp(i\mathbf{k}_{\mathbf{h}}\cdot\mathbf{x}_{\mathbf{h}})$ is related to the fact that the problem is autonomous with respect to variables x and z. When studying convection, we took normal modes in the form $\exp(i\mathbf{k}_{\mathbf{h}}\cdot\mathbf{x}_{\mathbf{h}}+st)$, with real wavevectors $\mathbf{k}_{\mathbf{h}}$ and a complex growth rate s, the real part of which accounted for the amplification/decay properties and the imaginary part for a possible oscillatory behavior. Here the introduction of the time dependence as $\exp(-i\omega t)$ is a matter of convenience and the two formulations are equivalent when taking $\omega \in \mathbb{C}$. We see in particular that the real part of $\omega = \omega_{\mathbf{r}} + i\omega_{\mathbf{i}}$ now accounts for the oscillation while the imaginary part $\omega_{\mathbf{i}}$ gives the growth rate ($s = \sigma - i\omega = -i(\omega_{\mathbf{r}} + i\omega_{\mathbf{i}}) = \omega_{\mathbf{i}} - i\omega_{\mathbf{r}}$). The appropriateness of choice (6.22) comes from the fact that, by contrast with plain Rayleigh-Bénard convection producing stationary roll patterns, here we will mainly have to deal with waves.

As before, the existence of a nontrivial solution to the stability problem (6.21) is subjected to a compatibility condition expressed as a relation between $\mathbf{k}_{\rm h}$ and ω function of the control parameter R, the dispersion relation that can be formally written as

$$\mathcal{L}(-i\omega, i\mathbf{k}_{\mathbf{h}}; R) = 0, \qquad (6.23)$$

where \mathcal{L} is a scalar functional relation while \mathcal{L} in (6.21) a linear operator. Typically, the relation $\mathcal{L} = 0$ arises as the existence condition of nontrivial solutions to an algebraic homogeneous linear system, see later §6.2.2.3, p. 230 and Exercise 6.4.4.

In fact, relation (6.23) can be read in different ways without any assumption on the real or complex character of the wavevector and the angular frequency. The approach taken when assuming $\mathbf{k}_{\rm h}$ real and ω complex is usually called *temporal* and was particularly relevant to the case of convection. The *spatial* approach, that assumes ω real and k complex, is more appropriate to open flows and allows one to account for the streamwise growth of perturbations in a flow forced by a spatially localized perturba-

tion that is periodic with some externally controllable angular frequency $\omega_{\rm f}$, as exemplified in the film "Flow Instabilities" [Mollo-Christensen (1972)].

We will come back to this issue in Section 6.2.4 but, for the moment, let us focus on the temporal approach and, introducing explicitly the fact that we have to deal with waves propagating at a (complex) phase velocity c, take normal modes (6.22) in the form

$$\mathbf{V}(x,y,z,t) = \{ ilde{\mathbf{v}}(y), ilde{p}(y)\} imes \exp\left[iig(k_x(x-ct)+k_zzig)
ight],$$

that we insert in (6.17–6.20). According to this definition, the modes evolve in time as $\exp(st)$ with

$$s=-i\omega=-ik_xc=-ik_x(c_{\mathrm{r}}+ic_{\mathrm{i}})=k_xc_{\mathrm{i}}-ik_xc_{\mathrm{r}}\,,$$

so that the flow is unstable when $k_x c_i > 0$, hence $c_i > 0$ with $k_x > 0$ by convention.

At this stage, it is useful to take advantage of the Squire theorem, subject of Exercise 6.4.2. It stipulates that the most dangerous perturbations with wavevectors $\mathbf{k}_{\rm h} = (k_x, k_z)$ and $k_x \neq 0$ are those with $k_z = 0$, and that $w \equiv 0$, accordingly. More precisely it shows that these *two-dimensional* modes are the most amplified ones in the inviscid case and have the lowest threshold when viscosity is taken into account.

For such reduced two-dimensional perturbations $(\tilde{u}, \tilde{v}, \tilde{p})$ with $k_x \equiv k$, one readily obtains the following simpler system

$$ik(U-c)\tilde{u} + \tilde{v}U' = -ik\tilde{p} + R^{-1}\left(\frac{\mathrm{d}^2}{\mathrm{d}y^2} - k^2\right)\tilde{u}$$
, (6.24)

$$ik(U-c)\tilde{v} = -\tilde{p}' + R^{-1}\left(\frac{\mathrm{d}^2}{\mathrm{d}y^2} - k^2\right)\tilde{v},$$
 (6.25)

$$ik\bar{u} + \tilde{v}' = 0, \qquad (6.26)$$

where the prime denotes differentiation with respect to the cross-stream variable y, e.g. $U' \equiv dU/dy$.

6.2.1.3 Orr-Sommerfeld and Rayleigh equations

Two-dimensional incompressible flow are usually solved by introducing a stream function ψ such that:

$$u=\partial_y\psi \qquad ext{and}\qquad v=-\partial_x\psi\,,$$

so that the continuity condition is automatically fulfilled.

Dropping the tildes, we introduce the normal modes as:

$$\psi(x, y, t) \mapsto \psi(y) \exp(ik(x - ct))$$
 (6.27)

in (6.24, 6.25), which leads to

$$ik(U-c)\psi' + (-ik\psi)U' = -ikp + R^{-1}\left(\frac{d^2}{dy^2} - k^2\right)\psi', \qquad (6.28)$$

$$ik(U-c)(-ik\psi) = -p' + R^{-1}\left(\frac{\mathrm{d}^2}{\mathrm{d}y^2} - k^2\right)(-ik\psi).$$
 (6.29)

The pressure is eliminated by differentiating (6.28) with respect to y, multiplying (6.29) by ik and subtracting the second equation from the first, which leads to the *Orr-Sommerfeld* equation:

$$(U-c)\left(\frac{d^2}{dy^2} - k^2\right)\psi - U''\psi = \frac{1}{ikR}\left(\frac{d^2}{dy^2} - k^2\right)^2\psi.$$
 (6.30)

This is a fourth-order differential equation in y, to which four boundary conditions must be applied, two at each end of the cross-stream domain, $y = y_{\min}$ and $y = y_{\max}$. The boundary conditions on ψ derive from those on the velocity perturbation at a solid boundary, *i.e.* u = v = 0, which gives:

$$0 = k\psi(y_{\min}) = k\psi(y_{\max}) = \psi'(y_{\min}) = \psi'(y_{\max}).$$
 (6.31)

In infinite domains, these conditions are replaced by non-divergence requirements as $y_{\min} \to -\infty$ and/or $y_{\max} \to +\infty$.

It is immediately remarked that the inviscid limit ' $\nu \rightarrow 0$ ' is singular since when $R \rightarrow \infty$, the small parameter R^{-1} is a factor of the terms containing the fourth-order derivative. As a matter of fact, in this limit the Orr-Sommerfeld equation simply reads:

$$(U-c)\left(\frac{d^2}{dy^2} - k^2\right)\psi - U''\psi = 0, \qquad (6.32)$$

and is called the *Rayleigh equation*. For this second order differential equation, the no-slip condition u = 0 is no longer applicable, while condition v = 0 remains:

$$0 = k\psi(y_{\min}) = k\psi(y_{\max}).$$
 (6.33)

On general grounds, the solutions of the Orr-Sommerfeld equation are built on four elementary functions and those of the Rayleigh on only two. When R is sufficiently large, out of these four functions, two will derive from those found for the corresponding Rayleigh equation. The other two, of viscous origin, will remain localized in regions of strong velocity gradient.

6.2.2 Inviscid flows

Let us start with the Rayleigh equation (6.32) and assume that we have found a complex solution $(\psi; c)$. Since the equation has real coefficients, it follows that $(\psi^*; c^*)$ is also a solution. So, to every stable solution $(c_i < 0)$ an unstable solution corresponds by complex conjugation, and reciprocally. This property comes from the time-reversal symmetry characteristic of the absence of dissipation. A sufficient stability condition is therefore the absence of solution with $c_i \neq 0$, so that we face three possibilities: either (6.32) has no solution other than the trivial one, or it admits only neutral modes $(c_i = 0)$, or it has also unstable modes $(c_i \neq 0)$. The problem is therefore to find the conditions under which the Rayleigh equation has unstable modes.

6.2.2.1 The Rayleigh theorem and related properties

A necessary condition for the existence of unstable solutions to Equation (6.32) is that the base flow profile displays inflection points (U'') passes through zero and change its sign).² The derivation of this result due to Rayleigh (1880) is a good example of a *global* method in the sense of Chapter 1, Section 1.3, and Chapter 2, Exercise 2.5.2.

Let us assume the existence of an unstable mode $(c_i \neq 0)$, then Equation (6.32) can be divided by (U - c) that never cancels. This leads to:

$$\left(rac{\mathrm{d}^2}{\mathrm{d}y^2}-k^2
ight)\psi=rac{U^{\prime\prime}}{U-c}\psi\,.$$

Multiplying by ψ^* and integrating the result by parts over $[y_{\min}, y_{\max}]$ using boundary conditions $\psi(y_{\min}) = \psi(y_{\max}) = 0$, one gets:

$$-\int_{y_{\min}}^{y_{\max}} \left(|\psi'|^2 + k^2 |\psi|^2 \right) \mathrm{d}y = \int_{y_{\min}}^{y_{\max}} \frac{U''}{U-c} |\psi|^2 \mathrm{d}y \,. \tag{6.34}$$

Expanding the right hand side yields:

$$\int_{y_{\min}}^{y_{\max}} rac{U''}{U-c} |\psi|^2 \mathrm{d}y = \int_{y_{\min}}^{y_{\max}} rac{U''(U-c_{\mathrm{r}}+ic_{\mathrm{i}})}{|U-c|^2} |\psi|^2 \mathrm{d}y \,.$$

Next, separating real and imaginary parts, and focusing on the latter, one

²Here we assume implicitly that U'' is not identically zero, which would be the case of plane Couette flow or, more generally, of linear-by-part velocity profiles, for which (6.32) has to be solved directly, see later p. 229.

finds:

$$c_\mathrm{i}\int_{y_\mathrm{min}}^{y_\mathrm{max}} rac{U''}{|U-c|^2} |\psi|^2 \mathrm{d}y = 0\,.$$

It remains to be noticed that the assumption $c_i \neq 0$ implies $\int_{y_{\min}}^{y_{\max}} (\dots) dy = 0$, which is possible only if the integrand goes through zero and changes its sign, and implies the same property for U'', hence the theorem. This is a necessary but *not* sufficient instability condition.

Stated another way, the Rayleigh theorem stipulates that unstable flow profiles (U'' = 0 somewhere in the cross-stream domain) display a vorticity extremum $\Omega_z(y) = -U'$. A more refined condition is obtained by working with the real part of the integrated Rayleigh equation (6.34). Following Fjørtoft (1950), one can show that a monotonic velocity profile with an infection point may have unstable modes only if $|\Omega_z(y)|$ displays a maximum at the inflection point. This criterion, is derived in Exercise 6.4.3 and illustrated in Figure 6.4.

Other results can be shown about neutral and unstable modes, if they exist. For example the *Howard semi-circle theorem* stipulates that the (complex) phase speeds of solutions to the Rayleigh equation, neutral or unstable, lie in the upper half-disk of the *c*-complex plane, centered on the real axis at $U_{av} = \frac{1}{2}(U_{min}+U_{max})$ with diameter $\Delta U = (U_{max}-U_{min})$, where



Fig. 6.4 Velocity profiles U(y) (left) and vorticity profiles |U'(y)| (right) corresponding to stable (top) and unstable (bottom) flows according to Fjørtoft criterion.



Fig. 6.5 Domain of unstable modes for an inviscid flow according to the Howard semicircle theorem.

 $U_{\min} = \min_y U(y)$ and $U_{\max} = \max_y U(y)$, Figure 6.5. For a derivation by a global method similar to that for the Rayleigh theorem, consult [Schmid and Henningson (2001)], p. 23ff.

Given a neutral mode with phase velocity c_r it may be remarked that, since $U_{\min} \leq c_r \leq U_{\max}$ there is a point y_c in the flow where $U(y_c) = c_r$. At such a point, called the *critical level*, the coefficient of the second order derivative in the Rayleigh equation is zero, which makes it singular. This singularity expresses a particular resonance between the base flow and the perturbation, since the latter travels at exactly the same speed as the fluid particles at that level.

We will not enter into a detailed discussion of this phenomenon, leaving it to more theoretically oriented works such as [Drazin and Reid (1981)] or [Schmid and Henningson (2001)], but just notice that it raises delicate analytical difficulties when y_c is different from the level y_s of the inflection point determined by solving $U''(y_s) = 0$. Instead, we now turn to a typical instability associated with the presence of an inflection point.

6.2.2.2 Kelvin-Helmholtz instability

When two flows merge at the end of a splitting plate, they form a mixing layer (Fig. 6.6). Let U_1 (U_2) be the speed of the fluid when y < 0 (y > 0) and large and forget about the boundary layers that form on each side of the plate. At some distance of the trailing edge, viscosity smoothes out the velocity profile U(y) that can be locally characterized by its vortical content. The vorticity thickness of the mixing layer is then defined by:

$$\delta_{\mathrm{vort}} = rac{U_2 - U_1}{U'_{\mathrm{max}}}$$

where $|U'_{max}|$ is the absolute value of the maximum of the spanwise vorticity



Fig. 6.6 Mixing layer and definition of the vorticity thickness.

component $\Omega_z = -U'(y)$. Close to the end of the plate, δ_{vort} thickens as the square root of the downstream distance X, owing to viscous diffusion.

At a given point X, the mixing layer can thus be seen as a quasiparallel flow characterized by speeds U_1 and U_2 , the thickness δ_{vort} , and the kinematic viscosity ν . Two dimensionless control parameters can be built with these quantities: the Reynolds number:

$$R = \frac{|U_2 - U_1| \, \delta_{\rm vort}}{\nu}$$

and the velocity ratio:

$$\varrho=\frac{U_2-U_1}{U_2+U_1}.$$

Obviously, this flow is a good candidate for a mechanical instability in the Rayleigh-Fjørtoft context. Locally, the profile is close to a hyperbolic tangent that can serve to model it analytically:

$$U(y) = U_{\mathrm{av}} + \frac{1}{2}\Delta U \tanh(2y/\delta_{\mathrm{vort}})$$
.

The spanwise vorticity component is then

$$|\Omega_z| = rac{\Delta U}{\delta_{
m vort} \, \cosh^2(2y/\delta_{
m vort})}.$$

In the limit $\nu \to 0$, which implies $\delta_{\text{vort}} \to 0$, we get $|\Omega_z| = \Delta U \, \delta_{\mathrm{D}}(y)$ where $\delta_{\mathrm{D}}(y)$ is the Dirac distribution. Stated another way, a velocity discontinuity (seen as an extreme case of inflexional profile) can be understood as a vortex sheet.

The Kelvin-Helmholtz (KH) instability that we now study is typical of inviscid vortex sheets. Let us consider a base flow profile displaying a velocity discontinuity at y = 0:

region (1)
$$y < 0$$
, $U(y) = U_{(-)} = U_{av} - \frac{1}{2}\Delta U$,
region (2) $y > 0$, $U(y) = U_{(+)} = U_{av} + \frac{1}{2}\Delta U$, (6.35)

and first discuss the physical instability mechanism following Batchelor's approach [Batchelor (1967), p. 511], sketched in Figure 6.7. The unperturbed vorticity sheet is represented by a regular distribution of equal intensity vortices in (a). Next, assume a periodic infinitesimal modulation of the intensity in (b) where points equivalent in a translation by one wavelength are labelled with the same letters and higher local vorticity is represented by arrowed circles with larger diameters.

Intuitively, the differential effect of vortices at A and C is to move the interface at B and D in the directions indicated in (c). At lowest order the effect of the base flow is to advect vorticity along the flow, to the left when the interface has been moved downwards and to the right in the opposite case. Typically, at B it is advected from B toward A and at D from D



Fig. 6.7 Mechanism of the KH instability interpreted in terms of vorticity. The frame is moving at the average speed U_{av} (coordinate $x' = x - U_{av}t$) so that the flow profile displays a jump from $-\frac{1}{2}\Delta U$ for y < 0 to $+\frac{1}{2}\Delta U$ for y > 0.

(a) Unperturbed vorticity sheet. (b) Modulation of the vorticity intensity. (c) Sinusoidal deformation of the interface and amplification of the modulation due to the advection of the vorticity by the base flow.



Fig. 6.8 Interpretation of the KH instability mechanism in terms of pressure.

toward A, thus increasing the intensity of the vortex at A and decreasing that of the vortex at C. In all cases the initial modulation is increased, hence the instability.

A second interpretation can be found in terms of pressure, as illustrated in Figure 6.8; consult, e.g. [Tritton (1990), p. 268]: Before the introduction of a perturbation, the trajectories of the fluid particles on both sides of the discontinuity are straight. When the discontinuity surface is undulated, the speed of the fluid is slightly augmented (reduced) in regions with negative curvature above (below) the surface while in regions with opposite curvature the reverse situation holds. A naive use of the Bernoulli theorem expressing the conservation of energy along streamlines for an inviscid fluid in quasi-steady state, $p + \frac{1}{2}\rho v^2 = \text{Cst}$, implies that when |v| increases, then p decreases. Consequently a pressure difference tends to build between the two sides of the discontinuity surface, that pushes it further away: the instability develops.

All along this discussion, the velocity ratio $\rho = (U_2 - U_1)/(U_2 + U_1)$ does not show up since the mechanism is analyzed in a frame moving at the average speed $U_{av} = \frac{1}{2}(U_2 + U_1)$. The physical conditions for the formation of the mixing layer are thus in some sense evacuated. In practice, the trailing edge of the splitting plate has a fixed position in the laboratory frame and, accordingly, a *spatial stability analysis* in the sense of §6.2.4 below should be developed. Intuitively, when downstream transport is dominant, *i.e.* $\rho \ll 1$, perturbations due to residual turbulence in the base flow get amplified by the mechanism but are blown away so that the instability is expected to be *convective*, the flow playing the role of a *noise amplifier*. In the opposite case $\rho \gg 1$, *i.e.* $U_2 \simeq -U_1$, the situation considered in the previous discussion of the mechanism prevails, the instability should develop on the spot, and thus be *absolute*. The flow is then a genuine *self-sustained oscillator*. The precise value of ρ at which the behavior change takes place depends on the base profile. For the simplest continuous linear-by-part approximation, (Exercise 6.4.4, second item) it happens at $\rho = 1$ and for the smooth hyperbolic tangent profile at $\rho \simeq 1.3$, consult Huerre and Rossi in [Godrèche and Manneville (1998), p. 169ff] for details and original references.

6.2.2.3 The Rayleigh equation with linear-by-part velocity profiles

The two previous "physical" approaches rely on a knowledge of basic fluid mechanics properties of inviscid flows relative to the transport of energy and vorticity that are not necessarily part of the reader's supposed background. It is accordingly valuable to choose a more analytical viewpoint on the KH instability, with just the assumption that it is legitimate to replace the actual base profile by a linear-by-part approximation, splitting the cross-stream interval $[y_{\min}, y_{\max}]$ into subintervals such that

$$U^{(j)}(y) = \alpha^{(j)} + \beta^{(j)}y$$
 for $y \in [y_j, y_{j+1}]$.

The second derivative of such profiles with respect to y is identically zero, except at discontinuity points where it is not defined. The Rayleigh equation then simply reads:

$$\left(\frac{\mathrm{d}^2}{\mathrm{d}y^2} - k^2\right)\psi = 0.$$
(6.36)

Over each subinterval, the solution of (6.36) is given by:

$$\psi^{(j)} = \sum_{(\pm)} A^{(j)}_{\pm} \exp(\pm kx) \,. \tag{6.37}$$

In order to find the full solution, it remains to express boundary conditions (6.33) at y_{\min} , y_{\max} and to match the different partial solutions (6.37) at the discontinuity points of U and/or U'. The matching conditions are obtained from the continuity of pressure in (6.24)

$$\left[\psi U' - (U-c)\frac{\mathrm{d}}{\mathrm{d}y}\psi\right]_{(j)} = \left[\psi U' - (U-c)\frac{\mathrm{d}}{\mathrm{d}y}\psi\right]_{(j+1)},\qquad(6.38)$$

and the continuity of the cross-stream velocity component v at the surface of discontinuity. This second condition derives from the definition of v as the cross-stream velocity at a point of the interface between domains (j) and (j + 1), with coordinate Y, hence $v = \frac{d}{dt}Y$ (Exercise 6.4.4, first item). It reads:

$$\left[\frac{\psi}{U-c}\right]_{(j)} = \left[\frac{\psi}{U-c}\right]_{(j+1)}.$$
(6.39)

Assuming that the interval $[y_{\min}, y_{\max}]$ has been cut into n pieces, 2n constants $(A_{\pm}^{(j)}, (j = 1, ..., n)$ have to be determined. One is left with two boundary conditions and 2(n-1) matching conditions, which indeed makes a system of 2n homogeneous linear equations. The dispersion relation is obtained from the condition that this system has nontrivial solutions.

Let us analyze the KH instability along these lines using the approximate, discontinuous base flow profile (6.35). Solutions to the Rayleigh equation (6.36) are taken in the form (6.37). Forbidding divergence when $y \to \pm \infty$ implies

$$A_{(-)}^{(1)} = 0 = A_{(+)}^{(2)}$$

while the matching conditions (6.38, 6.39) at y = 0 read

$$(+k) \left(U^{(1)} - c \right) A^{(1)}_{(+)} = (-k) \left(U^{(2)} - c \right) A^{(2)}_{(-)},$$
$$\frac{1}{U^{(1)} - c} A^{(1)}_{(+)} = \frac{1}{U^{(2)} - c} A^{(2)}_{(-)},$$

so that we are left with a homogeneous linear system of two equations for two unknowns. The corresponding compatibility condition:

$$(U_{(-)}-c)^2+(U_{(+)}-c)^2=0,$$

is solved to yield:

$$c = U_{\rm av} \pm i \frac{1}{2} \Delta U \,,$$

The instability of the flow directly derives from $c_i = \pm \frac{1}{2}\Delta U \neq 0$. Since there is no specific scale in this problem (unbounded medium, infinitely thin vorticity sheet) the instability condition is independent of the wavevector k and all modes are unstable. Their growth rates $\sigma = kc_i$, are larger and larger as k increases. Things are different for a confined layer $-\infty < y_{\min} < 0 < y_{\max} < +\infty$, as can be seen by solving Exercise 6.4.4 where several other linear-by-part profiles are also considered.

6.2.3 Viscous flows

6.2.3.1 Instability and viscous dissipation

For the KH instability, the destabilizing process is of mechanical origin and viscosity plays a "normal" stabilizing role. But the Rayleigh theorem shows that some flows do not have unstable modes at the inviscid limit. In order to learn what happens to them, we have to return to the Orr–Sommerfeld equation (6.30).

The problem is difficult from an analytical point of view owing to the singular character of the limit $R \to \infty$. As a matter of fact, whereas the absence of inflection point y_s such that $U''(y_s) = 0$ forbids the existence of unstable inviscid modes, neutral modes may exist, for which the Rayleigh problem is singular: at the critical level y_c , the phase speed c_r is equal to the speed $U(y_c)$ of the flow while $U''(y_c) \neq 0$. These neutral modes are the best candidates for becoming unstable when perturbed by viscous effects. Here we restrict ourselves to a qualitative presentation of the results, often obtained from a numerical solution of the Orr-Sommerfeld equation, and refer to *e.g.* [Drazin and Reid (1981)] for the relevant theory.

Let us first indicate why viscous dissipation could play a role in the destabilization of a mechanically stable flow. This hint is obtained by adopting a global point of view and considering the evolution of the kinetic energy contained in a specific two-dimensional mode with wavevector $k = 2\pi/\lambda$ and phase speed c_r . Let us compute $(6.17) \times u + (6.18) \times v$:

$$u\partial_t u + v\partial_t v + uvU' + U(v\partial_x v + u\partial_x u) = -u\partial_x p - v\partial_y p + R^{-1}[u(\partial_{xx} + \partial_{yy})u + v(\partial_{xx} + \partial_{yy})v], \quad (6.40)$$

and determine the average kinetic energy (per unit length in the spanwise direction) contained in a domain of length λ and height $(y_{\text{max}} - y_{\text{min}})$, in the frame moving at speed c_{r} labelled by coordinate $\tilde{x} = x - c_{\text{r}}t$. The quantity of interest is:

$$K_{
m pert} = \left\langle rac{1}{2} \left(u^2 + v^2
ight)
ight
angle \; ,$$

where $\langle \ldots \rangle$ denotes $\left[1/\lambda (y_{\max} - y_{\min}) \right] \int_{y_{\min}}^{y_{\max}} \mathrm{d}y \int_{\bar{x}_0}^{\bar{x}_0 + \lambda} (\ldots) \mathrm{d}\bar{x}.$

Averaging (6.40), we obtain:

$$\begin{aligned} \frac{\mathrm{d}}{\mathrm{d}t} K_{\mathrm{pert}} &+ \frac{1}{2} \left\langle U \partial_x \left(u^2 + v^2 \right) \right\rangle + \left\langle u v U' \right\rangle = \\ &- \left\langle (u \partial_x p + v \partial_y p) \right\rangle + R^{-1} \left\langle u (\partial_{xx} + \partial_{yy}) u + v (\partial_{xx} + \partial_{yy}) v \right\rangle. \end{aligned}$$

The second term on the l.h.s. cancels upon streamwise integration by virtue

of the periodicity in \bar{x} . On the right hand side we have $-u\partial_x p - v\partial_y p = -\partial_x(up) - \partial_y(vp) + p(\partial_x u + \partial_y v)$. The two first contributions cancel upon integration, either owing to streamwise periodicity or to cross-stream boundary conditions at y_{\min} and y_{\max} , and the last one from the continuity condition $\partial_x u + \partial_y v = 0$. Finally we get:

$$\frac{\mathrm{d}}{\mathrm{d}t}K_{\mathrm{pert}} = -\langle U'uv\rangle - R^{-1} \langle \left((\partial_x u)^2 + (\partial_y u)^2 + (\partial_x v)^2 + (\partial_y v)^2 \right) \rangle, \quad (6.41)$$

where the viscous terms (in R^{-1}) have been rearranged through integrations by parts exploiting the boundary conditions. This is the specific form taken by the *Reynolds-Orr equation* governing the evolution of the kinetic energy contained in the perturbation in the linear regime.³

Equation (6.41) symbolically reads:

$$\frac{\mathrm{d}}{\mathrm{d}t}K_{\mathrm{pert}} = P - D, \qquad (6.42)$$

where P is a production term and D, that contains all the terms in R^{-1} , accounts for viscous dissipation. Term D is manifestly always positive and works so as to decrease the energy contained in the perturbation. The flow can thus be unstable only if the production term P is positive and sufficiently large.

Let us consider perturbations in the form

$$egin{aligned} \{u,v\} &= rac{1}{2}igl(\{ ilde{u}(y), ilde{v}(y)\}\exp[i(k(x-ct)]+ ext{c.c.}igr) \ &= rac{1}{2}igl(\{ ilde{u}(y), ilde{v}(y)\}\exp(ikar{x})+ ext{c.c.}igr)\exp(kc_{ ext{i}}t)\,, \end{aligned}$$

with growth rate kc_i . It can then be checked that every term in (6.42) varies as $\exp(2kc_it)$ that can be factored out, leaving us with a discussion of the sign of

$$P \propto -\int U'(y) [ilde{u}(y) ilde{v}^*(y) + ext{c.c.}\,]\,\mathrm{d}y\,,$$

which is not determined in advance. On the contrary, this sign strongly depends on the shape of u and v, solutions to the full Orr-Sommerfeld equation (6.30). The asymptotic analysis in the limit ' $R \to \infty$ ' points out the role of the variation of the stream function $\tilde{\psi}$ close to the critical

³This equation is valid without modification in the nonlinear regime because the advection term conserves the energy. Terms of nonlinear origin and dropped upon linearization on the l.h.s. read $u^2 \partial_x u + uv \partial_y u + vu \partial_x v + v^2 \partial_y u$, further rewritten as $\frac{1}{2} \partial_x (u(u^2 + v^2)) + \frac{1}{2} \partial_y (v(u^2 + v^2))$ using the continuity condition. All these terms disappear in the averaging process. Even more, the result is not restricted to the two-dimensional case but also holds for three-dimensional perturbations.

level y_c , and shows the existence of specific unstable modes called *Tollmien-Schlichting (TS) waves*. Rather then starting a long and delicate discussion that would allow one to compute the behavior of the marginal stability curve in this limit (for an introduction, consult Huerre and Rossi in [Godrèche and Manneville (1998)]) we just present the main results obtained in part analytically and in part numerically.

6.2.3.2 Results

By contrast with convection-like instabilities (see Fig. 3.3, p. 81), marginal stability curves are conventionally presented in the plane (R, k) with R along the horizontal axis and k vertically. Whatever the origin of the instability, mechanical or viscous, they take the shape of loops with different aspects in the limit ' $R \to \infty$ ', as shown in Figure 6.9.

• When the flow is mechanically unstable (velocity profile with an inflection point and a vorticity maximum at the inflection point), the threshold is "low" and a whole band of wavevectors remains unstable as $R \to \infty$, see Figure 6.9 (left). Viscosity then plays its usual stabilizing role.

• When the base profile is mechanically stable (no inflection point), the threshold is "high" and the marginal stability curve has the shape of a hairpin that pinches as $R \to \infty$, see Figure 6.9 (right). Unstable modes are the TS waves resulting from the viscous mechanism alluded to above.



Fig. 6.9 Marginal stability curves typical of plane shear flows. In each case, the unstable domain corresponds to the interior of the loop. Left: Mechanically unstable flows; low threshold and finite-width band of unstable wavevectors when $R \to \infty$. Right: Mechanically stable flows (no inflection point); high thresholds and narrowing band of unstable wavevector when $R \to \infty$.

Plane Poiseuille flow in a channel is typical of flows without inflection point. Taking the maximum speed U_{max} on the centerline of the flow as velocity unit and the half-gap $h/2 = (y_{\text{max}} - y_{\text{min}})/2$ as length unit. the dimensionless base velocity profile is thus $U = 1 - y^2$ over the interval [-1, +1] and the Reynolds number is defined by $R = U_{\text{max}}h/2\nu$. The marginal stability curve looks like that in Figure 6.9 (right), with $R_c \simeq$ $5772 \gg 1$ and $k_c \simeq 1.02$ (critical wavelength $\lambda_c = (h/2)(2\pi/1.02) \simeq 3h).^4$ The phase speed of waves at threshold is $c_{r,c} \simeq 0.264$ (*i.e.* about 1/4 of U_{max}). To these numerical results one can add the asymptotic behavior of the marginal stability branches obtained analytically (see *e.g.* [Drazin and Reid (1981)] for a detailed presentation and results summarized in their Fig. 4.11, p. 190):

6.2.4 Instability and downstream transport

6.2.4.1 Theory vs. experiments

Let us come back to the dispersion relation (6.23). The Squire theorem tells us that the wavevector $\mathbf{k}_{\rm h}$ has a single relevant (streamwise) component noted k, so that we can write

$$\mathcal{L}(-i\omega, ik; R) = 0. \tag{6.43}$$

The presentation of results in the plane (R, k) corresponds to a *temporal* reading of this relation, further solved by assuming $k \in \mathbb{R}$ and ω complex, $\omega = \omega_{\rm r} + i\omega_{\rm i}$, so that $\omega_{\rm i}$ is the temporal growth rate (from the normal mode assumption: $\exp[i(kx - \omega t)] = \exp(\omega_{\rm i}t) \exp[i(kx - \omega_{\rm r}t)]$).

This approach was satisfactory in the case of convection that develops in an enclosure and emerges from the background noise, in the absence of any forcing (*absolute* instability). It is no longer appropriate for typical open flow situations like the one sketched in Figure 6.10, where downstream transport manifestly plays a dominant role (*convective* instability).⁵ In such a setup, a localized forcing, periodic with angular frequency ω_f is applied at

⁴S.A. Orszag, "Accurate solution of the Orr-Sommerfeld stability equation," J. Fluid Mech. **50** (1971) 689-703.

 $^{^5}$ The terms *absolute* and *convective* are the accepted ones. We avoided to use the word 'convective' all along Chapter 3 dealing with 'convection' which is 'absolute' in the absence of (sufficiently strong) through-flow.



Fig. 6.10 Experimental configuration used in the film "Flow Instabilities" [Mollo-Christensen (1972)]: A wind is blown over a water channel, produced by a vacuum cleaner downstream, and settled by the honeycomb upstream; its speed is measured by an anemometer. A wave generator hits the surface close to the entrance, producing a localized perturbation with given frequency. The height of the waves is measured by an appropriate gauge at a fixed distance from the wave generator.

a given point in the flow and the evolution of the perturbation is recorded at downstream stations as a function of the forcing characteristics and the Reynolds number. From the records, a downstream spatial amplification rate can be defined, negative when the perturbation is damped, positive otherwise.

The different behaviors observed, sketched in Figure 6.11, can be analyzed in terms of plane waves with angular frequency ω_f , $\exp[i(kx - \omega_f t)]$, modulated by a slowly decaying or growing envelope, $\exp(\mu x)$, with μ small, hence $\exp[i(kx - \omega_f t)] \exp(\mu x)$. The two exponential factors can be recombined as $\exp[i(k - i\mu)x - \omega_f t]$ to form a *complex* wavevector with real part k and imaginary part $-\mu$.

We are thus lead to a spatial reading of the dispersion relation (6.43) solved with $\omega = \omega_{\rm f}$ real and k complex. This simplistic presentation is not free from criticism. We have indeed avoided evoking what happens upstream of the forcing point and represent it in Figure 6.11. While $-k_i > 0$ corresponds to an amplification for x > 0 (which conventionally denotes the downstream direction), a mode with $-k_i < 0$ is amplified upstream and the picture is reversed. The nonequivalence of upstream and downstream directions due to advection has delicate analytical implications that we still mostly skip, asking the reader to turn to specialized references.⁶

⁶e.g. the review article by P. Huerre, "Open shear flow instabilities," in [Batchelor et al. (2000)] and references quoted therein.



Fig. 6.11 The spatial amplification rate of the waves is extracted from the response of the system to a localized periodic excitation. The marginal value $R_{\rm m}(\omega)$ of the control parameter separates the domain of damped waves $R < R_{\rm m}(\omega)$ from that of amplified waves $R > R_{\rm m}(\omega)$.

Without entering mathematical intricacies, following Gaster⁷ we can however relate the main characteristics of the spatial (S) problem to those of the temporal (T) problem. Let us consider the dispersion relation (6.43) and assume that it has been analytically solved for ω as a function of k, both complex, at given R:

$$\omega = \varpi(k; R) \,. \tag{6.44}$$

The expansion of the dispersion relation around some arbitrary point $(k^{(0)}, \omega^{(0)})$ solution to (6.43) reads

$$\omega_{\rm r} = \omega_{\rm r}^{(0)} + \frac{\partial \varpi_{\rm r}}{\partial k_{\rm r}} \delta k_{\rm r} + \frac{\partial \varpi_{\rm r}}{\partial k_{\rm i}} \delta k_{\rm i} + \dots \qquad (6.45)$$

$$\omega_{\mathbf{i}} = \omega_{\mathbf{i}}^{(0)} + \frac{\partial \varpi_{\mathbf{i}}}{\partial k_{\mathbf{r}}} \delta k_{\mathbf{r}} + \frac{\partial \varpi_{\mathbf{i}}}{\partial k_{\mathbf{i}}} \delta k_{\mathbf{i}} + \dots \qquad (6.46)$$

A nearly neutral T-mode is characterized by:

 $k_{
m r}{}^{({
m T})} = k_{
m r}{}^{(0)}\,, \qquad k_{
m i}{}^{({
m T})} = 0 ~~{
m (by~definition)}$

and the corresponding angular frequency

$$\omega_{\mathrm{r}}{}^{(\mathrm{T})} = \omega_{\mathrm{r}}{}^{(0)}, \qquad \omega_{\mathrm{i}}{}^{(\mathrm{T})} = \omega_{\mathrm{i}}{}^{(0)} \quad \mathrm{with} \quad |\omega_{\mathrm{i}}{}^{(\mathrm{T})}| \ll 1.$$

⁷M. Gaster, "A note on the relation between temporally increasing and spatiallyincreasing disturbances in hydrodynamic stability," J. Fluid Mech. 14 (1962) 222-224.

We are looking for the characteristics of an S-mode $(k_r^{(S)}, k_i^{(S)})$ with $k_r^{(S)} = k_r^{(T)}$ and angular frequency $\omega_r^{(S)}$ with $\omega_i^{(S)} = 0$ by definition. The T and S modes fulfill the dispersion relation (6.44) and are both nearly neutral, which means in particular $k_i^{(S)} = k_i \neq 0$ with $|k_i^{(S)}| \ll 1$. Inserting these assumptions in (6.45, 6.46) gives at lowest order

$$\omega_{\mathbf{r}}^{(S)} = \omega_{\mathbf{r}}^{(T)} + \frac{\partial \varpi_{\mathbf{r}}}{\partial k_{\mathbf{i}}} k_{\mathbf{i}}^{(S)} , \qquad (6.47)$$

$$0 = \omega_{i}^{(T)} + \frac{\partial \varpi_{i}}{\partial k_{i}} k_{i}^{(S)} .$$
(6.48)

Now, assuming that the dispersion relation (6.44) is analytic implies the existence of relations between the partial derivatives of the real and imaginary parts of ϖ , the so-called *Cauchy relations*:⁸

(a)
$$\frac{\partial \varpi_{\mathbf{r}}}{\partial k_{\mathbf{r}}} = \frac{\partial \varpi_{\mathbf{i}}}{\partial k_{\mathbf{i}}}$$
, (b) $\frac{\partial \varpi_{\mathbf{r}}}{\partial k_{\mathbf{i}}} = -\frac{\partial \varpi_{\mathbf{i}}}{\partial k_{\mathbf{r}}}$. (6.49)

Using (6.49a) in (6.48) yields:

$$\frac{\omega_{i}^{(T)}}{k_{i}^{(S)}} = -\frac{\partial \varpi_{r}}{\partial k_{r}}.$$
(6.50)

Strictly at threshold, this relation is an indeterminate ratio 0/0. Slightly off threshold, it allows us to convert spatial and temporal growth rates into each other. The quantity that shows up on the r.h.s. of (6.50) is the (real) group velocity of the normal modes, which is somewhat natural in view of the discussion on p. 85.

Furthermore, by definition of the threshold condition in the temporal case, ϖ_i reaches a maximum at $k = k_c$ for $R = R_c$. $\partial \varpi_i / \partial k_r$ is thus a small quantity of order $(k_r - k_c)$, which makes $\partial \varpi_r / \partial k_i$ also small from (6.49b) so that, from (6.47) at lowest significant order:

$$\omega_{\mathbf{r}}^{(\mathrm{S})} = \omega_{\mathbf{r}}^{(\mathrm{T})},$$

i.e. that the most dangerous forcing frequency is that corresponding to the most amplified temporal mode, which is intuitively expected.

The stability diagram corresponding to the spatial reading of the dispersion relation displays the forcing frequency on the vertical axis and the

⁸In order to derive relations (6.49), one takes advantage of the fact that $\varpi = \varpi_r + i \varpi_i$, as a function of $k_r = (k + k^*)/2$ and $k_i = -i(k - k^*)/2$, is only a function of k and not of its complex conjugate k^* (which is of course not the case of the most general function of the two variables k_r and k_i).
Reynolds number on the horizontal axis. Its experimental determination is concretely illustrated in the film "Flow instabilities" [Mollo-Christensen (1972)]. For mechanically stable flows, the marginal curve has again a hairpin shape, as illustrated in Figure 6.12.

Flows most often develop downstream, the plane Poiseuille flow being an exception. The Reynolds number is then usually a function of the downstream distance, *e.g.* the distance to the leading edge for a boundary layer as indicated on the horizontal axis of that figure. A first indication relative to the stability of the flow is then obtained by using the results of the linear theory locally, *i.e.* by neglecting the downstream dependence of the flow and assuming a plane parallel flow with a velocity profile identical to that at the considered position. Results will be relevant provided that the space-time growth rate of the normal mode remains large when compared to the space-time evolution rate of the flow itself. They can be corrected by developing a multiple-scale approach to account for the effects of non-parallelism or slow time dependence.

Testing for the stability of an open flow then comes to explore the plane (R, ω_f) by changing the angular frequency of the forcing as illustrated in the film. Of course, one should take the predictions of this linear approach with some caution. In particular, the boundary layer could be anticipated to be stable (relaminarization) from a naive reading of Figure 6.12 but this can happen only if, while crossing the unstable domain, perturbations are kept



Fig. 6.12 Marginal stability curve of a Blasius boundary layer flow submitted to a localized periodic forcing (vibrating ribbon experiment). As the flow develops downstream, R is a function of the distance X to the leading edge. A path in the (R, ω_f) plane is thus followed by moving the observation point at fixed infinitesimal forcing and perturbations are damped or amplified depending on its position. Far downstream the flow is expected to be stable provided that perturbations have no time to grow enough to invalidate the linear approach.

sufficiently small, so that no secondary instability (or even turbulence) sets in, which somehow leaves the *receptivity* problem open, *i.e.* how the system extracts the dangerous TS eigenmodes from the random perturbations forming the residual background turbulence (see later, §6.3.3.2).

6.2.4.2 Absolute and convective instabilities

The general stability problem is not only that of the resonant response of the flow to a periodic localized forcing at a fixed position. Previously, we considered temporal modes in the form of unlocalized waves $\exp(i(kx - \omega t)) = \exp(\omega_i t) \exp(i(kx - \omega_r t))$. The stability of the flow was then implicitly discussed in a frame moving at the phase speed of the waves $[kx - \omega_r t \equiv k(x - c_r t)]$: as long as no reference is made to fixed boundaries, this speed can freely be subtracted using Galilean invariance⁹ and the growth/decay properties can be obtained from the sign of ω_i . But neither the spatial approach nor the temporal approach are satisfactory when dealing with the "natural" transition that relates to the evolution of temporally incoherent and spatially localized small perturbations composing the residual turbulence, in the presence of walls or obstacles breaking Galilean invariance. In the most general case, a complete linear-response theory has to be developed. This implies a mastery of complex-variable analysis that goes beyond the prerequisites of this course, so we limit our ambition to giving a sketchy presentation that just keeps its spirit and suggest the reader to consult specialized works, e.g. Note 6, p. 235.

Since phenomena are ultimately detected in the laboratory frame, we have to account for the competition between the downstream transport of perturbations and their amplification by instability mechanisms. The fate of perturbations which leave the physical domain of interest before having triggered a change of behavior of the flow is indeed of minor importance. Accordingly, as already alluded to before, we shall say that:

• The instability is *convective* when fluctuations are advected downstream. In that case, its effects will be detectable depending on the level of triggering residual turbulence in the flow (or of external perturbations) and the flow behaves as a *noise amplifier*.

⁹The laboratory frame is attached to external characteristics of the flow that break the symmetry of the NS equation (in a infinite domain) with respect to a Galilean transformation: a change to a frame translating at constant speed $\bar{x} = x - Vt$, $\bar{t} = t$ and $v = \bar{v} + V$, *i.e.* $v(x,t) = \bar{v}(x - Vt,t) + V = \bar{v}(\bar{x},\bar{t})$, indeed leaves these equations unchanged owing to the special form of the *advection* term: $\partial_t v + v \partial_x v = \partial_t [\bar{v}(x - Vt, t) + V] + [\bar{v}(x - Vt, t) + V] \partial_x [\bar{v}(x - Vt, t) + V] = -V \partial_x \bar{v} + \partial_t \bar{v} + [\bar{v} + V] \partial_x \bar{v} = \partial_t \bar{v} + \bar{v} \partial_x \bar{v}$.

• The instability is *absolute* when its mechanism is sufficiently intense that perturbations can go against the stream and invade the whole experimental domain in spite of a general downstream transport by the base flow. The system then behaves as a *self-sustained oscillator*.

In the present context, it will be enough to express the result of the theoretical analysis in terms of the so-called *Briggs and Bers criterion* that, in some sense, extends Gaster's approach. Since the 'absolute/convective' discrimination relates to the laboratory frame, let us assume that we can restrict ourselves to the study of wave-packets composed of modes traveling at group velocity $v_{\rm g} = 0$. Now, $v_{\rm g}$ is defined as usual through $v_{\rm g} = \partial \varpi / \partial k$, where ϖ is here the complex function of the complex variable k obtained by solving the dispersion relation (6.43) for ω . Condition $v_{\rm g} = 0$ is thus an equation for k. Let $k^{(0)}$, be the solution to this equation and $\omega_i^{(0)} = \varpi_i(k^{(0)})$ the corresponding growth rate (if there are several solutions $k^{(0)}$, we consider only that with the largest $\omega_i^{(0)}$).

The criterion goes as follows:

- The flow is absolutely unstable if $\omega_i^{(0)} > 0$ (amplification on the spot).
- It is convectively unstable if $\omega_i^{(0)} < 0$, provided that $\omega_i > 0$ for some k.
- It is linearly stable when $\omega_i < 0$ for all k.

These situations are depicted in Figure 6.13 which displays perturbation profile as a function of space at several successive times. An elementary illustration of the use of the criterion is considered in Exercise 6.4.5.

The preceding discussion is typically *local* in that is applies to a quasiparallel flow provided that its x dependence can be neglected. In this perspective, when the flow develops downstream, the intensity of the instability mechanism varies in space so that its absolute/convective character may change downstream. In that case only a *global* analysis can help us understand the stability properties of the flow, which is another full story in itself. Finally, in all that precedes departures from the base flow were tacitly assumed to be wave-like and infinitesimal, making the linear approach legitimate (localized finite-size perturbations were discarded). Besides the identification of important instability mechanisms, a thorough understanding of the transition to turbulence in developing flows has to face difficulties arising from a combination of alternatives: absolute vs. convective, local vs. global, and linear vs. nonlinear.



Fig. 6.13 Space-time representation of the growth/decay of a perturbation introduced at x = 0 and t = 0. Top-left: The two edges of the wavepacket go in opposite directions and the perturbation grows on the spot, the instability is *absolute*. Top-right: The perturbation grows but is evacuated since the two edges of the wavepacket go in the same direction, the instability is *convective*. Bottom: The perturbation vanishes and the flow is *stable*.

6.3 Transition to Turbulence

6.3.1 Nonlinear development of instabilities

6.3.1.1 Saturation of the primary mode

In its principle, the approach is parallel to that followed in Chapter 3 for natural convection. In order to identify the different steps more clearly let us introduce subscripts, '0' for the base state, '1' for the primary perturbation, '2' for the secondary modes, etc. The linear stability theory of the parallel base flow V_0 yields a critical primary mode V_1 in the form of a two-dimensional wave with wavevector k aligned with the flow direction, phase speed c, and no spanwise flow component. Let A_1 be the amplitude of this mode. The first question to be answered relates to the saturation of A_1 beyond threshold, which is dealt with by means of an expansion in powers of the amplitude. This à la Landau approach has been introduced in the field of open flows separately by Stuart and Watson in 1960. Being a stationary cellular instability, convection is accounted for by a supercritical fork bifurcation governed by an equation for a single variable with real coefficients. By contrast, here the wavy behavior of the flow beyond threshold makes it time-oscillatory in the laboratory frame so that a Hopf bifurcation on a two-dimensional center manifold is expected, see Chapter 4, Eq. (4.16), p.128. With $s_1 = \sigma_1 - i\omega_1$ where σ_1 is the real growth rate of the primary mode (positive when unstable) and ω_1 the angular frequency, changing for a rotating amplitude by setting $A_1 = \tilde{A}_1 \exp(-i\omega_1 t)$, we get:

$$rac{\mathrm{d}}{\mathrm{d}t} ilde{A}_1=\sigma_1 ilde{A}_1-g| ilde{A}_1|^2 ilde{A}_1\,,$$

where the complex coefficient $g = g_r + ig_i$ describes the effective interaction of the unstable mode with itself (through the coupling with other, adiabatically eliminated, non-critical modes, see Chapter 4, §4.1).

Saturation is controlled by the sign of the real part g_r of g. When $g_r > 0$ the bifurcation is supercritical and the wave saturates at an amplitude that smoothly varies with the distance to threshold. In the opposite case, the wave does not saturate at lowest order, one must continue the expansion (see the case of the plane Poiseuille flow discussed later) but there is a risk that no stable branch exists at finite distance from the base solution, as recognized early by Landau (Note 9, p. 92).

6.3.1.2 Secondary instabilities

Let us now sketch what should be the next steps and assume that the nonlinear problem for the primary mode has been solved and that saturation at some amplitude A_1 has been obtained. The stability analysis is then repeated for the flow $\mathbf{V}_{01} = \mathbf{V}_0 + A_1 \mathbf{V}_1$ in a frame moving at the speed c_{nl} of the saturated wave (coordinate $\bar{x} = x - c_{nl}t$). In subscript '01' the '0' indicates that it is the new base flow and the '1' that it arises from the saturation of the primary perturbation.

Concrete computations are particularly involved. The standard linearization process leads to a partial differential system with coefficients periodic in \bar{x} with period $\lambda = 2\pi/k$. For such linear operators, the *Floquet* theory implies taking secondary normal modes V_2 in a form that explicitly isolates nontrivial space-time departures from a trivial space dependence in \bar{x} with period λ :

$$\mathbf{V}_2 = A_2 \exp[i(q_x \bar{x} + q_z z - \omega_2 t)] \widetilde{\mathbf{V}}_2(\bar{x}) \quad \text{with } \widetilde{\mathbf{V}}_2(\bar{x} + \lambda) = \widetilde{\mathbf{V}}_2(\bar{x}).$$
(6.51)

The exponential part accounts for a supplementary modulation introduced

by the putative secondary instability, with real wavevector $\mathbf{q} \equiv (q_x, q_z)$ and complex angular frequency $\omega_2 = \omega_{2,r} + \omega_{2,i}$ (temporal instability scheme).

The main possible cases of this conceptually well-posed problem can be identified without doing any explicit calculation, just by considering expression (6.51).

First it is not difficult to show that the value of q_x/k can be restricted to the interval [0, 1] since, setting $q_x = (\varsigma + \kappa)k$ with $\kappa \in \mathbb{N}$, one gets:

 $\exp(iq_x x) = \exp[2\pi i(\varsigma + \kappa)k\tilde{x}] = \exp(2i\pi\varsigma k\tilde{x})\exp(2i\pi\kappa k\tilde{x}),$

where the last exponential factor is periodic in \bar{x} with period $2\pi/\kappa k$, thus also periodic with period $2\pi/k$, so that it can be incorporated into $\tilde{\mathbf{V}}_2$ that accounts for the trivial part of the space dependence.

Next, possible secondary modes can be classified according to whether (i) $q_z = 0$ or $q_z \neq 0$, (ii) $q_x = 0$ or $q_x \neq 0$, and (iii) when $q_x \neq 0$, whether q_x is commensurate or not with k, *i.e.* ς rational or irrational. In particular: • When $q_z = 0$ and $q_x = k/2$ ($\varsigma = 1/2$), the secondary mode is still two-dimensional but has a wavelength twice that of the primary mode, this period doubling instability is called *pairing*.

• When $q_z \neq 0$ and $q_x = 0$, the secondary mode is most often called fundamental, while the term subharmonic is associated to the case $q_z \neq 0$ and $q_x = k/2$ (see later, Figure 6.18).

• When q_x is not commensurate to k (ς irrational), the situation is similar to that of quasi-periodic temporal systems considered in Chapter 4, §4.2.3, and can be transposed from it.

The physical mechanisms involved in secondary instabilities have been studied both theoretically and experimentally. One generally makes a distinction between viscous processes evolving over long time scales (in practice mostly the growth of TS waves) and *inertial* processes developing over short time scales. In this latter context, the *elliptical* instability affecting the core of vortical structures is of great importance. For a detailed discussion with references, see *e.g.* Kerswell¹⁰ or Huerre and Rossi in [Godrèche and Manneville (1998), p. 269ff].

Rather than considering the "natural" transition, the comparison of theory with experiments often bears on flows in which the spanwise dependence of the secondary mode is forced at a given q_z by some experimental trick, *e.g.* by grooving the wall in a boundary layer experiment, the splitting plate for a mixing layer, the lips of the slit for a plane jet, *etc.*

¹⁰R.R. Kerswell, "Elliptical instability," Ann. Rev. Fluid Mech. **34** (2002) 83-113.

The tertiary instability is in general hardly accessible to theory and small scale turbulence usually appears soon after secondary modes set in. The last step of the transition to turbulence is often interpreted as being due to inflection points in the velocity profiles resulting from the superposition of the primary and secondary modes to the base flow, thus promoting KH instabilities at the origin of the smaller scales.

On the whole, the situation is particularly involved since space and time aspects of the development of transitional flows are intermingled in a complicated way owing to downstream transport and evolution.

In the rest of this chapter we present salient features of the transition to turbulence at a phenomenological level resting on the dichotomy between inviscidly unstable and stable flows and the supercritical or subcritical character of the bifurcation. Rather than trying to be exhaustive, we focus on the most typical situations in each case.

6.3.2 Inviscidly unstable flows

6.3.2.1 General features

Two-dimensional base flows that are unstable at the inviscid limit display inflection points in their velocity profile (Rayleigh and Fjørtoft theorems). These are mixing layers (Fig. 6.6), jets and wakes (Fig. 6.2).

The transition to turbulence is characterized by primary instabilities that set in at low Reynolds numbers in the form of two-dimensional spanwise structures as expected from the linear theory. These structures then gently saturate beyond threshold to form developed vortices. Secondary modes are most often streamwise modulations $(q_z = 0)$ leading to quasiperiodic behavior and lockings, especially subharmonic locking that manifests itself as vortex pairing $(q_x = k/2)$. In the case of mixing layers, successive pairings may take place before a secondary instability can introduce some three-dimensional dependence. Small scale turbulence enters soon after the introduction of three-dimensional modes usually arising from the elliptical instability previously mentioned (Note 10). The plane jet, understood as two side-by-side mixing layers close to the outlet and merging further downstream, behaves in the same way.

Another general feature of this transition scenario is the persistence of *coherent structures* in the post-transitional regime. These are large vortices with lifetimes long when compared to their turn-over times, resembling KH vortices but now developing over a turbulent mean flow profile.

6.3.2.2 Wakes

In the following, we focus on the case of the wake of a cylinder, a quantitatively much studied system which combines specificities of flows displaying inflection points in the base flow to features arising from the absolute character of the instability which is responsible for its self-oscillatory nature. Here we borrow mainly from the review given by Williamson.¹¹

The distinctive feature of the wake behind a bluff body is the occurrence of a sharp Hopf bifurcation that marks the emergence of very regular periodic vortex shedding forming a pattern known as the Kármán vortex street after von Kármán (1911-12)¹² and already depicted in Figure 1.6, p. 16. The origin of this behavior can be understood by noticing that, at flow rates corresponding to the bifurcation point, a steady recirculation develops just behind the obstacle so that the velocity profile displays a region where downstream transport is compensated. By contrast, away from it, the flow has been smoothed by viscous effects and is closer to the uniform flow blowing from infinity around the body (Fig. 6.14). Accordingly, there is a region of the flow at the rear of the obstacle where the instability linked to the presence of inflection points can develop on the spot (hence 'absolute'), producing the oscillations. This situation is reminiscent of what happens in closed systems where confinement effects select specific modes and nonlinearities can be studied in the framework of dynamical systems theory (Chapter 3, $\S3.2.3$ and Chapter 4, $\S4.2$).

Let us consider a cylinder placed perpendicular to a uniform flow¹³ with speed U_{∞} . This experimental configuration is characterized by two dimensionless parameters. The first one is the Reynolds number:

$$R = U_{\infty} d / \nu \,,$$

where d is the diameter of the cylinder and ν the kinematic viscosity of the fluid. The second parameter is the aspect-ratio $\Gamma = L/d$ where L is the

¹¹C.H.K. Williamson, "Vortex dynamics in the cylinder wake," Annu. Rev. Fluid Mech. **28** (1996) 477-539.

¹²T. von Kármán, "Über den Mechanismus des Widerstandes, den ein bewegter Körper in ein Flüssikeit erzeugt," Nachr. Ges. Wiss. Göttingen, Math. Phys. Klasse (1911) 509-517, (1912) 547-556. Previous experimental observations by H. Bénard (1908) are reported in Notice sur les titres et travaux scientifiques de M. Henri Bénard (Gauthier-Villars, Paris, 1926). French chauvinism often associates the two names with the wake instability.

¹³In one of the last sequences of the film "Flow instabilities" [Mollo-Christensen (1972)], a cylinder is dragged through a viscous fluid at rest. In that case U_{∞} is of course the speed of the body.



Fig. 6.14 'Absolute/convective' transition in the wake of a cylinder (and, more generally, of a blunt body).

length of the cylinder. Here we assume $L \gg d$ (or $\Gamma \to \infty$), so that it seems legitimate to neglect end effects and to start with a two-dimensional base flow. Accordingly, we are left with R as the only parameter.

The flow structure can be understood from visualizations of the velocity field, while the different regimes are most conveniently identified through the variation of global observables. A first one is the oscillation frequency f, or in dimensionless form, the Strouhal number

$$S = rac{fd}{U_{\infty}}$$
 .

A second classical measure is the drag, *i.e.* the force $F_{\rm D}$ necessary to maintain the cylinder at its position, or rather the dimensionless drag coefficient $C_{\rm D} = F_{\rm D}/\frac{1}{2}\rho U_{\infty}^2 d$. Another related quantity is the base pressure, the pressure difference between a point at 180 degrees from the upstream stagnation point and a reference pressure, usually the static pressure at infinity. Here we will just discuss the early stages of the transition using the Strouhal number.

The transition to turbulence behind a cylinder is described at many places besides Williamson's review (Note 11), e.g. in [Tritton (1990)]. Beautiful pictures of the different laminar flow regimes by Taneda can be found in [van Dyke (1982)]. Creeping flow with flow lines smoothly sticking to the cylinder exists up to $R \sim 5-6$, beyond which a steady symmetrical bubble of recirculating fluid sets in. This time independent situation persists up to $R = R_c \simeq 48.5$ beyond which the wake oscillates, periodically emitting vortices parallel to the cylinder axis as illustrated in Figure 6.15.



Fig. 6.15 Top: Rerspective view of the shedding of parallel vortices, after one of Taneda's pictures in [van Dyke (1982)]. Bottom: Parallel vs. oblique shedding.

At threshold, the Strouhal number is of order 0.12, so that the streamwise wavelength $\lambda_0 = U_{\infty}/f$ is of order 8*d*.

Beyond threshold, the oscillation frequency increases regularly with the Reynolds number as shown in Figure 6.16. In fact the frequency selection problem is a difficult one, both experimentally and theoretically.

On the experimental side, it took some time before it was recognized that the dispersion of the observed frequencies was due to the occurrence of oblique shedding, i.e. shedding with the axis of the vortices at an angle θ with the cylinder axis. Assuming that the frequency of parallel shedding is f_0 , a simple geometrical construction (Fig. 6.15, bottom-right) shows that the frequency of oblique shedding (and with it, the Strouhal number S) is reduced by a factor $\cos \theta$: let λ_{θ} be the wavelength of the oblique vortex street as measured perpendicularly to the vortex axis and suppose that the instability mechanism makes it identical to that of parallel shedding, then the wavelength measured along the flow direction is longer by a factor $1/\cos \theta$; to get the result one just needs to convert wavelengths into oscillation periods using the speed of the flow U_{∞} . Rescaling frequencies accordingly yields the master curve displayed in Figure 6.16 (left).

The theoretical understanding of the neighborhood of the threshold is not as obvious as it may seen. At first, the wake can be thought of as a



Fig. 6.16 Variation of the Strouhal number S (dimensionless frequency of shedding) as a function of the Reynolds number R. Left: Two-dimensional shedding regime corrected for the effects of oblique shedding. Right: Transition to three-dimensionality. Modes modes A^{*} and B^{*} have the same structure than modes A and B but display dislocations; dashed lines correspond to unstable regimes; The transition '2D \leftrightarrow mode A^{*}' is hysteretic; the "natural" route is: 2d \rightarrow mode A^{*} \rightarrow mode B. After Williamson, note 11.

collection of Hopf oscillators, distributed all along the cylinder and coupled to their neighbors. An appropriate local model would then be (4.16), p. 128 which, once unfolded in the spanwise direction z, would yield a complex Ginzburg-Landau equation (5.36), p. 198. The dynamics of the wake close to the threshold is in fact reasonably well described by this equation whose coefficients can be fitted against experiments.¹⁴ Chevron patterns observed in case of oblique shedding, finite-size effects, or the presence of dislocations in the vortex system (viewed as phase defects for the complex amplitude), can be understood within this phenomenological approach. But this cannot be the whole story since the approach implicitly assumes that the streamwise (x) structure of the perturbation is frozen, which this is not the case. While the amplitude of the maximum of the wake velocity fluctuations indeed increases as the square root of the distance to threshold as expected for a Hopf bifurcation, the position of the maximum shifts upstream closer to the cylinder. Progress has been made recently regarding the variation of the streamwise shape of the vortex amplitude and the frequency selection problem by combining convective/absolute, global, and nonlinear issues.¹⁵

¹⁴M. Provansal, C. Mathis, L. Boyer, "Bénard-von Kármán instability: transient and forced regimes," J. Fluid Mech. **182** (1987) 1-22. Th. Leweke, M. Provansal, "The flow behind rings: bluff body wakes without end effects," J. Fluid Mech. **288** (1995) 265-310.

¹⁵See, e.g. B. Pier and P. Huerre, "Nonlinear self-sustained structures and fronts in spatially developing wake flows," J. Fluid Mech. **435** (2001) 145–174, B. Pier, "On the frequency selection of finite-amplitude vortex shedding in the cylinder wake," J. Fluid Mech. **458** (2002) 407–415, and references cited.

Beyond $R \approx 150$, the wake enters a transition regime where the regular vortex street may be disrupted by several different secondary instability modes, all involving spanwise modulation. The picture is further complicated by the presence of vortex dislocations. Two main modes have been identified,¹⁶ first experimentally, next theoretically through linear stability analyses.¹⁷ The first one, called 'mode A', arises as a deformation of the primary vortices with wavelength of the order of $0.6-0.8\lambda_0$, where λ_0 is the primary wavelength at the corresponding value of R. Since the Strouhal number is then about 0.2, the wavelength of mode A is of order 3-4d. By contrast the second one, called 'mode B', has a much shorter wavelength of order $0.2\lambda_0$, *i.e.* of the order of d. The physical origin of these modes cannot be explained in terms of the elementary KH mechanism discussed previously but rather relates to the elliptical instability (§6.3.1.2 and Note 10).

The transition regime extends up to $R \approx 260$. Beyond this point the intensity of fine structures increases and the nature of the instabilities involved in the primary vortex shedding mechanism changes gradually. Rather than deriving from the global mode attached to the recirculation bubble, as was the case at lower R, these instabilities have to do first with shear layer that forms downstream the detachment point and, at the highest Reynolds numbers, with the boundary layer that develops along the cylinder itself.

6.3.3 Inviscidly stable flows

6.3.3.1 Plane Poiseuille flow

This prototype of flows without inflection point has been much studied both theoretically and experimentally. The threshold of the instability against TW waves is $R_c \simeq 5772$ (see p. 234). As shown by Herbert¹⁸ the bifurcation is subcritical with unstable nonlinear steady states appearing below threshold, but the bifurcated branch turns back at $R_{nl} \simeq 2900$ (Fig. 6.17).

The next step should be the linear stability analysis of two-dimensional saturated TS waves against three-dimensional infinitesimal perturbations, as suggested in $\S6.2.1$. One would thus expect transversally modulated

 $^{^{16}}$ Ch. Williamson, "Three-dimensional wake transition behind a cylinder," J. Fluid Mech. (1995).

¹⁷D. Barkley, R.D. Henderson, "Three-dimensional Floquet stability analysis of the wake of a circular cylinder," J. Fluid Mech. **322** (1996) 215-241.

¹⁸T. Herbert, "Secondary instability of plane channel flows to subharmonic threedimensional disturbances," Phys. Fluids **26** (1983) 871-874.



Fig. 6.17 Bifurcation diagram of two-dimensional TW waves in plane Poiseuille flow. After Herbert, Note 18.

waves later decaying into small scale turbulence. Several difficulties hinder the observation of this scenario. First, the primary instability is convective so that detecting its presence in a finite length channel strongly depends on the level of background noise. Second, the bifurcation is subcritical so that working with a "clean" base flow inevitably leads to missing the stable bifurcated branch at finite distance from it, whereas triggering appropriate waves, *e.g.* using vibrating ribbons, is not so easy.

By contrast, a moderate high level of residual turbulence brings about a different scenario (*natural* transition) involving localized finite-size perturbations, called *turbulent spots*, *i.e.* limited patches of small-scale turbulent flow traveling amid laminar Poiseuille flow, growing in size as they process downstream.¹⁹ These turbulent spots are transient and rapidly decay into laminar flow below $R_{\rm g} \approx 1000$, but are sustained beyond this value, which is much lower than $R_{\rm c}$ or even $R_{\rm nl}$. The value $R_{\rm g}$ plays the role of a global stability threshold, hence the notation (see p. 9 for a refresher). Except for the existence of a linear threshold and a corresponding nonlinear branch, the situation is similar to that taking place in the plane Couette flow to be examined later in §6.3.4.3.

6.3.3.2 Boundary layer flows

The main drawback of the Poiseuille flow as a test case for continuous transition scenarios in open flows stems from its strongly subcritical character.

¹⁹D.R. Carlson, S.E. Widnall, M.F. Peeters, "A flow visualization of transition in plane Poiseuille flow," J. Fluid Mech. **121** (1982) 487-505.

Boundary layer flows may seem more favorable from this point of view and, being ubiquitous, have even more practical interest.

The Blasius velocity profile is another example of flow without internal inflection point. As we have seen, the thickness δ of the laminar boundary layer increases as the square root of the distance to the leading edge. Theory predicts the emergence of TS waves at $R = R_c \simeq 519.4$ with $k_c \simeq 0.303$ and $c_{\rm r,c} \simeq 0.397$ (all scalings based on δ). The bifurcation seems supercritical or nearly so, but the interpretation is made delicate by the downstream dependence of the flow. Anyway, linear stability has been tested at low residual turbulence levels in vibrating ribbon experiments and the agreement between predictions and observations is satisfactory once corrections due to the slight divergence of the flow are taken into account.

Several tricks have been used to attempt at controlling the spanwise wavevector q_z of the secondary modes in experiments (regularly spaced grooves or adhesive tapes aligned along the flow direction). In practice the most dangerous secondary modes either do not add streamwise modulation $(q_x = 0, \varsigma = 0, \text{ where } \varsigma \text{ was introduced in §6.3.1.2})$ or else introduce a spatial subharmonic $(q_x = k/2, \text{ that is } \varsigma = 1/2)$. In the latter case, resonant interactions between a mode triad formed with the primary mode (k, 0) and two oblique waves $(q_x = k/2, \pm q_z)$ may be expected from the quadratic coupling through the advection term $\mathbf{v} \cdot \nabla \mathbf{v}$ (Figure 6.18, top).

The velocity field resulting from the superposition of the base flow and the secondary modes displays streamwise bands with alternatively increased and reduced speed ("peak-valley" alternation) that, upon amplification, take the shape of so-called Λ -vortices, see Fig. 6.18, bottom.

A step-by-step transition is thus expected. Laminar flow first becomes unstable against two-dimensional TS waves, the amplitude of which slowly increases over a viscous time-scale. TS-waves then experience growth of secondary modes introducing three-dimensionality over a much shorter inertial time-scale, and a final breakdown into small scale turbulence.

In fact the kind of secondary mode that develops the fastest depends on the amplitude A_1 reached by the TS waves, which in turn is a function of the level of residual turbulence (through the exponential amplification of the primary mode over a limited time, converted into a distance by the downstream flow). For a detailed presentation consult the reviews by Herbert and/or Kachanov,²⁰ and also [Schmid and Henningson (2001)].

²⁰T. Herbert, "Secondary instabilities of boundary layers," Ann. Rev. Fluid Mech. **20** (1988) 487-526; Y.S. Kachanov, "Physical mechanisms of laminar-boundary-layer transition," Ann. Rev. Fluid Mech. **26** (1994) 411-482.



Fig. 6.18 Top: Resonance conditions for secondary modes. Bottom: (a) Aligned A-vortices, $q_x = 0$, 'K' mode. (b) Staggered A-vortices, $q_x = k/2$, 'H' mode.

Figure 6.19 adapted from [Schlichting (1979)] displays a global schematic image of these different steps.

• At the lowest residual turbulence levels, the amplitude of the TS waves staying below 0.2–0.3% of the speed at infinity, relaminarization is observed owing to the shape of the marginal stability curve, as already mentioned).

• When their amplitude is between 0.3% and 0.6%, the fastest emerging secondary mode is subharmonic, which produces staggered Λ -vortices, this is 'mode H' after Herbert's name, Note 18, p. 249.

• Finally, when their amplitude is larger than 0.6%, the prevalent secondary mode is with $q_x = 0$ and produces aligned Λ -vortices as early observed by Klebanoff and coworkers.²¹

A-vortices then breakdown into *spikes* that serve as germs for turbulent spots developing first in the cross-flow direction y and then laterally in the x-z plane while being advected downstream. Merging of the turbulent patches yields a fully developed turbulent boundary layer. Large streamwise rolls called *wall streaks*, with finite lifetime and well-defined wavelength can be identified within the turbulent boundary layer. They experience intermittent breakdown and regeneration. This phenomenon called *bursting*

²¹P.S. Klebanoff, K.D. Tidstrom, L.M. Sargent, "The three-dimensional nature of boundary layer transition," J. Fluid Mech. **12** (1962) 1-34.



Fig. 6.19 Side view and top view of a 'K mode' transitional boundary layer. After [Schlichting (1979)].

has been studied within the framework of chaotic dynamical systems using tools similar to those presented at the end of chapter 4, $\S4.4.^{22}$

These different steps are difficult to identify in the absence of control of secondary modes (natural transition) and *turbulent spots* often appear without apparent precursors. Such spots can also be introduced by perturbing the laminar boundary layer locally and sufficiently strongly. Of course, when the flow is very clean, one must not conclude that it never become turbulent, but that, when it does, the other scenario involving spots develops. A kind of competition between the standard sequences sketched above and a *bypass* transition not relying on secondary instability modes can thus be observed in the general case.

6.3.4 Turbulent spots and intermittency

6.3.4.1 Context

In several instances we have mentioned the fact that *turbulent spots*, *i.e.* bounded regions filled with strongly turbulent flow, could appear, move, and expand within laminar flow. This phenomenon can be understood by noticing that that solutions to the Navier-Stokes equations may not all be derived continuously from the thermodynamic branch by increasing the

²²N. Aubry, Ph. Holmes, J.L. Lumley, E. Stone, "The dynamics of coherent structures in the wall region of a turbulent boundary layer," J. Fluid Mech. **192** (1988) 115–173.



Fig. 6.20 (a, b) Super/sub-critical local bifurcation. The distance to the basic state is measured by the amplitude A of the primary instability mode beyond the corresponding threshold R_c . (c, d) Global bifurcation toward a branch of nontrivial states existing beyond $R > R_{\rm nt}$, disconnected from the basic state, Δ is the measure of the distance to that state.

Reynolds number after supercritical bifurcations, and that there may exist other fully nonlinear solutions belonging to other disconnected branches. Figure 6.20 illustrates this situation in a highly simplified way:

The bifurcation diagrams (a) and (b) correspond to *local* bifurcations and have the same precise technical meaning as in Chapter 4: variable A on the vertical axis is the amplitude of the bifurcated state.

By contrast, the nature of the states on the nontrivial branch in diagrams (c) and (d), resulting from *global* bifurcations, is left unspecified for the moment. Quantity Δ , which serves as an order parameter, is simply some idealized statistical measure of the distance to the base flow, hiding a multiplicity of (possibly irregular) flow configurations.

From these last two sub-figures, one will just retain (i) the possible existence of nontrivial states for $R > R_{\rm nt}$ ('nt' for 'nontrivial') and (ii) the fact that depending on cases, the local bifurcation can take place at values of R lower or higher than $R_{\rm nt}$. Diagram (d) represents a case with $R_c > R_{\rm nl}$ and $R_{\rm nl} > R_{\rm nt}$, *i.e.* a subcritical, primary, local bifurcation and nontrivial states at even lower values of R, reminiscent of our description of the plane Poiseuille flow transition. Diagram (c) with a supercritical primary bifurcation occurring first $R_c < R_{\rm nt}$, and a nontrivial branch soon becoming relevant, would rather model the case of the Blasius boundary layer, though the exact situation is difficult to appreciate due to the downstream development of the instability.

At any rate, the suggestion is made explicit that the globally supercritical cascade toward turbulence can be preempted by other mechanisms involved in the so-called bypass transition (that can thus be consistently termed globally subcritical). In terms of dynamical systems theory, the principal



Fig. 6.21 Transition to turbulence by growth of turbulent plugs/slugs in a cylindrical pipe. The growth, propagation, and merging of the slugs is suggested by sketches of snapshots taken at regularly spaced successive times.

difficulty is then to determine the boundary of the attraction basin of the nontrivial solution, that is to say not only the most dangerous directions in phase space, from the base state or related "trivial" nonlinear solutions, but also the amplitude of the perturbations that actually trigger the transition. Solving exercise 4.6.6 in Chapter 4 may contribute to the understanding of that difficulty in an exceedingly simplified case, the precise connection of that exercise with the present problem is discussed in the article from which it derives, Note 27, p. 174.

The case of plane Couette flow is, from this point of view, the most dramatic since the base flow is linearly stable for all R as shown by Romanov,²³ so that the branch that would correspond to the local bifurcation is pushed at infinity (in R).

This circumstance is shared by the Hagen-Poiseuille flow, the laminar flow in a cylindrical pipe, which is not a plane flow but is mentioned here owing to the historical role of Reynolds' experiment (1883) regarding the definition of the Reynolds number and the problem of the transition to turbulence.²⁴ It indeed turns out that the parabolic velocity profile (6.5) is also linearly stable for all R, though turbulent flow can be observed for sufficiently large R. The transition occurs without intermediate steps through the formation of *turbulent plugs* equivalent to the turbulent spots observed in plane flows. As suggested in Figure 6.21, a turbulent plug develops close to the tube wall, then invades the section and grows in the streamwise di-

 $^{^{23}}$ V.A. Romanov, "Stability of plane parallel Couette flow" (english transl.) Funktsional'nyi Analiz i Ego Prilozhaniya 7 (1973) 62-73.

 $^{^{24}}$ The experiment is repeated in the film Turbulence [Stewart (1972)]. There, the Reynolds number is modified by mixing water and glycerol in order to vary the viscosity, rather than by changing the flow rate.

rection to become a *turbulent slug*. These slugs grow in length and thus persist indefinitely, *i.e.* till the end of the pipe, provided R is large enough. In practice, the head of a slug moves faster than its tails for R greater than about 2300. The slugs become statistically longer and longer beyond this value and, if their birth probability is large enough, the pipe becomes turbulent along most of its length. The transition is thus highly sensitive to the care brought to the experimental conditions (shape of the entrance of the pipe, roughness of the wall, level of residual turbulence). The generation mechanism of the plug depends on the Reynolds number. Laminar flow can be maintained at $R > 10^5$ in experiments designed with extreme care but usually the natural transition is observed in pipes of moderate length for $R < 10^4$ due to detachment on sharp angles in the entrance section, or in the range $R \sim 10^4$ -10⁵ due to the instability of the boundary layer close to the wall of the pipe.

At any rate, the globally subcritical character of the transition toward nontrivial flow regimes, that implies coexistence *in phase space*, is essential to understand the coexistence of turbulent and laminar flow *in physical space* with well-defined fronts separating them at one and the same value of the control parameter.²⁵

6.3.4.2 Turbulent spots in Blasius and Poiseuille flows

Described first by Emmons (1951), turbulent spots developing in the Blasius boundary layer flow have the shape of an arrow head (Fig. 6.22, left).²⁶ They move rapidly in the flow, about 90% and 50% of the speed at infinity for the head and the tail respectively. They widen while advancing with a spreading angle of order 10° . They also thicken by entraining laminar fluid from the outside of the boundary layer.

In order to partially suppress problems linked to the space dependence of the base flow, Carlson *et al.* (Note 19, p. 250) have studied the development of spots artificially produced by local triggering in plane Poiseuille flow (R is now independent of the streamwise coordinate x). As already mentioned, turbulent patches persist within laminar flow for $R \ge 1000$, *i.e.* much below the value at which nonlinear TS waves bifurcate ($R_{nl} \simeq 2900$). As seen in Fig. 6.22 (right), triggered spots are oval. The head and tail of a spot move at speeds roughly 2/3 and 1/3 of the centerline velocity, re-

²⁵Y. Pomeau, "Front motion, metastability and subcritical bifurcations in hydrodynamics," Physica D 23 (1986) 3-11.

²⁶B. Cantwell, D. Coles, P. Dimotakis, "Structure and entrainment in the plane of symmetry of a turbulent spot," J. Fluid Mech. **87** (1978) 641-672.



Fig. 6.22 Turbulent spots in the Blasius flow at high R (left, after Cantwell *et al.* in [van Dyke (1982)]), and in plane Poiseuille flow (right, after Carlson *et al.*, Note 19).

spectively. The spreading angle is of the order of 8° . Further downstream, at distances ~ 150h (h = channel's height), a typical turbulent patch has a diameter of order 30-50h and then experiences *spot splitting*: a calm region appears, separating two turbulent spots that run side by side and further grow. Turbulent patches that develop spontaneously within the laminar flow at high levels of residual turbulence evolve similarly but have less regular shapes and, of course, appear randomly in space and time.

As a general rule, turbulent spots display different regions that are well visible in Figure 6.22: (1) the spot as a whole, a turbulent region with well-defined boundaries, that behaves as an obstacle in the flow since it moves somewhat more slowly than the neighboring laminar flow; (2) on the sides, trains of oblique waves seemingly produced by the motion of this obstacle; (3) an interior filled with small scale turbulence; and (4) a turbulent "wake" made of streamwise elongated streaks. These features are also present in Figure 1.7, p. 17, displaying a mature turbulent spot in plane Couette flow.

6.3.4.3 Plane Couette flow

Plane Couette flow, conceivably the simplest case of wall flow, is known to stay linearly stable for all R (Note 23). This guarantees the subcritical character of the transition, avoiding any risk of confusion with scenarios based on saturated TS waves. By contrast with Blasius or Poiseuille flows, there is further no problem with the downstream transport of spots when using the apparatus used in Saclay²⁷ and described in Figure 6.23 since, by construction, the mean advection cancels exactly for symmetry reasons. In

²⁷F. Daviaud, J. Hegseth et P. Bergé, "Subcritical transition to turbulence in plane Couette flow," Phys. Rev. Lett. **69** (1992) 2511-2514; O. Dauchot and F. Daviaud, "Finite amplitude perturbations and spot growth mechanism in plane Couette flow," Phys. Fluids **7** (1995) 335-343.



Fig. 6.23 Setup used to produce high-quality plane Couette flow in view of a quantitative study of turbulent spots. Left: The shear is created in the gap separating two portions of a wide plastic belt kept parallel at a constant distance 2h by four adjustable rolls, with counter-plates limiting mechanical vibrations, and entrained at linear speed Uby two large rolls. Right: Turbulent spots are generated by a thin jet that goes through the flow when holes in the belt and the counter-plates are aligned. After Daviaud and coll. (CEA Saclay).

fact, it is not so easy to produce a flow with a low level of residual turbulence and sophisticated image processing is necessary to reach a meaningful quantitative level.²⁸

Traditionally, the Reynolds number is defined as $\tilde{R} = Uh/\nu$ where U is the speed of the wall inducing the shear and 2h the width of the gap between the walls. It turns out physically more significant to define R from the shear rate U/h, the inverse of the advection time, and the viscous time over the gap, $\tau_{\rm v} = (2h)^2/\nu$, since experiments show that the structures involved in the transition occupy the full gap. The Reynolds number will accordingly be defined as $R = 4Uh/\nu = 4\tilde{R}$, allowing easier comparisons with, e.g. Blasius or Poiseuille flows.

Turbulent spots in plane Couette flow display characteristics similar to those in other wall flows. It is now well established that the plane Couette flow is globally stable for²⁹ $R < R_g \simeq 1300$. This value compares well with that in the plane Poiseuille flow case (~ 1000 for sustained spots), which is easily understandable when noticing that the latter can be viewed as two juxtaposed Couette flows (the centerline velocity $U_{\rm max}$ would be 2U and the half-height of the channel 2h, hence $R_{\rm Pois.} = 4Uh/\nu$ again).

 $^{^{28}}$ Equally interesting information can be obtained from direct numerical simulations of Navier-Stokes equations, *e.g.* A. Lundbladh and A.V. Johansson, "Direct simulations of turbulent spots in plane Couette flow," J. Fluid Mech. **229** (1991) 499-516.

²⁹For a review of experimental results, see P. M. and O. Dauchot, "Patterning and transition to turbulence in subcritical systems: the case of plane Couette flow," in *Coherent structures in classical systems*, D. Reguera, J.M. Rubi, L.L. Bonilla, eds. (Springer Verlag, 2001), pp. 58–79.



Fig. 6.24 Turbulent fraction function of time for different values of R. 'Q' indicates a *quench* experiment, and 'S' a *spot triggering* experiment. Top: Immediate unconditional relaxation toward laminar flow. Left: Relaxation usually after a long chaotic transient. Right: Sustained turbulence. Adapted from original results, courtesy S. Bottin and O. Dauchot (CEA Saclay, 1998).

The expression 'globally stable' is to be taken with precisely the meaning defined in Chapter 1, p. 9 and Figure 1.2. Experiments leading to the result are principally of two kinds:³⁰ either quench experiments ('Q') where a fully turbulent state, prepared at some large initial R, $R_{\text{init}} \gg R_{\text{g}}$, is suddenly decreased to some final value R_{fin} , or spot triggering experiments ('S') achieved as sketched in Figure 6.23 (right). In each case, the evolution of the turbulent fraction F_{t} , that is the relative surface occupied by the turbulent flow, is extracted from video recordings of the flow pattern, Figure 1.7, p. 17, and appropriate thresholding of the images. As illustrated in Figure 6.24, these experiments show that for $R \simeq 1120$ all perturbations relax rapidly, whatever their initial structure, shape, and amplitude. For R between $R_{\text{u}} \simeq 1250$ and $R_{\text{g}} \simeq 1300$, turbulent patches exist and live sufficiently long for the average measure of their surface as a function of time to make sense. This lifetime diverges as R gets closer and closer to

³⁰S. Bottin, F. Daviaud, P. M., O. Dauchot, "Discontinuous transition to spatiotemporal intermittency in plane Couette flow," Europhys. Lett. **43** (1998) 171-176.



Fig. 6.25 Turbulent fraction F_t as a function of the Reynolds number, from a compilation of experiments.

 $R_{\rm g}$ from below. Finally, turbulent patches persist indefinitely with finite probability for $R > R_{\rm g}$.

It can be noticed that even when turbulence is sustained (Fig. 6.24, bottom-right) some spots may relax immediately (label 'S2') though prepared in the same way as those that do not abort ('S1'). This feature is the trace of both the globally subcritical character of the bifurcation, and the sensitivity to initial condition expected from a basically chaotic process. Numerical simulations confirm this point by showing that, for Reynolds numbers in the corresponding range, the border of the attraction basin of the laminar flow is fractal.³¹ Furthermore, in this regime, the average value of the turbulent fraction does not depend on the type of experiment (Q or S), which qualifies it as an adequate order parameter.

The bifurcation diagram displayed in Figure 6.25 summarizes results on the average turbulent fraction F_t as a function of R. This diagram suggests that F_t increases regularly with R. As a matter of fact, the flow becomes uniformly turbulent beyond $R_t \simeq 1660$. In the range $R_g < R < R_t$ the intensity of turbulence appears to be modulated in the form of alternately laminar and turbulent oblique bands. Just beyond R_g , these bands are fragmented (Fig. 6.26, top) but, as R increases, for $R \sim 1400-1450$, fragments join together to form continuous bands (Fig. 6.26, bottom). Then the modulation progressively disappears as the regime of uniform turbulence is

³¹A. Schmiegel and B. Eckhardt, "Fractal stability border in plane Couette flow," Phys. Rev. Lett. **79** (1997) 5250-5253.



Fig. 6.26 Typical oblique turbulent stripes observed in the range $R_g < R < R_t$ in the plane Couette flow for R = 1360 (top) and R = 1432 (bottom). Courtesy A. Prigent (CEA Saclay).



Fig. 6.27 Stability diagram of the plane Couette flow.

reached. These results are recapitulated in Figure 6.27.

Turbulent bands are strikingly similar to the celebrated turbulent spirals observed in cylindrical geometry (Taylor-Couette flow, see Exercise 3.3.5, p. 113) when the cylinders rotate in opposite directions and sufficiently fast. Early observations of these spirals date back to Coles' experiments.³² A quantitative bifurcation diagram has been established by Andereck *et al*³³ for a ratio $\eta = r_o/r_i = 0.883$ where r_o and r_i are the radii of the outer and inner cylinders respectively. Further experiments performed by Dauchot and Prigent³⁴ with $\eta = 0.983$ show that bands and spirals take place in

³²D. Coles, "Transition in circular Couette flow," J. Fluid Mech. 21 (1965) 385-425.

 $^{^{33}}$ D. Andereck, Liu, H. Swinney, "Flow regimes in a circular Couette flow system with independently rotating cylinders," J. Fluid Mech. **164** (1986) 155–183.

 $^{^{34}}$ A. Prigent, G. Grégoire, H. Chaté, O. Dauchot, W. van Saarloos, "Large-scale finitewavelength modulation within turbulent shear flows," Phys. Rev. Lett. **89** (2002) 014501.

the same range of Reynolds numbers (weakly dependent on the average rotation rate as far as spirals are concerned), once computed, as in the case of the plane Couette flow, from the shear rate and the viscous time over the gap when curvature effects are small, $\eta \rightarrow 1$. A physical explanation for the occurrence of turbulent bands or spirals is still lacking at present.

The principal interest of this detailed study is to give a concrete expression to the conceptual presentation given in Figure 6.20, especially in connection with the nature of the states on the postulated disconnected *nontrivial branch* of the bifurcation diagram (compare with Figure 6.25).

6.3.4.4 Mechanisms?

Turbulent spots appear to play an essential role in transitional wall flows. They have common features that do not seem to rely on TS waves and their nonlinear development. First they are present at "intermediate" values of R, *i.e.* not too large so that the flow as a whole remains relatively coherent and not yet filled with small scale eddies as in fully developed turbulence (Chap. 7). For plane Couette flow this transitional regime extends in the range 1250 < R < 1440. Trying to identify a limited number of interacting modes and processes thus seems meaningful.

According to the Squire theorem and the conventional TS transition scenario, the most dangerous infinitesimal perturbations are two-dimensional, *i.e.* periodic in x, functions of y, and independent of z. Disturbances relevant to the non-conventional scenario are thus expected to have finite amplitude and be mostly spanwise, *i.e.* depend on z (and y) but not on x. Small perturbations of this kind are damped in the long term but transient linear amplification is not excluded owing to the *non-normal* character of the linear stability operator (see Appendix A, §A.3). The required finite amplitude can thus be reached provided that sufficiently large residual turbulence or intentional disturbances are present. (The linear scheme is therefore not completely abandoned since, on general grounds it is hopeless to find nontrivial nonlinear solutions from scratch.)

In practice it is not too difficult to imagine a mechanism by which streamwise structures can be sustained in the flow. As a matter of fact, a streamwise vortex [0, v(y, z), w(y, z)], even transient and weak, induces a redistribution of streamwise velocity $\delta U(y, z)$ by displacing fluid with speed U(y) to a different cross-stream coordinate y. This perturbation δU corresponds to a slowing down or a speeding up, depending on the local direction of displacement of the basic flow lines by the vortex (see



Fig. 6.28 Streamwise vortices and formation of streaks.

Figure 6.28). The presence of long, well visible, streamwise streaks in the flow is thus not so surprising. The superposition of the base flow and the streamwise component of the induced flow now displays inflection points that make it unstable to KH-like mechanism producing x-periodic waves, which in turn can trigger the transition.

The question is then to determine the initial amplitude of streamwise vortices able to do the job (to find the boundary of the attraction basin of the base state in dynamical systems terms). The answer has been searched mostly by direct numerical simulation of the NS equations and comparison of scenarios starting with different initial perturbations. For a detailed presentation with references, consult [Schmid and Henningson (2001), Chap. 9]. This points to the *receptivity* problem, *i.e.* the way the flow converts residual turbulence or controlled disturbances into effective perturbations.

In boundary layers, perturbations in the form of oblique waves (an anticipation of secondary instability modes of TS waves) seem to be most efficient. In the case of plane Couette or Poiseuille flows, with Reynolds numbers independent of the streamwise distance, the result is often presented in the form of a diagram relating the initial amplitude required for transition and the Reynolds number in the form

$$\Delta_{
m init} \propto R^{-\gamma}$$
 ,

where γ is an empirically determined exponent and Δ is a measure of the perturbation amplitude in the sense of Figure 6.20 (essentially the square-root of the kinetic energy contained in the disturbance). The above relation is supposed to hold for $R \gg 1$. This does not raise questions in the case of plane Couette flow which is linearly stable for all R. In the plane Poiseuille flow case, such a relation can be strictly valid only in the subcritical regime,

 $R < R_{\rm c} \simeq 5772$, since beyond it is linearly unstable, but fortunately $R_{\rm c}$ is quite large and turbulent patches are expected beyond $R \sim 1000$, which leave some room to the observation of the scaling behavior. For boundary layers, the relation makes sense only as long as the non-conventional transition through turbulent spots at high levels of residual turbulence preempts the conventional TS transition.

The value of the exponent γ depends on the flow considered and the type of perturbation introduced. Values such that $\gamma \approx 5/4$ (plane Couette) and $\approx 7/4$ (plane Poiseuille) have been extracted from direct numerical simulations of the NS equations.³⁵ A functional approach developed by Chapman³⁶ leads to $\gamma = 1$ for plane Couette flow and all kinds of perturbations, and to $\gamma = 3/2$ or $\gamma = 5/4$ for plane Poiseuille flow and streamwise vortices or oblique waves, respectively. It can be noticed that analytical values are slightly smaller than experimental ones, which means that the considered perturbations are effectively more dangerous than estimated (their amplitudes decrease faster than predicted when *R* increases).

The study of processes leading to the transition can also give ideas about the nature of states belonging to the nontrivial branch since it is sufficient to admit that the instability of the streamwise streaks produces new streamwise vortices, which closes the loop. Such a cycle has indeed been pointed out in numerical simulations. The model studied considered in Exercise 4.6.6 with its variables X and Y representing the amplitude of the streaks and the streamwise vortices respectively, possesses both trivial and nontrivial states, but is too simple to account for a realistic feedback. A more sophisticated model has been proposed by Waleffe³⁷ coupling four variables. It reads:

$$\begin{split} \dot{U} &= -\kappa_u R^{-1} U - \sigma_w W^2 + \sigma_u M V, \\ \dot{V} &= -\kappa_v R^{-1} V + \sigma_v W^2, \\ \dot{W} &= -\kappa_w R^{-1} W + \sigma_w U W - \sigma_m M W - \sigma_v V W, \\ \dot{M} &= -\kappa_m R^{-1} (M-1) + \sigma_m W^2 - \sigma_u U V, \end{split}$$

where M represents the mean flow, V the amplitude of the streamwise vor-

³⁵S.C. Reddy, P.J. Schmid, J.S. Baggett, D.S. Henningson, "On stability of streamwise streaks and transition thresholds in plane channel flows," J. Fluid Mech. **365** (1998) 269-303.

³⁶S.J. Chapman, "Subcritical transition in channel flows," J. Fluid Mech. **451** (2002) 35–97.

 $^{^{37}\}mathrm{F.}$ Waleffe, "On a self-sustaining process in shear flows," Phys. Fluids 9 (1997) 883-900.

tices, U that of the streaks, and W a supplementary variable accounting for three-dimensional features of the perturbed flow. Constants κ describe the viscous effects and coefficients σ are coupling constants arising from the NS advection term. Their values are such that the kinetic energy is conserved in the sense of Exercise 2.5.2. In addition to its base state (M = 1, U = V = W = 0), the model has several nontrivial states for R sufficiently large, representing possible operating points of the cycle mentioned above. In particular, in the perturbation equations around the base state, the equation for U contains a term in V (from the linearization of the term $\sigma_u MV$) that accounts for the growth of the streak component induced by the streamwise vortices at the origin of transient energy growth. The boundary of the attraction basin of the base state has been studied in detail for this system and complex transients reminiscent of what is observed in laboratory experiments below the global stability threshold have been shown to exist.³⁸ More realistic modelings are the subject of ongoing work.

In the following we will no longer consider the instability mechanisms at the origin of unsteadiness in turbulent flows but rather focus on the statistical aspects of the flows that develop beyond the transitional regime.

6.4 Exercises

6.4.1 Velocity-vorticity perturbation equations

Starting from Equations (6.17-6.20) for the perturbation, show that the pressure fulfills a Poisson equation:

$$\nabla^2 p = -2v\partial_y U \tag{6.52}$$

by computing the divergence of the momentum equations and using the continuity equation. Applying the three-dimensional Laplacian to the equation for v, next eliminate the pressure variable thanks to (6.52) and show that this yields:

$$(\partial_t + U\partial_x)\nabla^2 v - \partial_{yy}U\partial_x v = R^{-1}\nabla^4 v.$$
(6.53)

Finally, by cross differentiation and subtraction of equations for u and w show that the equation for $\Omega_y = \partial_z u - \partial_x w$ reads

$$(\partial_t + U\partial_x)\Omega_y - R^{-1}\nabla^2\Omega_y = -\partial_y U\partial_z v.$$
(6.54)

³⁸O. Dauchot, N. Vioujard, "Phase space analysis of a dynamical model for subcritical transition to turbulence in plane Couette flow," Eur. Phys. J. B **14** (2000) 377–381.

Notice that (6.53) is formally identical to the Orr-Sommerfeld equation (6.30) and is closed for v, while (6.54), called the *Squire equation*, is not closed for Ω_y but contains a forcing term in v.

Show that the no-slip condition u = v = w = 0 at a solid wall implies the boundary conditions $v = \partial_y v = 0$ and $\Omega_y = 0$.

6.4.2 Derivation of the Squire theorem

The theorem states that for a one-directional plane flow $\mathbf{v}_0 = U(y)\hat{\mathbf{x}}$ (see Figure 6.1a), the most dangerous perturbations are two-dimensional, *i.e.* without spanwise velocity component and independent of the transverse coordinate z.

1) Write down the linearized NS equations (6.17–6.20) for a normal mode taken as

$$[\mathbf{v}, p](x, y, z, t) = [\bar{\mathbf{v}}, \bar{p}](y) \exp[i(k_x(x - ct) + ik_z z)].$$

2) Notice the symmetrical role of $k_x u$ and $k_z w$ in the continuity equation and derive the equation governing $k_x u + k_z w$ by appropriate combinations of the equations for u and w.

3) Check that the equations for $k_x u + k_z w$ and v are those of a twodimensional problem in x and y (6.24-6.26) for some wavevector \tilde{k} and a Reynolds number \tilde{R} to be identified.

4) Conclude the argument by comparing the critical Reynolds numbers corresponding to a three-dimensional mode (k_x, k_y) and the associated twodimensional mode $(\tilde{k}, 0)$ in the viscous case, and the growth rates of these modes in the inviscid case.

6.4.3 Derivation of Fjørtoft's theorem

Rayleigh's theorem is obtained by working with the imaginary part of equation (6.34). Fjørtoft's improvement is gained by manipulating its real part.

1) Use Rayleigh's theorem to show that the real part of the phase velocity c_r can be replaced by any real speed \overline{U} in the real part of (6.34).

2) Choose the speed $\overline{U} = U_s = U(y_s)$ at an inflection point $(y_s \text{ such that } U''(y_s) = 0)$ as a reference speed and expand $U(y) - U_s$ and U'' in Taylor series in the neighborhood of y_s .

3) Insert these expansions in the identity for the real part of (6.34) and,

assuming that the velocity profile U(y) is monotonic with a single inflection point, derive a necessary condition for instability by the argument used to obtain Rayleigh's criterion. Express the result by noticing that U'(y) is the opposite of the base flow vorticity, $(\nabla \times \mathbf{v})_z = \partial_x v_y - \partial_y v_x$, Fig. 6.4.

6.4.4 Stability of some linear-by-part velocity profiles

Instability modes of profiles in the form (6.36-6.39) are searched here by direct computation. (For unbounded domains, the boundary conditions at infinity are obtained by expressing that the solutions do not diverge exponentially.)

6.4.4.1 Derivation of matching conditions (6.38) and (6.39)

1) Express the continuity of pressure from equation (6.24) for the streamwise velocity component u at discontinuity points of U and/or U' and derive condition (6.38).

2) Write down the equation governing the motion of a material point with cross-stream coordinate Y belonging to the interface between the two flows, first in full generality and next for a normal mode (*i.e.*, for $Y = \tilde{Y} \exp(ik(x - ct))$). Noticing that by definition of the cross-stream velocity component one has dY/dt = v, obtain \tilde{Y} as a function of $\tilde{\psi}$ and derive the matching condition (6.39).

6.4.4.2 Kelvin-Helmholtz instability and mixing layer

1) Find the dispersion relation of normal modes for a confined mixing layer with velocity profile:

$$egin{aligned} U(y) &= U_1 & ext{for} & 0 < y < y_{ ext{max}}\,, \ U(y) &= U_2 & ext{for} & y_{ ext{min}} < y < 0\,. \end{aligned}$$

where $y_{\min} < 0$ and $y_{\max} > 0$ are the cross-stream bounds of the fluid vein. 2) Same question for the smoothed velocity profile:

Take $y_{\min} = -y_{\max}$. Consider first y_{\max} large and next $y_{\max} > 1$ but finite.

3) Same question for the monotonic profile defined as

$$U(y)=lpha y+rac{1}{2}(1-lpha)ig(|y+1|-|y-1|ig)$$

on the interval [-b;b], b > 1 and discuss the solutions according to the value of α .

- a) Show that, for all α , modes with $k \to \infty$ are not unstable.
- b) Show that there cannot be unstable modes when $\alpha > 1$.

c) For $\alpha < 1$, show that there exist unstable modes when $k \to 0$ and $\alpha \approx 1$ (the complete dispersion relation should be determined outside this neighborhood). Interpret these results in view of Fjørtoft's theorem.

6.4.4.3 Stability of the plane jet

The base velocity profile of a plane jet (with speed U = 1) of some inviscid fluid entering a large vessel of the same fluid at rest can be taken in the form

$$egin{aligned} U(y) &= 0 & ext{ for } y < -1\,, \ U(y) &= 1 & ext{ for } -1 < y < 1\,, \ U(y) &= 0 & ext{ for } 1 < y\,. \end{aligned}$$

1) Compute the dispersion relation while taking care of the parity of the solutions in the cross-stream coordinate y.

2) Discuss the solutions and explain why the even mode is called 'sinuous' and the odd mode 'varicose' (cf. Fig. 6.29).



Fig. 6.29 Streamlines of the unstable modes of the plane jet.

6.4.5 Convective and absolute instabilities in the CGL equation

Consider the linearized Ginzburg-Landau equation in the following form:

$$\partial_t A - V \partial_x A = rA + (1 + i\alpha) \partial_{x^2} A$$

where A is the amplitude of the envelope of the primary waves generated by some instability mechanism and r is the corresponding control parameter, diffusion is normalized to 1 and α accounts for the dispersion of the waves (see Chapter 5, p. 198). A supplementary term $V\partial_x A$ has been introduced to mimic the effects of an average flow at some speed V on the pattern.

1) Determine the dispersion relation of plane waves solutions in the form $A \propto \exp(i(kx - \omega t))$ and derive their growth rate. When are they unstable according to the definition and what is their (real) phase speed?

2) When $V \neq 0$, the question arises of the convective or absolute nature of the instability. Compute the group velocity of plane waves $v_{\rm g} = \partial \omega / \partial k$, then the complex wavevector $k^{(0)}$ for which $v_{\rm g} = 0$, and find its growth rate $\omega_{\rm i}^{(0)}$ as a function of r and V.

3) Conclude by applying the Briggs-Bers criterion, and discuss the respective roles of the instability mechanism (through r) and the advection (through V) on the absolute/convective character of the instability.

This page intentionally left blank

Chapter 7

Developed Turbulence

In the previous chapters, we focused our attention on the transition to turbulence in closed systems and then in open flows. This problem was essentially studied from the viewpoint of mechanisms and associated thresholds for chosen idealistic base flows. However when one tries to evaluate the values of the control parameters of systems at hand, one finds that, most of the time they are far beyond the thresholds determined theoretically. This is in particular the case of most flows of geophysical or engineering interest (flows in tubes, around vehicles or obstacles, combustion, *etc.*) while the opposite situation, laminar flow, is quite exceptional (highly viscous fluids, lubrication, flows in capillaries of porous media, *etc.*)

From the study of instability mechanisms, we can expect at most hints on the *details* of how things evolve *locally* and *instantaneously* in a turbulent flow. But this "microscopic" account, as useful as it can be to the understanding of irregularities and chaos in the flow, must be replaced by a statistical approach able to deal with their "macroscopic" regularities, in much the same way as, in gases, collisions between molecules lose interest when thermodynamics is at stake. As we will see this analogy is fruitful but must not be pushed too far. We indeed have to stress immediately that turbulence is not a property of a fluid (in some new state of matter) but of its *motion* at large scales when compared to molecular dimensions.

As far as applications are concerned, turbulence is first of all characterized by strong *mixing* properties due to intense velocity fluctuations over a wide range of scales (§7.1). One of the objectives of the statistical theory to be built is thus to isolate the mean flow from the fluctuations (§7.2). Turbulent mixing implies statistical smoothing that is tempting to understand as a diffusion. As current experience tells us, it is more efficient by several orders of magnitude than molecular diffusion (sugar in a stirred cup of coffee). Though it is not well founded theoretically, the notion of *turbulent* diffusivity is then introduced ($\S7.3$) and we illustrate its use in the example of a fully turbulent boundary layer to predict the shape of the mean flow. We conclude the chapter by hinting at why and how going beyond this elementary approach to deal with situations of practical interest.

In this chapter we are essentially interested in completing the overall perspective centered on the dynamics of complex systems (mostly involving flowing fluids) by giving a few ideas of what happens when a hierarchy of interacting structures is involved, and developed turbulence should just be taken as an example in continuation with what precedes. However, this chapter just touches on a wide subject, many expositions of which can be found in the literature by more qualified authors. Here, let us mention a few books and defer to the corresponding part of the general bibliography for more references. [Tennekes and Lumley (1972)] or [Lesieur (1997)] may first serve as thorough introductions. The monograph [Frisch (2001)], in the perspective of Kolmogorov's work, has a more theoretical flavor. Finally, [Pope (2000)] brings a particularly clear and interesting light on modeling and numerical issues.

7.1 Scales in Developed Turbulence

Let us come back to fundamental characteristics of developed turbulence and the different spatio-temporal scales involved. Their origin has been briefly and schematically illustrated in Figure 1.8, p. 18. Here we reexamine them in a more quantitative way.

7.1.1 Production scale

Having noticed that turbulent flows develop at high Reynolds numbers, let us try to be a little more precise and call ℓ_0 and \tilde{v}_0 the external characteristics of a given flow. Subscript '0' here indicates the scale of motions at the starting point of the hierarchy evoked above and generated by the nonlinear advection term $\mathbf{v} \cdot \nabla \mathbf{v}$. Tildes are attached to fluctuating quantities, the separation of the mean flow from the fluctuations will be discussed later.

Let \tilde{v}_0 measure the order of magnitude of velocity fluctuations and ℓ_0 be the typical size of eddies produced by the instabilities leading to local chaos. The relation between these scales and the apparent external characteristics of the flow may not be immediate. For example in grid turbulence (Fig. 7.1),



Fig. 7.1 Turbulent flow behind a grid: From production to decay. After a picture in [van Dyke (1982)].

 ℓ_0 may a priori depend on the diameter of the bars, their wakes producing the eddies, on the distance between the bars which moves the point where different wakes merge to produce a homogeneous turbulent flow, as well as on the typical diameter L of the channel that undoubtedly gives an upper bound to the size of the eddies present in the flow. For its own part, \tilde{v}_0 should be related to the transverse velocities generated by the vortices shed by the bars, which in turn scale with the average speed of the flow U. Some uncertainty thus exists about the value of the quantities entering the definition of the Reynolds number $R_0 = \tilde{v}_0 \ell_0 / \nu$. This is what will later encourage us, on p. 278, to define a Reynolds number conventionally noted R_{λ} and based on a length called the *Taylor micro-scale*, which can be derived from experimental measurements.

Let us notice that the hierarchy of scales gets wider as turbulence gets more developed. The multiplicative process from which it results will be analyzed in a logarithmic perspective that will not be sensitive to the starting point (ℓ_0, \tilde{v}_0) . Even if it differs from the naive nominal Reynolds computed from U and L by some unknown proportionality factor, we will assume that the R_0 built on the production scale (ℓ_0, \tilde{v}_0) and the fluid's kinematic viscosity ν is always very large compared to one (powers of 10!) in situations of developed turbulence. This assumption dismisses an intermediate regime called *soft turbulence* still too close to the transitional regime and where the nature of the concrete mechanisms that lead to local chaos cannot be ignored, a regime we alluded to in the case of turbulent convection in §3.2.5 (see especially Figure 3.14, p. 103).
At the production scales, the viscous relaxation time of velocity fluctuations \tilde{v}_0 is large when compared to their turn-over time $\tau_0 = \ell_0/\tilde{v}_0$, since we have $\tau_{\mathbf{v}} = \ell_0^2/\nu$, so that the ratio $\tau_{\mathbf{v}}/\tau_0 = R_0$ is by assumption very large. Viscous dissipation can thus be neglected, leaving us just with the nonlinear advection term $\mathbf{v} \cdot \nabla \mathbf{v}$. As already discussed in Chapter 1, p. 19, when operating on eddies of typical size $k_0 \sim 1/\ell_0$, it tends to produce eddies of typical size $2k_0 \sim \ell_0/2$. This implies an energy transfer toward scales $\ell < \ell_0$ that should be independent of viscosity.

Following a well-established tradition, the energy flux from large to small scales is denoted ϵ . The fundamental assumption that allows us to estimate its value is of dimensional nature (see the exercises). It consists of saying that during a turn-over time τ_0 of eddies with size ℓ_0 , a fraction (fixed on average) of the kinetic energy *per* unit mass present in these eddies $\sim \tilde{v}_0^2$ is transfered to smaller ones:

$$\epsilon \sim \tilde{v}_0^2 / \tau_0 = \tilde{v}_0^3 / \ell_0 \,.$$
(7.1)

7.1.2 Inertial scales and Kolmogorov spectrum

Since the advection term produces scales of smaller and smaller sizes, of order ℓ_0 , $\ell_0/2$, $\ell_0/4$... let us examine its effects on an eddy of size $\ell \ll \ell_0$ and velocity \tilde{v}_ℓ , with turn-over time $\tau_\ell = \ell/\tilde{v}_\ell$. The corresponding Reynolds number reads $R_\ell = \tilde{v}_\ell \ell/\nu$ and, even if $\ell \ll \ell_0$, as long as $R_\ell \gg 1$, the corresponding motions carry the energy and transfer part of it but have no time to dissipate it into heat, hence their name of *inertial* scales. As the scale division process is iterated, correlations among eddies are expected to decrease, so that it can easily be accepted that fluctuations at inertial scales are *homogeneous* and *isotropic*, that is to say invariant by translation and rotation, see Figure 7.2.

At steady state the energy cannot accumulate in some given scale. Accordingly, the amount that is extracted *per* unit time from larger scales has to be transfered to smaller ones by the nonlinear advection term. Assuming that the instantaneous transfer does not fluctuate and is equal to ϵ all along the cascade, *i.e.* independent of ℓ , one can estimate the velocity fluctuation \tilde{v}_{ℓ} using the argument leading to (7.1) for scale ℓ_0 , which gives:

$$\epsilon \sim \tilde{v}_\ell^2 / \tau_\ell = \tilde{v}_\ell^3 / \ell \,,$$

from which \tilde{v}_{ℓ} can be derived and thus the Reynolds number at the corre-



Fig. 7.2 Sufficiently far from the production region, here a grid, one can consider that turbulence is locally homogeneous and isotropic when the Reynolds number is large enough. The *homogeneity* and *isotropy* concepts have to do with the statistical invariance of the flow with respects to translation and rotation, as suggested by the arrows supposed to move the circled region. After a picture in [van Dyke (1982)].

sponding scale:

$$\tilde{v}_{\ell} \sim \tilde{v}_0 \left(\ell/\ell_0\right)^{1/3} = (\epsilon \ell)^{1/3}$$
(7.2)

and

$$R_{\ell} = \frac{\tilde{\nu}\ell}{\nu} = \frac{\tilde{\nu}_0(\ell/\ell_0)^{1/3}\ell}{\nu} = \frac{\tilde{\nu}_0\ell_0}{\nu} \frac{\ell}{\ell_0} \left[\frac{\ell}{\ell_0}\right]^{1/3} = R_0 \left(\ell/\ell_0\right)^{4/3}$$
(7.3)

This energy transfer through the *inertial cascade* can continue as long as R_{ℓ} stays much larger than one, so that the neglect of the viscous dissipation remains legitimate (see Figure 7.3).

Turbulence in the inertial regime is usually characterized by its energy spectrum E(k) defined through $K = \int_0^\infty E(k) dk$, where quantity K is the total kinetic energy *per* unit mass contained in the flow. Splitting the k-space in concentric shells, one can write

$$K = \sum_p K_p \,,$$

where p labels scale ℓ_p with wavevector $k_p = 2\pi/\ell_p$. The energy spectrum is homogeneous to an energy *per* unit wavevector (and *per* unit mass). On dimensional grounds this gives:

$$K_p = k_p E(k_p) \sim \tilde{v}_{\ell_p}^2 , \qquad (7.4)$$



Fig. 7.3 Inertial cascade in wavevector space $(k = 2\pi/\ell)$ from production to dissipation.

so that using (7.2) one gets:

$$K_p \sim ilde{v}_0^2 (\ell_p/\ell_0)^{2/3}$$
 .

Coming back to k-space, omitting index p, and substituting this expression in relation (7.4) one obtains:

$$E(k) = C\epsilon^{2/3}k^{-5/3}.$$
(7.5)

This is the famous Kolmogorov spectrum. Constant $C \simeq 1.5$ from empirical adjustment with experiments) is supposed to be universal.

In experiments, what is usually recorded is the time series of fluctuations at a given point rather than a chart of the whole fluctuation field at a given time. In order to pass from wavevectors to frequencies, it is sufficient to observe that the time dependence of a quantity associated with a small scale contains a trivial part due to the transport by the large scale motion. This is the essence of Taylor's *frozen turbulence hypothesis*. Especially in wind or water tunnels, with average speed V at the location of measurement, one gets $\omega = Vk$ and thus $d\omega = Vdk$, so that, in k as well as in ω , a power law with exponent -5/3 is expected, as illustrated in Figure 7.4.

7.1.3 Dissipation scales

The turn-over time of eddies τ_{ℓ} can be obtained from $\epsilon \sim \tilde{v}_{\ell}^2/\tau_{\ell} \sim (\ell/\tau_{\ell})^2/\tau_{\ell} \sim \ell^2/\tau_{\ell}^3$ which yields $\tau_{\ell} \sim (\ell^2/\epsilon)^{1/3}$, whereas the viscous diffusion time is still given by $\tau_{\rm v} \sim \ell^2/\nu$. As long as the first one is much shorter than the second one, viscosity has no sufficient time to smooth out



Fig. 7.4 Kinetic energy $E(\omega)$ contained in the longitudinal velocity fluctuations in the ONERA-S1 wind tunnel in Modane, France. Courtesy Y. Gagne (PhD Dissertation, Grenoble University, 1987).

fluctuations and inertial effects dominate. This is the regime considered up to now. But, as the cascade proceeds, τ_{ℓ} decreases as $\ell^{2/3}$ while $\tau_{\rm v}$, initially larger, decreases much more rapidly as ℓ^2 . Viscosity ceases to be negligible when these two times have the same order of magnitude. This leads directly to the definition of the Kolmogorov dissipation scale $\ell_{\rm K}$:

$$\ell_{
m K}^2/\epsilon = \left[\ell_{
m K}^2/
u
ight]^3$$

or

$$\ell_{\rm K} = \left(\nu^3/\epsilon\right)^{1/4} = \left(\nu^3 \ell_0/\tilde{v}_0^3\right)^{1/4}.$$
(7.6)

see Figure 7.3. It is easily checked that the Reynolds number corresponding to this scale is of the order of one, so that the different regimes match correctly: imposing $R = R_{\rm K} = 1$ in (7.3) for the Reynolds number in the inertial range gives the same result that can be rewritten as

$$\ell_{\rm K}/\ell_0 = R_0^{-3/4} \,. \tag{7.7}$$

This leads us to point out that the difference between similar turbulent flows manifests itself only at the smallest scales (exercise 7.5.3). The film [Stewart (1972)] illustrates this feature using cinema *special effects* as an example: the eye is mostly sensitive to large structures and is easily mistaken about the order of magnitude of the Reynolds number, which allows specialists to blast models rather than full-scale film sets.

Structures with sizes $\ell \ll \ell_K$, smoothed by dissipation, are completely enslaved to larger scales. In developed turbulence, eddies sufficiently inde-

pendent to act as degrees of freedom must be at least as large as $\ell_{\rm K}$. The number of such eddies that can be piled up in a volume of order ℓ_0^3 is thus:

$$N_{\rm turb} \propto \left(\ell_0/\ell_{\rm K}\right)^3 \sim R^{9/4}\,,\tag{7.8}$$

which is Landau's estimate of the number of degrees of freedom in developed turbulence.

7.1.4 Remarks

7.1.4.1 Taylor micro-scale

Starting from the dissipation rate ϵ and an estimate of the intensity of large scale velocity fluctuations \tilde{v}_0 , one can define another characteristic length λ called the *Taylor micro-scale*. The order of magnitude of velocity gradients is indeed given by $\nabla \tilde{v} \sim \tilde{v}_0 / \lambda$. The dissipation is homogeneous to $[\nu] [\nabla \tilde{v}]^2$, hence $\epsilon = \nu \tilde{v}_0^2 / \lambda^2$, up to numerical factors, so that:

$$\lambda/\ell_0=R_0^{-1/2}\,,$$

by substitution. The Reynolds number R_{λ} is then defined as $R_{\lambda} = \tilde{v}_0 \lambda / \nu$, so that:

$$R_{\lambda}=R_0^{1/2}$$
 .

On the other hand, comparing λ with $\ell_{\rm K}$ one gets:

$$\lambda/\ell_{
m K} \propto R_0^{1/4} \propto R_\lambda^{1/2}$$

Taking the square root to pass from R_0 to R_{λ} leads to less impressive numbers than estimates from primitive quantities. In practice, R_{λ} is also a less subjective quantity since λ and \tilde{v}_0 can both be derived from experiments by separate measurements of the root mean square value at a point of the velocity fluctuations (\tilde{v}_0) and of the velocity gradient fluctuations (\tilde{v}_0/λ), which makes it appealing to characterize the turbulence level empirically.

7.1.4.2 Decaying turbulence

To conclude this section, let us insist on the fact that, without an energy supply sustaining it, turbulence ultimately decays since its develop in a dissipative medium. In the absence of permanent forcing, a flow starting from a given set of turbulent initial conditions always returns to the rest state at the end of a long *turbulent transient* (Exercise 7.5.4). For open flows such as those in Figure 7.1 and 7.2, this happens naturally as the fluid is followed downstream. Close to the grid, turbulence is intense but it decays while being advected since no further perturbation is introduced in the flow. Since small structures are more heavily damped than large ones, they disappear shortly and only large eddies remain far downstream.

7.2 Mean Flow and Fluctuations

7.2.1 Statistical approach

The turbulent flow now being conceived as a stochastic process, it becomes necessary to specify its statistical characteristics. The nature of the corresponding mathematical theory rests on the definition of probability distributions representing *statistical ensembles*, that is to say a supposedly infinite collection of systems submitted to the same constraints, prepared in the same way, and among which a given observed state is drawn "at random." In this context, ensemble averages are defined as *ensemble average*:

$$\langle Y \rangle = \sum_{i} Y_{i} \varpi_{i} , \qquad (7.9)$$

where Y_i denotes the value of observable Y in state *i* with probability ϖ_i , which is determined as the fraction of systems in the ensemble that are in state *i*, the specific flow configuration realized at the considered time $(\varpi_i = \lim_{N \to \infty} N_i/N)$. Higher order statistical moments are defined in a similar way. Here the nature of the observable Y is not specified, this can be for example the value of a given velocity component at some particular location. What matters is our ability to prepare the appropriate ensemble since a large number of similar experiments should be realized in principle. This might be less difficult in numerical experiments than in the laboratory despite the fact that N is anyway supposed to tend to infinity. Anyway, the statistical approach using ensembles remains mostly a (good) theoretical concept to which it is advisable to oppose an empirical viewpoint resting on measurements taken on a single specific realization that has developed in the course of time. *Time averages* are defined as:

$$\overline{Y}(t) = \lim_{T \to \infty} \frac{1}{T} \int_{t}^{t+T} Y(t') \,\mathrm{d}t' \,, \tag{7.10}$$

where Y is the considered observable, t the current time, and T a finite but long duration that in principle should be made arbitrarily large. According to this viewpoint, it is assumed that during a typical experiment the flow explores all the accessible configurations, weighted by the fraction of time it spends in a given state (which finds its foundation in the instability of individual trajectories for a non-predictable chaotic system, as already discussed in Chapter 4, $\S4.3$).

The regime is *stationary* when the so-defined average quantity $\overline{Y}(t)$ converges toward a value independent of t as T increases. When the flow depends on time, \overline{Y} defines a sliding average at fixed T. This definition remains satisfactory provided that T can be kept sufficiently small when compared to the typical evolution time of the flow while being long enough with respect to time scales of the turbulent fluctuations. The theoretical prediction of the flow's statistical properties takes for granted the equivalence of the two types of calculation:

ensemble average \equiv time average

(*ergodic* hypothesis). In what follows we will consider ensemble averages (7.9) while not forgetting that they will be evaluated using time averages (7.10), which is reasonable in the stationary case but may raise difficulties otherwise.

7.2.2 Reynolds averaged Navier-Stokes (RANS) equations

Let us apply the ensemble averaging procedure defined above to the continuity and Navier-Stokes equations for an incompressible flow. For the sake of compactness, here and as often as possible in the rest of the chapter, equations will be written using the convention of implicit summation over repeated subscripts and simplified notations: $\partial_{x_j} \equiv \partial_j$, j = 1, 2, 3 with $x_1 = x$, $x_2 = y$, $x_3 = z$, so that $\partial_j v_j \equiv \sum_{j=1}^3 \partial_j v_j$, and so on. However, when the specification of coordinate axes will be needed, the velocity components will recover the names they had in the previous chapter, *i.e.* u for x_1 (streamwise), v for x_2 (cross-stream), and w for x_3 (spanwise).

The primitive equations then read:

$$\partial_j v_j = 0, \qquad (7.11)$$

$$\rho(\partial_t + v_j \partial_j) v_i = \partial_j \sigma_{ij} , \qquad (7.12)$$

where σ is the stress tensor. Viscous effects are parameterized by the kinematic viscosity $\nu = \mu/\rho$ where the shear viscosity μ and the density ρ are supposed to be constant. For a Newtonian fluid we thus simply get:

$$\sigma_{ij} = -p \,\delta_{ij} + 2\mu s_{ij} \qquad \text{with} \qquad s_{ij} = \frac{1}{2} (\partial_i v_j + \partial_j v_i) \,. \tag{7.13}$$

Following Reynolds, we isolate the fluctuations from the mean flow by requiring that they average to zero. For convenience and in order to make the distinction more obvious, we denote all fluctuating quantities with tilded lower-case letters and mean values with upper-case letters:

$$\mathbf{v} = \mathbf{V} + \tilde{\mathbf{v}}, \qquad p = P + \tilde{p}. \tag{7.14}$$

By definition, we have:

$$\langle \mathbf{v} \rangle \equiv \mathbf{V}, \quad \langle p \rangle \equiv P \quad \text{and} \quad \langle \tilde{\mathbf{v}} \rangle = 0, \quad \langle \tilde{p} \rangle = 0.$$
 (7.15)

Averaging defined by (7.9) is a linear operation that commutes with differentiation, yielding:

$$\langle \partial_t \mathbf{v}
angle = \partial_t \langle \mathbf{v}
angle = \partial_t \mathbf{V}$$

 $(\equiv 0 \text{ in the stationary case})$ and

$$\langle \mathbf{\nabla} \mathbf{v}
angle = \mathbf{\nabla} \langle \mathbf{v}
angle = \mathbf{\nabla} \mathbf{V}$$

Averaging the continuity equation (7.11) leads to:

$$0 = \langle \partial_j v_j \rangle = \langle \partial_j (V_j + \tilde{v}_j) \rangle = \langle \partial_j V_j \rangle + \langle \partial_j \tilde{v}_j \rangle = \partial_j V_j + \partial_j \langle \tilde{v}_j \rangle$$

that is to say:

$$\partial_j V_j = 0 \tag{7.16}$$

and, by subtraction of (7.11):

$$\partial_j \tilde{v}_j = 0. \tag{7.17}$$

The introduction of (7.14) in the advection term $(V_j + \tilde{v}_j)\partial_j(V_i + \tilde{v}_i)$ gives four terms, $V_j\partial_jV_i$, $\tilde{v}_j\partial_jV_i$, $V_j\partial_j\tilde{v}_i$, and $\tilde{v}_j\partial_j\tilde{v}_i$. Only the first and the last terms survive to the averaging. Like for primitive quantities, we can define tensors associated to the average flow and the fluctuations:

$$\Sigma_{ij} = -P \,\delta_{ij} + 2\mu S_{ij} \qquad \text{with} \qquad S_{ij} = \frac{1}{2} \left(\partial_j V_i + \partial_i V_j \right) \tag{7.18}$$

and

$$\tilde{\sigma}_{ij} = -\tilde{p}\,\delta_{ij} + 2\mu\tilde{s}_{ij}$$
 with $\tilde{s}_{ij} = \frac{1}{2}\left(\partial_j\tilde{v}_i + \partial_i\tilde{v}_j\right)$. (7.19)

The averaged Navier–Stokes equation then reads:

$$ho(\partial_t V_i + V_j \partial_j V_i + \langle ilde v_j \partial_j ilde v_i
angle) = \partial_j \Sigma_{ij} \, .$$

The third term on the left hand side can be transformed as:

$$\langle \tilde{v}_j \partial_j \tilde{v}_i \rangle = \langle \partial_j (\tilde{v}_j \tilde{v}_i) - \tilde{v}_i \partial_j \tilde{v}_j \rangle = \partial_j \langle \tilde{v}_j \tilde{v}_i \rangle - \langle \tilde{v}_i (\partial_j \tilde{v}_j) \rangle = \partial_j \langle \tilde{v}_j \tilde{v}_i \rangle,$$

using continuity condition (7.17) to cancel the second term of the third member of this series of equivalences.

In component form, the *Reynolds stress tensor* is defined as^1

$$\tau_{ij} = -\rho \langle \tilde{v}_i \tilde{v}_j \rangle \,. \tag{7.20}$$

By identification with the mean stress tensor one then gets the mean flow equation in the form:

$$\rho(\partial_t + V_j \partial_j) V_i = \partial_j \left(-P \delta_{ij} + 2\mu S_{ij} + \tau_{ij} \right) . \tag{7.21}$$

It is immediately remarked that the set (7.16, 7.21) is not closed since the r.h.s. of (7.21) cannot be expressed entirely as a function of P and Vbut refers to the Reynolds stress, *i.e.* the average of a fluctuating term that remains to be evaluated. This would be achieved by writing down the equations for the fluctuations by subtracting (7.21) from (7.12), which gives:

$$\rho(\partial_t \tilde{v}_i + V_j \partial_j \tilde{v}_i + \tilde{v}_j \partial_j V_i + \tilde{v}_j \partial_j \tilde{v}_i) = \partial_j (-\tilde{p} + 2\mu \, \tilde{s}_{ij} - \tau_{ij}) \,, \qquad (7.22)$$

then by deriving equations for products $\tilde{v}_i \tilde{v}_j$ from $\partial_t (\tilde{v}_i \tilde{v}_j) = \tilde{v}_i \partial_t \tilde{v}_j + \tilde{v}_j \partial_t \tilde{v}_i$, adding equations such as (7.22) after multiplication by the appropriate fluctuating velocity components, then averaging and solving the so-obtained system.

These operations produce triple products in \tilde{v}_i , \tilde{v}_j , \tilde{v}_k , leaving us with triple correlations after averaging, and so on. We have therefore to face the well known *closure* problem of turbulence theory: new terms appear in equations at a given order that can be determined only by continuing the computation at higher orders. In a number of applications the problem is settled by imposing a *closure relation* among moments at a given order, that is to say by postulating relations that give the new unknown quantities in terms of already determined quantities.

Instead of considering the equations governing every component of the Reynolds stress tensor we will limit ourselves to the derivation of the equation for its trace which, up to a factor 1/2, is nothing but the average

¹The density ρ is still present in this definition so that, dimensionally, τ_{ij} is indeed a force *per* unit surface. Furthermore we do not follow a current practice that consists in omitting the minus sign from the definition, and thus in calling 'Reynolds stress' the quantity $\langle \tilde{v}_i \tilde{v}_j \rangle$.

fluctuating energy

$$ar{e} = rac{1}{2}
ho\langle ilde{\mathbf{v}}^2
angle = rac{1}{2}
ho\langle ilde{v}_i ilde{v}_i
angle = rac{1}{2} au_{ii}$$
 .

7.2.3 Energy exchanges in a turbulent flow

The distinction between mean flow and turbulent fluctuations leads us to examine how these two components of the flow exchange energy. Directly from the definitions, we get:

$$E_{
m tot} = rac{1}{2}
ho\langle {f v}^2
angle = rac{1}{2}
ho\langle ({f V}+ ilde{f v})^2
angle = rac{1}{2}
ho\left[\langle {f V}^2
angle + 2\langle {f V}\cdot ilde{f v}
angle + \langle ilde{f v}^2
angle
ight]$$

Noticing that $\langle \mathbf{V}^2 \rangle = \mathbf{V}^2$ and $\langle \mathbf{V} \cdot \tilde{\mathbf{v}} \rangle = \mathbf{V} \cdot \langle \tilde{\mathbf{v}} \rangle = 0$, we simply get:

$$E_{
m tot} = rac{1}{2}
ho {f V}^2 + rac{1}{2}
ho \langle ilde{f v}^2
angle = E + ilde{e}\,.$$

In order to compute the kinetic energy of the mean flow, Eq. (7.21) for component V_i is multiplied by V_i itself, and next the sum over the subscript *i* is performed. After a few manipulations and using definition (7.20) for the Reynolds stress, we obtain:

$$(\partial_t + V_j \partial_j)E = -\partial_j \left(PV_j - 2\mu S_{ij}V_i + \tau_{ij}V_i\right) - 2\mu S_{ij}S_{ij} - \tau_{ij}S_{ij}.$$
 (7.23)

This equation is valid locally. The first term on the r.h.s., $-\partial_j(\ldots)$, corresponds to a flux term that, in a global balance equation, can be integrated out as a surface term. Let us consider the other terms. The next one, $-2\mu S_{ij}S_{ij}$, obviously accounts for the energy dissipated by the mean flow through viscous friction. It is most often completely negligible since the large scale mean shear is small when compared to the shear from the small scale eddies. The most interesting term is the last one, $-\tau_{ij} S_{ij}$, which represents the work done by the mean flow on the fluctuations, and thus the energy transferred to the smaller scales.

Let us now perform the complementary computation for the fluctuating kinetic energy. From the equation for the total energy, after averaging and subtraction of the mean flow kinetic energy equation just determined we get:

$$(\partial_t + V_j \partial_j) \,\tilde{e} = -\partial_j \left(\langle \tilde{v}_j \tilde{p} \rangle - 2\mu \langle \tilde{s}_{ij} \tilde{v}_i \rangle + \frac{1}{2} \rho \langle \tilde{\mathbf{v}}^2 \tilde{v}_j \rangle \right) - 2\mu \langle \tilde{s}_{ij} \tilde{s}_{ij} \rangle + \tau_{ij} \, S_{ij}.$$

$$(7.24)$$

Discarding again the flux gradient term and considering what remains, one easily identifies the fluctuating viscous dissipation $-2\mu \langle \tilde{s}_{ij} \tilde{s}_{ij} \rangle$, which is a negative definite contribution and the turbulent energy production $+\tau_{ij}S_{ij}$, directly subtracted to the energy of the mean flow.

A stationary regime can be obtained only if these two terms balance each other, otherwise, the evolution is transient (decaying turbulence). We now turn to the determination of the mean flow.

7.3 Mean Flow and Effective Diffusion

7.3.1 Mixing length and eddy viscosity

As we have already pointed out, turbulence is characterized by diffusivities that are considerably augmented with respect to their microscopic counterparts, viscosity in Stokes' law, molecular diffusivity in Fick's law. It is however advisable to understand how the molecular properties arise in kinetic theory of gases.

On general grounds, the macroscopic diffusion equation for the density of some microscopic physical quantity Y reads

$$\partial_t Y = D_Y \partial_{x^2} Y \,,$$

where D_Y is the relevant diffusivity, homogeneous to $[L]^2[T]^{-1}$. It describes the smoothing of inhomegeneities by redistribution of that quantity as it is randomly transported by the microscopic agents (molecules). The diffusivity can be estimated through a dimensional argument in which microscopic times and lengths involved in their motions are introduced. The kinematic viscosity related to the redistribution of linear momentum is for example given by:

$$u \sim rac{\xi^2}{ au_{
m c}} = \xi \, ar c$$

in which ξ is the mean free path, τ_c the average time between collisions, and $\bar{c} = \xi/\tau_c$ the average speed of molecules at the considered temperature. A detailed computation would give $\nu = \xi \bar{c}/3$ since on average only one third of the molecules contribute to the transfer of momentum in the direction of the gradient of velocity. Furthermore, in a gas all molecular diffusivities have the same order of magnitude since the physical quantities are transported by the same agents, the molecules themselves. Hence, the Prandtl number, ratio of the kinematic viscosity (diffusivity of momentum) to the thermal diffusivity (for energy) is of order one. Things are different in viscous condensed media where heat diffusion is fast since it also involves vibration modes of a local pseudo-lattice. An analogous situation holds in mixtures where the diffusivity of a solute can be very slow when compared to viscous

damping when the solute's molecules are much bigger than the solvent's molecules (Schmidt number $S = \nu/D \gg 1$, see also Exercise 3.3.3 where the Lewis number $L = D/\kappa$ was defined).

The idea underlying the introduction of an *eddy viscosity* rests on a heuristic argument that simply replaces the thermal velocity \bar{c} by the typical *mixing velocity* \tilde{v}_{mix} , *e.g.* the root-mean-square of \tilde{v} at some relevant place, and the mean free path ξ by a corresponding length ℓ_{mix} , called the *mixing length* (Prandtl):

$$u_{\rm t} = ilde{v}_{\rm mix} \, \ell_{\rm mix}$$

This concept is useful but its use is delicate because its justification is purely dimensional: the specific values of its two ingredients are not specified a priori, by contrast with the case of gases where the thermal speed and the mean free path are unambiguously defined and where the latter remains usually extremely small when compared to the scale of the applied gradients. We already know in particular that, in turbulence, many eddies are active over a whole range of scales in turbulence, and that there is no real gap between the scales typical of the mean flow and those typical of turbulent fluctuations. This caveat being stated, below we elaborate a little more on the concept of eddy viscosity, noticing that the argument can sometimes be developed in terms of time scales rather than space scales (e.g. in Exercise 7.5.5).

Up to now we have done as if the scales (speed, length) were constant but most flows of practical interest develop downstream, boundary layers, jets, wakes, mixing layer.

In the case of a laminar flow controlled by the molecular viscosity, the width of the sheared region $\delta_{\rm m}$ in the cross-stream direction y depends on the streamwise coordinate x in a way correctly predicted by the dimensional argument in which, the advection $U\partial_x U$ is balanced by the viscous diffusion $\nu \partial_{y^2} U$ in the Navier–Stokes equations. For plane flows, this yields $U^2/X = \nu U/\delta_{\rm m}^2$, where X is the distance from the origin of the flow (leading edge of the plate for a boundary layer). Physically X/U can be understood as the advection time from this origin to the point of interest and $\delta_{\rm m}$ as the width of the region affected by viscous damping during that time.

When trying to apply the dimensional argument to a turbulent flow in the x direction with a mean flow profile U(y) at position X, the estimate of the advection time does not change (X/U) while the momentum fluctuations are no longer diffusively damped but rather advected by eddies with characteristic speed \tilde{v}_{mix} . This speed is not an equivalent of the well-defined thermal speed in a near-equilibrium gas, but on the contrary a scale-dependent quantity. A crude guess, to be refined in the Sec. 7.3.2 below, would be to assume $\tilde{v}_{\rm mix}$ constant and to evaluate the thickness of the layer affected by turbulence dimensionally as $\delta_t = \tilde{v}_{\rm mix}(X/U)$. Unfortunately, in flows developing downstream, this assumption is not tenable and that supposedly constant $\tilde{v}_{\rm mix}$ has to vary (slowly) with X. The widening of the turbulent layer cannot be estimated without pushing the argument further or taking into account experimental observations (see Sec. 7.4, later on). The behavior of $\delta_t/X = \tilde{v}_{\rm mix}/U$ indeed depends on the class of flow considered. In boundary layers $\tilde{v}_{\rm mix}/U \to 0$ whereas for jets and mixing layers $\tilde{v}_{\rm mix}/U \to C$ ste.

7.3.2 Application to the determination of the mean flow

In what follows we restrict ourselves to the case of a fully developed turbulent boundary layer along a plane wall in direction x at stationary state $(\partial_t U \equiv 0)$, in the absence of pressure gradient $(\partial_x P \equiv 0)$. We assume that we are far enough from the leading edge, so that its thickness δ_t , defined as the distance at which the speed is roughly that of the external flow, is very large, and the fluctuations have sufficient room to develop over a wide range of scales. We also neglect the thickening of the layer so that the mean flow V is reduced to its streamwise component $U\hat{\mathbf{x}}$.

From now on, we consider only quantities *per* unit mass, which means that Eqs. (7.21–7.24) have been divided by ρ , with the replacements $\mu \mapsto \nu$, $(P/\rho, \tilde{p}/\rho) \mapsto (P, \tilde{p})$, and that stresses or pressures are homogenous to velocities squared, especially the Reynolds stress defined by (7.20), $\tau_{ij} \mapsto -\langle \tilde{v}_i \tilde{v}_j \rangle$.

Close to the wall, the no-slip condition imposes that all the velocity components tend to zero. Accordingly the flow can be characterized by the mean shear at the wall $\partial_y U|_{y=0}$, or rather by the friction at the wall, a stress *per* unit mass denoted τ^* :

$$\tau^* = \nu \partial_y U|_{y=0} \,,$$

since the Reynolds stress does not contribute to the stress *at* the wall due to the cancellation of the transverse fluctuating velocity component there. Experimentally, it is not possible to evaluate τ^* using this relation since the mean flow profile is not measured with sufficient precision in the limit $y \to 0$ (a derivative has to be taken further, which does not improve the situation). However at very large Reynolds numbers this quantity is transferred to the

bulk without losses, so that it can be evaluated from the measurement of the mean flow profile and the Reynolds stresses farther from the wall. Anyway, this stress is dimensionally homogeneous to the square of a speed, called the *friction velocity* and here denoted \tilde{v}^* , so that we set:

$$\tau^* = (\tilde{v}^*)^2.$$

Using this speed and the kinematic viscosity ν one can construct a length unit y^* through:

$$y^* =
u / ilde{v}^*.$$

The wall units \tilde{v}^* and y^* then serve to rewrite the problem close to the wall in dimensionless variables:

$$y^+ = y/y^*$$
 and $U^+ = U/\tilde{v}^*$. (7.25)

In the immediate vicinity of the wall, $y^+ \sim 1$, in a region called the *viscous sublayer*, the mean velocity gradient is very large, so that the viscous contribution of the mean strain tensor to the mean stress tensor is not negligible (see later). But it is empirically observed that this gradient decreases very rapidly and that one can neglect it somewhat beyond the viscous sublayer. We now focus on the determination of the mean velocity profile in that region where the stress tensor is reduced to its Reynolds stress part τ_{ij} .

Modeling the Reynolds stress using an *eddy viscosity*, one obtains:

$$au_{xy} =
u_{\mathbf{t}}(y) \partial_y U$$
 .

The assumptions that have been made simplify the mean flow equation (7.21) considerably since one is left with:

$$\partial_y au_{xy} = 0$$
, hence: $u_{\mathbf{t}}(y) \partial_y U = \operatorname{Cst}$,

where the constant on the right hand side cannot be something other than the wall stress $\tau^* = (\tilde{v}^*)^2$ introduced earlier.

The variation of ν_t with y remains to be fixed. One then assumes that most of the turbulent dispersion at distance y from the wall is due to the largest possible eddies that can transfer momentum to the wall from that position (Fig. 7.5). Quantities \tilde{v}^* and y are the only ones we have at our disposal to construct ν_t . Therefore we get:

$$\nu_{\rm t}(y) = \chi \, y \, \tilde{v}^* \,, \tag{7.26}$$



Fig. 7.5 Fully developed boundary layer. Outside the viscous sublayer with thickness of the order of y^* , the most natural length scale of the eddies that disperse momentum efficiently is the distance y to the wall for dimensional reasons. The logarithmic mean profile follows.

where χ is a dimensionless proportionality coefficient called the *Kármán* constant. Integrating $(\chi y \, \tilde{v}^*) \, \partial_y U = (\tilde{v}^*)^2$, that simplifies as:

$$\partial_y U = \frac{1}{\chi} \frac{\tilde{v}^*}{y}, \qquad (7.27)$$

we obtain:

$$U(y) = \frac{\tilde{v}^*}{\chi} \ln(y) + \text{Cst}. \qquad (7.28)$$

At this stage it is useful to turn to wall units, which gives us the universal law:

$$U^{+}(y^{+}) = \frac{1}{\chi} \ln(y^{+}) + C. \qquad (7.29)$$

Kármán's logarithmic profile gives a satisfactory description of the bulk of the fully developed boundary layer, which justifies the assumptions made to obtain it *a posteriori*. Empirical fits against experiment give us:

 $\chi \simeq 0.41$ and $C \simeq 5$.

These constants are also supposed to be universal.

The logarithmic behavior predicted by (7.29) cannot be extrapolated indefinitely, neither for $y \to 0$ nor for $y \to \infty$ since the logarithm diverges in both limits. It is only valid in the internal part of the layer called the *inertial* layer, and sufficiently far beyond the viscous sublayer where we must come back to the full expression of the stress in (7.21) but this time,



Fig. 7.6 Mean flow profile in a fully developed turbulent boundary layer as it comes out of the mixing length theory.

neglect the contribution of the Reynolds stress. We then simply get:

$$u \partial_y U = \tau^*$$
,

which we integrate to obtain:

$$U(y) = (au^*/
u)y = [(ilde{v}^*)^2/
u]y = ilde{v}^*(ilde{v}^*y/
u) = ilde{v}^*(y/y^*)$$

(there is no integration constant since $U|_{y=0} = 0$). Turning to wall units this reads

$$U^+ = y^+ \, .$$

Experiments show that the log law is acceptable for $y^+ > 40$ and the linear law for $y^+ < 3$. The matching of these two variation laws is achieved in the region $3 < y^+ < 40$ called the *buffer layer*.

The logarithmic divergence of the profile for y large is equally unacceptable. The bulk log-layer flow has now to be matched with the exterior flow $U = U_{\infty}$ for $y > \delta_t$. In the region where this matching takes place, the velocity scale is still \tilde{v}^* (we have nothing else) but the length scale is now the thickness δ_t of the layer. Dimensionally, we must have

$$\partial_y U = rac{ ilde{v}^*}{\delta_{
m t}} f(y/\delta_{
m t})$$

where f is an unknown function of the dimensionless variable $\bar{y} = y/\delta_t$.

This expression of $\partial_y U$ in the external region must be matched with that given by (7.27), valid in the interior region. This implies $f(\bar{y}) \sim 1/\chi \bar{y}$ for $\bar{y} \ll 1$. The speed of the exterior flow being denoted as U_{∞} , the velocity in the external region of the layer is thus obtained from the integration of:

$$\partial_{ar{y}}U=rac{ ilde{v}^{*}}{\chiar{y}}\,,$$

that is:

$$U(ar{y}) = U_\infty + rac{ ilde{v}^*}{\chi}\log(ar{y})$$

(so that $U = U_{\infty}$ for $\bar{y} = 1$).

Since we are interested in the domain $y \leq \delta_t$, it is advisable to consider the velocity defect $[U_{\infty} - U(y)]$ normalized by \tilde{v}^* and to measure y using its natural unit δ_t , hence:

$$\Delta U^* = rac{U_\infty - U(y)}{ ilde v^*} = -rac{1}{\chi}\log(ar y)$$

The calculation can be adapted to other situations, e.g. the determination of the pressure head necessary to drive a fully turbulent flow in a plane channel. This can be done along the same lines by considering two turbulent boundary layers placed face to face, each with its viscous sublayer, buffer, inertial, and external layers, which implies sufficiently large Reynolds numbers. The essential assumption is the existence of a single velocity scale v^* . The argument breaks down when several scales are relevant.

7.4 Beyond the Elementary Approach

In this presentation of developed turbulence, we sticked to statistical characteristics of the flow at the lowest possible order, relying on the mean flow/fluctuations decomposition (in fact the root mean square of fluctuations). Things become much more complicated beyond this elementary level. For example if dimensional arguments give the essential of the Kolmogorov spectrum (7.5) that accounts for two-point correlations of the velocity field² The analysis of higher order correlations is more delicate since the *closure problem* cannot be escaped. A limitation comes from Landau's old objection to the Kolmogorov theory, according to which the dissipation rate *per* unit mass ϵ cannot be constant but rather fluctuates in time and

²See Chapter 4, §4.4.1, for the relation between Fourier spectra and correlations.

space, which is called *inertial range intermittency*. The excitation of eddies at a given scale is indeed sporadic and the energy is distributed at random (local chaos) between the different daughter-eddies (size $\ell/2$) generated by nonlinearities from the mother-eddy (size ℓ). In fact triads of wavevectors are implied, such that $\mathbf{k}_1 + \mathbf{k}_2 + \mathbf{k}_3 = 0$. Elementary arguments developed previously assume, *e.g.* $|\mathbf{k}_1| \sim |\mathbf{k}_2| \sim k$ and $|\mathbf{k}_3| \sim 2k$, but the combination also produces wavevectors such that $|\mathbf{k}_3| \ll k$: a transfer also takes place toward larger scales though the transfer toward smaller scales is statistically dominant, so that the simplest image of the energy cascade subsists.

The fine statistical characterization of turbulence rests on the study of velocity increments as functions of the distance:

$$\Delta_i v_j(r) = v_j(\mathbf{x} + r \hat{\mathbf{x}}_i) - v_j(\mathbf{x}) \,,$$

r being the distance measured in direction i indicated by the unit vector $\hat{\mathbf{x}}_i$, parallel (i = j) or perpendicular $(i \neq j)$ to the velocity component j considered. The analysis has to deal with statistical moments of the distribution of these increments:

$$D_n = \langle \Delta_i v_j^n \rangle.$$

The argument that leads to (7.2) would predict

$$D_n = C_n (\epsilon r)^{n/3} \,,$$

which is experimentally well verified for n = 2. For n = 3, the result is exact, as shown by Kolmogorov who finds $C_3 = 4/5$ (the '4/5 law'). When searching to adjust moments D_n to power laws r^{ζ_n} , one observes that experiments deviate from the simple prediction $\zeta_n = n/3$ due to the intermittency phenomenon mentioned earlier. This problem is reviewed in detail in [Frisch (2001)].

Other quantities defined in terms of gradients of the velocity field are also interesting:

$$M_n^{[ij]} = \frac{\langle (\partial_i \tilde{v}_j)^n \rangle}{\langle (\partial_i \tilde{v}_j)^2 \rangle^{n/2}} \,,$$

and most especially the *skewness* coefficient $S = M_3^{[ii]}$ and the *kurtosis* coefficient $K = M_4^{[ii]}$ of longitudinal gradients (i = j) that would be respectively equal to 0 and 3 if turbulence were a *bona fide* centered Gaussian random process.

7.4.1 Turbulence modeling

The approach followed above in order to determine the turbulent boundary layer mean profile rests on the intuitive concept of *eddy viscosity*. Trying to close the hierarchy of equations right at lowest level using appropriate extensions to this model seems both natural and desirable in order to deal also with other more complicated situations. An excellent introduction to that problem is [Pope (2000)] and a thorough discussion of models used in numerical fluid dynamics is [Wilcox (2000)].

According to Boussinesq (1897), Reynolds stresses can be modeled by means of an eddy viscosity assumption expressed as:

$$-\langle \tilde{v}_i \tilde{v}_j
angle + rac{2}{3} k \delta_{ij} = 2
u_{\mathbf{t}} S_{ij} , \qquad (7.30)$$

where $S_{ij} = \frac{1}{2}(\partial_i V_j + \partial_j V_i)$ is the mean strain tensor and where, according to common practice, k here denotes the fluctuating kinetic energy per unit mass.³ Reynolds averaged equations then read:

$$(\partial_t + V_j \partial_j) V_i = -\partial_i \left(P + \frac{2}{3}k \right) + \partial_j \left[2(\nu + \nu_t) S_{ij} \right].$$
(7.31)

Depending on the approach, the formulation can be complete or not, *i.e.* according to whether the expression of the eddy viscosity ν_t is totally (or partially) fixed from the outside (incomplete model), or derived from the mean flow properties through specific equations (complete model).

The simplest possible assumption consists in prescribing ν_t from the local characteristics of the external one-directional flow along x at position X: $\nu_t(X) = U(X)\ell_0(X)/R_t$, where ℓ_0 is some outer scale and R_t a turbulent Reynolds number. But this has a narrow range of applicability since it does not take the least account of the specificities of turbulence.

One step further, the mixing length theory used to treat the case of the boundary layer is a less trivial example of incomplete model founded on a clear physical assumption. Coming back to (7.26), p. 287, it was assumed that $\nu_t = \tilde{v}_{mix} \ell_{mix}$ where $\tilde{v}_{mix} = \tilde{v}$ and ℓ_{mix} are two quantities relative to the turbulent mixing. More generally, Prandtl suggested taking $\tilde{v}_{mix}/\ell_{mix} \propto |\partial_y U|$ for a mean flow U along x sheared along y, which is dimensionally correct though this proposal has no real physical grounds. We thus get:

$$u_{\mathbf{t}} = \ell_{\min}^2 |\partial_y U|,$$

³The risk of confusion of this quantity with a wavevector is small in the present context. The presence of term $\frac{2}{3}k\delta_{ij}$ in expression (7.30) guarantees that the trace of the tensor is zero, owing to the mean flow incompressibility condition, $\partial_i V_i = 0$.

that has been extended to general flows by Smagorinsky (1963) in the form

$$\nu_{\rm t} = \ell_{\rm mix}^2 (2S_{ij}S_{ij})^{1/2} \,. \tag{7.32}$$

The two formulations are easily checked to be equivalent when used in Prandtl's conditions: evaluating the second form with a mean strain tensor S_{ij} reduced to $S_{12} = S_{21} = \frac{1}{2}\partial_1 V_2$, one gets: $\nu_t = [2(S_{12}S_{12} + S_{21}S_{21})]^{1/2} = [4(\frac{1}{2}\partial_1 V_2)^2]^{1/2} = |\partial_1 V_2|.$

All this however does not make a complete model since an expression for $\ell_{\rm mix}$ is still requested. It is thus advisable to turn to complete models, still resting on the eddy viscosity assumption, but free of external specification of $v_{\rm mix}$ and/or $\ell_{\rm mix}$ on a case-by-case basis. These models can be developed at several levels depending on the number of additional dynamical equations closing the Reynolds averaged equations. From this viewpoint, mixing length models are zero-equation models (or 'algebraic' models). One-equation models determine the velocity scale $\tilde{v}_{\rm mix}$ from the turbulent kinetic energy k by setting $\tilde{v}_{\rm mix} \propto \sqrt{k}$. The latter quantity is then obtained from Eq. (7.24), with appropriately modeled unknown terms. This equation is rewritten here in the generic form:

$$(\partial_t + V_j \partial_j) k = \partial_j \mathcal{F}_j^{(k)} + \mathcal{P}^{(k)} - \mathcal{D}^{(k)} , \qquad (7.33)$$

where $\mathcal{F}^{(k)}$, $\mathcal{P}^{(k)}$ and $\mathcal{D}^{(k)}$ represent flux, production, and dissipation terms, respectively. The actual flux term is replaced by a conventional diffusion term:

$$oldsymbol{\mathcal{F}}^{(k)} = -\kappa_{ ext{t}}^{(k)} oldsymbol{
abla} k \, ,$$

where the turbulent diffusivity of energy κ_t is related to the eddy viscosity (turbulent diffusivity of momentum) ν_t by means of an effective Prandtl number traditionally noted σ_k , *i.e.* $\kappa_t = \nu_t/\sigma_k$. In full generality, the production term reads $\mathcal{P} = -\langle \tilde{v}_i \tilde{v}_j \rangle S_{ij}$. The eddy viscosity assumption turns it into:

$$\mathcal{P}^{(k)} = 2\nu_{\rm t} S_{ij} S_{ij} \,,$$

which is positive definite. The dissipation term, corresponding to the energy ϵ transferred from the large scales is modeled by (7.1), which yields:

$$\mathcal{D}^{(k)} = C_{\mathrm{D}} rac{k^{3/2}}{\ell_{\mathrm{mix}}}\,,$$

where $C_{\rm D}$ is one of the constants in the model. The presence of $\ell_{\rm mix}$ in that equation, as well as in the relation that defines the eddy viscosity, that is $\nu_{\rm t} \propto k^{1/2} \ell_{\rm mix}$, still makes this type of modeling incomplete. This limitation is raised by adding a supplementary equation.

The best known two-equation model is the k- ϵ model that collects all that precedes but, instead of adding an equation governing ℓ_{mix} , uses the fact that the missing length can be constructed from k and ϵ as $\ell_{\text{mix}} \propto k^{3/2}/\epsilon$, which is just Eq. (7.1) solved for ℓ given ϵ . The eddy viscosity is then taken as:

$$u_{
m t} = C_{\mu} rac{k^2}{\epsilon} \, ,$$

where C_{μ} is a constant. The equation for ϵ that closes the system is copied from (7.33):

$$(\partial_t + V_j \partial_j)\epsilon = \partial_j \mathcal{F}_j^{(\epsilon)} + \mathcal{P}^{(\epsilon)} - \mathcal{D}^{(\epsilon)}, \qquad (7.34)$$

whose different terms are modeled in a fully parallel way:

$${\cal F}^{(\epsilon)}=-\kappa^{(\epsilon)}_{
m t}oldsymbol{
abla}\epsilon\,,$$

with $\kappa_t^{(\epsilon)} = \nu_t / \sigma_{\epsilon}$, where σ_{ϵ} is the Prandtl number relative to the "diffusion" of ϵ . The production and dissipation terms are taken further in the form

$${\cal P}^{(\epsilon)}=C_{\epsilon 1}rac{{\cal P}^{(k)}\epsilon}{k}\,,\qquad {\cal D}^{(\epsilon)}=C_{\epsilon 2}rac{\epsilon^2}{k}\,,$$

which can be justified by noticing that quantity k/ϵ is homogeneous to a time τ and that a dimensionally consistent way to pass from (7.33) to (7.34) is by dividing (7.33) by this time and introducing phenomenological dimensionless constant to account for the differences between the processes governing the dynamics of k and ϵ . Constants appearing in the model, C_{μ} , $C_{\epsilon 1}$, $C_{\epsilon 2}$, σ_k and σ_{ϵ} are generally adjusted by demanding that the predictions in few test cases (decaying turbulence, log law in boundary layers, *etc.*) correspond to empirical data. To go further one should consult [Pope (2000)], Chapter 10.

7.4.2 Large eddy simulations

The variables introduced above refer to ensemble averages. However, quantitative information collected from the study of specific realizations (structure and temporal history) are at least as important as the properties of statistical ensembles ideally corresponding to these experiments. There is therefore a pressing need for numerical simulations of Navier–Stokes equations at large Reynolds numbers, in industrial or geophysical configurations for example. The evolution of small scales is of little interest, except that they have to be present to guarantee energy dissipation. On the contrary, large scale motions that carry the energy and achieve the mixing are of more interest. This suggests performing a *filtering* of the equations that eliminates small scales, as implemented in *large eddy simulations* ('LES'). The need for a filtering of primitive equations is clearly understood from Landau's estimate (7.8) of the number of degrees of freedom relevant to developed turbulence, which might open a pessimistic perspective: despite the present observed increase of computational power, Reynolds numbers accessible to direct numerical simulation of Navier–Stokes equations will, for long, stay modest (see Exercise 7.5.1). Considering spectral approaches to the simulation problem (cf. Appendix B, §B.2.2) will allow us to better understand why and how one can develop an alternative strategy.

In practice large scale eddies are reconstructed from a set of small wavevectors of the order of the inverse of ℓ_0 , the scale in which energy is injected in the flow by the main instability mechanism. Few steps in the cascade are necessary to reach complete disorder. These steps depend little on the precise value of the Reynolds number. Let us call⁴ $k_{\rm LE}$ the maximum wavevector necessary to account for large eddies (hence subscript 'LE'). One thus gets $k_{\rm LE} \propto k_0$, with a rather small proportionality factor, typically 2^n for n steps, with n = 3 or 4. In order of magnitude, the number of such modes is thus given as the volume of a sphere with radius $k_{\rm LE}$ in Fourier space: $N_{\rm LE} \propto k_{\rm LE}^3$. By contrast the value of $k_{\rm K}$, the wavevector associated to the Kolmogorov scale (7.6) marking the end of the inertial cascade, is strongly dependent on $R: k_{\rm K} \propto R^{3/4}$. The corresponding number of active modes, given by the Landau estimate $N_{\rm diss} \propto k_{\rm K}^3 \sim R^{9/4}$, is rapidly diverging. Let us denote $k_{\rm max}$ the largest wavevector included in the simulation (typically, $k_{\text{max}} = \pi/\Delta x$, where Δx measures the space resolution). In principle the injected energy will be correctly dissipated at the given value of R if $k_{\max} \ge k_{\rm K}$ and thus $N_{\max} \propto k_{\max}^3 \ge N_{\rm diss}$, which is impossible for the values of R of interest in applications.

In fact, the large scales will be correctly described provided that the energy is correctly transferred through the scale derived from $k_{\rm LE}$. Accordingly, simulating the whole range of scales between $k_{\rm LE}$ and $k_{\rm K}$ in the inertial cascade is necessary only if one does not know how to parame-

⁴From now on k no longer refers to the fluctuating kinetic energy but serves again to label wavevectors.

terize this transfer through $k \sim k_{\rm LE}$. The filtering is thus an operation by which numerous "useless" modes $(N_{\rm diss} - N_{\rm LE} \gg N_{\rm LE})$ are eliminated to the benefit of few modes below $k_{\rm LE}$ "useful" to the description of large scale motions. A detailed operational presentation of the method is out of question within the limited scope of these notes —for this one may consult [Pope (2000)], Chapter 13— but just give a qualitative idea of it, resting on, and extending the discussions in the previous subsection.

The ensemble averaging leads to a splitting of the hydrodynamic fields into two components (7.14) that fulfill (7.15). Here the flow is similarly decomposed into filtered quantities $(\mathbf{v}, p)^{\mathrm{f}}$ (superscript 'f') and residual quantities $(\mathbf{v}, p)^{\mathrm{r}}$ (superscript 'r'). The filtered velocity field in physical space is obtained from the *convolution* of the primitive field with a filter function G_{Δ} :

$$\mathbf{v}^{\mathrm{f}}(\mathbf{x},t) = \int G_{\Delta}(\mathbf{y},\mathbf{x})\mathbf{v}(\mathbf{x}-\mathbf{y},t) \,\mathrm{d}^{3}\mathbf{y}, \qquad (7.35)$$

where $G_{\Delta}(\mathbf{y}, \mathbf{x})$ is some appropriately normalized integral kernel $(\int G_{\Delta}(\mathbf{y}, \mathbf{x}) d^3 \mathbf{y} = 1)$ that is localized in space over a region of typical size Δ . The filter is *homogeneous* if it does not depend on position \mathbf{x} which is usually the case except close to a wall). Furthermore it is *isotropic* if it depends only on the length $|\mathbf{y}|$ and not the orientation of \mathbf{y} . The simplest homogeneous isotropic filter corresponds to the averaging over a sphere with diameter Δ (the averaging over a cube of side Δ would not be isotropic since the result would depend on the local flow direction with respect to the coordinate axes). Smoother filters, *e.g.* Gaussian, are usually considered.

Inspired by the initial remark about modes in Fourier space, one can also work in spectral space and perform the filtering in that space where the convolution product of two functions in physical space is just the ordinary product of their Fourier transforms. The Fourier transform of the filter function in physical space is called the *transfer function*. The transfer function of a Gaussian function is another Gaussian function, whereas the transfer function of a square window displays damped oscillatory wings in Fourier space. Conversely, the filter function of the low-pass square filter in spectral space also displays wings in physical space. The simplest spectral filtering is indeed achieved by such low-lass filtering that amounts to projecting the dynamics onto the subspace of Fourier modes such that $k \leq k_{\Delta}$, where k_{Δ} is the *cut-off* wavevector (*i.e.* truncating the Fourier decomposition of the fields beyond k_{Δ}). The residual velocity field is by definition the difference between the primitive field and the filtered field:

$$\mathbf{v}^{\mathbf{r}}(\mathbf{x},t) = \mathbf{v}(\mathbf{x},t) - \mathbf{v}^{\mathbf{f}}(\mathbf{x},t).$$

The filtering is somewhat less simple than the ensemble averaging. First it is to be noted that except when it comes to a projection over a set of modes, and contrary to averaging (see (7.15), p. 281), filtering is not an *idempotent* operation. Applying the filter to an already filtered quantity does not leave it unchanged, since filtering a residual does not give zero. One gets:

$$\mathbf{v} \equiv \mathbf{v}^{f} + \mathbf{v}^{r} \Rightarrow \mathbf{v}^{f} = \left[\mathbf{v}^{f}\right]^{f} + \left[\mathbf{v}^{r}\right]^{f} \text{ with } \left[\mathbf{v}^{r}\right]^{f} \neq 0 \text{ (in general).}$$

In the same way, it is easy to show that time differentiation commutes with filtering but that space differentiation commutes only if the filter is homogeneous, which is shown from a direct computation starting with (7.35):

$$\partial_i v_j^{\mathrm{f}} = \left[\partial_i v_j\right]^{\mathrm{f}} + \int v_j(\mathbf{x} - \mathbf{y}) \partial_i G_\Delta(\mathbf{y}, \mathbf{x}) \,\mathrm{d}\mathbf{y} \,.$$

Except in special cases (e.g. close to a wall) one can work with a homogeneous filter and try to get, in this simple case, the equations governing the filtered fields to be explicitly resolved in a simulation. Filtering the continuity equation immediately gives:

$$\left[\partial_j v_j\right]^{\mathrm{f}} = \partial_j v_j^{\mathrm{f}} = 0 \,,$$

and, by subtraction:

$$\partial_j v_j^{\mathbf{r}} = 0$$

for the residue. Difficulties with the momentum equations are analogous to those that led to the introduction of the Reynolds stresses, p. 282. In full generality, these equations read:

$$\rho \left(\partial_t v_i^{\rm f} + v_j^{\rm f} \partial_j v_i^{\rm f}\right) = \partial_j \left(\sigma_{ij}^{\rm f} + \tau_{ij}^{\rm f}\right) \,, \tag{7.36}$$

where σ_{ij}^{f} is the stress tensor defined in the usual way with the filtered quantities:

$$\sigma^{\mathrm{f}}_{ij} = -p^{\mathrm{f}} \delta_{ij} + \eta s^{\mathrm{f}}_{ij} \qquad \mathrm{with} \qquad s^{\mathrm{f}}_{ij} = rac{1}{2} \left(\partial_i v^{\mathrm{f}}_j + \partial_j v^{\mathrm{f}}_i
ight).$$

The supplementary term τ_{ij}^{f} plays the role of the Reynolds stress in the ensemble averaging approach. Its presence is due to the fact that the term

 $[v_i v_j]^{\rm f}$ that appears upon filtering the advection term is not equal to the product $v_i^{\rm f} v_j^{\rm f}$ of the filtered components of the velocity. Introducing the decomposition $\mathbf{v} = \mathbf{v}^{\rm f} + \mathbf{v}^{\rm r}$, one readily gets:

$$[v_{j}v_{i}]^{f} = [v_{j}^{f}v_{i}^{f}]^{f} + [v_{j}^{f}v_{i}^{r}]^{f} + [v_{j}^{r}v_{i}^{f}]^{f} + [v_{j}^{r}v_{i}^{r}]^{f} .$$
(7.37)

The first term on the r.h.s. cannot be expressed as a product of resolved (i.e. filtered) quantities but the difference $\left[v_j^f v_i^f\right]^f - v_j^f v_i^f$, called the *Leonard stress*, is usually quantitatively small. The two terms next involve resolved and residual scales, and the last term, called *subgridscale* though at this stage no computational grid has been introduced, involves non-resolved scales only. A closure problem of a nature somewhat different from that in ensemble averaging is then posed. It rests on a modeling of the additional tensor τ_{ij}^f in (7.36), which is further decomposed into an isotropic part that contributes to modify the filtered pressure and an anisotropic residue. This anisotropic part is most often modeled using an eddy viscosity assumption that closes the system at this level. A Smagorinsky formula (7.32) may be chosen, where the strain tensor is of course defined in terms of the filtered fields.

In spite of an apparent parallelism, this approach is rather different from that developed in the previous subsection. As a matter of fact, here the filtered fields are fluctuating quantities while before they were averaged non-fluctuating quantities. The presence of time in equations (7.21-7.24), and later in the k- ϵ model (7.31-7.34) has indeed a different meaning related to the specific nature of the ensemble statistics.

More or less sophisticated models —here we have just suggested the simplest formulation—used in LES present themselves as systems of equations for a "turbulent fluid" with specific rheological properties that would depend on the local state of the flow. They are still largely under development and demand to be validated by comparisons with direct simulations (moderate R) and laboratory experiments (high R), since they settle the actual complexity of turbulent exchanges, cf. (7.37), via a closure assumption.

The strategies used to solve the problem of interaction between resolved and residual motions might inspire the scientific approach to the nonlinear dynamics of complex systems such as the climate system to be considered as an example at the end of the next chapter, after a brief summary of the topics touched on up to now.

7.5 Exercises

7.5.1 Scales in turbulence and numerical simulations

Direct numerical simulations of Navier-Stokes equations in a situation of developed turbulence are considered, in view of the amount of computer resources required for reliable results as a function of the nominal Reynolds number. The linear size of a cubic domain where a turbulent flow is L, the typical velocity U, ν is the kinematic viscosity and we assume that the Reynolds number can be varied through ν at given L, U.

1) Recall Kolmogorov's estimate of the dissipation scale $\ell_{\rm K}$ relative to L at given R.

2) From a numerical point of view, but without reference to a specific numerical scheme (as discussed in Appendix B), Δx and Δt denoting the space and time integration step, respectively, it is assumed that the small scales are correctly resolved provided that $\ell_{\rm K} \sim 2\Delta x$. Determine the variation with R of the number of grid points needed to resolve the smallest scales.

3) Numerical stability considerations come and limit the time step Δt . In a system dominated by advection (term $\mathbf{v} \cdot \nabla(\mathbf{i})$) the stability is controlled by a *CFL (Courant-Friedrich-Lewy) stability criterion* stipulating that the propagation speed of local (numerical) information $\Delta x / \Delta t$ must remain larger than the speed of transport of physical information by the velocity field. Numerical stability thus requires $\Delta t < \Delta x / U$. Derive the variation of the number of time steps necessary to perform a simulation over a time interval T with space step Δx , with $T \sim L/U$ (turn-over time).

4) Evaluate the amount of computational power as a function of R in a typical full 3D simulation, and the relative increase of power implied by a doubling of R.

7.5.2 Efficiency of turbulent mixing.

1) A radiator is installed to heat up a room with typical size L (volume L^3). Determine the time necessary to reach thermal equilibrium by assuming pure thermal diffusion using a dimensional argument. Take L = 5 m, and $\nu = 15 \, 10^{-6} \text{ m}^2/\text{s}$. For a gas, the Prandtl number is $\nu/\kappa \simeq 0.7$.

2) Consider now turbulent heating with largest eddies of the size of the room generating smaller eddies driving the process. Still dimensionally, estimate the order of magnitude of the air speed above the radiator by assuming the kinetic energy is converted from the potential gravitational energy (buoyancy) over a height h above the radiator. Take $g = 10 \text{ m/s}^2$, a temperature increase $\delta\theta = 10^{\circ}$ above the ambient temperature, h = 0.1 m. The thermal expansion coefficient is $\alpha = 1/300 \text{ K}^{-1}$.

3) A fraction of this energy is available to stir the air in the room. Assuming that the effective stirring speed is only 20% of the speed computed above, determine the characteristic stirring time for the scale of the room.

7.5.3 Developed turbulence in a cumulus cloud

1) Consider the motions inside a typical small cumulus with diameter of the order of $\ell_0 \sim 300$ m. Assuming that the speed of eddies at this scale is $\tilde{v}_0 \sim 3$ m/s, compute the corresponding Reynolds number $(\nu = 1510^{-6} \text{ m}^2/\text{s})$.

2) Estimate of the energy dissipation rate *per* unit mass and the total power (in kW) dissipated in the cloud ($\rho = 1.25 \text{ kg/m}^3$). Evaluate the internal scale $\ell_{\rm K}$ at which energy is effectively dissipated by viscous friction.

3) An observation plane goes through the cloud at speed 180 km/h, carrying a hot-wire velocity probe. This apparatus is made of a small length sensor of length $\delta l = 0.5$ mm. Assuming that it can resolve velocity fluctuations at scales of the order $3\delta l$, what is the maximum frequency of turbulent signals that can recorded by the device. Like in grid turbulence, one can suppose that the relative velocity of the probe with respect of the fluid is the essential factor in converting spatial fluctuations into temporal fluctuations according to Taylor's *frozen turbulence hypothesis*. Compare the Kolmogorov scale $\ell_{\rm K}$ to the scale of the correctly sampled fluctuations. Estimate the turn-over time of large eddies and justify the hypothesis.

7.5.4 Decaying turbulence

Consider a container with volume L^3 containing an initially well stirred fluid that is left to settle. With $\ell = 1$ m, and $\nu = 15 \, 10^{-6} \text{ m}^2/\text{s}$, and assuming initial velocity fluctuations of order $\tilde{v}_0 = 1$ m/s (\tilde{v} denotes here the root mean square velocity in one direction of space), compute the Reynolds number at the beginning of the experiment.

1) Assume that the characteristic scale of turbulent motions remains ℓ and derive the evolution equation for the kinetic energy $\frac{3}{2}\tilde{v}^2$, the evolution law of the velocity and the decay law of the Reynolds number that follows.

2) Compute the time necessary to reach $R \sim 10$ and derive the corresponding fluctuation velocity. At this stage, the viscous dissipation of the large scales is no longer negligible and one must abandon the inertial expression of ϵ to, replace it with the viscous form $\epsilon_{\text{visc.}} = c\nu \tilde{v}^2/\ell^2$. Determine the value of the constant C that assures the continuity of the matching of the two laws at R = 10 and the law governing the final stage of settling.

7.5.5 Turbulence in the atmospheric boundary layer

The atmospheric boundary layer is submitted to the Earth's rotation. The angular velocity of a frame linked to the surface is given by $\Omega \sin(\lambda)$, where λ is the latitude. (Coriolis acceleration is given by $2\mathbf{\Omega} \times \mathbf{v}_r$, where \mathbf{v}_r is the relative velocity; $f = 2\Omega \sin(\lambda)$ is called the 'Coriolis parameter'.)

1) Compute the time scale corresponding to this motion at latitude 45° and next, by a dimensional argument, the thickness of the atmospheric layer that feels the friction on the ground in the laminar case. Take $\nu = 15 \times 10^{-6} \text{ m}^2/\text{s}$.

2) In fact the atmosphere is turbulent. Estimate the thickness $\ell_{\rm mix}$ of the boundary layer affected by turbulent fluctuations of the order of 3% of wind velocity (typically 10 m/s) over the same time scale. Determine the corresponding eddy viscosity given by $\nu_{\rm t} = \tilde{v}_{\rm mix} \ell_{\rm mix}$.

7.5.6 Two-dimensional turbulence

The spectrum of velocity fluctuations in the two-dimensional case is different from that in three-dimensional turbulence. In the latter case, the energy cascade toward small scales generates the Kolmogorov $k^{-5/3}$ spectrum. In two dimensions it is admitted that the turbulent processes are controlled by the transport and dissipation of vorticity. In two dimensions the vorticity has a single component $\omega = \partial_x v_y - \partial_y v_x$ (along the normal to the plane in which the motions are supposed to take place). The enstrophy is this then defined as $W = \frac{1}{2}\omega^2$.

1) Consider an eddy of arbitrary size ℓ and velocity v_{ℓ} , estimate its vorticity ω_{ℓ} and the associated enstrophy.

2) Let $\omega_{\rm I}$ be the vorticity present in the injection scales $\ell_{\rm I}$ and assume that, *per* unit time, eddies loose a fixed fraction of the enstrophy present at these scales. Paralleling the argument for energy injection in three dimensions, estimate the rate β of injection of enstrophy *per* unit time.

3) During the cascade, the size of an eddy is reduced but its vorticity is conserved. Evaluate the typical velocity v_{ℓ} function of ω_{I} and ℓ .

4) Dimensionally, the energy spectrum E(k) is an energy *per* unit wavevector (and *per* unit mass). Show that the inertial enstrophy cascade leads to the Kraichnan spectrum $E(k) \propto \beta^{2/3} k^{-3}$.

Chapter 8

Summary and Perspectives

Through all these pages, our main aim has been to present an introduction to the study of complex systems from the viewpoint of nonlinear dynamics. An abstract approach generalizing the classical treatment of oscillators has been developed and applied to several concrete situations encountered in hydrodynamics. The validity of this theory eventually rests first on the recognition of space-time macroscopic *coherence* induced by instability mechanisms that optimally develop with specific scales, as opposed to the incoherence of motions involved in plain relaxation to thermodynamic equilibrium at microscopic scales. In turn this coherence reduces the multiplicity of possible dynamical behaviors, up to a point where the context of low-dimensional dynamical systems become an appropriate setting. Complexity then mainly means chaos, temporal or spatio-temporal. Such instabilities are generic in macroscopic systems driven far from thermodynamic equilibrium by external stresses. The tendency to return to uniform local configurations above implies a resistance to change expressing the dissipative character of the global dynamics, which is not incompatible with slow large-scale unsteady evolution. In fluid mechanics, this has been illustrated by the emergence of behaviors, from well-ordered cells or waves, to randomly disorganized, turbulent eddies.

After a brief summary of the results obtained so far $(\S8.1, 8.2)$ we examine whether the approach that has been followed can be transposed to other situations where "aggregates" of different nature are in interaction over some range of spatio-temporal scales. Leaving aside applications of technical interest (e.g. flows around obstacles with complex geometries), we focus the discussion on a topic where problems already arise at the modeling stage and next when one wants to draw reliable conclusions from this modeling. Macroeconomics evolution or biodiversity dynamics in ecology

belong to this class of problems. Of equal interest to the future of our society is the problem of *climate change*, which we choose to consider in $\S8.3$, mostly since it is still rather close to physics and mechanics (though it might not be completely free of geopolitical afterthoughts).

8.1 Dynamics, Stability, and Chaos

The concept of *dynamics* governing a time evolution has been modeled as an *initial value problem* for a differential system written in the form:

$$\frac{\mathrm{d}}{\mathrm{d}t}\mathbf{X} = \mathcal{F}(\mathbf{X}), \qquad \mathbf{X}(t^{(0)}) = \mathbf{X}^{(0)}.$$
(8.1)

Next, some operating point called the *base state* and denoted here X_b is supposed to be given. The notion of *stability*, that occupies a central position, is easier to approach in a linear context where the departure from the base state $X' = X - X_b$ is infinitesimal, which allows one to truncate the perturbation expansions beyond the first significant order, leading to problems in the form

$$\frac{\mathrm{d}}{\mathrm{d}t}\mathbf{X}' = \mathcal{L}\mathbf{X}',\tag{8.2}$$

where \mathcal{L} is a linear operator to which one can apply the familiar tools of linear algebra, and especially *normal mode* analysis. Normal modes are taken in the form:

$$\mathbf{X}' = \hat{\mathbf{X}} \exp(st),$$

where s is the growth rate. This leads to an eigenvalue problem:

$$(\mathcal{L} - s\mathcal{I})\hat{\mathbf{X}} = 0 \tag{8.3}$$

yielding a series of eigenvalues $\in \mathbb{C}$ called its *spectrum*:

$$s_n = \sigma_n - i\omega_n$$
, $n = 1, 2, \dots$

further ordered by decreasing values of the σ 's: $\sigma_1 \geq \sigma_2 \geq \sigma_3 \geq \ldots$

The principal result of this analysis is a representation of the dynamics in a infinitesimal neighborhood of the base state in terms of a superposition of normal modes $\hat{\mathbf{X}}_n$, each with an amplitude $A_n(t)$, function of time and $A_n(t^{(0)}) = A_n^{(0)}$. In the simplest cases, *i.e.* non-degenerate, this superposition can be written as:

$$\mathbf{X}'(t) = \sum_{n} A_n^{(0)} \exp\left[s\left(t - t^{(0)}\right)\right] \hat{\mathbf{X}}_n.$$

The stability of the base state \mathbf{X}_{b} can thus be discussed as a function of the sign of the real parts σ_{n} of the complex growth-rates s_{n} of the eigenmodes. From the linear viewpoint a given mode n is thus stable when $\sigma_{n} < 0$, neutral or marginal when $\sigma_{n} = 0$, and unstable when $\sigma_{n} > 0$. A sufficient condition for the instability of \mathbf{X}_{b} is thus $\sigma_{1} > 0$. The imaginary part ω_{n} of s_{n} next allows one to distinguish between stationary modes with $\omega_{n} = 0$ from oscillatory modes with $\omega_{n} \neq 0$.

In full generality, \mathcal{F} , \mathbf{X}_{b} , \mathcal{L} and its spectrum, are functions of control parameters accounting for the different ways of acting on the system, which we symbolize by a single variable r in the present formal setting. A *bifurcation* takes place when the real part of the growth rate of the most dangerous mode crosses zero from below as r is varied. The *critical* condition is thus given by $\sigma_1(r) = 0$ which can be solved for r to define the *instability threshold* r_c .

When the base state is unstable, the linear analysis is valid only as long as the system stays sufficiently close to its base state, which does not hold long since some perturbations diverge exponentially. Terms neglected until now in the perturbation expansion must be taken into account, expressing the interaction of the unstable modes between themselves and with all other modes.

In this domain, the essential result is the possibility to eliminate the stable modes $\hat{\mathbf{X}}_{s}$ ('s' for 'stable'), with amplitudes collectively called \mathbf{A}_{s} , and only keep the *master* modes $\hat{\mathbf{X}}_{c}$ with amplitudes \mathbf{A}_{c} ('c' for 'critical' of 'central', *i.e.* either stable or unstable but near-marginal, see below or Chapter 4, §4.1.2). The reason is that the former are not independent of the latter but rather *enslaved* to them. This elimination, sketched in Figure 4.3, p. 120, leads one to define an *effective dynamics* only coupling the central modes between themselves:

$$rac{\mathrm{d}}{\mathrm{d}t}\mathbf{A}_{\mathrm{c}} = \mathcal{L}_{\mathrm{c}}\mathbf{A}_{\mathrm{c}} + \mathcal{N}_{\mathrm{c,eff}}(\mathbf{A}_{\mathrm{c}})$$
 .

Apart from during a brief transient (exponential relaxation of stable modes), the system then evolves on a manifold defined by some relation

$$\mathbf{A}_{\mathbf{s}} = \boldsymbol{\mathcal{G}}_{\mathrm{eff}}(\mathbf{A}_{\mathrm{c}})\,,$$

that can usually be determined asymptotically in the long time limit and for small departure from the critical conditions by a well-defined technique called the *center manifold reduction*.

Technically, a gap in the spectrum has to be assumed, such that $\sigma_{\rm s} < 0$

and $\min |\sigma_{\rm s}| \gg |\sigma_{\rm c}|$ with $\sigma_{\rm c} \sim 0$. But, in practice, the domain of interest of this reduction is larger than it seems because the *normal form* that is obtained has most of its structure imposed by the symmetries of the problem and the nature of the critical modes. This *universality* opens the door to a *phenomenological modeling* that relies only on these ingredients. This is particularly interesting because the number of possible forms and the manner to perturb them away from criticality are both generically small. Let us mention the most frequent ones.

The first one relates to the bifurcation of a single real amplitude A displaying a symmetry with respect to the change $A \mapsto -A$. At criticality it is governed by:

$$\frac{\mathrm{d}}{\mathrm{d}t}A = -A^3$$

and, upon adding the most general perturbation, by:

$$\frac{\mathrm{d}}{\mathrm{d}t}A = rA - A^3 + h\,,\tag{8.4}$$

where r measures the distance to the bifurcation point in the control parameter space and h is a small perturbation that breaks the initial symmetry. (In this equation r and h are two real parameters.) This bifurcation, illustrated in Figure 4.4, p. 127, is the dynamical translation of one of Thom's seven *elementary catastrophes* and was introduced earlier in the theory of thermodynamic phase transitions by Landau.

The second example is the *Hopf bifurcation* of a pair of complex conjugate modes that accounts for the birth of self-sustained oscillations, e.g. in the van der Pol system. Close to the threshold, the normal form reads

$$\frac{\mathrm{d}}{\mathrm{d}t}Z = (r - i\omega_{\mathrm{c}})Z - g|Z|^2 Z, \qquad (8.5)$$

where $Z = A_r + iA_i$ is the complex amplitude of one of the two interacting modes, ω_c is the angular frequency at threshold, r the distance to threshold and g a complex coefficient. Its real part g_r describes the saturation of the amplitude beyond threshold while its imaginary part g_i , the nonlinear dispersion coefficient, accounts for the dependence of the frequency on the amplitude, a typical feature of nonlinear oscillators, see Figure 4.5, p. 129.

The story then continues with the emergence of more complex behavior, still crucially involving stability considerations. The state resulting from this first bifurcation can indeed be considered as a new base state, that may become unstable with respect to secondary modes, and then tertiary modes, *etc.* This point of view refers to a situation where an ever larger number of modes initially in the stable group ($\sigma_n < 0$) become critical and next unstable ($\sigma_n > 0$), therefore increasing the dimension of the effective dynamics. Stationary modes (*fixed point* in phase space) described using a single real amplitude, e.g. (8.4), are then usually followed by periodic behaviors (*limit cycle* attractors) governed by (8.5), which in turn may become unstable with respect to other oscillatory modes yielding quasi-periodic behavior (*torus* attractor), etc.

Ruelle and Takens have shown that, contrary to the original intuition of Landau who argued in terms of *superpositions* of oscillatory behaviors, complexity arises from the *interaction* of modes, with *chaos* as a nontrivial consequence. These two competing scenarios were sketched in Figure 3.5, p. 93.

In order to understand the emergence of chaos from a regular behavior, it is most appropriate to take the *Poincaré section* of trajectories, a procedure that generalizes the stroboscopic analysis of forced periodic behaviors and, at any rate, reduced the initial continuous time system to an iteration (*discrete time* system).

The essential property of chaotic systems is the *instability of trajectories* on the corresponding attractor, then said to be *strange*, which expresses the loss of long term predictability resulting from the *sensitivity to initial conditions* and small perturbations. This property is understood most easily when considering the dyadic iteration, the simplest model of chaotic system:

$$X_{k+1} = 2X_k \pmod{1},$$

that perfectly illustrates the (exponential/geometrical) growth of the distance between neighboring trajectories: $\delta X_k = 2^k \delta X_0 = \exp(n \ln 2) \delta X_0$. In the general case, this divergence can be measured by means of *Lyapunov* exponents that extend the notion of growth rate initially introduced to deal with the stability of regular trajectories. For one-dimensional iterations in the form

$$X_{k+1}=\mathcal{F}(X_k)\,,$$

one then defines:

$$\lambda = \lim_{K o \infty} rac{1}{K} \sum_{k=0}^{K-1} \ln \left(\left| rac{\mathrm{d} \mathcal{F}}{\mathrm{d} X}(X_k)
ight|
ight) \, ,$$

where $\{X_k, k = 0, 1, ...\}$ is a reference trajectory, so that the distance between neighboring trajectories varies as $(\exp \lambda)^k$ for k large enough. In addition, strange attractors generally display a *fractal* structure contrasting the smooth aspect of regular attractors. Finally, it has been seen that the transition to chaos develops through one out of a small number of universal *scenarios*, the most celebrated being the *subharmonic cascade* that leads to an aperiodic behavior at the end of a series of period doublings (period 2, $4, \ldots, 2^m, \ldots$) with successive thresholds geometrically converging towards a finite value, as illustrated in Figure 4.13, p. 145.

8.2 Continuous Media, Instabilities, and Turbulence

For a discrete system, (8.1) is a finite-dimensional differential system. For a continuous medium, it is a system of partial differential equations. The linearized operator defined by (8.2) also contains partial derivatives with respect to (physical) space coordinates.

In order to simplify the analysis, one often considers the limit of translationally invariant systems that allows one to solve (8.3) by means of a spatial Fourier transform, by making the change

$$\hat{\mathbf{X}}\mapsto ilde{\mathbf{X}}\exp(ikx)$$
 .

Branches of normal modes are then parameterized by the wavevector k. The relation $s = s_n(k;r) = \sigma_n(k;r) + i\omega_n(k;r)$ is called the *dispersion* relation relative to mode n. The condition $\sigma_n(k;r) = 0$, once solved for r, gives the corresponding marginal stability condition:

$$r=r_n^{(\mathrm{m})}(k).$$

As before, one is interested in the most dangerous mode, the mode with the largest σ . Let us assume that mode '1' is the most dangerous for all values of k (no branch exchange) and further drop the subscript. Usually, the wavevector of the perturbation cannot be controlled and the growth rate displays a quadratic maximum around some specific k_c called the *critical wavevector*. Generically the real part can thus be written as:

$$\tau_0 \sigma = r - r_{\rm c} - \xi_0^2 (k - k_{\rm c})^2. \tag{8.6}$$

In this equation, τ_0 is the natural evolution time of the fluctuations and ξ_0 , homogeneous to a length, is called the *coherence length*. This relation asserts the optimal character of couplings implied in the instability mechanism for the specific value $k = k_c$ (the growth rate of neighboring modes is smaller). The marginal stability conditions close to the instability threshold (r_c, k_c) then reads

$$r = r_{\rm c} + \xi_0^2 (k - k_{\rm c})^2 \,. \tag{8.7}$$

The values of r_c , k_c , τ_0 , ξ_0 have to be determined on a case-by-case basis and can be grossly evaluated by physical and dimensional arguments, but the universal contents of (8.7) remains.

The instability is stationary or oscillatory depending on whether the critical angular frequency ω_c is identically zero or not. A stationary instability with $k_c \neq 0$ is called *cellular*, while when $\omega_c \neq 0$ it is a *wave*. In this context, the study of natural (Rayleigh-Bénard) convection, Chapter 3, has shown that it is a simple example of cellular instability, with a particularly intuitive mechanism involving buoyancy and dissipation. As to shear flows, the theory shows rather easily that they are unstable against spanwise-uniform streamwise-propagating waves when the base flow displays an inflection point.

In the case of open flows, a stationary instability with respect to a framework at rest with the mean flow is trivially seen as a wave in the laboratory frame. This leads to an important distinction according to whether the mean advection is able or not to evacuate perturbations downstream or not, whether the instability is *convective* or *absolute*, respectively (see Figure 6.13, p. 241). Except for few immediate elementary facts, we got past the corresponding theory, owing to its delicate analytical character.

When the continuous medium is not translationally invariant, usually due to end effects, the role of confinement has to be discussed. To this aim, one defines *aspect ratios*:

$$\Gamma = \ell / \lambda_{\rm c}$$
,

where ℓ is the typical size of the system in the direction of the instability wavevector, and $\lambda_c = 2\pi/k_c$ is the critical wavelength of the instability.

When $\Gamma \sim 1$ (confined systems), the instability mechanism produces a small number of elementary cells that play the role of effective degrees of freedom since confinement effects are strong enough to maintain the space coherence. Adiabatic elimination of stable modes evoked earlier helps one to derive an effective low-dimensional dissipative dynamical system, which makes the theory of *deterministic chaos* fully relevant, especially in regard of its universal aspects, see §3.2.3 where the case of natural convection was presented in detail.
The opposite case of *extended systems*, $\Gamma \gg 1$ corresponds to a situation where the bifurcated solution remains coherent at a local scale only, typically few $\lambda_c s$. Another strategy has to be employed to account for them, in terms of *envelopes* describing the slow, long-wavelength modulations brought to an ideally uniform nonlinear solution. Technically these envelopes are obtained through a rather boring *multiple scale* strategy which can be short-circuited through a phenomenological approach making an immoderate use of the *universal properties* of the system close to its instability threshold. This universality is linked to its physical symmetries and the nature of the primary bifurcation. The system is then modeled by unfolding the normal form that governs the amplitude of the critical mode, thus becoming an envelope. This is achieved by adding a dynamics, often simply diffusive, driving spatial modulations to the amplitude, which yields equations of complex-Ginzburg-Landau type, see §5.4.

While being restricted to the weakly nonlinear range, these theoretical developments, sketched in Chapter 5, satisfactorily account for topological defects and universal secondary instabilities. Defects (singular points in the envelopes) and *phase* instabilities linked to the local translation and rotation invariances in space, both contribute to the large-time long-distance disorganization of the *patterns* at the heart of *spatio-temporal chaos* as a route to turbulence in extended systems.

This route to turbulence is relatively progressive and can thus be called 'globally supercritical' in the sense that the bifurca*ting* state (which develops beyond the bifurcation point) stays close to the bifurca*ted* state (which is unstable) and replaces it. Some physical systems, and especially convection even at large aspect ratios, become turbulent in that way.

The case of shear flows is relatively more delicate. As a matter of fact, one has to distinguish flows that are *mechanically unstable*, *i.e.* unstable at the inviscid limit, from those that remain stable in this limit. The prototype of the first class is the shear layer experiencing a Kelvin-Helmholtz instability related to the presence of an inflection point in the base profile. The second class of flow is formed with those that can possibly be unstable but through a subtle process in which viscous dissipation is crucially involved, producing *Tollmien-Schlichting waves*.

The consequences of this classification in the strongly nonlinear regime are important. In particular, whereas the flows belonging to the first class follow a globally *supercritical* route to turbulence that develops at moderate Reynolds numbers, in the second case, the linear instability threshold, if it exists at all, is high, so that there is room for other typically nonlinear solutions to the Navier-Stokes equations at intermediate Reynolds numbers. This explains that the transition to turbulence, globally *subcritical* according to this scenario also called the *bypass transition*, appears to be "wilder" than in the first case. Actually, *turbulent spots* concentrate, locally in space amid laminar flow, that special kind of solution that does not derive continuously from the base state, see Figure 6.20, p. 254.

Practical situations are often complicated: convective/absolute instability, super/sub-critical bifurcation, regular/chaotic motion, modulations and spatio-temporal chaos. A good understanding of the processes at stake can help in mastering them, that is to say, controlling the transition to turbulence. A toolbox has been presented, and its use demonstrated in some typical cases. It contains most of the implements necessary to explore the richness that surges as soon as nonlinearities cannot be neglected, whatever the field of knowledge considered.

We have said little about the regime of *turbulence developed* that sets in well beyond the transitional region. In this circumstance, the deterministic approach is no longer of help since many degrees of freedom interact in a cascade over a large range of space scales, from the large ones where the turbulent energy is extracted from the mean flow by the instability mechanisms to the small ones where it is dissipated by viscosity effects, see Figure 7.3, p. 276. Elementary dimensional arguments allowed us to estimate the rate of transfer of the energy (per unit mass) through the *inertial range* where viscous dissipation can be neglected and energy distribution is ruled by Kolmogorov's energy spectrum $E(k) \sim k^{-5/3}$, down to the *dissipation scale* where viscous effects cease to be negligible.

The irregular (chaotic) character of turbulent flows lead us to introduce statistical methods. Two fundamental components of the flow, the *average flow* and the *fluctuations* were identified following Reynolds, but the *closure problem* arose immediately through the definition of *Reynolds stresses*. A very preliminary solution to this problem was presented, in terms of an *eddy viscosity*, itself evaluated within a *mixing length* hypothesis. The approach was then used to determine the mean profile of a fully developed turbulent boundary layer, yielding Kármán's logarithmic law, see Figure 7.6, p. 289.

The two main results obtained by dimensional arguments, Kolmogorov's spectrum and Kármán's logarithmic law are remarkably verified by experiments but remains zero-order approximations that do not thoroughly treat the problem of exchanges between eddies with different scales. Empirical divergences appear as soon as one tries to go beyond this elementary level. In concrete configurations of interest to applications, a numerical approach is often necessary, with a crucial modeling step to deal with the closure problem $(k-\epsilon \text{ model}, \text{ Large Eddy Simulations},...)$.

In the context considered up to now, the origin of the complexity was mostly in the nonlinear character of the dynamics of systems with "simple" definitions. More "complex" interactions of processes can also take place, e.g. involving rotation and magnetic fields in conducting fluids (dynamo effect), combustion, *etc.* possibly demanding separate modeling efforts. In the concluding section we rather introduce a case where complexity arises from the interaction of a large variety of sub-systems, exchanging many physical quantities over a large range of space and time scales, the Earth's climate system.

8.3 Approach to a Complex System: the Earth's Climate

In the traditional geographical sense, the study of climate was limited to a description of meteorological phenomena such as temperature, wind, amount of rain, *etc.*, that characterize the mean state of the atmosphere throughout a typical year. This meaning of course remains but, in the perspective of a rapid change induced by human activities, one has to enlarge the viewpoint to also include how this average situation arises,¹ and to understand the individual dynamics of many "actors," the atmosphere, the ocean, the cryosphere (ice caps), the biosphere (mainly the vegetation), and at, the end of the chain, our society.

The main physical media involved are fluids in motion, transporting energy in different forms. Energy exchanges play an important role but are constrained by rigorous physical conservation laws. A large variety of nonlinear feedbacks and space-time scales are involved in climate phenomena, which explains why one could naively try to tackle them using our toolbox. We therefore first extend the framework of previous approaches to evoke a few tangible applications of the concepts introduced earlier, before going to more speculative grounds.

Climate can first be understood through its meteorological dimension, variable and to a certain extent random. In French, there is a sailor saying that: Mentira bien souvent/ Qui prédira le temps,/ Mais beaucoup

 $^{^{1}}$ A nice introduction is [Philander (2000)]; at a more technical level one can cite [Peixoto and Oort (1992)], and [Henderson-Sellers and McGuffie (1987)] for modeling issues.

moins pourtant/S'il est bon observant, which essentially means that you are certainly a lier when you predict weather but much less of one if you make good observations. Current wisdom of old sailors as expressed in this saving is thus an excellent anticipation of the concept of deterministic chaos, originating from long term unpredictability (the lie) in spite of predictable short term evolution (reasonable extrapolation of accurate observations) as it was formally introduced by Ruelle in the Seventies, not long after the informal quotation by Lorenz —a great meteorologist, by the way- in the Sixties. The key-concept of chaos theory is the indefinite amplification of tiny differences between initial conditions, that forbids long term prediction. In the case of weather, this prediction rests on the direct integration of equations governing the motions of the atmosphere. These equations are implemented in an essentially satisfactory way in the best routines of modern Meteorological Offices but at the regional scale that we consider presently, the reliability of the results depends on an accurate account of initial conditions in the bulk of the atmosphere (pressure, winds, temperature, humidity) and boundary conditions at the surface (sea surface temperature, relief,...). This data is known with a limited precision at the nodes of a sparse network of observation point, whereas the problem is set in terms of partial differential equations, which leaves aside all local events of small amplitude. Figure 8.1 illustrates the fact that radically different predictions are obtained beyond some predictability range, today



Fig. 8.1 States at day+7 (bottom line) of three simulations starting from initial conditions all compatible with the same ensemble of physical observations (top). After Palmer, in [Hall (1992)].



Fig. 8.2 METEOSAT image of a storm centered at the entrance of the English Channel, September 6, 1995; © Météo-France, www.meteo.fr, with permission.

still much less than fifteen days (Lorenz's "butterfly effect"). As a first step, one indeed has to proceed to *data assimilation*, *i.e.* put data in an appropriate form to start the simulation, after some pre-treatment optimizing the short-time evolution and making the dynamical extrapolation reliable over a longer period of time, in accordance with chaos theory that tells us that the prediction (in abstract phase space) is more accurate if initial conditions are better known, but that small initial uncertainties always reveal themselves. Recent progress may lead us to think that the detailed midterm (few days) prediction of extreme events such as the one illustrated in Figure 8.2, their conditions of occurrence and their development, is within reach and can only improve in parallel with computing power which allows one to follow individual trajectories ever better and make statistics over ensembles of trajectories, while long-term unpredictability is inescapable.

Keeping the dynamical systems perspective, let us mention the use of nonlinear empirical reconstruction techniques introduced at the end of Chapter 4, §4.4. Takens's delayed coordinate method was in particular applied to the quantitative analysis of time series measuring the intensity of the El Niño phenomenon, as an attempt to predict its evolution one or two years in advance. El Niño is a sequence of climatic "anomalies" that affect firstly the inter-tropical Pacific Ocean.² It is characterized by a cyclic modification of heat exchanges and zonal motions of water masses together with a swing of the pressure field between the East-Pacific and the West-Pacific called the Southern Oscillation (hence the acronym ENSO meaning

²For an introduction, consult Chapter 9 of [Philander (2000)].



Fig. 8.3 Variations of the MEI measuring the intensity of the ENSO phenomenon.

El Niño-Southern Oscillation). The cycle is locked on the yearly seasonal periodicity but has a variable duration between two and seven years, with warm episodes (El Niño) and cold ones (La Niña). The warm phase is characterized by warm waters off the coasts of Peru before Christmas, due to a weakening of trade winds that should push these waters further to the West. In the old days the El Niño phenomenon was considered as a gift (at the regional scale) since abundant rains made Peruvian deserts somehow fertile. Nowadays, we have a better understanding of at-a-distance effects, and intense ENSO episodes (e.g. 1982-83, 1997-98) are considered as calamities since large regions in the World suffer from them: augmented rains and floods in western South America, drought and forest fires in the West-Pacific and Australia, not mentioning suspected consequences on the global climate. The intensity of the phenomenon is measured by several indices³ (Fig. 8.3). The aim of the game is, starting from the time series of a given index, to predict the occurrence of an intense episode six months or one year in advance. Adaptative filtering techniques (singular spectrum analysis, Note 22, p. 164) and nonlinear techniques of dynam-

³The first one, and the oldest is the Southern Oscillation Index, SOI, that directly measures the pressure difference between Darwin, in the north of Australia and Papeete in Tahiti (see e.g. the web site of the Australian Commonwealth Bureau of Meteorology, <http://www.bom.gov.au/climate/current/> that displays monthly values of this index from 1876 on). Another one, produced by the Japan Meteorological Agency (JMA, <ftp://www.coaps.fsu.edu/pub/JMA_SST_Index/>), is a sliding average over five months of sea surface temperature anomalies in a portion (4°S-4°N, 150°W-90°W) of the Pacific Ocean. Data was gathered since 1949 but could be reconstructed in the past from 1868 on. A third one is elaborated by K. Wolter, at NOOA's Climate Diagnostic Center, as a compound of six different measures called MEI for Multivariate ENSO Index. It covers the contemporary period since 1950, and can be found at the web address <http://www.cdc.noaa.gov/~kew/MEI/>.

ics reconstruction now seem to do better than the best linear statistical extrapolation techniques.⁴ In paleoclimatology, time series of the temperature in the ocean has been studied in the same way, pointing out oscillatory behaviors⁵ that we will consider later.

The previous approach rests on the analysis of the time series of a single variable. A more ambitious strategy, also more computer demanding, is to account for meteorological fields in their space-time dimension. This is done by extending the classical *principal components analysis* (see [Peixoto and Oort (1992)] for the atmospheric context) by taking into account the time ordering of pictures serving to compute the principal components, in order to include dynamical aspects. This multichannel combination of singular spectrum analysis and reconstruction leads to the objective identification of *weather regimes*, centers of action, regular oscillations (30-days in the North-Atlantic), blockings (understood as trajectories being close to a saddle fixed point in phase space), at-a-distance connections between regional phenomena, *etc.*,⁶ each contributing to the understanding of climate processes and to the improvement of mid-term weather forecasts.

Up to now we have considered the meteorological facet of climate. The approach just described, local and deterministic in essence, cannot be pursued without adaptation when studying the system at a global scale and in the long term (several years to several hundred of years, not to say even more if ice ages are of interest). The problem is again of going from a "microscopic" perspective to "macroscopic" one, which is analogous to (but less well defined than) the determination of the mean flow in developed turbulence. The understanding of the Earth's climate and its evolution is however not a pure intellectual exercise, since the impact of human activities has to be taken into account and political choices have to be made, that will undoubtedly orient our near future existence.

As we have already mentioned, many different actors interact on a wide range of space and time scales, at the origin of the complexity of the dynamics of our natural environment. The description of this complexity relies on a hierarchy of climatic variables defined as statistical averages over geo-

⁴M. Ghil, M. Kimoto, J.D. Neelin, "Nonlinear dynamics and predictability in the atmospheric sciences," Review of Geophysics Supplement (1991) 46-55.

 $^{{}^{5}}$ R. Vautard and M. Ghil, "Singular spectrum analysis in nonlinear dynamics, with applications to paleoclimatic time series," Physica D **35** (1989) 395-424.

⁶G. Plaut and R. Vautard, "Spells of low-frequency oscillations and weather regimes in the northern hemisphere," J. Atmos. Sc. **51** (1994) 210–236. P.A. Michelangeli, R. Vautard and B. Legras, "Weather regimes: recurrence and quasi-stationarity," J. Atmos. Sc. **52** (1995) 1237–1256.

graphical regions of various sizes (local, regional, global) and over different lengths of time (day, month, season, year). These variables are generally defined as sliding averages over some duration (usually thirty years) in order to suppress the short term variability. Other quantities of interest are the statistics of the departures from the means, called *anomalies*, and of extreme events. By construction, fastest meteorological fluctuations are filtered out but slower trends remain, which makes it difficult (and subject to controversy) to separate "natural" climatic variability from climate change induced by human activities.

Let us briefly examine the ingredients to this variability. The global climate system is sketched in Figure 8.4. It is composed of different tightly linked elements with various badly matched individual time constants. External forcing by the sun is essential and, while it is easy to attribute the basic driving mechanism to the pole-ward gradient of the energy exchanged with outer space —mostly received in the equatorial region while little is received and much radiated out near the poles— this does not help much understanding the details of the redistribution (somehow reminiscent of what occurs in natural convection) which mainly involves two fluids, the atmosphere and the ocean, with different physical characteristics, in a rotating framework.

Time scales introduced by the solar forcing extend over a very wide range. The highest and most obvious frequencies are of course given by the daylight and season cycles that are most important in the short term. But the sun-spot cycle (11 years) may play a role and the variation of



Fig. 8.4 Different components of the climate system, with an indication of their characteristic time constants.

astronomic parameters of the Earth's orbit, and in particular, the precession of the equinox, with typical periods ranging from 10^4 to 4×10^5 years, are thought to drive the ice age cycles on a geological time-scale.

The atmosphere is naturally sensitive to the daily forcing but most of its evolution takes place on a weekly basis, which corresponds to the average motion/instability of air masses at mid latitudes. On longer time-scales, it is affected by the sea surface temperatures (SST) that follow the seasonal cycle. In addition to the tidal rhythm, the ocean as a whole evolves on quite longer times, from a few years for the upper ocean to hundreds of years much below. The surface of the sea ice, important since it reflects sun light in a region that receives little, varies essentially on a seasonal basis. Polar ice caps, that store and return fresh water, change on time scales of the order of a century up to periods fixed by the longest astronomical scales. Except for punctual effects linked to volcano eruptions that introduce dust particles and gases in the atmosphere (e.g. the Pinatubo in 1991), geophysical phenomena linked to continental drift are also important by changing the shape of ocean basins or erecting mountain chains over tens of millions of years, but the effect of closing or opening straits may have more immediate consequences by changing oceanic currents.

The biosphere is of course sensitive to the seasonal cycle but, on durations of order tens to hundreds of years, the flora follows the slow displacement of climatic zones and its role should not be neglected in the analysis of feedbacks rooting the climate variability. As to the human activity, its effects can be felt on short time scales at the local scale (agricultural practices, deforestation), or on longer scales in particular by modifying the chemical composition of the atmosphere, introducing greenhouse gases in "anomalous" proportions, especially since the middle of the Nineteenth Century. The greenhouse gases (GG) are all gases with molecules containing at least three atoms (mainly water vapor, carbon dioxide, methane) even at the trace level (CFCs). Their climatic effect is linked to infra-red absorption properties due to rotation-vibration resonances. The molecules are transparent to visible and ultra-violet radiations in the incoming solar flux but opaque to the infra-red thermal reemission from the ground. The resulting accumulation of heat in the lower atmosphere is called the greenhouse effect, first put forward by Arrhenius (1896).

Paleoclimate studies focus on the very long term variability of climate. Our knowledge of past climates rests on the record of data derived from the study of cores drilled in ocean or lake sediments and in polar ice caps. Quantitative information is obtained on climatic conditions in given regions by analyzing the isotopic composition of samples⁷ or the correlations between the plankton/pollen contents of sediments with the type of climate. The relation between the GG concentrations and the temperature has also been studied from the air bubbles captured in the ice. Though the number/accuracy of these records is limited, their time series yield a precise identification of the most important events, as seen in Figure 8.5, next page.

The alternation of glacial (cold) and interglacial (warm) episodes at mid and high latitudes, with an approximate period of order 10^5 years, shows that the variability may have a large amplitude. Though it is generally believed that this alternation is related to insulation fluctuations induced by the already mentioned small variations of the Earth's orbital parameters (obliquity, eccentricity, position of the perihelion) —Milankovitch's theory (1924)— the reason the system entered this alternating regime, and the mechanisms controlling the periodicity are not fully understood and still subject to controversy.⁸ At any rate, the forcing at 10^5 years is extremely weak and the most important contributions lie at 41, 19, and 23 kilo-years, and are effectively found, without much surprise, in the Fourier spectrum of the temperature signal. This idea of a superposition of simple responses to some external forcing is linear in essence. As such it is valid only for systems close to equilibrium or, which comes to the same, to strongly dissipative systems. In fact, a forced nonlinear system far from equilibrium is likely to experience partial lockings or chaos, which makes any analysis resting on too short a time series questionable. In addition, the small scales of the system play the role of a noise that can contribute to erase or amplify the response to the forcing depending on its amplitude.⁹ The astronomical theory is also powerless to account for rapid climate changes recently pointed out, and to which we now turn.

⁷For example, the oxygen isotopic fraction O^{18}/O^{16} depends on the physical conditions at the time of formation of sediments/ice and gives information on the temperature or the global ice volume.

⁸I.J. Winograd *et al.* "Continuous 500,000-year climate record from vein calcite in Devils hole, Nevada," Science **258** (1992) 255-260; J. Imbrie, A.C. Mix, and D.G. Martinson, "Milankovitch theory viewed from Devils Hole," Nature **363** (1993) 531-553.

⁹Discussing stochastic systems is much beyond our purpose in this course. The phenomenon alluded to here is called "stochastic resonance." (For a review see: L. Gammaitoni, P. Hänggi, P. Jung, and F. Marchesoni, "Stochastic resonance," Rev. Mod. Phys. **70** (1998) 223–287.) This phenomenon —the enhanced response of a periodically forced bistable system (here glacial/interglacial)— was suggested as an explanation of the occurrence of the 10^5 -years period by R. Benzi, *et al.*, "Stochastic resonance in climatic change," Tellus **34** (1982) 10–16, but without decisive argument in its favor.



Fig. 8.5 Variation of the temperature in Antarctica during the last ice cycle as derived from isotopic measurements from a core drilled at the Vostok site. Analysis of the data by Takens delayed coordinate method combined to singular spectrum analysis, allows one to isolate the main components of the signal. The continuous thin line through the fluctuating data is the signal reconstructed from a projection onto the five first components. The projection onto the two next components displays a clear oscillation with modulated amplitude (dotted line). Data provided by P. Yiou, LSCE, CEA-Saclay.

The detailed study of the last ice cycle, that begins 115 000 years ago and culminates 21 000 years ago, has effectively pointed out shorter periods superposed to the large amplitude saw-tooth modulation. Figure 8.5 suggests a period of the order of 6 000 years. This picture is given to prove that variability may also have fast variability yet to explain. It may however be misleading in that the oscillation pointed out seems smooth, an effect of the filtering procedure, whereas the variations are rapid, a feature which took time to be recognized.¹⁰ These fast variations have been identified in the isotopic analysis of ice cores, as well as by carefully examining the size distribution of sediments. At roughly regular intervals, the North-Atlantic Ocean witnesses massive iceberg surges identified through layers of materials scratched on the North-American continent and transported as far as the point of melting, where they are incorporated in the sediments (Heinrich events). These episodes are seemingly well correlated to temperature fluctuations over Greenland called Dansgaard–Oeschger oscillations.

¹⁰For a historical presentation with references, consult S. Weart, "The discovery of rapid climate change," Physics Today **56** (2003) 30-36., also on the Internet http://www.aip.org/history/climate/.



Fig. 8.6 Dansgaard-Oeschger oscillations and Heinrich events (provided by D. Paillard, LSCE, CEA-Saclay).

Several indicators testifying for these rapid changes are displayed in Figure 8.6. Shaded vertical bands through the data signal the anomalies of size distribution in the sediments. The upper curve (isotopic fraction anomaly for oxygen in Greenland ice) is interpreted in terms of the temperature at which the snow is formed over the land. The two other curves are related to the waters in the North-Atlantic ocean, respectively indicative surface temperature (middle) and salinity (bottom), and both derived from the Oxygen isotopic composition of specific plankton micro-organisms (iceberg melting brings fresh waters and changes the salinity; plankton composition is finely tuned to temperature and salinity). Most recent evidences of these oscillations give a period of 1 500 years with cycles skipped (hence 6 000 years would appear as a filtered subharmonic). The origin of these oscillations is searched in abrupt changes of the oceanic circulation, and convincing sophisticated models have been proposed.¹¹ Later we will examine a much simpler "conceptual" model.

Closer to the present time, important climate fluctuations have been observed, just to mention the 'climate optimum' (around 1000) when Vikings saw Greenland as a green land, and 'the little ice age' (Sixteenth to Eighteenth Century). At the human lifetime scale, climate variability finds

¹¹A. Ganopolski and S. Rahmstorf, "Abrupt glacial climate changes due to stochastic resonance," Phys. Rev. Lett. 88 (2002) 038501. See also Note 9.

its trace, most of the time catastrophic, in the already mentioned ENSO phenomenon, persistent droughts implying some extension of the deserts (Sahel), frequency change of cyclones, exceptional floods, and so on.

Understanding climate variability and learning to cope with it, in the perspective of global change, with converging pieces of evidence for warming, is an important objective of climate research.¹² Let us examine different types of approaches in which nonlinear science might be involved apriori. First it should be noticed that basic science (mechanics, physics, chemistry, biology) surely have their word to add in accounting for exchanges from the most microscopic stage to the macroscopic level. However the most detailed description is presumably impossible and further useless, so that some sort of reduction is necessary. This reduction cannot attain the level of rigor reached in dynamical systems theory that allows the asymptotic separation of slow active variables from fast enslaved variables, but it is appealing to try to understand climate as some effective dynamics with times scales much slower than those of individual contributing processes. Classification of variables is however not easy. For example atmospheric modes are only partially entrained by oceanic modes via sea surface temperatures, and the ocean responds in a filtered way to wind stresses generated by unstable fast atmospheric modes, hence complicated feedbacks.

Climate modeling can be developed at various levels, from simple energy balance models (EBM) governing few variables averaged over the whole surface of the Earth, to general circulation models (GCM) that try to mimic at closest (partial differential equations) the dynamics of one of the main elements, the atmosphere or the world ocean, as forced by the other and vice versa, and to coupled models that involve also the cryosphere and the biosphere in the most general context, which is now made possible by the tremendous power increase of present-day computers. For a first approach, consult [Henderson-Sellers and McGuffie (1987)]. The level of modeling is tightly linked to the degree of realism of the description to which one aims, and perhaps in inverse proportion to the degree of understanding one would reach.

EBMs are more specifically adapted to the study of large scale equilibria, hypothesized fixed points of a global climate system (see Exercise 8.4). If they can help make up one's mind about the sensitivity of climate to different variations, e.g. albedo, the fraction of what is reflected toward

¹²For a critical point of view, see: R.S. Lindzen, "Climate dynamics and global change," Ann. Rev. Fluid Mech. **26** (1994) 353-378.

space to what is received, by nature they can hardly account for any time dependence, at least in their zero space-dimension. One dimensional extensions, *i.e.* zonal models in one space-dimension, the latitude, with averaging on the longitude and the altitude) improve the situation slightly but cannot account for features of geographical origin.¹³ These effects can be partially incorporated in so-called *box models*, whose variables characterize large geographic units and interact through selected processes supposed to be relevant and important to the phenomenon under study. Such models are often said to be *conceptual*, in order to insist on their heuristic role. By construction they present themselves as systems of ordinary differential equations and therefore perfectly fit the framework of low-dimensional dissipative dynamical systems theory, bifurcation and chaos theory.

As an example, let us consider the conceptual model of ocean circulation and ice dynamics sketched in Figure 8.7 and introduced by Paillard and Labeyrie¹⁴ as an element of explanation to Heinrich events mentioned earlier. The idea rests on relaxation oscillations linked to a bistable dynamics of the North-Canada (Laurentide) ice cap during a glacial period: a slow accumulation regime and a fast ice surge stage, coupled to the thermohaline circulation in the North-Atlantic ocean. The oceanic circulation is an important vector of heat transfer between low and high latitudes. It is induced by density differences due to temperature and salinity variations between large water bodies (hence the name 'thermohaline', see also Exercise 3.3.3 in Chapter 3 where this circumstance was considered as a possible source of instability).

At the level of fractions of ocean basins, water bodies can be identified by their physical properties as subsystems called 'boxes', exchanging mass, momentum, heat, salt, with their neighbors.¹⁵ Here the model has three boxes. Box (2) and (3) correspond to low latitudes, box (2) to the upper ocean that receives an amount Q_2 of solar energy, and box (3) to the deep ocean. Box (1) account for high latitude waters, extends over the whole ocean depth, and carries out the sinking of surface waters coming from the

¹³For a detailed discussion, see: G.R. North, "Lessons from energy balance models," in *Physically based modelling and simulation of climate and climatic change*, M.E. Schlesinger, Ed. (Kluwer, 1988), vol. II, pp. 627–651.

¹⁴D. Paillard and L. Labeyrie, "Role of the thermohaline circulation in the abrupt warming after Heinrich events," Nature **372** (1994) 162-164.

¹⁵Such models were introduced by H.M. Stommel, "Thermohaline convection with two stable regimes of flow," Tellus **13** (1961) 224–230. For a recent sophisticated example, see R.-X. Huang and H.M. Stommel, "Convective flow patterns in an eight-box cube driven by combined wind stress, thermal and saline forcing," J. Geophys. Res. **97** (1992) 2347–2364.



Fig. 8.7 Conceptual model of interaction between the North-Atlantic ocean and the Canadian ice cap during a glacial episode (after Paillard, 1994).

low latitudes owing to gravitational instability (warm waters become denser while cooling). An energy Q_1 is returned to space from this box. The variable m measures the intensity of the circulation through these three boxes (advection). Effective transfer coefficients describe the turbulent exchanges at their common borders.

Another element of this model is a fictitious atmosphere transporting heat and humidity towards the North. Water vapor and latent heat fluxes delivered to the ice cap are measured by variables V and L.

Finally, fresh water can be stored in the last element of the model, an ice cap characterized by its thickness H and whose evolution displays the two regimes of slow accumulation and fast surge. The change from one regime to the other is controlled by an auxiliary variable, the temperature $T_{\rm b}$ of the ice on the bedrock, that decides whether it is rough or slippery.

Once the model is constructed the rest of the study is a standard problem of dynamical systems theory, ending in a bifurcation diagram depending on the phenomenological parameters introduced.¹⁶ When the latter have realistic order of magnitudes, after elimination of the transients (that appear to be very short), one obtains a periodic behavior like that illustrated in Figure 8.8. The fresh water cycle is seen to be enslaved to the dynamics of accumulation/surge periodicity, which "explains" the sudden heating of the inland temperature of Greenland just after an event. At first sight, this seems paradoxical since the ocean has just been cooled by the melting ice but this is boosted by the thermohaline circulation that grows rapidly

 $^{^{16}}$ D. Paillard, "The hierarchical structure of glacial climatic oscillations: interaction between ice-sheets and climate," Climate Dynamics **11** (1995) 162–177 (and PhD dissertation).



Fig. 8.8 The permanent regime of the model sketched in Figure 8.7 is rapidly reached. For the considered values of its parameters, an alternation of events with different amplitudes is observed. The graph on the top line accounts for ice surges more or less visible in Figure 8.4. Notice the temporary heating when the thermohaline circulations starts again. (Data provided by D. Paillard, LSCE, CEA-Saclay)

and brings warmer waters. Different more or less simple behaviors can be obtained by tuning the models parameters (existence of oscillations with alternatively large and small amplitudes). This kind of approach may thus help us understand the stability of the global oceanic circulation and some observed counter-intuitive phenomena, provided that we accept tuning parameters appropriately.

Next, an intermediate level of modeling is possible, more detailed than box models but with still a regional geographical scope, in terms of (simplified) partial differential equations. For the ENSO phenomenon, see the review article by Neelin *et al.*¹⁷ But, in order to study the general dynamics of climate and not only a specific phenomenon, EBMs and box models are not refined enough, regional models are not global enough, so that coupled GCMs have to be considered. They explicitly solve as many processes as possible over the full range of scales available. These models are close to the primitive equations of hydrodynamics but, like in turbulence, and in an acute way, the problem of subgridscale parameterization has to be faced, *i.e.* how to account for events at scales that cannot (and will never

¹⁷J.D. Neelin, M. Latif, and F.-F. Jin, "Dynamics of coupled ocean-atmosphere models. The tropical problem," Ann. Rev. Fluid Mech. **26** (1994) 617–659.

be) represented explicitly. This irreducible part of modeling is particularly difficult because of the variety of processes acting inside each grid element.

For example, whereas a simplified description of radiative exchange within the atmosphere may be sufficient for accurate forecasts of its motion over a few days, it is not the same in the long term and a slightly defective parameterization will lead to an unacceptable systematic drift that will show up on, say, monthly averages. On the time scales at which climate evolves, these energetic exchanges are particularly sensitive to the amount of greenhouse gases (GG) and to the locally prevalent kind of clouds. Obtaining a robust parameterization of these two factors is an important challenge in view of a correct extrapolation of the present state, the more as their effects are intimately linked.

Apart from CFCs that are now banned (owing to their catalytic effects on the destruction of the stratospheric ozone), most GGs have a natural origin. However, their abundance may be largely modified by human activity. The most important GG is water vapor; carbon dioxide and methane are also important but their effects are indirect, through a positive feedback: more of them increases the temperature of air that in turn can contain more water vapor, which increases the greenhouse effect. The problem is a difficult one, because the cycle of water is complicated. To stay with the atmosphere, the greenhouse effect of clouds depends on their nature. In equatorial regions, strongly convective thick clouds made of water droplets are thought to reflect the solar radiation toward space (cooling effect). On the other hand, high altitude thin ice clouds (cirrus) should take part in the greenhouse effect by letting the solar radiation enter and stopping the infrared emission from the ground.

Other elements are equally difficult to take into account, for example dust in the high atmosphere, in general and especially after a violent volcanic eruption (cooling observed after Pinatubo's eruption already mentioned), the mechanical effects of geographical relief and vegetation cover, the level of biomass production through photosynthesis, etc.

Presently, coupling of the atmosphere and the ocean often remains a problem owing to the difference in their respective time constants. While on a given period each CGM, atmospheric or oceanic, works reasonably well when forced in the appropriate way, the coupled models may drift when they are not called back toward some observed climate state. But then, what about coping with an unknown future? As a matter of parameterization reliability test the models can be run in conditions corresponding to different reference periods, e.g. the last glacial maximum (21 000 years ago) or the climatic optimum (6000 years ago) for which geological records are available to compare with. Another type of experiment is the so-called 'doubling of the carbon dioxide concentration', for which different models with different implementations of the subgrid processes may give outputs differing by a factor of two on a global scale and even more locally, which is considerable. In the theory of dynamical systems, the behavior of the solution may change qualitatively close to a bifurcation point where the system considered looses robustness. The question therefore arises as to whether the dispersion of the results depends only quantitatively on the parameterization (that should be improved or better tuned) or it is the trace of a deeper qualitative problem relative to the operating point of the climatic system, besides its already known and much studied glacial/interglacial bistability.

The plausibility of climate evolution scenarios for the Twenty-first Century and beyond,¹⁸ taking into account the "perturbations" brought by human activity to the "natural" operation of the climate system, is a crucial stake of the development of models. Even if general circulation models, more focused on *describing* than *understanding* the machinery, seem the better placed to bring useful answers, one might think that dynamical systems theory should get better involved. Of course it applies strictly speaking to low-dimensional systems, the conceptual models alluded to above. But its demand of robustness, the distinction between enslaved modes and master modes it makes, and concept of effective dynamics that it proposes, should not be discarded a priori. It is indeed tempting to understand the climate as the effective slow dynamics of a coupled system atmosphereocean-cryosphere-biosphere whose fast variables (atmosphere, biosphere) would be eliminated to the benefit of fewer slow variables (ocean, ice). Computer technology permits us to consider ever more sophisticated models but the risk of a forward escape towards ever more complexity really exists. Dynamical theory provides an alternate framework apt to structure our reasoning and criticize it from the inside.

Difficulties that arise in the study of climate are but one example of those that appear due to nonlinearities at work in complex systems. Our limited intuition of their effects, potentially unattended and/or exacerbated, should incite us to devote a sustained attention with a tinge of humility and caution to them.

¹⁸IPCC Third Assessment Report, Climate Change 2001, Intergovernemental Panel on Climate Change, http://www.ipcc.ch, and the numerous Web sites dedicated to "global change."

8.4 Exercise: Ice ages as catastrophes

We consider an ultra-simplified model of climate attempting to describe it in terms of a single quantity: the average temperature T at the surface of the Earth.

The average temperature is determined by a balance of energy that reads:

$$C\dot{T} = R_{\rm uv} - R_{\rm ir}, \qquad (8.8)$$

which expresses the fact that the power stored in the atmosphere (l.h.s.; dimensionally, C is some effective specific heat) is the difference between the power absorbed in the short wavelengths R_{uv} and the power emitted in the long wavelengths R_{ir} .

The absorbed power R_{uv} is a fraction of the power received from the Sun, the so-called solar constant, here noted Q_0 :

$$R_{\rm uv} = Q_0(1-\alpha) \tag{8.9}$$

On general grounds, the fraction α directly reflected toward outer space, called the *albedo*, is a function of the nature of the ground and the degree of cloud cover. Here it is taken to be a function of T only. The reflection capacity of the ground is indeed mostly a function of the presence of ice or snow, itself function of the temperature. Here we assume that α is continuous and linear by parts (Sellers' hypothesis¹⁹):

$$\begin{split} T < T_{\mathbf{i}} : & \alpha = \alpha_{\mathbf{i}} , \\ T_{\mathbf{i}} < T < T_{\mathbf{s}} : & \alpha = \alpha_{\mathbf{i}} + \beta \left(T - T_{\mathbf{i}} \right) & \beta = \frac{\alpha_{\mathbf{s}} - \alpha_{\mathbf{i}}}{T_{\mathbf{s}} - T_{\mathbf{i}}} , \\ T_{\mathbf{s}} < T : & \alpha = \alpha_{\mathbf{s}} . \end{split}$$

Taking into account previous remarks, one expects $\alpha_{\rm s} < \alpha_{\rm i}$. Constant $T_{\rm i}$ corresponds to a temperature below which snowy ground is surely found $(\alpha_{\rm i} \text{ high})$ and $T_{\rm s}$ a temperature above which the ground is free of ice or snow $(\alpha_{\rm s} \text{ low})$.

One could think that the Earth behaves as a black body that radiates according to the Stephan law $R_{\rm ir} \propto \sigma T^4$ but the actual situation is more complex since the atmosphere is not transparent to IR radiations (greenhouse gases absorb part of this radiation). The Stephan law is therefore

 $^{^{19}}$ W.D. Sellers, "A global climate model based on the energy balance of the earth-atmosphere system," J. Appl. Meteor. 8 (1969) 396-400.

replaced by the simplified relation

$$R_{\rm ir} = a + b(T - T_{\rm i}), \qquad (8.10)$$

where a and b are two empirical constants (Budyko's hypothesis²⁰). We assume b > 0 so that R_{ir} remains a growing function of T.

1) The parameters in (8.9) being given, draw on the same figure R_{uv} and R_{ir} as functions of T and discuss the number of fixed points of system (8.8–8.10) graphically, depending on the value of a at fixed b, $b < \beta Q_0$ (the variation of a could be due to some extrinsic, e.g. anthropogenic, modification of the greenhouse effect).

2) Discuss this again while assuming that the parameters of $R_{\rm ir}$ are fixed and that $R_{\rm uv}$ varies through Q_0 (e.g. astronomic forcing).

3) Examine the stability of the different solutions to (8.8) when it has three roots (the difference $R_{uv} - R_{ir}$ will be drawn as a function of T). Discuss the physical origin of the mechanisms that guarantee the stability of extreme solutions (*i.e.* interpret the negative character of feedbacks).

4) Show that when the orbital parameters (term Q_0) or the greenhouse effect (coefficient a) vary with a sufficient amplitude but quasi-statically (*i.e.* slowly enough so that T is the steady-state solution to Problem (8.8) at the corresponding instantaneous value of its parameters) the climate of the Earth can pass from a cold solution (ice-age) to a warm solution (interglacial state) and vice versa.

The Budyko–Sellers model (8.8-8.10), completed by a noise term, was used in the specific work mentioned in Note 9 to exemplify stochastic resonance.

 $^{^{20}}$ M.I. Budyko, "The effect of solar radiation variations on the climate of the earth," Tellus **21** (1969) 611-619.

This page intentionally left blank

Appendix A

Linear Algebra

We recall here a few elements of linear algebra frequently used in applications. Two perspectives are taken, the first one focuses on the decomposition of vectors on eigen-bases, at the heart of linear stability theory (\S A.2), the other tackles essentially the same questions but from an "energetic" point of view via the definition of appropriate scalar products (\S A.3). Beforehand, Section A.1 is devoted to the enumeration of a few elementary properties linked to the matrix representation of operators and changes of bases. Among the many reference books on the subject, let us just mention [Hirsch and Smale (1974)] that fits our purpose particularly well.

A.1 Vector Spaces, Bases, and Linear Operators

We suppose that the definition and immediate properties of vector spaces are known. So, let us consider such a space, noted \mathbb{X} with elements (vectors) **X**. Scalars x entering linear combinations can be real or complex. To every vector space on \mathbb{R} , one can associate a complex extension formed with the same vectors but in which the scalars (and thus also the components of the vectors) are in \mathbb{C} .

A vector $\mathbf{X} \in \mathbb{X}$ can be specified by its components in a given basis $\{\mathbf{E}_i, i = 1, \ldots, d\}$:

$$\mathbf{X} = \sum_{i=1}^{d} x_i \mathbf{E}_i \,, \tag{A.1}$$

where d is the dimension of X. A linear form is a linear function that associates a scalar to every vector in the space. The set of forms also has a vector space structure called the *dual space*. Forms extracting every component of the vectors on a given basis makes the *dual basis*.

A linear operator \mathcal{L} can be defined by its action on the basis vectors:

$$\mathcal{L}\mathbf{E}_{j}=\sum_{i}l_{ij}\mathbf{E}_{i}\,,$$

which defines a $(d \times d)$, two-dimensional array called a *matrix*, here denoted [L]; by convention, the first (second) subscript labels the lines (columns). The column vectors of [L] are the images of the basis vectors. The result of the action of \mathcal{L} on an arbitrary vector then reads

$$y_j = (\mathcal{L}\mathbf{X})_j = \sum_i l_{ji} x_i \,. \tag{A.2}$$

The representation of a linear operator changes with the basis. Let us define the new basis vectors $\{\mathbf{F}_j, j = 1, ..., d\}$ by their components in the old basis through:

$$\mathbf{F}_j = \sum_i t_{ij} \mathbf{E}_i \,. \tag{A.3}$$

This builds a matrix [T] with elements t_{ij} representing the operator \mathcal{T} expressing the change of basis. \mathcal{T} is a linear *invertible* operator, whose column vectors are the components of the new basis vectors in the old basis. Components $\{x_i\}$ and $\{x'_i\}$ of some given vector \mathbf{X} in the two bases $\{\mathbf{E}_i\}$ and $\{\mathbf{F}_i\}$ are then related by:

$$x_i = \sum_j t_{ij} x'_j \,, \tag{A.4}$$

while the new components are given as a function of the old ones by:

$$x'_j = \sum_k u_{jk} x_k \,, \tag{A.5}$$

where the u_{ij} are the elements of the matrix [U] inverse of [T] and denoted $[T]^{-1}$. For this reason, the components x_i are said to be *contra-variant*.

Concretely, the inverse of a given square matrix [A] can be obtained by hand as the transposed of the matrix of cofactors, divided by the determinant det([A]) of the matrix. The cofactor of element a_{ij} is $\alpha_{ij} = (-1)^{i+j} \det([\tilde{A}_{ij}])$, where $[\tilde{A}_{ij}]$ is the $(d-1) \times (d-1)$ matrix obtained by suppressing line *i* and column *j* of matrix [A]. Matrix [U] can be obtained from [T] in this way.

From (A.1), (A.3) and (A.4), one readily gets:

$$y'_j = \sum_k u_{jk} \sum_l l_{kl} \sum_i t_{li} x'_i = \sum_i l'_{ji} x'_i,$$

hence

$$l'_{ji} = \sum_{k} \sum_{l} u_{jk} l_{kl} t_{li} , \qquad (A.6)$$

so that the elements of the matrix representing \mathcal{L} in the new basis are given by $l'_{li} = \sum_k \sum_j u_{lk} m_{kj} t_{ji}$, *i.e.*

$$[L'] = [T]^{-1}[L][T].$$

Matrices [L'] et [L] are said to be *similar*. It is easy to see that

$$[L']^k = [T]^{-1}[L]^k[T].$$

Remark. While developing Quantum Mechanics, Dirac introduced notations useful to linear algebra. In his setting, vectors are denoted as $|\mathbf{X}\rangle$ and linear forms as $\langle \mathbf{Y}|$. Let $\{|\mathbf{E}_i\rangle, i = 1, ..., d\}$ be a basis and $\{\langle \mathbf{E}_i|, i = 1, ..., d\}$ the dual basis. The projector on basis vector $|\mathbf{E}_i\rangle$ reads $|\mathbf{E}_i\rangle\langle\mathbf{E}_i|$, and the fact that the basis is complete leads to $\sum_i |\mathbf{E}_i\rangle\langle\mathbf{E}_i| = \mathcal{I}$, where \mathcal{I} is the identity operator in X. Coming back to (A.1), one gets $|\mathbf{X}\rangle = \left[\sum_i |\mathbf{E}_i\rangle\langle\mathbf{E}_i|\right] |\mathbf{X}\rangle$, so that, by identification, $\langle \mathbf{E}_i|\mathbf{X}\rangle$ is the *i*th contra-variant component of $|\mathbf{X}\rangle$.

The action of the linear operator \mathcal{L} on a vector $|\mathbf{X}\rangle$ gives the vector

$$|\mathbf{Y}\rangle = \mathcal{L}|\mathbf{X}\rangle = \left[\sum_{j} |\mathbf{E}_{j}\rangle\langle\mathbf{E}_{j}|\right] \mathcal{L}\left[\sum_{i} |\mathbf{E}_{i}\rangle\langle\mathbf{E}_{i}|\right] |\mathbf{X}\rangle,$$

which allows the identification $l_{ji} = \langle \mathbf{E}_j | \mathcal{L} | \mathbf{E}_i \rangle$, cf. (A.2).

In a change of basis, vectors of the new basis are defined from the old one by $|\mathbf{F}_j\rangle = \sum_i |\mathbf{E}_i\rangle \langle \mathbf{E}_i |\mathbf{F}_j\rangle$, which defines operator \mathcal{T} . Its inverse $\mathcal{U} = \mathcal{T}^{-1}$ is represented by the matrix with elements $u_{ji} = \langle \mathbf{F}_j | \mathbf{E}_i \rangle$. The transformation (A.4) of the components of vector $|\mathbf{X}\rangle$ is then immediately obtained. In the same way the elements of the matrix representing \mathcal{L} in basis $\{|\mathbf{F}_j\rangle\}$ are given by: $\langle \mathbf{F}_j | \mathcal{L} | \mathbf{F}_i \rangle = \sum_k \sum_l \langle \mathbf{F}_j | \mathbf{E}_k \rangle \langle \mathbf{E}_k | \mathcal{L} | \mathbf{E}_l \rangle \langle \mathbf{E}_l | \mathbf{F}_i \rangle$, which is easily identified to (A.6). At this stage Dirac's notations are essentially of mnemotechnical interest.

MATLAB notations can also be useful. In this framework, all kinds of arrays are placed between square brackets. The comma (or a blank space) is used to separate elements in a line, the semi-colon to indicate the line change (equivalent to a 'carriage return'). A vector, traditionally represented in column form is thus a $(d \times 1)$ array $\mathbf{V} = [V_1; V_2; ...]$. The line representation, transposed of the column representation then reads: $[V_1, V_2, ...] \equiv [V_1; V_2; ...]^t$. With these conventions, the canonical scalar product can be written $\mathbf{V} \cdot \mathbf{W} = [V_1, V_2, ...][W_1; W_2; ...] = \sum V_k W_k$. By contrast the tensorial product of two vectors $\mathbf{V} \otimes \mathbf{W}$ is a $d \times d$ array with elements $V_i W_j$, which one can write $[V_1; V_2; ...][W_1, W_2, ...]$.

A.2 Structure of a Linear Operator

As we have seen in Chapter 2, and more specifically in Section 2.2, the solution to the initial value problem for a linear differential system:

$$\dot{\mathbf{X}} = \mathcal{L}\mathbf{X}, \quad \mathbf{X}(t=0) = \mathbf{X}_0,$$
 (A.7)

leads to the evaluation of

$$\mathbf{X}(t) = \exp(t\mathcal{L})\mathbf{X}_0 \tag{A.8}$$

with, by definition:

$$\exp \mathcal{Z} \equiv \mathcal{I} + \sum_{k=1}^{\infty} \frac{1}{k!} \mathcal{Z}^k$$
(A.9)

for an arbitrary linear operator \mathcal{Z} . The computation of this exponential is made easier by resolving the structure of the operator \mathcal{Z} , that is to say by decomposing the full space X into a direct sum of nontrivial *invariant* subspaces.

One says that a vector subspace \mathbb{X}' of \mathbb{X} is invariant for \mathcal{L} if the image of every $\mathbf{X} \in \mathbb{X}'$ is in \mathbb{X}' . Trivial invariant subspaces are \mathbb{X} itself, the null space \mathbb{O} , the image of \mathbb{X} by \mathcal{L} (set of vectors $\mathbf{Y} = \mathcal{L}\mathbf{X}$ for all $\mathbf{X} \in \mathbb{X}$) and the kernel of \mathcal{L} (set of vectors $\mathbf{Z} \in \mathbb{X}$ such that $\mathcal{L}\mathbf{Z} = \mathbf{0}$).

The kernel of \mathcal{L} is reduced to \mathbb{O} when \mathcal{L} is invertible, which is the case if its determinant is nonzero. One then says that the matrix has 'maximal rank'. (The rank of a general matrix, *i.e.* not necessarily square, is the dimension of the largest square sub-matrix with nonzero determinant that can be extracted from it by suppressing lines and columns.)

The simplification brought by the so-obtained decomposition of space \mathbb{X} is apparent through the change of basis that makes it explicit: Let $\mathbb{X} = \mathbb{X}_1 \oplus \mathbb{X}_2$ and \mathcal{L} the operator for which \mathbb{X}_1 is invariant, then for $\mathbf{X}_1 \in \mathbb{X}_1$, $\mathcal{L}\mathbf{X}_1 \in \mathbb{X}_1$ and, in a basis of \mathbb{X} formed with a basis of \mathbb{X}_1 appropriately completed by a basis in the supplementary space \mathbb{X}_2 , \mathcal{L} takes on a block structure:

$$\mathcal{L} = egin{bmatrix} \mathcal{L}_{11} & \mathcal{L}_{12} \ \mathcal{O} & \mathcal{L}_{22} \end{bmatrix},$$

where \mathcal{O} is the null operator ($\mathcal{O}\mathbf{X} = \mathbf{0}$ whatever \mathbf{X}). If in addition \mathbb{X}_2 is

also invariant, \mathcal{L} acquires a block-diagonal structure:

$$\mathcal{L} = \begin{bmatrix} \mathcal{L}_{11} & \mathcal{O} \\ \mathcal{O} & \mathcal{L}_{22} \end{bmatrix} \,.$$

The eigen-direction attached to an eigenvalue s of \mathcal{L} , such that:

$$\mathcal{L}\mathbf{X} = s\mathbf{X}$$

is the prototype of the sought invariant subspaces (hence the kernel of operator $\mathcal{M} = \mathcal{L} - s\mathcal{I}$). The eigenvalues are the roots of the *characteristic polynomial* obtained by expanding the determinant:

$$\det(\mathcal{L}-s\mathcal{I})=0$$
 .

The fundamental theorem of algebra asserts that, in dimension d, this degree-d polynomial has d roots in \mathbb{C} , possibly degenerate. One can thus write:

$$\det(\mathcal{L} - s\mathcal{I}) = a_0 + a_1 s + \ldots + a_{d-1} s^{d-1} + s^d = 0 = \prod_j (s - s_j)^{d_j},$$

where d_j is the multiplicity of eigenvalue s_j , with $\sum_i d_j = d$.

In order to treat complex eigenvalues $(s \in \mathbb{C})$ of a linear operator acting in a vector space on \mathbb{R} , one must work with its *complex extension*. For example a rotation in a two-dimensional subspace is associated to a pair of simple conjugate purely imaginary roots, solutions to $s^2 + \omega^2 = 0$, and, though there are no real eigenvectors, there are two eigen-directions in the complex extension.

In the general case, at least one eigenvector $\widehat{\mathbf{X}}_j$ can be found for a given (real or complex) eigenvalue s_j in the complex extension. If s_j is non degenerate, the associated eigen-subspace is one-dimensional. It is generated by the eigenvector $\widehat{\mathbf{X}}_j$ solution to $(\mathcal{L} - s_j \mathcal{I}) \widehat{\mathbf{X}}_j = \mathbf{0}$ (see Exercise A.4).

If s_j is degenerate, which corresponds in dynamics to a linear resonance condition, the problem is more complicated and one must search for a special basis in which the matrix representing the restriction of the operator to this subspace is in its *normal* form.

A.2.1 Jordan normal form

The main tool to determine the Jordan normal form of an operator is the theory of operator polynomials, *i.e.* polynomials in the form $\mathcal{Q} = \sum_{n=0}^{n_{\max}} a_n \mathcal{L}^n$. The decomposition of the vector space X into a direct sum of

invariant subspaces rests on the determination of the kernels of operators $(\mathcal{L} - s_j \mathcal{I})^{d'}$, where d' is a trial dimension, as a generalization of the eigendirection defined as the kernel of $\mathcal{L} - s_j \mathcal{I}$. These kernels are called the *generalized eigen-subspaces* associated to eigenvalue s_j .

Without entering the derivation, we now give some practical results about them. First they form a series of embedded subspaces of increasing dimensions, the kernel of $(\mathcal{L} - s\mathcal{I})^2$ containing that of $\mathcal{L} - s\mathcal{I}$, etc. The largest of them, called the *principal subspace*, has dimension d_j equal to the multiplicity of the eigenvalue, it is the kernel of $(\mathcal{L} - s_j\mathcal{I})^{d_j}$. Next, the Cayley-Hamilton theorem stipulates that \mathcal{L} fulfills its own characteristic polynomial, that is:

$$\prod_j (\mathcal{L} - s_j \mathcal{I})^{d_j} \equiv 0 \,,$$

which can be understood as the most direct translation of this decomposition into a direct sum of principal subspaces.

The numerical approach to multiples eigenvalues as it is implemented in usual computer routines can lead to difficulties in concrete cases. It may therefore be interesting to present the main analytical steps of the computation. Let X_j be the principal subspace associated to s_j , with dimension d_j , and \mathcal{L}_j the restriction of \mathcal{L} to X_j . Define also $\mathcal{M}_j = \mathcal{L}_j - s_j \mathcal{I}_{d_j}$, where \mathcal{I}_{d_j} is the identity operator in X_j . Then:

• The index of a principal vector $\mathbf{X} \ (\in \mathbb{X}_j)$ is defined as the smallest nonnegative integer m such that $\mathcal{M}_j^m \mathbf{X} = 0$. This index m is such that $1 \le m \le d_j$; the lower bound m = 1 corresponds to the case of the eigenvector ($\mathcal{M}\mathbf{X} = 0$); the upper bound d_j is because \mathbf{X} belongs to \mathbb{X}_j by assumption ($\mathcal{M}_j^{d_j}\mathbf{X} = 0$). It can easily be seen that when \mathbf{X} is a vector with index m then $\mathcal{M}_j^n \mathbf{X}$ as index m - n for all n < m.

• Exploring X_j , one begins with finding the maximum index m_{\max} of vectors in this subspace by computing the kernel of $\mathcal{M}^{d'}$ for increasing trial dimensions d'. When $m_{\max} = d_j$, one gets a vector with maximal index \mathbf{Y}_{d_j} by explicitly solving the linear problem

$$(\mathcal{L}_j - s_j \mathcal{I}_{d_j})^{d_j} \mathbf{X} = 0.$$

• One then considers the series of d_j vectors \mathbf{Y}_{d_j} , $\mathbf{Y}_{d_j-1} = \mathcal{M}_j \mathbf{Y}_{d_j}$, $\mathbf{Y}_{d_j-2} = \mathcal{M}_j \mathbf{Y}_{d_j-1} = \mathcal{M}_j^2 \mathbf{Y}_{d_j}$, ..., $\mathbf{Y}_1 = \mathcal{M}_j \mathbf{Y}_2 = \mathcal{M}_j^{d_j-1} \mathbf{Y}_{d_j}$, that one can write in the more condensed form:

$$\left\{\mathbf{Y}_{d_j-n} = \mathcal{M}_j^n \mathbf{Y}_{d_j}; \quad n = 0, 1, \dots, d_j - 1\right\},$$

noticing that $\mathbf{Y}_0 = \mathcal{M}_j \mathbf{Y}_1 = \mathcal{M}_j^{d_j} \mathbf{Y}_{d_j} \equiv 0$ since \mathbf{Y}_{d_j} has maximal index by assumption. The series of vectors $\{\mathbf{Y}_1, \ldots, \mathbf{Y}_{d_j}\}$ is an acceptable basis for the principal subspace \mathbb{X}_j attached to s_j since one has d_j linearly independent vectors. This property can easily be checked by showing that any linear combination $\mathbf{Z} = \sum_{n=1}^{d_j} \mu_n \mathbf{Y}_n$ cancels if and only if all its coefficients cancel. Computing first $\mathcal{M}_j^{d_j-1}\mathbf{Z}$, one indeed gets a single non-identically zero term $\mu_{d_j}\mathbf{Y}_{d_j}$, which implies $\mu_{d_j} = 0$. Next, the computation of $\mathcal{M}^{d_j-2}\mathbf{Z}$ gives $\mu_{d_j-1} = 0$, and so on, down to μ_1 .

• The Jordan normal form directly derives from the matrix representation of \mathcal{L}_j in this basis: Vector \mathbf{Y}_1 (index 1) is along eigenvector $\widehat{\mathbf{X}}_j$ and, by construction of the \mathbf{Y}_n , one gets:

$$\mathbf{Y}_{n-1} = (\mathcal{L}_j - s_j \mathcal{I}_{d_j}) \mathbf{Y}_n \,,$$

that is:

$$\mathcal{L}_j \mathbf{Y}_n = s_j \mathbf{Y}_n + \mathbf{Y}_{n-1}$$
.

The expression of the elementary Jordan block directly derives from this expression:

	n-1	\boldsymbol{n}	n + 1	
(
	0	:		
	1	0	:	
n-1	s_{j}	1	0	
n	0	s_j	1	
n+1	÷	0	s_j	
		÷	0	
(

Introducing the operator \mathcal{N}_{d_j} represented in this basis by a matrix with a series of ones just above the diagonal and zeroes everywhere else, one gets

$$\mathcal{L}_j = s_j \mathcal{I}_{d_j} + \mathcal{N}_{d_j}$$
.

• When the maximal index is such that $m_{\max} < d_j$, one first constructs the subspace generated by the \mathbf{Y}_n just found. This subspace has dimension m_{\max} , and one has to redo the same work in the supplementary subspace with dimension $d_j - n_{\max}$, *i.e.* find the maximal index of vectors in this new



Fig. A.1 (a) Jordan block structure in the basis of generalized eigenvectors, example: s_1 is non-degenerate; s_2 is double but without diagonal form; s_3 is with multiplicity four, decomposed into two subspaces with dimensions one and three; s_4 is also with multiplicity four (zero entries are left blank). (b) Possible cases in dimension d = 3: Three distinct eigenvalues, possibly only in the complex extension if two eigenvalues out of the three are complex conjugate (b1). Two real distinct eigenvalues, in diagonal form (b2) or not (b3). Eigenvalue with multiplicity three and vectors with maximum index $n_{\max} = 1, 2, \text{ or } 3$, giving three Jordan blocks of order 1 (b4), one of order 1 and one of order 2 (b5), or one of order 3 (b6).

subspace, find one vector with maximal index, build the associated basis, etc. up to the point where the full subspace X_j is decomposed. Using the so-obtained partial bases to build the basis of X_j one can represent \mathcal{L}_j as a direct sum of Jordan blocks aligned along the diagonal. This result is illustrated in Figure A.1(a). In applications, one rarely deals with problems in a dimension higher than three analytically. The work involved in this systematic approach is therefore never as fastidious as it could seem. Figure A.1(b) illustrates all the possible cases for d = 3. Notice also that in case (b1), when the three distinct roots are s_1 plus a pair of complex eigenvalues $s_{\pm} = \sigma \pm i\omega$, the diagonal form holds in the complex extension only, otherwise one has a square block, *e.g.*

$$\begin{bmatrix} s_1 & 0 & 0 \\ 0 & \sigma & -\omega \\ 0 & +\omega & \sigma \end{bmatrix}.$$

The two-dimensional case was illustrated in Chapter 2 without mentioning the index concept. This gap is filled in Exercise A.4.2.

A.2.2 Exponential of a matrix

As far as the linear dynamics evolution governed by (A.7) is concerned, the aim of all that precedes is to turn operator \mathcal{L} into the best adapted form in view of the computation of its exponential (A.8) defined as a power series (A.9). In the basis where \mathcal{L} is in its Jordan form one gets:

$$\exp(t\mathcal{L}_j) = \exp\left[t\left(s_j\mathcal{I}_{d_j} + \mathcal{N}_{d_j}\right)\right] = \exp\left(ts_j\mathcal{I}_{d_j}\right)\exp\left(t\mathcal{N}_{d_j}\right)$$

Here the exponential of the sum is simply the product of the exponentials because \mathcal{I}_{d_j} commutes with any operator and thus also with $\mathcal{N}_{d_j}^n$. One obviously gets $\exp(ts_j\mathcal{I}_{d_j}) = \exp(ts_j)\mathcal{I}_{d_j}$, and one is left with the computation of the second exponential, which is an easy matter since operator \mathcal{N}_{d_j} is *nilpotent*, *i.e.*, $(\mathcal{N}_{d_j})^{d_j} = \mathcal{O}$, so that we are left with a finite number of terms in the power series. Considering the case $d_j = 3$ as an example and dropping all useless indices, we get:

$$\mathcal{N} = \begin{bmatrix} 0 & 1 & 0 \\ 0 & 0 & 1 \\ 0 & 0 & 0 \end{bmatrix}, \quad \mathcal{N}^2 = \begin{bmatrix} 0 & 0 & 1 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{bmatrix}, \quad \mathcal{N}^3 = \begin{bmatrix} 0 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{bmatrix},$$

so that

$$\exp(t\mathcal{N})=\mathcal{I}+t\,\mathcal{N}+rac{1}{2}t^2\mathcal{N}^2=egin{bmatrix} 1&t&rac{1}{2}t^2\ 0&1&t\ 0&0&1 \end{bmatrix}$$

and thus

$$\exp(t\mathcal{L}) = \begin{bmatrix} \exp(st) & t \, \exp(st) & \frac{1}{2}t^2 \, \exp(st) \\ 0 & \exp(st) & t \, \exp(st) \\ 0 & 0 & \exp(st) \end{bmatrix}$$

The terms $(t^j/j!) \exp(st)$ are said to be *secular*. They account for a slow algebraic drift with respect to a dominant exponential behavior.

A.2.3 Perturbation of a linear problem

Degeneracy of eigenvalues is a singularity that usually demands to be raised by introducing perturbations in the most general way. It turns out that, the most general perturbation $\delta \mathcal{L}$ brought to some operator \mathcal{L} , *i.e.* $\mathcal{L}' = \mathcal{L} + \delta \mathcal{L}$ may be decomposed in two parts, one that does not change its structure and another one ("true" perturbations) that does something nontrivial, either by changing its eigenvalues or the fact that it can be turned into diagonal form or not.

Examples of operations that do nothing to a linear system are changes of bases, and more generally invertible transformations \mathcal{T} , such that $\mathcal{L}' =$ $\mathcal{T}^{-1}\mathcal{L}\mathcal{T}$, in which case \mathcal{L} and \mathcal{L}' are said to be *similar*. In such operations, the characteristic polynomial is indeed left invariant since:

$$det [(\mathcal{T}^{-1}\mathcal{L}\mathcal{T}) - s\mathcal{I}] = det [\mathcal{T}^{-1}(\mathcal{L} - s\mathcal{I})\mathcal{T}]$$

= det [$\mathcal{T}\mathcal{T}^{-1}(\mathcal{L} - s\mathcal{I})$] = det($\mathcal{L} - s\mathcal{I}$)

Only the part of perturbation $\delta \mathcal{L}$ that makes \mathcal{L}' depart from staying similar to \mathcal{L} is of interest. The complementary part, which leaves them similar, is sought for as a transformation \mathcal{T} close to identity:

$$\mathcal{T} = \mathcal{I} + \Gamma$$

i.e., with γ_{ij} small so that \mathcal{T} remains invertible. The transformed operator is then given by

$$\mathcal{L}' = \mathcal{T}^{-1}\mathcal{L}\mathcal{T} = (\mathcal{I} + \Gamma)^{-1}\mathcal{L}(\mathcal{I} + \Gamma).$$

We need the inverse of $\mathcal{I} + \Gamma$ but, by identification, it can be checked that

$$(\mathcal{I} + \Gamma)^{-1} = \mathcal{I} - \Gamma + \Gamma^2 - \dots + (-1)^k \Gamma^k + \dots$$

so that neglecting all terms beyond first order in Γ , by substitution and in full generality we get:

$$\mathcal{L}' = \mathcal{L} + (\Gamma \mathcal{L} - \mathcal{L} \Gamma).$$

The operator $\Gamma \mathcal{L} - \mathcal{L}\Gamma = [\Gamma, \mathcal{L}]$ is called the *commutator* of \mathcal{L} and Γ . The "true" perturbation is therefore $\delta \mathcal{L} - [\Gamma, \mathcal{L}]$ and one must next ask for the number of independent parameters on which it depends.

Nontrivial answers are already found in dimension two. Let

$$\mathcal{L}' = \mathcal{L} + \delta \mathcal{L},$$

where $\delta \mathcal{L}$ is represented by the most general matrix:

$$\delta \mathcal{L} = \begin{bmatrix} \delta l_{11} & \delta l_{12} \\ \delta l_{21} & \delta l_{22} \end{bmatrix} = \delta l_{11}[\mathbf{U}_1] + \delta l_{12}[\mathbf{U}_2] + \delta l_{21}[\mathbf{U}_3] + \delta l_{22}[\mathbf{U}_4],$$

where

$$\left\{ \begin{bmatrix} \mathbf{U}_1 \end{bmatrix} = \begin{bmatrix} 1 & \mathbf{0} \\ \mathbf{0} & \mathbf{0} \end{bmatrix}; \quad \begin{bmatrix} \mathbf{U}_2 \end{bmatrix} = \begin{bmatrix} \mathbf{0} & \mathbf{1} \\ \mathbf{0} & \mathbf{0} \end{bmatrix}; \quad \begin{bmatrix} \mathbf{U}_3 \end{bmatrix} = \begin{bmatrix} \mathbf{0} & \mathbf{0} \\ \mathbf{1} & \mathbf{0} \end{bmatrix}; \quad \begin{bmatrix} \mathbf{U}_4 \end{bmatrix} = \begin{bmatrix} \mathbf{0} & \mathbf{0} \\ \mathbf{0} & \mathbf{1} \end{bmatrix} \right\}$$

A. Linear Algebra

form a canonical basis of the space of (2×2) matrices. Operator Γ is further defined by

$$\boldsymbol{\Gamma} = \begin{bmatrix} \gamma_{11} & \gamma_{12} \\ \gamma_{21} & \gamma_{22} \end{bmatrix}, \qquad (A.10)$$

whose coefficients are adjustable parameters (by contrast with the δl_{ij} which are given).

Let us first consider the case when \mathcal{L} can be turned into diagonal form:

$$\mathcal{L} = \begin{bmatrix} s_1 & 0 \\ 0 & s_2 \end{bmatrix}$$
.

Computation of the commutator leads to:

$$[\mathbf{\Gamma}, \mathcal{L}] = \begin{bmatrix} 0 & \gamma_{12}(s_2 - s_1) \\ \gamma_{21}(s_1 - s_2) & 0 \end{bmatrix},$$

and therefore

$$\delta \mathcal{L} - \left[\, \mathbf{\Gamma}, \mathcal{L} \,
ight] = egin{bmatrix} \delta l_{12} & \delta l_{12} - \gamma_{12}(s_2 - s_1) \ \delta l_{21} & -\gamma_{21}(s_1 - s_2) & \delta l_{22} \end{bmatrix} \, .$$

When $s_1 = s_2$, \mathcal{L} is a multiple of \mathcal{I} and commutes with any Γ . The part of $\delta \mathcal{L}$ that derives from a similarity is thus identically zero so that any perturbation $\delta l_{jj'} \neq 0$ modifies the dynamics, generically by raising the degeneracy (as an exercise one can look for cases where only the diagonalizable character is broken)

By contrast when $s_1 \neq s_2$, there exists a continuous two-parameter family of perturbations that gives an operator similar to the original one since one can choose $\gamma_{jj'} = \delta l_{jj'}/(s_{j'} - s_j)$ for (jj') = (12) or (21). True perturbations form a complementary two-parameters family $(\delta l_{11}, \delta l_{22})$ that modifies each eigenvalue separately.

Things are a little less trivial when \mathcal{L} cannot be turned to diagonal form, but only into the Jordan form:

$$\mathcal{L} = \begin{bmatrix} ar{s} & 1 \\ 0 & ar{s} \end{bmatrix}$$

The commutator then reads:

$$\left[\, oldsymbol{\Gamma}, oldsymbol{\mathcal{L}} \,
ight] = egin{bmatrix} -\gamma_{21} & \gamma_{11} - \gamma_{22} \ 0 & \gamma_{21} \end{bmatrix}$$

and we see that γ_{12} has disappeared, so that there remains only two parameters γ_{21} and $\gamma_{11} - \gamma_{22}$ to cancel as many terms as possible in:

$$oldsymbol{\delta L} - \left[\, oldsymbol{\Gamma}, \mathcal{L} \,
ight] = egin{bmatrix} \delta l_{11} + \gamma_{21} & \delta l_{12} - (\gamma_{11} - \gamma_{22}) \ \delta l_{21} & \delta l_{22} - \gamma_{21} \end{bmatrix}.$$

The term δl_{12} can always be suppressed by choosing $\gamma_{11} - \gamma_{22}$ appropriately, and there is enough freedom to cancel one of the two other terms, either δl_{11} or δl_{22} , whereas term δl_{21} can never be suppressed. The most general perturbation that *unfolds the singularity* of this degenerate problem can thus be taken as

$$\mathcal{L}' = \mathcal{L} - \eta \mathcal{U}_3 - \eta' \mathcal{U}_4 \equiv \begin{bmatrix} \bar{s} & 1 \\ -\eta & \bar{s} - \eta' \end{bmatrix},$$
 (A.11)

i.e. $\delta l_{21} = -\eta$ and $\delta l_{22} = -\eta'$ where the minus signs are introduced for convenience. The characteristic polynomial then reads:

$$(s-\bar{s})(s-\bar{s}+\eta')+\eta = s^2 - (2\bar{s}-\eta')s + \bar{s}(\bar{s}-\eta')+\eta = 0$$

The discriminant of this quadratic polynomial is $\Delta = {\eta'}^2 - 4\eta$. How the degeneracy is raised is illustrated in Figure A.2, and one can see that, depending on the path in the parameter space, one can get either real or complex eigenvalues, which account for the dynamics in the vicinity of an improper node (Figure 2.7(b), p. 41): trajectories resemble those close to a focus as well as close to a node (Figure 2.5(a) and Fig. 2.6(a)).



Fig. A.2 Raising the degeneracy at a double non-diagonalizable eigenvalue. The most general perturbation depends on parameters η and $\eta' \ (\ll |s|)$. Γ is a path in parameter space.

The generalization in dimension d:

	\bar{s}	1	0		0	0]
	0	\bar{s}	1	•••	0	0
	:	:	÷	۰.	÷	÷
	0	0	0	•••	$ar{s}$	1
,	$\lfloor -\eta_1 \rfloor$	$-\eta_2$	$-\eta_3$	• • •	$-\eta_{d-1}$	$\bar{s} - \eta_d$

is called the Jordan-Arnold normal form.

A.3 Metric Properties of Linear Operators

A.3.1 Scalar products, adjoints, normal and non-normal operators

The definition of a *scalar product* adds an Euclidean structure to vector spaces on \mathbb{R} . A scalar product is a bilinear form with general expression

$$\mathcal{E}(\mathbf{X},\mathbf{Y}) = \sum_i \sum_j g_{ij} x_i y_j \, ,$$

that is symmetric $(g_{ij} = g_{ji})$ and such that $\mathcal{E}(\mathbf{X}, \mathbf{X})$ is definite positive, *i.e.* $\mathcal{E}(\mathbf{X}, \mathbf{X}) \geq 0$ and $\mathcal{E}(\mathbf{X}, \mathbf{X}) = 0$ if and only if $\mathbf{X} = \mathbf{0}$.

The canonical scalar product relative to a given basis simply reads:

$$\mathcal{E}(\mathbf{X},\mathbf{Y}) = \sum_{i} x_i y_j$$

By construction, the basis is orthonormal with respect to the so-defined scalar product since one has: $\mathcal{E}(\mathbf{E}_i, \mathbf{E}_j) = \delta_{ij}$, where δ_{ij} is the Kronecker symbol ($\delta_{ii} = 1$ and $\delta_{ij} = 0$ for $i \neq j$).

The elements of the matrix [T] representing the change from an orthonormal basis $\{\mathbf{E}_i\}$ to another $\{\mathbf{F}_j\}$ are given by $t_{ij} = \mathcal{E}(\mathbf{E}_i, \mathbf{F}_j)$, those of the inverse matrix [U] by

$$u_{ij} = \mathcal{E}(\mathbf{F}_i, \mathbf{E}_j) = \mathcal{E}(\mathbf{E}_j, \mathbf{F}_i) = t_{ji}.$$

Such matrices are said to be orthogonal.

In the case of vector spaces on \mathbb{C} , the scalar product is defined through a Hermitian form

$$\mathcal{H}(\mathbf{X},\mathbf{Y}) = \sum_i \sum_j g_{ij} x_i^* y_j \, ,$$

whose coefficients g_{ij} now fulfill $g_{ij} = g_{ji}^*$, so that

$$\mathcal{H}(\mathbf{X},\mathbf{Y}) = \mathcal{H}(\mathbf{Y},\mathbf{X})^*.$$

The quadratic form $\mathcal{H}(\mathbf{X}, \mathbf{X})$ must also be definite positive (Hermitian norm). The canonical scalar product relative to a given basis reads:

$$\mathcal{H}(\mathbf{X},\mathbf{Y}) = \sum\nolimits_i x_i^* y_j$$
 .

The change from an orthonormal basis to another one yields

$$u_{ij} = \mathcal{H}(\mathbf{F}_i, \mathbf{E}_j) = \mathcal{H}(\mathbf{E}_j, \mathbf{F}_i)^* = t_{ji}^*.$$

Such transformations are said to be unitarian.

Let us consider the general case of a Hermitian scalar product. An operator \mathcal{L} being given, its *adjoint* \mathcal{L}^{\dagger} is defined by

$$\mathcal{H}(\mathbf{Y}, \mathcal{L}^{\dagger}\mathbf{X}) = \mathcal{H}(\mathcal{L}\mathbf{Y}, \mathbf{X}) = \mathcal{H}(\mathbf{X}, \mathcal{L}\mathbf{Y})^{*}.$$
(A.12)

Adjoint operators are thus represented by matrices such that: $(l^{\dagger})_{ij} = l_{ji}^{*}$.

An operator \mathcal{L} is said to be *normal* if it commutes with its adjoint, *i.e.* $\mathcal{L}^{\dagger}\mathcal{L} \equiv \mathcal{L}\mathcal{L}^{\dagger}$. It is said to be *self-adjoint* or *Hermitian* if it is identical to its adjoint: $\mathcal{L}^{\dagger} \equiv \mathcal{L}$.

Normality and hermiticity properties interfere with those arising from the eigenvalue decomposition. It can be shown that a normal operator \mathcal{L} possesses an orthogonal eigen-basis in which it is represented by a diagonal matrix. If in addition it is Hermitian, all its eigenvalues are real.

When the operator is non-normal (*i.e.* does not commute with its adjoint), these properties are lost. Not only the operator can have complex eigenvalues, but there is no orthogonal basis in which it is represented by a diagonal matrix (cf. Jordan normal form). On the other hand, it is possible to find a double series of basis vectors $\{\mathbf{E}_j, \mathbf{F}_j, j = 1, \ldots, d\}$ such that $\{\mathbf{E}_i\}$ is an eigen-basis of \mathcal{L} and $\{\mathbf{F}_j\}$ an eigen-basis of \mathcal{L}^{\dagger} such that

$$(s_j^* - s_i)\mathcal{H}(\mathbf{F}_j, \mathbf{E}_i) = 0,$$

i.e. vectors \mathbf{E}_i and \mathbf{F}_j are orthogonal except when s_j and s_i are complex conjugate. The series is said to be *bi-orthogonal*. Unfortunately, this says nothing about scalar products $\mathcal{H}(\mathbf{E}_j, \mathbf{E}_i)$ and $\mathcal{H}(\mathbf{F}_j, \mathbf{F}_i)$, see Exercise A.4.

Remark: Non-normality and transient growth of the energy. Let a linear differential dynamical system be governed by

$$\mathbf{X} = \mathcal{L}\mathbf{X},$$

one can interpret the canonical Hermitian norm

$$2K = \mathcal{H}(\mathbf{X}, \mathbf{X})$$

as an energy. This interpretation is particularly appropriate when considering systems of mechanical origin, whose dependent variables are velocities, *e.g.* fluids governed by the Navier-Stokes equations linearized around some base state for which the kinetic energy $K = \frac{1}{2} \int_{\mathcal{V}} (u'^2 + v'^2 + w'^2) dx dy dz$ contained in a perturbation has all the properties requested for a norm.

On general grounds, asymptotic stability properties (long time limit) of the fixed point at the origin are obtained from the spectrum of eigenvalues s of \mathcal{L} (*i.e.*, stability, neutrality, or instability for $\mathcal{R}e(s) < 0$, = 0 or > 0, respectively). When the considered system is unstable, the energy contained in an arbitrary (but small) perturbation may decrease when the trajectory are called back the fixed point along strongly attracting eigendirections but always escape exponentially fast in the unstable subspace since the long time behavior is controlled by eigenvalues with positive real parts. The case of asymptotically stable systems is less transparent because, while they always approach the fixed point at late stage, trajectories may initially depart from it. This going away of trajectories, in the sense of the norm, can be interpreted as a transient growth of the energy for some initial conditions. One can indeed have $\dot{K} > 0$ in some sectors of the tangent phase space, which may be surprising for a stable system. This is because the quadratic form:

$$\dot{K} = \frac{1}{2}[\mathcal{H}(\dot{\mathbf{X}}, \mathbf{X}) + \mathcal{H}(\mathbf{X}, \dot{\mathbf{X}})] = \frac{1}{2}[\mathcal{H}(\mathcal{L}\mathbf{X}, \mathbf{X}) + \mathcal{H}(\mathbf{X}, \mathcal{L}\mathbf{X})] = \mathcal{R}e(\mathcal{H}(\mathbf{X}, \mathcal{L}\mathbf{X}))$$

is not necessarily definite negative when \mathcal{L} is non-normal, even when all eigenvalues are negative or have negative real parts. A simple example is treated in Exercise A.4.4.

The interest of this property in hydrodynamics derives from the fact that nonlinearities conserve the kinetic energy, so that the initial growth of the energy in the perturbations around the base state is controlled by the linear part of the tangent operator whatever their amplitude.²¹ In practice, transient effects are all the more marked that diagonal terms are small when compared to non-diagonal terms. If one notices that diagonal terms mainly account for viscous dissipation, one understands that the situation is highly degenerate at large Reynolds numbers, so that this growth has essentially

 $^{^{21}}$ D.S. Henningson & S.C. Reddy, "On the role of linear mechanisms in transition to turbulence," Phys. Fluids 6 (1994) 1396-1398.
the same origin as the sub-dominant algebraic evolution called 'secular' in the study of the Jordan normal form. See also the introduction of Chapter 6 and §6.3.4.

A.3.2 Fredholm Alternative

Let us now consider the role played by the adjoints in the resolution of inhomogeneous linear problems (as they appear for example in perturbation theory, Chapter 2, p. 55ff):

$$\mathcal{L}\mathbf{X} = \mathbf{F} \,. \tag{A.13}$$

The solution is unique and given by $\mathbf{X} = \mathcal{L}^{-1}\mathbf{F}$ for all \mathbf{F} , as long as \mathcal{L} is invertible, that is to say as long as its kernel is trivial (\mathcal{L} of maximal rank). The homogeneous problem

$$\mathcal{L}\mathbf{X} = \mathbf{0} \tag{A.14}$$

then admits only the null vector $\mathbf{X} = \mathbf{0}$ as a solution. When this is not the case, *i.e.* \mathcal{L} is not of maximal rank and has a nontrivial kernel, the adjoint homogeneous problem

$$\mathcal{L}^{\dagger}\mathbf{X} = \mathbf{0} \tag{A.15}$$

also has nontrivial solutions $\tilde{\mathbf{X}} \neq \mathbf{0}$ (Fredholm alternative). The inhomogeneous problem (A.13) then has solutions only if the right hand side \mathbf{F} is orthogonal to the kernel of the adjoint (Fredholm theorem). Vector $\tilde{\mathbf{X}}$ being in the kernel of \mathcal{L}^{\dagger} , this reads

$$\mathcal{H}(\tilde{\mathbf{X}}, \mathbf{F}) = 0, \qquad (A.16)$$

see Exercise A.4.3. When this condition is fulfilled the solution exists but is not unique. This indeterminacy can be raised by imposing, *e.g.* the orthogonality of the solution to the kernel of \mathcal{L}^{\dagger} , which, in perturbations problems presents the solution to the inhomogeneous problem as a true correction (in the sense of the scalar product) to the solution to the homogeneous problem.

A.3.3 Boundary value problems and adjoint operators.

Up to now we have considered only finite dimensional spaces. Let us have a look as spaces whose elements are real or complex functions defined on some interval [a, b]. The natural extension of the sum appearing in the definition of the scalar product is now an integral:

$$\mathcal{H}(g,f) = \int_a^b g(x)^* f(x) \,\mathrm{d}x \,.$$

The case of interest to us is when these functions fulfill a differential boundary value problem of order n, with the differential operator \mathcal{L} written as:

$$\mathcal{L}f=\sum_{m=0}^n a_m(x)f^{(m)}\,,$$

 $f^{(m)}$ denoting the *m*th derivative of *f* with respect to *x* and $f^{(0)} \equiv f$. In order to formulate a well posed problem, we must add *n* boundary conditions:

$$\sum_{m=0}^{n-1} \alpha_{jm} f^{(m)}(a) + \beta_{jm} f^{(m)}(b) = u_j, \qquad j = 1, \dots, n.$$

In many practical cases, these boundary conditions apply separately at each end of the interval, that is $\beta_{jm} = 0$ when $\alpha_{jm} \neq 0$ and vice versa. When $u_j = 0$ for all j, boundary conditions are said to be homogeneous.

Returning to the definition of the adjoint operator (A.12), we see that the expression of \mathcal{L}^{\dagger} has to be determined from:

$$\int_a^b g^*(x) \mathcal{L}^{\dagger} f(x) \, \mathrm{d}x = \int_a^b (\mathcal{L}g(x))^* f(x) \, \mathrm{d}x \,,$$

by removing g^* from the action of the differential operator on the right hand side of this identity. This is done by means of a series of integrations by parts:

$$\int_{a}^{b} a_{m}^{*}(g^{*})^{(m)} f \, \mathrm{d}x = \left[(g^{*})^{(m-1)} a_{m}^{*} f \right]_{a}^{b} - \int_{a}^{b} (g^{*})^{(m-1)} (a_{m}^{*} f)' \, \mathrm{d}x$$

This operation progressively lowers the order of derivatives acting on g^* and must be pursued down to zero. The formal expression of \mathcal{L}^{\dagger} is then easily obtained as:

$$\mathcal{L}^{\dagger}f = \sum_{m=0}^{n} (-1)^{m} (a_{m}^{*}f)^{(m)}$$

However, the integrations by parts leaves us with a complicated boundary form made of integrated terms evaluated at the boundaries and involving derivatives of f, the a_m^* , and g^* . Inserting the boundary conditions of the direct problem in this linear form and imposing the cancellation of the residue gives the set of boundary conditions to be applied to the functions of the adjoint problem.

A.4 Exercises

Among the results recalled in this Appendix, several find a nontrivial illustration already in dimension two. The following, extremely elementary exercises may help one better understand the generalizations stated without derivation in the text. Vectors \mathbf{X} are represented in the canonical basis by column arrays:

$$\mathbf{X} = \begin{bmatrix} x_1 \\ x_2 \end{bmatrix}$$

and the linear operator \mathcal{L} by a matrix [L]:

$$\mathcal{L} = \begin{bmatrix} l_{11} & l_{12} \\ l_{21} & l_{22} \end{bmatrix}, \qquad (A.17)$$

The complex extension of the vector space on \mathbb{R} (*i.e.*, $x_1 \in \mathbb{R}$, $x_2 \in \mathbb{R}$) is the set of vectors with complex coefficients (*i.e.*, $x_1 \in \mathbb{C}$, $x_2 \in \mathbb{C}$). When necessary, the canonical scalar product is defined as:

$$\mathcal{H}(\mathbf{Y},\mathbf{X})=y_1^*x_1+y_2^*x_2\,.$$

A.4.1 Eigenvalues and eigenvectors, non-degenerate case

1) Write down the eigenvalue problem $\mathcal{L}\mathbf{X} = s\mathbf{X}$ and the corresponding characteristic equation. Rewrite the latter using the sum S and the product P of the roots $(S = \text{trc}(\mathcal{L}) = l_{11} + l_{22}, P = \det(\mathcal{L}) = l_{11}l_{22} - l_{21}l_{12}$. Discuss the number and the nature of the roots in the plane of parameters (S, P) and identify the possible bifurcations upon variation of a control parameter r, S = S(r) and P = P(r). [see Fig. A.3].

2) Identify the domain of parameters corresponding to two real distinct eigenvalues. Compute the eigenvalues, determine the corresponding eigenvectors and the matrix expressing the change from the canonical basis to the eigen-basis. Compute the inverse of this matrix.

3) Determine trajectories of $\dot{\mathbf{X}} = \mathcal{L}\mathbf{X}$ with $\mathbf{X}(0) = \mathbf{X}_0$ in the canonical basis by changing back and forth to the eigen-basis. (As a concrete example, take $\dot{X}_1 = 2X_1 + X_2$, $\dot{X}_2 = 4X_1 - X_2$.)

4) Find the adjoint \mathcal{L}^{\dagger} to \mathcal{L} for the canonical scalar product (general case, [L] with coefficients in \mathbb{C}). Show that when [L] has real coefficients, \mathcal{L}^{\dagger} is normal $(\mathcal{L}\mathcal{L}^{\dagger} \boxminus \mathcal{L}^{\dagger}\mathcal{L})$ only when it is symmetric $(l_{12} = l_{21})$ or such that $l_{11} = l_{22}$ and $l_{12} = -l_{21}$. Find the eigenvectors (E₁, E₂) of \mathcal{L} and (F₁, F₂)



Fig. A.3 Nature of the roots of the characteristic equation of the two-dimensional problem as a function of S and P.

of \mathcal{L}^{\dagger} . Check that, in all cases, $\mathcal{H}(\mathbf{F}_1, \mathbf{E}_2) = 0 = \mathcal{H}(\mathbf{F}_2, \mathbf{E}_1)$ but that one has $\mathcal{H}(\mathbf{E}_1, \mathbf{E}_2) = 0 = \mathcal{H}(\mathbf{F}_1, \mathbf{F}_2)$ only when \mathcal{L} is normal.

5) When there are two complex conjugate eigenvalues, $s = \sigma \pm i\omega$, find a basis in which the system is represented by (2.31), Chapter 2, p. 39.

A.4.2 Degenerate case and Jordan normal form

1) Find the condition fulfilled by the l_{ij} when the characteristic polynomial has a double root. Find eigenvectors and determine the condition for any vector of the plane to be an eigenvector.

2) In the general case, this condition is not satisfied and there is only one eigen-direction, generated by a single vector \mathbf{E} , the components of which will be determined. Compute the image of an arbitrary vector \mathbf{X} and check that $\mathbf{Y} = (\mathcal{L} - \bar{s}\mathcal{I})\mathbf{X}$ is parallel to \mathbf{E} . Derive from this an illustration of the Cayley-Hamilton theorem.

3) Show that in the basis { $\mathbf{Y} = (\mathcal{L} - \bar{s}\mathcal{I})\mathbf{X}$; \mathbf{X} }, the matrix representing \mathcal{L} is in canonical Jordan form, which justifies writing a two-dimensional dynamical system with degenerate eigenvalues as (2.35, 2.36), p. 40.

4) Unfold the singularity by setting $s_{(\pm)} = \bar{s} \pm \delta s$, where δs is a small perturbation, and consider the dynamical system:

$$rac{\mathrm{d}^2}{\mathrm{d}t^2}X - 2ar{s}\,X + ig(ar{s}^2 - \delta s^2ig)\,X = ig(rac{\mathrm{d}}{\mathrm{d}t} - s_{(+)}ig)ig(rac{\mathrm{d}}{\mathrm{d}t} - s_{(-)}ig)\,X = 0\,,$$

integrate it explicitly by solving successively $\left(\frac{d}{dt} - s_{(+)}\right)Y = 0$ and next $\left(\frac{d}{dt} - s_{(-)}\right)X = Y$, then show how secular terms associated to linear resonance appear by taking the limit $\delta s \to 0$.

A.4.3 Fredholm Alternative

- 1) Compute the kernel of operator \mathcal{L} in (A.17).
- 2) Solve the inhomogeneous problem $\mathcal{L}\mathbf{X} = \mathbf{F}$ taken in the form

$$egin{aligned} &l_{11}x_1+l_{22}x_2=f_1\,,\ &l_{21}x_1+l_{22}x_2=f_2\,, \end{aligned}$$

when the kernel of \mathcal{L} is not trivially reduced to the null vector. Observe that the problem has solutions only when the two equations are proportional and express this condition.

3) Determine the adjoint \mathcal{L}^{\dagger} of \mathcal{L} and its kernel. Then check that the existence condition obtained can be written in the form (A.16).

A.4.4 Transient growth of the energy

Consider $\dot{\mathbf{X}} = \mathcal{L}\mathbf{X}$ defined by:

$$\dot{X}_1 = -aX_1 + X_2 , \ \dot{X}_2 = -bX_2 ,$$

where constants a and b are positive and b > a (already considered in Exercise 4.6.6, p. 172).

1) This system is non-normal, and thus has no orthogonal eigen-basis. Find the eigenvalues and the eigen vectors of \mathcal{L} and its adjoint \mathcal{L}^{\dagger} (for the canonical scalar product).

2) Consider the energy $K = \frac{1}{2}(X_1^2 + X_2^2)$ and determine the condition on a and b rendering \dot{K} definite negative (monotonic stability). When this condition is not satisfied, identify the region of the (X_1, X_2) plane where $\dot{K} > 0$ and the direction $\rho_{\max} = X_2/X_1$ that maximize the initial growth of the energy. Examine in particular the case $a \ll 1$, $b \ll 1$, so that the behavior of the system is dominated by the out-of-diagonal term.

3) Compute the solution (X_1, X_2) issued from the initial condition $X_1^{(0)}, X_2^{(0)} = \rho X_1^{(0)}$, the evolution of the energy associated to this trajectory and the value of ρ that maximizes $K(t)/K^{(0)}$.

Appendix B

Numerical Approach

Present-day computers make numerical simulations of complex physical systems feasible. Their flexibility and accuracy allow us to gather quantitative information on situations of practical interest but also to obtain more qualitative hints about the nature of interactions at work. In this appendix we focus on the simplest, low-cost, approaches most particularly appropriate to the second aspect, leaving the first one to specialists. We present some classical numerical schemes used in simulations of *initial value problems* defined in terms of *ordinary* (ODEs) or *partial* (PDEs) *differential equations*. The level is kept very elementary and theoretical discussions are avoided, though numerical stability properties derive directly from the application of techniques developed in other parts of the present work, as can be seen by working out the proposed exercises.

In practice, the evolution of the system under consideration is determined at a series of regularly spaced times $t_n = n \Delta t$. The corresponding integration methods, appropriate to ODEs as well as PDEs, are introduced in §B.1. The physical-space dependence inherent in PDEs implies a specific coupling between degrees of freedom. This can be treated in the most naive way by estimating the values of the different fields at the nodes of a discrete lattice, $x \rightarrow x_j = j \Delta x$. Replacing partial derivatives by their finite differences yields the numerical schemes presented in §B.2.1. Another description can be achieved in terms of projection of the dynamics on a functional basis spanning the whole physical space. These so-called spectral methods are briefly introduced, using Fourier modes in §B.2.2.

Finite element methods and their variants specifically developed for fluid mechanics are left completely aside despite their accuracy and their versatility regarding complex geometries encountered in technical applications for which commercial softwares are fully adapted. Instead of relying on such black-boxes, we prefer present simple algorithms that can be straightforwardly translated into programming languages such as MATLAB. To our belief, the required investment is light but rewarding in terms of the insight gained in the complexity arising from nonlinearities.

B.1 Treatment of the Time Dependence

We consider the following general initial value problem:

$$\frac{\mathrm{d}}{\mathrm{d}t}v = f(v,t) \qquad \text{with} \quad v(t_0) = v_0 \,. \tag{B.1}$$

here written as a non-autonomous ODE but the case of PDEs is the same as far as time dependence is concerned.

Expanding the solution as a Taylor series in the neighborhood of t_n , we get:

$$v_{n+1} = v_n + \Delta t \frac{d}{dt} v \Big|_n + \frac{1}{2} \Delta t^2 \frac{d^2}{dt^2} v \Big|_n + \frac{1}{6} \Delta t^3 \frac{d^3}{dt^3} v \Big|_n + \cdots$$
(B.2)

where the successive derivatives of v, total derivative with respect to t, can be evaluated at t_n recursively, using (B.1):

$$\begin{aligned} \frac{\mathrm{d}}{\mathrm{d}t}v\big|_{n} &= f_{n}, \\ \frac{\mathrm{d}^{2}}{\mathrm{d}t^{2}}v\big|_{n} &= \left.\frac{\mathrm{d}}{\mathrm{d}t}\frac{\mathrm{d}}{\mathrm{d}t}v\big|_{n} &= \left.\frac{\mathrm{d}}{\mathrm{d}t}f(v,t)\big|_{n} \\ &= \left.\left(\partial_{v}f\frac{\mathrm{d}}{\mathrm{d}t}v + \partial_{t}f\right)\right|_{n} = \left(\partial_{v}f_{n}\right)f_{n} + \partial_{t}f_{n} \\ \frac{\mathrm{d}^{3}}{\mathrm{d}t^{3}}\big|_{n} &= \cdots, \end{aligned}$$

provided that we can evaluate the successive partial derivatives of f(v,t) analytically, which is hardly conceivable beyond the lowest order without the help of formal algebra softwares. The idea is thus to look for numerical approximations of these derivatives.²²

Truncating expression (B.2) beyond first order, we get the most elementary formula (Euler, cf. Fig. B.1):

$$v_{n+1} = v_n + \Delta t f(v_n, t_n). \tag{B.3}$$

This integration scheme is said to be *first order* because it is exact up to correction of order Δt^2 , and *explicit* because the knowledge of v and f(v, t) at previous steps is sufficient to compute the solution at the next step.

 $^{^{22}}$ For elementary formulas consult [Abramowitz and Stegun (1972)]. A detailed presentation can be found in, *e.g.* [Lapidus and Seinfeld (1971)].

Another first order scheme is obtained by evaluating v_n using quantities computed at t_{n+1} :

$$v_n = v_{n+1} - \Delta t \left. \frac{\mathrm{d}}{\mathrm{d}t} v \right|_{n+1} + \frac{1}{2} \Delta t^2 \left. \frac{\mathrm{d}^2}{\mathrm{d}t^2} v \right|_{n+1} - \frac{1}{6} \Delta t^3 \left. \frac{\mathrm{d}^3}{\mathrm{d}t^3} \right|_{n+1} + \dots, \quad (B.4)$$

which leads to:

$$v_{n+1} = v_n + \Delta t f(v_{n+1}, t_{n+1})$$

This scheme, still exact up to corrections $\mathcal{O}(\Delta t^2)$, is termed *implicit* since v_{n+1} is obtained from an implicit equation:

$$v_{n+1} - \Delta t f(v_{n+1}, t_{n+1}) = v_n .$$
 (B.5)

Subtracting (B.4) from (B.2) and noticing that

$$\frac{\mathrm{d}^2}{\mathrm{d}t^2}v\Big|_{n+1} = \left.\frac{\mathrm{d}^2}{\mathrm{d}t^2}v\Big|_n + \Delta t \left.\frac{\mathrm{d}^3}{\mathrm{d}t^3}v\right|_n + \mathcal{O}(\Delta t^2)\,,$$

one can see that the term $\mathcal{O}(\Delta t^2)$ disappears, so that:

$$v_{n+1} = v_n + \frac{\Delta t}{2} \left(f(v_n, t_n) + f(v_{n+1}, t_{n+1}) \right) + \mathcal{O}(\Delta t^3) \,. \tag{B.6}$$

This *implicit second order* formula is called the *Crank-Nicholson* scheme (cf. Fig. B.1).



Fig. B.1 Solution of $\frac{d}{dt}v = -v$ at $t = \Delta t = 0.5$ with v(0) = 1: Exact result compared to estimates from the explicit Euler (EE), implicit Euler (IE) and Crank-Nicholson (CN) formulas, Eqs. (B.3), (B.5), and (B.6), respectively.

Let us compare the error terms using f(v) = -v with $v_0 = 1$ as a simple example. We get $v = \exp(-t)$ and after a time step

$\mathrm{Exact}\ (\exp(-\Delta t)):$	$1 - \Delta t + \frac{1}{2}\Delta t^2 - \frac{1}{6}\Delta t^3 + \dots$
Explicit Euler (B.3) :	$1-\Delta t$
Implicit Euler (B.5) :	$rac{1}{1+\Delta t}=1-\Delta t+\Delta t^2-\Delta t^3+\dots,$
Crank–Nicholson (B.6) :	$\frac{1-\frac{1}{2}\Delta t}{1+\frac{1}{2}\Delta t}=1-\Delta t+\frac{1}{2}\Delta t^2-\frac{1}{4}\Delta t^3+\ldots,$

which clearly shows the improvement brought by the Crank-Nicholson scheme (cf. exercise B.3.1). The obvious computational drawback of implicit iteration schemes, largely compensated by better stability properties as shown later, is that the solution at t_{n+1} is not expressed simply in terms of the solution at t_n . The inversion is trivial only if v is a single scalar variable and f is linear as in the example chosen. When this is not the case, specific techniques are required, see §B.2.1.

Up to now we have used only *one-step* formulas, requiring the knowledge of the solution at a single time. A simple improvement of (B.3) is obtained by inserting:

$$\frac{\mathrm{d}^2}{\mathrm{d}t^2} v \Big|_n = \frac{1}{\Delta t} \left(\frac{\mathrm{d}}{\mathrm{d}t} v |_n - \frac{\mathrm{d}}{\mathrm{d}t} v |_{n-1} \right) + \mathcal{O}(\Delta t)$$

in (B.2), which leads to the second-order Adams-Bashford formula:

$$v_{n+1} = v_n + \Delta t \left(\frac{3}{2} f(v_n, t_n) - \frac{1}{2} f(v_{n-1}, t_{n-1}) \right) , \qquad (B.7)$$

exact up to $\mathcal{O}(\Delta t^3)$. This formula is one of the simplest examples of *multi-step* schemes, computing v_{n+1} from the knowledge of v at several previous times, here t_n and t_{n-1} .

An equally simple possibility is to write (B.4) at times t_{n-1} and t_n instead of t_n and t_{n+1} , which yields

$$v_{n-1} = v_n - \Delta t \, \frac{\mathrm{d}}{\mathrm{d}t} v|_n + \frac{1}{2} \Delta t^2 \frac{\mathrm{d}^2}{\mathrm{d}t^2} v|_n - \frac{1}{6} \Delta t^3 \frac{\mathrm{d}^3}{\mathrm{d}t^3} v|_n + \dots ,$$

so that after subtraction of (B.2) one gets another second order formula, also exact up to $\mathcal{O}(\Delta t^3)$:

$$v_{n+1} = v_{n-1} + 2\Delta t f(v_n, t_n), \qquad (B.8)$$

called the *leap-frog* scheme.

In addition to accuracy requirements, the choice of an integration scheme is also subjected to stability requirements. As a matter of fact, the numerical scheme must not amplify round-off errors so that the solution obtained is the physical one and not some parasitic solution. The problem is important since when, for example, one uses a two-step formula to integrate a first-order differential equation, one obtains an iteration that starts with two initial conditions instead of one for the time-continuous problem. The added mode must be damped in order that the simulation keeps its meaning. Let us compare schemes (B.7) and (B.8) used to integrate $\frac{d}{dt}v = -v$:

• Iteration (B.7) reads

$$v_{n+1} = v_n - \Delta t \left(\frac{3}{2} v_n - \frac{1}{2} v_{n-1} \right)$$

or, better,

$$egin{aligned} & u_{n+1} = v_n \, , \ & v_{n+1} = rac{1}{2} \Delta t \, u_n + ig(1 - rac{3}{2} \Delta t ig) v_n \, , \end{aligned}$$

which, from standard linear stability analysis (Exercise B.3.2), leads to the characteristic equation:

$$s^2 - s\left(1 - \frac{3}{2}\Delta t\right) - \frac{1}{2}\Delta t = 0$$
.

At lowest order in Δt , the two roots are $s_{(+)} = 1 - \Delta t$ and $s_{(-)} = -\frac{1}{2}\Delta t$. The first one clearly corresponds to the physical solution and the second one to the numerical mode. Recalling that the differential equation is here replaced by a discrete-time system like those introduced in Chapter 3 (see in particular Figure 4.10, p. 140) we obtain that the latter mode is damped provided that $|s_{(-)}| < 1$, which can be achieved by taking Δt sufficiently small, here $\Delta t < 2$. This does not mean that the numerical solution obtained with Δt just below the limit is a good approximation, but just that it is not polluted by the numerical mode. The Adams–Bashford scheme is said to be conditionally stable.

• Iteration (B.8) reads:

$$v_{n+1} = v_{n-1} - 2\,\Delta t\,v_n,$$

that is:

$$u_{n+1} = v_n ,$$

$$v_{n+1} = u_n - 2 \Delta t v_n ,$$

which leads to

$$s^2 + 2\,\Delta t\,s - 1 = 0\,,$$

so that $s_{(+)} \simeq 1 - \Delta t$ as before. But now one has $s_{(-)} \simeq -1 - \Delta t$ and since $|s_{(-)}| > 1$, the numerical mode is always unstable. The leap-frog scheme is thus always unconditionally unstable. Since $s_{(-)}$ is negative, successive iterates alternate from positive to negative values. This subharmonic instability is due to the fact that the scheme leaves even time steps uncoupled to odd ones. Such a behavior is frequently the signature of a numerical instability (see later Figure B.16). Scheme (B.8) is however little demanding and its use is not forbidden provided the growth of errors is controlled. In general, the numerical mode is tamed by averaging over two iterations every p computation, with $p \gg 1$ but still small enough, so that its amplitude remains small when compared to the truncation errors.

High order accuracy can be reached with sophisticated multi-step formulas. Their drawback stems from the need to generate numerical initial conditions from the physical ones, especially when the time step has to be changed for accuracy reasons. Explicit formulas of Runge-Kutta type avoid this problem. The idea is to use the basic first-order Euler formula but with a better estimate of the slope. Second order accuracy is obtained by averaging the slopes evaluated at t_n and t_{n+1} but the resulting (Crank-Nicholson) scheme is implicit. To avoid such a difficulty one can first estimate v_{n+1} with the first-order explicit formula:

$$v' = v_n + \Delta t f(v_n, t_n), \qquad (B.9)$$

then compute the slope at this point $f(v'_{n+1}, t_{n+1})$, and take the average with the slope at (v_n, t_n) as an improved guess. This yields:

$$v_{n+1} = v_n + \frac{1}{2}\Delta t \left(f(v_n, t_n) + f(v', t_{n+1}) \right).$$
(B.10)

The correction is sufficient to achieve second order accuracy while keeping an explicit scheme (Exercise B.3.1). This case is illustrated in Figure B.2, left.

Another way to achieve the same goal is to introduce an intermediate time $t_{n+1/2}$, to compute v' with the first-order formula

$$v' = v_n + \frac{1}{2}\Delta t f(v_n, t_n), \qquad (B.11)$$

and to use the slope at that point to extrapolate the solution at time t_{n+1} ,



Fig. B.2 Achieving second order accuracy with Runge-Kutta formulas (B.9, B.10) and (B.11, B.12).

as shown in Figure B.2, right:

$$v_{n+1} = v_n + \Delta t f(v', t_{n+1/2}). \tag{B.12}$$

High-order formulas combine these two basic techniques to get the best estimate of the slope. The classical fourth-order scheme

$$v_{n+1} = v_n + \frac{1}{6}\Delta t \left(f(v_n, t_n) + 2f(v', t_{n+1/2}) + 2f(v'', t_{n+1/2}) + f(v''', t_{n+1}) \right)$$
(B.13)

involves three intermediate estimates:

$$v' = v_n + \frac{1}{2}\Delta t f(v_n, t_n),$$

$$v'' = v_n + \frac{1}{2}\Delta t f(v', t_{n+1/2}),$$

$$v''' = v_n + \Delta t f(v'', t_{n+1/2}).$$

One full time step thus requires four evaluations of f, which may be numerically demanding, but the scheme stays explicit and starts with a single initial condition so that the time step can easily be changed.

Predictor-corrector methods are multi-step schemes that are often preferred to Runge-Kutta methods because they require a smaller number of evaluations of f to reach the same accuracy though it is more difficult to change the time step. For more information, consult [Lapidus and Seinfeld (1971)] or [Press *et al.* (1986)].

Explicit methods can easily be extended to treat differential systems coupling several variables, see the examples considered in §B.4.

B.2 Treatment of Space Dependence in PDEs

B.2.1 Finite difference methods

The derivation of an integration scheme is made of two main steps. First, a consistent approximation of space differential operators has to be found. Next a time-integration scheme is chosen. The resulting space-time iteration is then studied for stability. A supplementary step would be to demonstrate convergence in the limit of infinitesimal space and time steps. For *linear* partial differential problems, it can be shown that *consistence* and *stability* implies *convergence*. Here we do worry about convergence and are happy with numerical solutions that look "physical."

We consider only systems that extend in one space direction. The solution is sought for at regularly spaced nodes $\{x_j\}$ of a periodic lattice with spacing $\Delta x = \ell/N$ where ℓ is the length of the domain and N + 1 the number of points, including end points. The numerical solution $v_{n,j}$ is thus noted with two subscripts but the first one, corresponding to time, will be dropped where it is not necessary.

B.2.1.1 Space discretization and consistence

Finite difference approximations of space derivatives are obtained in the same way as those of time derivatives.²³ For example, combination of two first order approximations of the first order derivatives:

$$\partial_x v|_j = \left(v_{j+1} - v_j\right)/\Delta x \quad \text{ and } \quad \partial_x v|_j = \left(v_j - v_{j-1}\right)/\Delta x \quad (\mathrm{B.14})$$

yields the second order centered formula:

$$\partial_x v|_j = (v_{j+1} - v_{j-1})/2\Delta x.$$
 (B.15)

In the same way, the expression of the second derivative at order Δx^2 reads:

$$\partial_{xx} v|_j = (v_{j+1} - 2v_j + v_{j-1}) / \Delta x^2$$
, (B.16)

Higher order accuracy requires more points, e.g. five at fourth order:

$$\partial_{xx} v|_{j} = \left(-v_{j+2} + 16v_{j+1} - 30v_{j} + 16v_{j-1} - v_{j-2}\right)/12\Delta x^{2}.$$
(B.17)

The second-order approximation of the fourth derivative reads:

$$\frac{\partial_{xxxx}v|_{j} = (v_{j+2} - 4v_{j+1} + 6v_{j} - 4v_{j-1} + v_{j-2})/\Delta x^{4}.$$
(B.18)

²³Useful formulas are found in [Abramowitz and Stegun (1972)]. A complete presentation is given by [Richtmyer and Morton (1967)]. Consult also [Acton (1970)].

The order of *consistency* of some finite difference approximation is the value of the exponent that controls the decay of the global truncation error when Δx decreases. It serves nothing to approximate different derivatives appearing in a model by formulas at different orders. In practice, the representation of differential operators by finite differences is not very good since truncation errors decay only algebraically, as a power of Δx . From this point of view spectral methods briefly introduced in Section B.2.2 behave in a much better way.

B.2.1.2 Boundary conditions

Consistency considerations also affect the treatment of boundary conditions when they involve derivatives (Neumann conditions). Since it is in general easy to achieve consistency at order Δx^2 for interior points —for example by preferring (B.15) to (B.14) for ∂_x — it would be a pity to spoil the quality of an approximation by a bad account of boundary conditions, even if their effects do not propagate beyond some narrow "numerical boundary layer."

Consider for example the condition $\partial_x v = 0$ at one end point of the x interval. Using (B.14) we get an expression accurate at order Δx , which simply implies $v_j = v_{j+1}$, with j = 0 and 1 for the boundary point and the first interior point, respectively. To achieve second order consistency, we can use:

$$\partial_x v|_j = (-3v_j + 4v_{j+1} - v_{j+2})/2\Delta x$$
 (B.19)

so that canceling $\partial_x v$ at the boundary gives $3v_0 - 4v_1 + v_2 = 0$. Another possibility is to add a fictitious point outside, j = -1 and associate a variable v_{-1} whose value is forced to that of the first interior point at all times, $v_{-1} = v_1$, as dictated by formula (B.15).

B.2.1.3 Time discretization and stability

For more specificity we consider the one-dimensional diffusion equation:

$$\partial_t v = \partial_{xx} v \,, \tag{B.20}$$

with boundary conditions that need not be given here. The temporal scheme is obtained according to one the recipes of section B.1 (here we restore subscript n). From the spatial viewpoint we assume second order consistency, *i.e.* using (B.16) to discretize $\partial_{xx}v$. The most important question is that of the numerical stability of the scheme, as determined from the growth rate of the "noise" generated by truncation and round-off

errors. Here, we only compare the case of the two simplest schemes deriving from (B.3) and (B.5). In the *explicit* case, the space derivative $\partial_{xx}v$ is evaluated at t_n , that is:

$$v_{n+1,j} = v_{n,j} + \frac{\Delta t}{\Delta x^2} \left(v_{n,j+1} - 2v_{n,j} + v_{n,j-1} \right) , \qquad (B.21)$$

while it is evaluated at t_{n+1} in the *implicit* case, which yields:

$$-\frac{\Delta t}{\Delta x^2}v_{n+1,j+1} + \left(1 + \frac{2\Delta t}{\Delta x^2}\right)v_{n+1,j} - \frac{\Delta t}{\Delta x^2}v_{n+1,j-1} = v_{n,i}.$$
 (B.22)

The explicit method is straightforward and requires less computational work than the implicit method that requires the inversion of a linear system with a band structure and only three near-diagonal terms. As we will see, this amount of work is rewarding.

The stability analysis of finite-difference systems is performed in the same way as that of physical systems (see Chap. 2, §3.1.3 and Chap. 3, §4.2). The evolution of a perturbation $\delta v_{n,j}$ to a given numerical solution $v_{j,n}$ is studied by means of a discrete Fourier transform:

$$\delta \! v_{n,j} = \delta \! ilde{v}_{n,k} \exp(ikj)$$
 .

Inserting this expression into equations (B.21) and (B.22), we obtain:

Explicit:
$$\delta \tilde{v}_{n+1,k} = \left(1 - \frac{2\Delta t}{\Delta x^2} (1 - \cos(k))\right) \delta \tilde{v}_{n,k}$$
, (B.23)

Implicit:
$$\left(1+\frac{2\Delta t}{\Delta x^2}(1-\cos(k))\right)\delta \tilde{v}_{n+1,k}=\delta \tilde{v}_{n,k}$$
, (B.24)

both of the general form:

$$\delta \tilde{v}_{n+1,k} = \xi(k) \, \delta \tilde{v}_{n,k} \, ,$$

and, as we know, the perturbation grows when $|\xi(k)| > 1$ and decays when $|\xi(k)| < 1$.

For the explicit scheme, this gives:

$$1 - rac{4\Delta t}{\Delta x^2} \leq \xi_{ ext{exp}} = 1 - rac{2\Delta t}{\Delta x^2} (1 - \cos(k)) \leq 1$$
,

The upper bound condition brings nothing, whereas the lower bound implies $-1 < \xi_{exp}(\pi)$ for stability. This leads to:

$$-1 < 1 - rac{4\,\Delta t}{\Delta x^2} \quad \Rightarrow \quad rac{4\,\Delta t}{\Delta x^2} < 2 \quad \Rightarrow \quad \Delta t < rac{\Delta x^2}{2} \,,$$

which means that the time step must be small enough at given spacing: the explicit scheme is thus *conditionally stable*.

By comparison, the growth rate of perturbations with the implicit scheme:

$$\xi_{imp} = \left(1 + \frac{2\Delta t}{\Delta x^2} (1 - \cos(k))\right)^{-1}$$

is such that $0 < \xi_{imp} \leq 1$ for all k since $(1 - \cos(k)) \geq 0$. This scheme is therefore *unconditionally stable*, so that Δt can be chosen for temporal accuracy requirement only, whatever the value of Δx .

For the explicit scheme the most unstable perturbations are those with $k = \pi$. Going back to physical space, one finds $\delta v_{n,j} = \delta \tilde{v}_n (-1)^j$: the behavior of the solution at even nodes is the opposite of that at odd nodes, which is typical of numerical instabilities. At this stage, the instability can be tamed by decreasing the time step until the stability condition is fulfilled but it is better to turn to an implicit scheme, especially when higher derivatives are involved (Exercise B.3.3). Higher order time discretization would be studied in the same way.

B.2.1.4 Efficient treatment of implicit schemes

The supplementary work needed to find the solution at time step t_{n+1} by the implicit scheme (B.22) is easily achieved by a simple numerical "double sweep" method since the matrix to be inverted has only a few terms close to the diagonal. This is in fact a special case of the classical "LU" decomposition used to solve linear systems.

Let the initial system be $\mathcal{M}\mathbf{V} = \mathbf{F}$, the operator \mathcal{M} is then written as the product $\mathcal{L}\mathcal{U}$ of two operators, \mathcal{L} represented by a lower triangular matrix [L] with elements $l_{ij} = 0$ when i < j, and an operator \mathcal{U} represented by an upper triangular matrix [U] such that $u_{ij} = 0$ when i > j. The decomposition is unique upon requiring that the diagonal elements of [U] are scaled to unity, $u_{ii} = 1$. The starting matrix [M] is three-diagonal, *i.e.* $m_{ij} \neq 0$ for j = i and $j = i \pm 1$, [L] and [U] have a band structure with two non-zero diagonals only: $l_{ij} \neq 0$ for j = i and j = i - 1, $u_{jj} = 1$, and $u_{ij} \neq 0$ for j = i + 1.

The problem which reads:

$$(\mathcal{L}\mathcal{U})\mathbf{V} = \mathcal{L}(\mathcal{U}\mathbf{V}) = \mathcal{L}\mathbf{W} = \mathbf{F}$$

is then replaced by two equations

$$\mathcal{L} \mathbf{W} = \mathbf{F}$$
 and $\mathcal{U} \mathbf{V} = \mathbf{W}$,

that are solved through two first-order recursion relations, first a *forward* iteration to obtain W:

$$l_{11}w_1 = f_1 \,,$$

 $l_{i\,i-1}w_{i-1} + l_{ii}w_i = f_i \,, \qquad i = 2, \dots, N \,,$

and second a *backward* iteration to get V from W:

$$v_N = w_N \,,$$

 $v_i + u_{i\,i+1} v_{i+1} = w_i \,, \qquad i = N-1, \dots, 1 \,.$

where the coefficients l_{ij} and u_{ij} are computed once for all using two simple first-order recursion relations. The general term of the product reads $\sum_{k} l_{ik} u_{kj} = m_{ij}$, with two special cases for i = 1 and j = N. One readily obtains

$$l_{11}=m_{11}\,,$$
 $l_{11}u_{12}=m_{12}\,,$

then for i = 2 to N - 1

$$egin{aligned} & l_{ii-1} = m_{ii-1} \,, \ & l_{ii-1} u_{i-1\,i} + l_{ii} = m_{ii} \,, \ & l_{ii} u_{ii+1} = m_{ii+1} \,, \end{aligned}$$

and finally for i = N

$$l_{NN-1} = m_{NN-1},$$

 $l_{NN-1}u_{N-1N} + l_{NN} = m_{NN}.$

These relations are easily implemented numerically and can be generalized to the case with any finite and small number of elements close to the diagonal. This happens for example when the model contains fourth order derivatives, hence five terms from (B.18), which leads to similar but second-order recursion relations, see exercise B.3.3 and B.4.4.2 focusing on the Swift-Hohenberg model.

For problems that are two-dimensional in space, implicit schemes lead to linear systems where the matrix [M] is sparse with a block-diagonal structure. Instead of generalizing the previous algorithms to block matrices, which is possible but requires a lot of storage, one usually turns to iterative methods. The description of corresponding algorithms, over-relaxation, conjugate-gradients, *etc.*, goes beyond our present purpose.²⁴

B.2.1.5 Treatment of nonlinear terms

Up to now we have considered linear problems. In most cases of interest, nonlinearities are present, so let us write formally $\partial_t v = \mathcal{L}v + \mathcal{N}(v)$. The linear part is often the most dangerous part from the point of view of numerical stability since it usually contains partial derivatives of higher order than the nonlinear part, the dangerous character of which comes from other sources as we know. In order to preserve second order accuracy while avoiding a fully implicit treatment of nonlinearities, one can develop quasi-linearized schemes of the Crank-Nicholson type by replacing v_{n+1} by $v_n + (v_{n+1} - v_n)$ in the nonlinear term, expanding it to first order in $(v_{n+1} - v_n)$ assumed to be small enough:

$$\mathcal{N}(v)\big|_{n+1} \approx \mathcal{N}(v)\big|_n + \mathcal{Q}_n(v_{n+1} - v_n),$$

where $Q_n = \delta \mathcal{N} / \delta v \big|_n$, the functional derivative of \mathcal{N} with respect to v is a coefficient evaluated at time t_n . When nonlinearities are not numerically dangerous, another possibility is to treat the linear and nonlinear terms separately, using an implicit Crank-Nicholson scheme for the linear part and an explicit Adams-Bashford scheme for the nonlinear part

$$v_{n+1} = v_n + \frac{1}{2}\Delta t \left[\left(\mathcal{L}v \big|_{n+1} + \mathcal{L}v \big|_n \right) + \left(3\mathcal{N}(v) \big|_n - \mathcal{N}(v) \big|_{n-1} \right) \right]$$

B.2.2 Spectral methods

Spectral methods rest on a conversion of the partial differential problem into an ordinary differential problem for the amplitudes of modes obtained by projection onto a complete functional basis. For simplicity, we consider here only the case of periodic boundary conditions at the ends of an interval of length ℓ , which allows one to use Fourier modes:

$$v(x) = \sum_{m \in \mathbb{Z}} \hat{v}_m \exp(2\pi i m x/\ell), \qquad (B.25)$$

²⁴The interested reader is invited to consult specialized references, *e.g.* J.H. Wilkinson, "Solution of linear algebraic equations and matrix problems by direct methods," Chapter 2.2 in *Digital computer user's handbook*, M.K. Klerer and G.A. Korn, Eds. (McGraw-Hill, 1967).

so that, when $v \in \mathbb{R}$, one must have $\hat{v}_{-m} = \hat{v}_m^*$. The coefficients of the Fourier series are given by:

$$\hat{v}_m = \frac{1}{\ell} \int_0^\ell v(x) \exp(-2\pi i m x/\ell) \,\mathrm{d}x, \quad m \in \mathbb{Z}.$$
 (B.26)

We recall that:

$$\frac{1}{\ell} \int_0^\ell \exp\left(2\pi i (m'-m)x/\ell\right) \mathrm{d}x = \delta_{mm'}, \qquad (B.27)$$

where $\delta_{mm'}$ is the Kroneker symbol (= 1 when m = m' and zero otherwise), and that, reciprocally,

$$\sum_{m=-\infty}^{+\infty} \exp\left(2\pi i m (x-x')/\ell\right) = \delta_{\mathrm{D}}(x-x'), \qquad (B.28)$$

where $\delta_{\rm D}(x)$ is the Dirac distribution (generalized function) such that

$$\int_{-\infty}^{+\infty} f(x') \delta_{\mathrm{D}}(x-x') \,\mathrm{d}x = f(x) \,.$$

For the diffusion equation (B.20), the evolution of the solution is easily obtained by integrating the differential equations for the Fourier coefficients $\hat{v}_m = \hat{v}_m(t)$:

$$rac{\mathrm{d}}{\mathrm{d}t}\hat{v}_m = -k_m^2\hat{v}_m \qquad ext{with} \qquad k_m = 2\pi m/\ell\,,$$

which yields

$$\hat{v}_m(t)=\hat{v}_m(0)\exp(-k_m^2t)\,,\quad m\in\mathbb{Z}.$$

Let us notice that, here, the differential operator is diagonal in Fourier space. This may not be the case with other boundary conditions requiring other basis functions, e.g. Chebyshev polynomials, but, in general, an exact representation of the differential operator in spectral space can still be found.

In practice, the series is truncated beyond some maximal value N/2, so that $\sum_{m=-\infty}^{+\infty}$ is replaced by $\sum_{-N/2+1}^{+N/2}$, see Fig. B.3(a, b). This truncation introduces some approximation forbidding us to resolve the dynamics of structures with wavelengths shorter than $\lambda_{\min} = 2\pi/k_{\max} = 2\ell/N$, which can be understood as the result of sampling the solution on a regular lattice with spacing $\Delta x = \ell/N$. One may say that the method is of *infinite order* since the error term decreases exponentially with N, as $\exp\left(-(N\pi/\ell)^2 t\right)$ at given t, *i.e.* faster than any power of Δx .



Fig. B.3 (a) Spectrum $|\hat{v}_m|^2$ of a periodic function. (b) Same spectrum, truncated beyond N/2. (c) Discrete Fourier transform of the function that has the same value as the solution at the collocation points.

In fact we have not yet said how integral (B.26) is computed. In practice the solution is only represented at the nodes of the lattice mentioned above and called the *collocation points*. This allows us to pass from continuous Fourier transforms to discrete ones.²⁵ Let $\{x_j = j\Delta x; j = 0, ..., N-1\}$ be the set of these points, (B.26) is then modified to read

$$\hat{v}_m^{(d)} = \frac{1}{N} \sum_{j=0}^{N-1} v_j \exp\left(-2\pi i \frac{mj}{N}\right),$$
 (B.29)

while the solution (B.25) is written as

$$v_j = \sum_{m=-N/2+1}^{+N/2} \hat{v}_m^{(d)} \exp\left(2\pi i \frac{mj}{N}\right).$$
(B.30)

Notice that, depending on the computer program, the 1/N factor may be placed in (B.30) or in (B.29), or one can find a factor $1/\sqrt{N}$ in each.

Relations (B.27, B.28) then read:

$$\frac{1}{N}\sum_{j=0}^{N-1}\exp\left(2\pi i\frac{(m'-m)j}{N}\right) = \delta_{mm'}$$

²⁵A brief but nice introduction to discrete Fourier transforms is given by R.J. Higgins, "Fast Fourier transform: an introduction with some mini computer experiments," Am. J. Physics 44 (1976) 766-773. For more information, of course consult [Gottlieb and Orszag (1977)] or [Canuto et al. (1988)].

 and

$$\sum_{m=-N/2+1}^{N/2} \exp\left(2\pi i \frac{m(j'-j)}{N}\right) = N\delta_{jj'}.$$

It can be noticed that the series of Fourier coefficients $\hat{v}_m^{(d)}$ is periodic with period N since:

$$\hat{v}_{m+N}^{(d)} = \sum_{j=0}^{N-1} v_j \exp[-2\pi i(m+N)j/N] \boxminus \sum_{j=0}^{N-1} v_j \exp(-2\pi i m j/N) = \hat{v}_m^{(d)},$$
(B.31)

whereas \hat{v}_m is not periodic, as illustrated in Figure B.3. When v is a real quantity, it is also observed that not only $\hat{v}_0^{(d)}$ is real (like \hat{v}_0), but that $\hat{v}_{N/2}^{(d)}$ is also real since one has $(\hat{v}_{N/2}^{(d)})^* = \hat{v}_{-N/2}^{(d)} = \hat{v}_{-N/2+N}^{(d)} = \hat{v}_{N/2}^{(d)}$. The Fourier transform thus changes the N real quantities v_j , $j = 0, \ldots, N-1$, into a set of 2 real quantities, $\hat{v}_0^{(d)}$ and $\hat{v}_{-N/2}^{(d)}$, and N/2 - 1 complex quantities $\hat{v}_m^{(d)}$, $m = 1, \ldots, N/2 - 1$, so that the total number of degrees of freedom describing the solution is preserved.

The periodicity of the discrete spectrum induces what is known as the aliasing phenomenon: two Fourier modes distant by N in the spectrum are aliases of each other. This is another way of saying that, having no information on modes with m > N/2, one cannot resolve structures with scales smaller than the lattice spacing.

The important point is that we know how to go from v_j to $\hat{v}_m^{(d)}$ and vice versa by means of "fast" algorithms with operation counts of order $N \log N$ and not N^2 . The Fast Fourier Transform (FFT) takes advantage of the standard trigonometric relations to organize the information flow so as to minimize the number of arithmetic operations. Commercial softwares such as MATLAB are effective in computing FFTs when N is a power of two but also when N can be decomposed as $N = 2^{n_2}3^{n_3}5^{n_5}\dots$ with prime factors that are not too large. This possibility of a fast passage from physical to spectral space and back allows one to treat nonlinearities in a clever way by means of so-called *pseudo-spectral* methods, provided that aliasing is appropriately dealt with.

The difficulty with nonlinearities is that products in physical space are transformed into convolution sums in spectral space, as exemplified here using quadratic nonlinearities, *i.e.* a product vw. For the exact solution

we have:

$$\widehat{vw}_m = \sum_{m'=-\infty}^{+\infty} \hat{v}_{m-m'} \hat{w}_{m'},$$

which derives from the Fourier analysis of

$$egin{aligned} v(x) &= \left(\sum_{m'=-\infty}^{+\infty} \hat{v}_{m'} \exp(2\pi i\,m'x/\ell)
ight) \left(\sum_{m''=-\infty}^{+\infty} \hat{w}_{m''} \exp(2\pi i\,m''x/\ell)
ight) \ &= \sum_{m'=-\infty}^{+\infty} \sum_{m''=-\infty}^{+\infty} \hat{v}_{m'} \hat{w}_{m''} \exp[2\pi i\,(m'+m'')x/\ell]\,, \end{aligned}$$

see Figure B.4(a).

In the perspective of a truncation beyond $\pm N/2$, one should only keep the combinations such that m', m'', and m = m' + m'' belong to the considered interval (Fig. B.4(b), gray-shaded region). The computation of the Fourier transform of a formally quadratic term thus involves N sums with N^2 products each. A clever way to get around the rapid growth of the operation count with N is to take advantage of fast transforms to compute the product in physical space instead staying in the spectral space, with a reduction factor of order $\log(N)/N$.

Things are however not simple since one must take care of aliasing. If nothing special is done, the Fourier evaluation of a product computed in physical space contains spurious terms (Fig. B.4c). Let us indeed start with:

$$v_j = \sum_{m=-N/2+1}^{N/2} \hat{v}_m^{(\mathrm{d})} \exp{\left(2\pi i rac{mj}{N}
ight)}, \qquad w_j = \sum_{m'=-N/2+1}^{N/2} \hat{w}_{m'}^{(\mathrm{d})} \exp{\left(2\pi i rac{m'j}{N}
ight)},$$

next define :

$$\widehat{vw}^{(\mathrm{d})}_m = rac{1}{N}\sum_{j=0}^{N-1} (v_j w_j) \exp\left(-2\pi i rac{mj}{N}
ight)$$

and expand the right hand side. We get:

$$\widehat{vw}_{m}^{(d)} = \frac{1}{N} \sum_{j=0}^{N-1} \sum_{m'=-N/2+1}^{N/2} \sum_{m''=-N/2+1}^{N/2} \widehat{v}_{m'}^{(d)} \widehat{w}_{m}^{(d)} \exp\left(2\pi i \frac{(m'+m''-m)j}{N}\right)$$

with $(m', m'') \in [-N/2 + 1, N/2]$ and m = -N + 2, ..., N. Terms with m = m' + m'' outside the band [-N/2 + 1, N/2] are sent back inside the



Fig. B.4 (a) The computation of the convolution involves terms such that m = m' + m''. (b) For the truncated system only modes with |m'| < N/2, |m''| < N/2, |m| < N/2should be kept (the gray-shaded region). (c) For the discrete transformation, aliasing send each term in the triangular gray-shaded corners in the corresponding region close to the center, which perturbs the result.

band by the periodicity property (B.31) so that we get

$$\widehat{vw}_{m}^{(d)} = \sum_{m'+m''=m} \widehat{v}_{m'}^{(d)} \widehat{w}_{m''}^{(d)} + \sum_{m'+m''=m\pm N} \widehat{v}_{m'}^{(d)} \widehat{w}_{m'}^{(d)}$$

with m = -N/2 + 1, ..., 0, ..., N/2. When the Fourier coefficients $\hat{v}_m^{(d)}$ and $\hat{w}_m^{(d)}$ decrease sufficiently fast as |m| increases, the correction due to the spurious terms may be negligible but it is preferable to get rid of them ("unaliased" scheme). A popular way to achieve this purpose is illustrated in Figure B.5. Starting with the spectrum of v and w with $-N/2 + 1 \leq m \leq N/2$ one turns to a spectrum with $-M/2 + 1 \le m \le M/2$ and M > N by adding zeroes for $-M/2 + 1 \le m \le -N/2$ and $N/2 + 1 \le m \le M/2$ (in the figure, all products in the band outside the interior square with side N cancel). Going back to the physical space one obtains an evaluation of v and w on a lattice with M regularly spaced points, finer than the original lattice. The product vw is then computed at the nodes of this lattice and next Fourier transformed. It remains to drop the Fourier coefficients outside interval [-N/2+1, N/2] to get the unaliased spectrum of the product. The most dangerous non-zero term is with m' = m'' = N/2. Its alias, m' + m'' - M, gets off the band if m' + m'' - M < -N/2 + 1, thus if 3N/2 - 1 < M, so that one can take M = 3N/2 ("3/2 rule"). Despite manipulations implied by the size increase from N to M, the evaluation of convolutions involved in the treatment of nonlinear terms is freed from a systematic error.



Fig. B.5 As in Figure B.4c, but now with M instead of N, the aliased terms belong to the gray-shaded triangular regions. In the regions with uniform shading, they arise from terms that are zero by construction (|m| outside the [-N/2+1, N/2] interval). The non-trivial aliased terms belong to hatched regions. Some still belong to the domain bounded by the thick line when M < 3N/2 (left), which is no longer the case for $M \ge 3N/2$ (right).

B.3 Exercises

B.3.1 Truncation and round-off errors

Consider the differential equation $\frac{d}{dt}v = -v$ with initial condition v(0) = 1and exact solution $v(t) = \exp(-t)$ and point out the role of the order of a numerical scheme on the precision of the numerical integration by comparing the exact result at t = 1, v = 1/e, to approximations obtained with first-order iterations (B.3) and (B.5), second-order schemes (B.6), (B.7), (B.8), (B.9, B.10) and (B.11, B.12). In particular, show formally that the RK schemes are of order Δt^2 .

Write down all the corresponding programs and determine the distance $|v_N - 1/e|$ between the exact solution and the numerical solution v_N obtained performing N successive iterations of each scheme with a time-interval $\Delta t = 1/N$. Use logarithmic scales to plot this distance as a function of Δt and to identify the regime dominated by truncation errors from that dominated by the accumulation of round-off errors. Test also the fourth-order Runge-Kutta scheme (B.13).

B.3.2 Stability of multi-step schemes

Consider the integration of $\frac{d}{dt}v = f(v)$ by means of second-order schemes (B.7) and (B.8) and assume that a solution $\{v_n, n = 0, 1, ...\}$ is known.

Neighboring solutions $\{\tilde{v}_n, n = 0, 1, ...\}$ can be written as series of perturbations $\{u_n, n = 0, 1, ...\}$, with $u_n = \tilde{v}_n - v_n$ governed by the linearized map $g_n = df/dv|_n$. Show that the two numerical schemes respectively lead to:

$$u_{n+1} = u_n + \frac{1}{2}\Delta t \left(3g_n u_n - g_{n-1} u_{n-1} \right), \qquad (B.32)$$

$$u_{n+1} = u_{n-1} + 2\Delta t \, g_n u_n \,. \tag{B.33}$$

Then write (B.32) and (B.33) as two-dimensional maps for the two variables u_n and $v_n = u_{n-1}$ and compute their eigenvalues. Determine which is the physical eigenvalue λ_{phys} controlling the solution to the linearized problem $\frac{d}{dt}u = g(v)u$, where v is the solution to the primitive problem $\frac{d}{dt}v = f(v)$, and which is the eigenvalue λ_{num} corresponding to the spurious numerical mode. The instability takes place when $|\lambda_{num}| > 1$. Assuming that |g| is bounded by some g_{max} for the values of v of interest, discuss the stability properties of the two schemes.

B.3.3 Finite difference schemes for the SH model

Derive second-order consistent finite difference approximations to the linear part of the Swift-Hohenberg (SH) model:

$$\partial_t v = rv - (\partial_{xx} + 1)^2 v \equiv (r - 1)v - \partial_{xxxx}v - 2\partial_{xx}v.$$
(B.34)

Consider explicit-Euler, implicit-Euler and Crank-Nicholson temporal schemes and use (B.18) to discretize the fourth-order derivative. Study their stability by introducing discrete Fourier normal modes analogous to those leading to conditions (B.23) and (B.24).

Write down quasi-linearized Crank-Nicholson schemes for the original SH model, *i.e.* (B.34) with nonlinear term $\mathcal{N}(v) = -v^3$ added to its r.h.s. by expanding the nonlinear terms $\mathcal{N}(v)$ as:

$$\mathcal{N}(v)|_{n+1} = \mathcal{N}(v)|_n + \delta \mathcal{N}/\delta v|_n (v_{n+1} - v_n).$$

Consider also the modified SH model completed by a term $\mathcal{N}(v) = -v\partial_x v$ on the r.h.s. (*i.e.* an advection term $v\partial_x v$ on its l.h.s).

B.4 Case studies

The working sessions are organized around two themes: ODEs and PDEs. To deal with ODEs, we choose the simplest fourth-order Runge-Kutta scheme (B.13) and apply it to dynamical systems with few degrees of freedom, the periodically forced pendulum (two-dimensional, non-autonomous, §B.4.1) and the Lorenz model (three-dimensional, autonomous, §B.4.2). Other examples are illustrated but not studied in detail, the Rössler system, §B.4.3.1, and the Chua circuit, §B.4.3.2. All this can serve to materialize concepts introduced in Chapters 2 and 4.

Problems arising in EDPs are illustrated using two variants of the SH model introduced in some exercises of Chapters 3 and 4, and considered in exercise B.3.3 above. They may help to better understand pattern formation studied in Chapter 5. The approach can be extended to illustrate the generation of dissipative waves. Boundary conditions of Neumann-Dirichlet type are better adapted to finite difference methods and will mainly serve us to illustrate numerical stability problems. Periodic boundary conditions are ideal for a first approach of spectral methods used to illustrate space-time chaotic regimes. Computational Fluid Dynamics (CFD) will be left to specialists.

B.4.1 ODEs 1: Forced pendulum

We consider a pendulum submitted to a sinusoidal external force:

$$\frac{\mathrm{d}^2}{\mathrm{d}t^2}\theta + \eta \frac{\mathrm{d}}{\mathrm{d}t}\theta + \sin(\theta) = f\cos(\omega t) \tag{B.35}$$

where θ is the angle that the pendulum makes with the vertical, η is the damping coefficient, f is the intensity of the forcing and ω its angular frequency.

Write down a second-order Runge-Kutta program (B.11, B.12) for (B.35) written as a system of two first-order ODEs for θ and $\phi = \dot{\theta}$, taking care of the explicit time dependence introduced by the forcing. Then turn to a fourth-order scheme (B.13).

1) Use that routine to draw the projection of the trajectories in the plane (θ, ϕ) . Choose for example $\omega = \frac{2}{3}$, $\eta = 0.5$ and f variable in the range [0.5, 3]. Consider in particular cases f = 1.07 and 1.15. See Figure B.6.

2) Perform a stroboscopic analysis (Poincaré section) of the system at the forcing period $T = 2\pi/\omega$ and observe the attractor in particular for the previous parameter values mentioned. See Figure B.7, left.



Fig. B.6 Forced pendulum for $\omega = 2/3$, f = 1.15. Top-left: Time series of θ . Top-right: Time series of $\dot{\theta} \equiv \phi$. Bottom-left: ϕ as a function of θ . Bottom-right: Three-dimensional representation in the space (t, θ, ϕ) , ϕ -axis vertical, θ -axis to the left, t-axis to the right.

3) When the intensity of the forcing f increases, one finds parameter ranges where the attractor is periodic and other ranges where it is chaotic. Draw the bifurcation diagram obtained by varying f by recording the values reached by variables ϕ on the surface of section (for every $\Delta t = T$), at given $\omega \in [\omega_1, \omega_2]$, for a sufficiently long trajectory and after having eliminated points corresponding to the transient. See Figure B.7, right.

For all these simulations, it will be interesting to choose the time step Δt such that $\Delta t = T/N$ so that a full oscillation of the forcing corresponds exactly to an integer number of time steps. N will be sufficiently large, so that one has $\Delta t \ll T$ and $\Delta t \ll T_0 = 2\pi$, the natural period of the pendulum in the vicinity of its equilibrium position. A detailed study would show that the value of the bifurcation points is sensitive to the numerical resolution (compare results with the 2nd-order and 4th-order schemes).



Fig. B.7 Left: Poincaré section of the forced pendulum for f = 1.15. Right: Bifurcation diagram for $f \in [1, 1.25]$ obtained when increasing f (the result would be slightly different when decreasing f because some bifurcations are sub-critical, different attractors may coexist in some limited ranges of the parameter f, and hysteresis takes place correspondingly).

B.4.2 ODEs 2: Lorenz model

In Chapter 3, §3.2.1 we introduced a simplified model of nonlinear convection initially proposed by E.N. Lorenz, cf. Note 5, p. 88 rewritten here for convenience using standard notations:

$$\frac{\mathrm{d}}{\mathrm{d}t}X = \sigma(Y - X)\,,\tag{B.36}$$

$$\frac{\mathrm{d}}{\mathrm{d}t}Y = -XZ + rX - Y, \qquad (B.37)$$

$$\frac{\mathrm{d}}{\mathrm{d}t}Z = XY - bZ \,. \tag{B.38}$$

1) Display the solution in the three-dimensional space (X, Y, Z), draw the time series of Z and study the attractor obtained for the values initially chosen by Lorenz: $\sigma = 10$, b = 8/3 and r = 28 (Figure B.8).

Then vary r on the interval $r \in [145, 170]$, a range where the chaotic attractor decays into a periodic cycle through a subharmonic cascade and returns to chaos by intermittency (Figures B.9, and B.10).

2) Determine the Lorenz map defined as $Z_{k+1} = f(Z_k)$ where Z_k is defined by the condition that Z(t) goes through a maximum, *i.e.* $\frac{d}{dt}Z = 0$ with $\frac{d^2}{dt^2}Z < 0$. To do so, at each iteration check the value of $D_n = X_nY_n - bZ_n$, identify intervals $[t_n, t_{n+1} = t_n + \Delta t]$ where D_n changes sign from positive to negative. By linear interpolation, find the time τ_n where the change of sign takes place. Restarting from the result at t_n , perform



Fig. B.8 Left: The Lorenz attractor for r = 28 with $\Delta t = 0.01$. Initial conditions $X_0 = 6 = Y_0, Z_0 = 27$. Right: Time series Z(t) in the same conditions.



Fig. B.9 Left: "Intermittent" Lorenz attractor for r = 166.5 with $\Delta t = 0.01$. Initial conditions: $X_0 = 12 = Y_0$, $Z_0 = 165.5$. Lines joining successive points are not drawn to better point out the existence of a "ghost" trajectory recurrently visited by the system. Right: Limit cycle for r = 166 with $\Delta t = 0.01$. Initial conditions: $X_0 = 12 = Y_0$, $Z_0 = 165$. The cycle is reached after a transient that is not represented here.

a single Runge-Kutta step with time interval $\tau_n - t_n$ to obtain the set of coordinates of the corresponding point. Continue the simulation from t_{n+1} determined previously. The Lorenz map for r = 28 is displayed in Figure B.11. This method to obtain Poincaré maps with essentially the same accuracy as the integration itself was proposed by Hénon.²⁶ The same strategy can thus be used for Poincaré sections other than that defined by the condition D = 0. A popular one in early studies of the Lorenz model was the surface defined by Z = r - 1 and crossed from above (for example). 3) Construct a bifurcation diagram for the Lorenz model by plotting $Z_k - r + 1$ as a function of r (Figure B.12). Sets of isolated points at given

 $^{^{26}\}mathrm{M}.$ Hénon, "On the numerical computation of Poincaré maps," Physica D 5 (1982) 412-414.



Fig. B.10 Left: Z(t) for r = 166, periodic regime after a brief transient. Right: Z(t) for r = 166.5, intermittent regime.

r correspond to periodic behavior, continuous rows of inhomogeneously distributed points to chaotic behavior. The bifurcation around r = 320 is not a period doubling but a symmetry breaking bifurcation as shown by drawing trajectories in three dimensions before and after the bifurcation.



Fig. B.11 Lorenz map $Z_{k+1} = f(Z_k)$ for r = 28.



Fig. B.12 Bifurcation diagram of the Lorenz model.

B.4.3 ODEs 3: Rössler and Chua models

B.4.3.1 Rössler model

Another simple three-dimensional autonomous dynamical system with a strange attractor is the Rössler model.²⁷

$$egin{array}{lll} rac{\mathrm{d}}{\mathrm{d}t}X&=-Y-Z\,,\ rac{\mathrm{d}}{\mathrm{d}t}Y&=X+aY\,,\ rac{\mathrm{d}}{\mathrm{d}t}Z&=b+(X-c)Z\,. \end{array}$$

Notice that its definition is even simpler than that of the Lorenz model since it has a single nonlinear quadratic term and no obvious symmetry. The dynamics has two ingredients: approximately periodic behavior at some distance from an unstable spiral fixed point and fast excursions away from the (X, Y)-plane. It displays periodic and chaotic regimes. Typical trajectories are shown in Figure B.13. Use the same tools as for the Lorenz model to study its bifurcation diagram as a function of $c \in [1, 20]$.

²⁷O.E. Rössler, "An equation for continuous chaos," Physics Lett. A 57 (1976) 397–398. Also, "Continuous chaos - Four prototype equations," Ann. NY Acad. Sc. 316 (1979) 376–392. See also: J.C. Sprott, "Simplest dissipative chaotic flows," Physics Lett. A 228 271–274.



Fig. B.13 The Rössler attractor for a = 0.1, b = 0.1. Top-left: Limit cycle for c = 12. Top-right: Limit cycle after a period doubling for c = 12.7. Bottom: Chaotic attractor beyond the accumulation point of the subharmonic cascade for c = 13.6.

B.4.3.2 Chua circuit

The Chua circuit is the last example proposed here. It is defined as a prototype of analogical systems displaying chaos that can be implemented as electronic circuits with good accuracy. It is also three-dimensional and autonomous. Its main properties are reviewed in a collective work edited by R.N. Madan.²⁸ It reads

$$\frac{\mathrm{d}}{\mathrm{d}t}X = \alpha(Y - X - F(X)),$$

$$\frac{\mathrm{d}}{\mathrm{d}t}Y = X - Y + Z,$$

$$\frac{\mathrm{d}}{\mathrm{d}t}Z = -\beta Y.$$

where F(X) is an odd function of X that is linear by part: $F(X) = m_1 X + \frac{1}{2}(m_0 - m_1)(|X + 1| - |X - 1|)$. The "double scroll" attractor is displayed in Figure B.14. Establish the bifurcation diagram of the Chua system from a Lorenz map of variable Z, with $\beta \in [1, 25]$ and other parameters fixed as in the caption of Figure B.14. Display other typical attractors.

²⁸R.N. Madan, Ed., Chua's circuit: a paradigm for chaos (World Scientific, 1993).



Fig. B.14 Two views of the Chua "double scroll" attractor for $\alpha = 9$, $\beta = 100/7$, $m_0 = -8/7$, $m_1 = -5/7$.

B.4.4 PDEs 1: SH model, finite differences

Finite difference methods are illustrated here using the original variant of the Swift-Hohenberg model with a cubic nonlinear term in one space dimension:

$$\partial_t v = rv - (\partial_{xx} + 1)^2 v - v^3 \tag{B.39}$$

with boundary conditions

$$v(0) = v(\ell) = \partial_x v(0) = \partial_x v(\ell) = 0 \tag{B.40}$$

The two parameters are r and ℓ . The model derives from a potential and is a good tool to study the formation of cellular structures in simple cases.

B.4.4.1 Explicit scheme

Develop the simplest possible, first order in time, second order in space, explicit numerical scheme. To maintain second order consistency at the boundaries, add fictitious exterior points enslaved to the first inner points as explained in §B.2.1.2. Observe how the numerical instability develops when the condition on Δt obtained in Exercise B.3.3, namely $\Delta t < \Delta x^4/8$ for Δx small, is violated.

The effect of the space resolution is shown in Figure B.15. In order to reach the finest grid, the time step should be considerably reduced in order to avoid the numerical instability illustrated in Figure B.16. Notice that in certain cases depending on initial conditions, the nonlinear term may be



Fig. B.15 Explicit-Euler numerical simulation of the Swift-Hohenberg model with r = 0.3; random initial conditions with amplitude 0.01; solution at $t_f = 300$. Left: $\ell = 64$, n = 64, $\Delta x = 1.0$, $\Delta t = 0.1$. Right: $\ell = 64$, n = 128, $\Delta x = 0.5$, $\Delta t = 0.008$, numerically stable.



Fig. B.16 Numerical instability of the Swift-Hohenberg model with $\ell = 64$, n = 128, $\Delta x = 0.5$, $\Delta t = 0.01$, r = 0.3, random initial conditions with amplitude 0.01. Left: State at iteration #20. Right: The same numerical instability at three successive time steps, iterations #18 (dots), #19 (dashes), and #20 (solid line).

sufficiently strong to prevent the numerical divergence but the so-obtained solution is not physical in that it has a space period very different from the expected one $\simeq \lambda_c = 2\pi$ (in practice much shorter).

B.4.4.2 Implicit scheme

Develop an implicit Euler scheme, first order in time, second order in space, by solving the linear problem as explained in §B.2.1.4. Take $\ell = 64$ ($\simeq 10\lambda_c$) in order to deal with a sufficiently extended system. The number of grid points N will be varied to check the effect of the numerical resolution



Fig. B.17 Swift-Hohenberg model with $\ell = 64$, $\Delta t = 0.1$, r = 0.3, random initial conditions with amplitude 0.01; solution at $t_f = 300$. (a) n = 64, $\Delta x = 1.0$. (b) n = 128, $\Delta x = 0.5$. (c) n = 256, $\Delta x = 0.25$. (d) zoom on the left boundary showing the accuracy of the numerical account of boundary condition $\partial_x w = 0$.

(Figure B.17). Second order schemes can also be developed, either of the quasi-linearized Crank-Nicholson type (Exercise B.3.3) or according to an Adams-Bashford scheme.

Since numerical stability problems are dismissed, focus on the physics of the formation of the cellular structure. Identify the initial phase of exponential growth of perturbations selected by the linear dynamics, then the saturation phase where nonlinearities come in the foreground, and the final stage where most effects compensate so that a slow residual motion (phase diffusion) towards the final time-independent state is observed.

B.4.5 PDEs 2: SH model, pseudo-spectral method

Consider now the modified Swift-Hohenberg model:

$$\partial_t v + v \partial_x v = \left[r - (\partial_{xx} + 1)^2 \right] v,$$
 (B.41)

with periodic boundary conditions at the ends of an interval of length ℓ . This model, which does not derive from a potential and thus can have unsteady behavior, is particularly appropriate to study the growth of spacetime chaos in cellular structures.

Derive a Fourier pseudo-spectral algorithm implementing the model taking advantage of the fact that $v \partial_x v = \frac{1}{2} \partial_x (v^2)$, according to the scheme:

$$\hat{v}_m \begin{cases} \mapsto \widehat{\mathcal{L}v}_m = s_m \hat{v}_m = \left(r - (k_m^2 - 1)^2\right) \hat{v}_m \\ \mapsto v(x) \mapsto \frac{1}{2} v(x)^2 \mapsto \frac{1}{2} \widehat{v}_m^2 \mapsto \widehat{v\partial_x v}_m \equiv \frac{1}{2} i k_m \widehat{v}_m^2 = \widehat{\mathcal{N}}_m \end{cases}$$

where $\widehat{\mathcal{N}}_m$ must be computed as discussed earlier in order to avoid aliasing errors. Next choose a time evolution scheme (§B.1). Here the linear part can be integrated exactly since it is diagonal in Fourier space. For the nonlinear part a second order Adams-Bashford scheme can be chosen: The final result is

$$\hat{v}_{n+1,m} = \exp(s_m \Delta t) \hat{v}_{n,m} + \Delta t \left[\frac{3}{2} \widehat{\mathcal{N}}_{n,m} - \frac{1}{2} \widehat{\mathcal{N}}_{n-1,m} \right] \,.$$

When r = 1 the Swift-Hohenberg model happens to be a variant of the Kuramoto-Sivashinsky equation. In a slightly different form:

$$\partial_t v + \frac{1}{2} (\partial_x v)^2 + \partial_{xx} v + \partial_{xxxx} v = 0,$$
 (B.42)

this equation appears in the treatment of turbulent interface propagation, e.g. flame fronts, as shown in Figure B.18.



Fig. B.18 Kuramoto-Sivashinsky equation for $\ell = 64$, n = 128, $\Delta x = 0.5$, with a time interval $\Delta T = 1$ between two successive snapshots.
This page intentionally left blank

Bibliography

Dynamical systems, bifurcations and catastrophes, chaos

- Abarbanel, H.D.I. (1996). Analysis of Observed Chaotic Data, (Springer).
- Abraham, R.H. and Shaw, C.D. (1992). Dynamics: the Geometry of Behavior (Addison-Wesley).
- Alligood, K.T., Sauer, T.D., and Yorke, J.A. (1996). Chaos, an Introduction to Dynamical Systems (Springer).
- Baker, G.L. and Gollub, J.P. (1996). Chaotic Dynamics, an Introduction, 2nd Edition (Cambridge Univ. Press).
- Beck, Ch. and Schlögl, F. (1995) Thermodynamics of Chaotic Systems (Cambridge Univ. Press).
- Bergé, P., Pomeau, Y., and Vidal, Ch. (1986). Order within Chaos, towards a Deterministic Approach to Turbulence (Wiley).
- Cladis, P.E. and Palffy-Muhoray, P., Editors (1995) Spatio-Temporal Patterns in Nonequilibrium Complex Systems (Addison-Wesley).
- Cross, M.C. and Hohenberg, P.C. (1993). Pattern formation outside equilibrium, Rev. Mod. Phys. 65 (1993) 851-1112.
- Cvitanović, P., Editor (1989). Universality in Chaos (Adam Hilger) [reprint collection].
- Devaney, R.L. (1989) An Introduction to Chaotic Dynamical Systems, 2nd edition (Addison-Wesley).
- Drazin, P.G. (1992). Nonlinear Systems (Cambridge Univ. Press).
- Eckmann, J.P. and Ruelle, D. (1985). Ergodic theory of chaos and strange attractors. Rev. Mod. Phys. 57 (1985) 617-656.
- Frøyland, J. (1992). Introduction to Chaos and Coherence, (IoP Publishing).
- Glansdorff, P. and Prigogine, I. (1971). Thermodynamics of Structures, Stability and Fluctuations (Wiley).
- Guckenheimer, J. and Holmes, Ph. (1983). Nonlinear Oscillations, Dynamical Systems, and Bifurcations of Vector Fields (Springer).
- Haken, H. (1983). Synergetics, 3rd edition (Springer).
- Hao, B.-l., Ed. (1987-...). Directions in Chaos (World Scientific) [several volumes]. Hao B.-l., Ed. (1990). Chaos II (World Scientific) [reprint collection].

- Hirsch, M.W. and Smale, S. (1974). Differential Equations, Dynamical Systems and Linear Algebra (Academic Press).
- Kaneko, K., Editor (1993). Theory and Application of Coupled Map Lattices (Wiley).
- Kanz, H. and Schreiber, Th. (1997). Nonlinear Time Series Analysis, Cambridge Nonlinear Science Series 7 (Cambridge Univ. Press).
- Lefschetz, S. (1977). Differential Equations: Geometric Theory (Dover).
- Mandelbrot, B.B. (1982). The Fractal Geometry of Nature (Freeman).
- Moon, F.C. (1992). Chaotic and Fractal Dynamics, an Introduction for Applied Scientists and Engineers (Wiley).
- Nayfeh, A.H. and Mook, D.T. (1979). Nonlinear Oscillations (Wiley).
- Ott, E. (1993). Chaos in Dynamical Systems (Cambridge Univ. Press).
- Poston, T. and Stewart, I. (1978). Catastrophe Theory and its Applications (Pitman).
- Rabinovich, M.I., Ezersky, A.B., and Weidman, P.D. (2000). The Dynamics of Patterns (World Scientific).
- Schuster, H.G. (1988). Deterministic Chaos, an Introduction 2nd edition (VCH).
- Schuster, H.G., ed. (1999). Handbook of Chaos Control (Wiley).
- Steeb, W.-H. (1991). A Handbook of Terms used in Chaos and Quantum Chaos (BI, Mannheim).
- Strogatz, S.H. (1994). Nonlinear Dynamics and Chaos, with Applications to Physics, Chemistry and Engineering (Addison-Wesley).
- Weigend, A.S. and Gershenfeld, N.A., Editors (1993). Time Series Prediction: Forecasting the Future and Understanding the Past (Addison-Wesley).
- Wolfram, S., Editor (1986). Theory and Application of Cellular Automata (World Scientific).

Hydrodynamics, instabilities, and turbulence

- Batchelor, G.K. (1967). An Introduction to Fluid Dynamics (Cambridge Univ. Press).
- Batchelor, G.K., Moffatt, H.K., and Worster, M.G., Editors (2000). Perspectives in Fluid Dynamics. A Collective Introduction to Current Research (Cambridge Univ. Press).
- Betchov, R. and Criminale, W.O. (1967). Stability of Parallel Flows (Academic Press).
- Chandrasekhar. S. (1961). Hydrodynamic and Hydromagnetic Stability (Clarendon Press).
- Drazin, P.G. (2002). Introduction to Hydrodynamic stability (Cambridge University Press).
- Drazin, P.G. and Reid, W.H. (1981). Hydrodynamic Stability (Cambridge Univ. Press).
- Frisch, U. (2001). Turbulence. The legacy of A.N. Kolmogorov, 2nd edition (Cambridge Univ. Press).
- Godrèche, Cl. and Manneville, P., Editors (1998). Hydrodynamics and Nonlinear

Instabilities (Cambridge Univ. Press).

- Landau, L.D. and Lifshitz, E.M. (1987). Fluid Mechanics Course of Theoretical Physics, Vol. 6, 2nd edition (Butterworth-Heinemann).
- Lesieur, M. (1997). Turbulence in Fluids (Kluwer).
- Lesieur, M., Yaglom, A., and David, F., Editors (2001) New Trends in Turbulence. Les Houches, session LXXIV (Springer).
- Mollo-Christensen, E.L. (1972). Flow Instabilities, film produced by [NCFMF (1972)].
- NCFMF (1972). Series of films produced by the National Committee for Fluid Mechanics, distributed by Encyclopedia Britannica Educational Corp.; accompanying book: Illustrated Experiments in Fluid Mechanics (MIT Press).
- Pope, S.B. (2000). Turbulent flows (Cambridge Univ. Press).
- Rothman, D.H. and Zaleski, S. (1997). Lattice-gas Cellular Automata. Simple Models of Complex Hydrodynamics (Cambridge Univ. Press, 1997).
- Schlichting, H. (1979). Boundary Layer Theory (McGraw-Hill).
- Schmid, P.J. and Henningson, D.S. (2001). Stability and Transition in Shear Flows (Springer-Verlag).
- Stewart, R.W. (1972). Turbulence, film produced by [NCFMF (1972)].
- Swinney, H.L. and Gollub, J.P., Editors (1985). Hydrodynamic Instabilities and the Transition to Turbulence (Springer).
- Tabeling, P. and Cardoso, O., Editors (1994) Turbulence. A Tentative Dictionary (Plenum Press).
- Tennekes, H. and Lumley, J.L. (1972). A First Course in Turbulence (MIT Press). Tritton, D.J. (1990) Physical Fluid Dynamics (Cambridge Univ. Press).
- van Dyke, M. (1982). An Album of Fluid Motion (Parabolic Press).
- Wilcox, D.C. (2000). Turbulence Modeling for CFD (DWC Industries).

Numerical modeling

- Abramowitz, M. and Stegun, I.A., Editors (1972). Handbook of Mathematical Functions (National Bureau of Standards) [Chapter 22: Numerical interpolation, differentiation, and integration].
- Acton, F.S. (1970). Numerical Methods that Work (Harper and Row).
- Canuto, C., Hussaini, M.Y., Quarteroni, A., and Zang, T.A. (1988). Spectral Methods in Fluid Dynamics (Springer).
- Finlayson, B.A. (1972). The Method of Weighted Residuals and Variational Principles, with Application in Fluid Mechanics, Heat and Mass Transfer (Academic Press).
- Gottlieb, D. and Orszag, S.A. (1977) Numerical Analysis of Spectral Methods: Theory and Applications (SIAM Publications).
- Lapidus, L. and Seinfeld, J.H. (1971). Numerical Solution of Ordinary Differential Equations (Academic Press).
- Press, W.H., Flannery, B.P., Teukolsky, S.A., and Vetterling, W.T. (1986). Numerical Recipes, the Art of Scientific Computing (Cambridge Univ. Press) [program sources available].

Richtmyer, R.D. and Morton, K.W. (1967) Difference Methods for Initial Value Problems (Interscience).

Miscellaneous

- Anderson, P.W., Arrow, K.J., and Pines, D., Editors (1988). The Economy as an Evolving Complex System, SFI Studies in the Sciences of Complexity (Addison-Wesley).
- Hall, N., ed. (1992). The NewScientist Guide to Chaos (Penguin Books).
- Henderson-Sellers, A. and McGuffie, K. (1987). A Climate Modelling Primer (Wiley).
- Peixoto, J.P. and Oort, A.H. (1992). Physics of Climate (AIP Publishing).
- Philander, S.G. (2000) Is The Temperature Rising? The Uncertain Science of Global Warming (Princeton Univ. Press).
- Stanley, H.E. (1988). Introduction to Phase Transitions and Critical Phenomena (Oxford University Press). [reprint original edition, Pergamon Press, 1971]

Winfree, A.T. (1990). The Geometry of Biological Time (Springer).

Index

adiabatic elimination, 89, 119, 139, 169, 203, 309 advection, 9, 74, 202, 239, 274, 285 aliasing, 366 analytical mechanics, 32 aspect ratio, 93, 116, 309 large, 99–101, 117, 181 small, 94, 95, 102, 117 attractor, 31 fixed point, 13, 31, 307 limit cycle, 48, 129, 307 limit torus, 115, 178, 307 strange, 115, 143, 147, 151, 307 averaging method, 52, 69 baker map, 147, 150 dissipative, 153 Benjamin-Feir instability, 201 bifurcation, 1, 5, 43, 48, 63, 89, 305 Hopf, 128, 197, 242, 245, 306 of limit cycle, 142 perfect/imperfect, 127, 170 saddle-node, 170, 171 subharmonic, 97 super/sub-critical, 13, 14, 17, 89, 129, 184, 185, 242, 249, 254 transcritical, 170, 184 bifurcation diagram, 13, 50, 126, 137, 144, 170boundary layer, 15, 215, 250, 285, 301 Blasius (laminar), 216 stability & transition, 238, 251, 254

thickness, 216, 217, 251 turbulent, 19, 102, 286, 288 Kármán log-law, 288, 289 Boussinesq approximation, 105 Briggs-Bers criterion, 240, 269 Burgers eq., 23, 202, 208 Busse balloon, 92, 99 bypass transition, 253, 254, 311 Cantor set, 151, 153, 176 catastrophe theory, 1, 34, 128, 306, 328, 384cellular instability, 85, 104, 169, 182, 191, 193, 195, 199, 309, 378, 380, 381chaos, 5, 14, 86, 94, 117, 143, 147, 148, 168, 303, 307, 309, 313 chemical kinetics, 20 Chua circuit, 377 climate change, 317, 327 closure problem, 282, 293, 298, 311 coherence length, 84, 193, 198, 308 configuration space, 27 confinement effects, 14, 93, 116, 168, 309 conjugate variables, 27 continuous medium, 6, 14, 76, 308 control parameter, 5, 9, 28, 75, 119, 182, 305Couette flow (cylindrical), 113, 261 Couette flow (plane), 213, 215 stability & transition, 17, 257

cross-roll instability, 195, 200, 208 defect (in pattern), 98, 196, 310 defect mediated turbulence, 203 degree of freedom, 3, 13 effective, 12, 13, 84, 116, 309 in analytical mechanics, 27, 64 in developed turbulence, 278, 295 demodulation, 156 determinism, 4, 27, 60, 147 diffusion, 8, 11, 22, 284 molecular (Fick law), 7, 110 thermal (Fourier law), 6, 71 turbulent, 272, 285 viscous (Stokes law), 8, 11, 71 dimension correlation, 166 embedding, 159, 163, 166 fractal, 147, 152, 153, 166, 176 topological, 153, 176 dispersion relation, 76, 84, 182, 191, 198, 220, 230, 236, 240 dissipative structure, 11 Duffing oscillator, 2, 50, 58, 66, 69 forced, 132 dyadic map, 148 dynamical system, 2, 27, 30, 303, 323 autonomous/forced, 27, 130, 131 conservative/dissipative, 31, 115 discrete-time, 4, 28 gradient, 34, 196, 206 linear, 37, 63, 334, 344 Earth's climate, 312, 317 Eckhaus instability, 199–201 eddy viscosity, 285, 287, 292, 311 elliptical instability, 243 energy method (stability), 42, 62 ensemble average, 279, 280, 294 ENSO phenomenon, 315

- envelope equation, 191, 197, 310 ergodic hypothesis, 280 Euler eq., 8 false neighbor method, 163
- Fast Fourier Transform (FFT), 366

Fick law (molecular diffusion), 7 first harmonic approximation, 49, 133 fixed point, 13, 36, 89, 115 center (elliptic point), 39, 40, 45 focus (spiral point), 39, 40, 46 improper node, 41 node, 38 saddle, 38, 45 stable/unstable, 31, 38 Fjørtoft criterion, 224, 266 fluid particle, 7 forcing external, 28 parametric, 27 periodic, 131, 136, 234, 317, 371 Fourier eq., 6, 22, 108 fractal structure, 116, 151, 176, 308 Fredholm alternative, 56, 346, 350 frequency locking, 95, 137-139, 143, 174 frozen turbulence (Taylor), 276, 300 Galerkin method, 50, 88, 107, 119 Ginzburg-Landau eq., 198, 248, 269 global mode, 16 globally super/sub-critical transition, 254, 256, 259, 260, 310 gradient flow, 34, 196, 206 greenhouse effect, 318, 326 group velocity, 85, 198 Hénon map, 152 Hamilton equations, 33 harmonic oscillator, 2, 25, 39

incompressible flow, 8, 213, 221 inertial cascade, 18, 274, 275, 291, 311

Hopf bifurcation, 128, 129, 197, 242,

damped, 29, 31

Hilbert transform, 157

Howard theorem, 224

homogeneous instability, 85

forced, 132

245, 306

instability convective/absolute, 16, 212, 228, 234, 239-241, 246, 248, 309 Briggs-Bers criterion, 240, 269 local/global, 16, 240, 248 phase/amplitude, 199 primary/secondary, 10, 14, 90 stationary/oscillatory, 85 instability cascade, 10, 14, 90, 125, 254, 308 instability threshold, 11, 81 intermittency, 97, 146, 373, 374 in developed turbulence, 291 spatio-temporal, 204, 253 jet, 215, 217, 244, 268 turbulent, 286 Kármán log-law, 19, 289, 311 Kelvin-Helmholtz instability, 212, 227 kernel of an operator, 334 Kolmogorov scale, 277 Kolmogorov spectrum, 19, 276, 311 Korteweg-de Vries eq., 202 Kuramoto-Sivashinsky eq., 201, 381 Lagrange equations, 32 Landau expansion, 13, 34, 126, 127, 169, 306 Langevin eq., 28, 160 Lewis number, 110 limit cycle, 48, 115, 129 limit set, 115 limit torus, 142 linear response, 1, 128 linearization, 36, 74, 168 Lorenz map, 373 Lorenz model, 88, 373 Lotka–Volterra system, 22 Lyapunov exponent, 147, 165 Lyapunov functional, 62 master/slave mode, 89 material derivative, 7 micro/meso/macroscopic scale, 5, 18,

303

mixing layer, 15, 225, 244, 267 turbulent, 286 mixing length, 285, 311 momentum space, 27 multiple scale method, 57, 310 mutual information, 162 Navier-Stokes eq., 8, 108, 213, 280, 311Newell criterion, 202 Newell-Whitehead-Segel eq., 193 noise extrinsic, 4, 28, 160, 166 intrinsic, 97, 99 noise amplifier, 16, 228, 240 non-normal operator, 262, 343, 344, 350normal form, 13 Jordan, 37, 40, 63, 335, 339, 349 Jordan-Arnold, 343 nonlinear, 122, 123, 198 normal mode, 75 oscillatory, 42, 77, 128 stationary, 42, 77 numerical scheme Adams-Bashford, 354, 363, 380 Crank-Nicholson, 363, 370 Euler, 352, 370 explicit/implicit, 352, 353, 360, 370 finite difference, 358, 360, 378 first/second order, 352–354 leap-frog, 354 guasi-linearized, 363 Runge-Kutta, 356, 357, 369, 371 spectral & pseudo-spectral, 363, 366, 380 Nusselt number, 101 open flow, 15, 211, 234, 309 order parameter, 86, 128 Orr-Sommerfeld eq., 222, 231, 266 paleoclimate, 316, 318 pattern, 11, 93, 117, 181, 189, 310 modulation, 192 selection, 190, 197

pendulum, 2, 44 damped, 45 phase instability, 199, 310 phase portrait, 28 phase space (state space), 3, 13, 26 phase transition (thermodynamic), 34, 86, 127, 306 phase turbulence, 202 phase velocity, 85 phase-winding solution, 207 Poincaré section & map, 4, 132, 140, 307, 374 Poincaré-Bendixson theorem, 130 Poincaré-Lindstedt method, 55, 69 Poiseuille flow (channel) laminar, 215 stability & transition, 234, 249, 250, 254 Poiseuille flow (pipe) laminar, 214 stability & transition, 17, 255 Prandtl number, 82, 90, 284 proper orthogonal decomposition, 165 qualitative dynamics, 28 Rössler model, 376 rank of a matrix, 334 Rayleigh criterion, 15, 223, 266 Rayleigh eq., 222, 229 Rayleigh number, 73, 81 Rayleigh-Bénard instability, 11, 71 repellor, 31 resonance, 40, 124, 135, 137, 143, 335 primary/secondary, 133, 135 spatial, 188–190, 251, 252 stochastic, 319 weak/strong, 143, 144, 146 resonance tongue, 138 resonant term, 56, 60, 122, 123, 183, 186, 188Reynolds averaging, 19, 280, 293, 311 Reynolds number, 9, 99, 211, 218, 219, 226, 245, 258, 273 Taylor's R_{λ} , 278 Reynolds stress, 19, 282, 287, 292, 311

Reynolds-Orr eq., 232 saddle point, 34 scenario, 115, 142, 308 intermittency, 97, 146, 373, 374 Landau, 92, 307 Ruelle-Takens, 92, 95, 143, 307 subharmonic, 94, 144, 308, 373 secular term, 41, 55, 56, 339, 350 self-sustained oscillator, 16, 47, 48, 229, 240 sensitivity to initial conditions, 147 separatrix, 45 singular value decomposition, 165 soliton, 65 spatio-temporal chaos, 86, 93, 98, 181, 310 Squire theorem, 221, 234, 262, 266 stability, 4 asymptotic, 42, 62 conditional, 10 global, 9, 250, 265 global method, 42, 62, 223 linear, 9, 38, 73, 219, 304 numerical, 355 CFL criterion, 299 orbital, 42, 62 structural, 5, 39, 43 state space (phase space), 3, 27 statistical ensemble, 279 Stokes law (viscous diffusion), 8 stroboscopic analysis, 4, 28, 131, 371 Strouhal number, 246 subharmonic cascade, 94, 95, 144, 145, 308, 373 susceptibility, 1, 128 Swift–Hohenberg model, 103, 370, 371, 378, 380synergetics, 119 Taylor-Couette instability, 114, 261 temporal vs. spatial stability analysis, 220, 234, 236 texture, 97, 98, 117, 191 thermal conduction, 6, 10, 71, 72 thermal convection, 10, 71

thermodynamic branch, 36, 253 thermodynamic equilibrium, 5, 303 local, 6 neighborhood, 1, 9, 36 thermohaline circulation, 323 thermohaline convection, 109 time average, 149, 155, 279, 280 Tollmien-Schlichting wave, 212, 233 transient chaos, 178 transient energy growth, 262, 344, 350 turbulent convection, 101 turbulent scale, 18, 271 dissipation, 18, 277, 311 inertial, 18, 274 production, 18, 272, 274 Taylor, 273, 278 turbulent spot, 213, 253, 311 in boundary layer, 252, 256, 257 in plane Couette flow, 17, 257, 258 in Poiseuille channel flow, 250, 256 Turing pattern, 112

universality in developed turbulence, 276, 288 in dynamical systems, 13, 115, 124, 306, 309 in pattern dynamics, 191, 195, 198, 199, 309, 310 in phase transitions, 128 unpredictability, 5, 116, 149, 313, 314 van der Pol oscillator, 46, 54, 59 forced, 137 variation calculus, 196 vector field, 13, 26, 27, 30, 34, 35, 130 divergence, 30, 61 reconstruction, 165 viscous friction, 29, 45, 65, 68 wake, 16, 215, 218, 244, 245 wave dispersion, 85, 198 weather forecast, 313, 316 zigzag instability, 199-201